Poverty, Income Fluctuations and Work: Argentina 1991-2002

Guillermo Antonio Cruces PhD Thesis, Economics Department London School of Economics and Political Science University of London

May 2005

UMI Number: U194921

All rights reserved

INFORMATION TO ALL USERS

The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



UMI U194921 Published by ProQuest LLC 2014. Copyright in the Dissertation held by the Author. Microform Edition © ProQuest LLC. All rights reserved. This work is protected against unauthorized copying under Title 17, United States Code.



ProQuest LLC 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106-1346

THESES F 8393 993842

ABSTRACT

ntish Library of Politica nd Economic Science

> This thesis presents results on the economics of poverty and labour markets, using data from Argentina in the 1991-2002 period for empirical applications. The thesis is divided into two Parts.

> The first Part concentrates on poverty and the effects of income fluctuations on well-being. The 1995-2002 period in Argentina was characterised by recurring economic crises that produced large fluctuations in household income. The empirical applications of this Part rely on a rotating panel dataset from the Greater Buenos Aires region to study the effects of this variability.

> The first Chapter introduces the data and its characteristics, and describes the economic context of the period. The second Chapter defines a family of indicators of well-being, based on the theory of choice under uncertainty, that account for the negative impact of income fluctuations on household welfare. Chapters 3 and 4 present risk adjusted measures of income and the transient-chronic poverty decomposition, respectively, two methodologies for the study of poverty with panel data which are related to the indicators defined in Chapter 2. In Chapters 3 and 4, the household characteristics associated with income fluctuations and their impact on well-being are identified through regression analysis.

> Part II deals with fertility and women's labour supply from an empirical perspective, and uses data from the 1991 Argentine Census for its applications. Chapter 5 presents the theoretical and econometric framework employed to deal with the endogeneity of the fertility decision. This identification strategy exploits parental sex preferences as instrumental variables for further child-bearing. Chapter 6 discusses its validity in developing countries, and provides empirical evidence for Argentina. Chapter 7 presents the main results of the estimation, which state that additional children cause a reduction in their mother's labour supply. Finally, Chapter 8 proposes a new test for the generality of instrumental variables results, which is illustrated with the same dataset.

TO LAURA AND CLARITA

٠

ACKNOWLEDGEMENTS

Many thanks to Frank Cowell, my supervisor, for his support throughout the writing of this thesis, which would not have materialised without his continuous challenge to intellectual conformism.

I truly benefited from joint work with Quentin Wodon, who generously shared his knowledge, energy and experience with me. Sebastián Galiani was a great teacher and provided advice and motivation during the writingup process.

Stefan Dercon's observations on an early version of Chapter 3 played an important role in the origins of Chapter 2. Martin Ravallion made comments on Chapter 3 in the form of a peer review, and he also provided an insightful discussion that contributed to Chapter 2. I am also indebted to Ramses Abul-Naga and Giovanni Ko for comments on that Chapter. Alan Manning and Marco Manacorda provided detailed comments on Part II, and contributed with helpful inputs at different stages of the dissertation. Alberto Abadie was an excellent teacher, and he generously shared the JTPA data used in Part II. I often received valuable comments from presentations at different seminars and conferences: my thanks to the attendants. The usual disclaimer applies: all remaining errors, alas, are mine.

I also wish to thank Camila Arza, Sami Berlinski, Claudia Giacometti, Raquel Mayer, Juan Martín Moreno, Edgar Poce and Lucas Siga, who offered their generous help at different stages of this thesis with data, references and suggestions. For their help in academic matters (but more than that in most cases) I would like to thank Paolo Benedetti, Alejandro Bonvecchi, Michella Cella, Juan De Laiglesia, Carlo Fiorio, Pablo Gerchunoff, Markus Goldstein, Colin Lewis, Ceema Namazie, Silvia Pezzini, Juan Pablo Rud, Miguel Sánchez, Barbara Veronese and Tito Winograd. Jonathan Chuei and Lucrecia Fittipaldi provided excellent research assistance. I also want

ACKNOWLEDGMENTS

to thank all members of STICERD, who create a challenging and pleasant work environment, and especially Leila Alberici, Sue Coles, Jane Dickson, Nic Warner and all the staff who provided invaluable assistance.

Diana, Silvia, Natalia, Sebastián and Camilo hold a special place in these acknowledgements, and I want to thank their love and constant support throughout all these years. Finally, for their support and encouragement in general, I would like to thank Fernando García, Carlos Manuel Grosso, Ana Langer, Enrique Martínez, Julia Samuel and Bill Wells. My fragile memory certainly left out people who deserve to be mentioned in these acknowledgements: to them, my thanks and my sincere apologies.

Financial support from the Economics Department (LSE), Suntory and Toyota Centres for Economics and Related Disciplines (STICERD), the Anglo-Jewish Association (Karten Scholarship), the Rayne Foundation, Comisión Nacional de Investigaciones Científicas (CONICET), Fundación Antorchas, the British Council and the Foreign and Commonwealth Office is gratefully acknowledged.

The greatest of thanks to Laura for always standing by my side and for showing the rare gift of seemingly infinite patience. Without her – and her patience – this thesis would have been impossible. The thesis is dedicated to her and to the memory of Clarita, who left us her unconditional love and the example of her struggle.

CONTENTS

Co	onten	ts	6
Li	st of '	Tables	10
Li	st of :	Figures	12
In	trodu	iction	14
Ι	Po	verty and Income Fluctuations, Argentina 1995-2002	17
1	Arg	entina's Crises and the Poor	18
	1.1	Introduction to Part I: Key Economic Events, Argentina 1991-	
		2002	18
	1.2	Household Data and Measurement of Poverty in Argentina .	22
		1.2.1 National cross sections and the GBA panel	22
		1.2.2 Income aggregate and equivalence scale	24
		1.2.3 Poverty lines and regional heterogeneity	28
	1.3	Poverty Trends and Short Term Dynamics	29
		1.3.1 Income, prices and poverty lines	29
		1.3.2 Poverty trends	33
		1.3.3 Poverty transitions and short term dynamics	38
	1.4	Guide to the Literature on Poverty in Argentina	42
		1.4.1 Income distribution and poverty in Argentina	42
		1.4.2 Economic crises, household welfare and coping	40
		strategies	43
		1.4.3 Longitudinal studies of poverty and well-being in Ar-	
		gentina	46
	1.5		48
2	Inco	ome Fluctuations, Poverty and Well-Being Over Time	50
_	2.1	Introduction	50
	2.2	Ex-ante and Ex-post Income Variability	52
		2.2.1 Prospective evaluation of well-being: ex-ante utility	
		and income risk	52

CONTENTS

		2.2.2	Retrospective evaluation of well-being: ex-post utility	
			and income fluctuations	54
	2.3	A Fran	nework for the Evaluation of Income Fluctuations	56
		2.3.1	The structure of the evaluation function	56
		2.3.2	Evaluation of well-being and variability over time	60
		2.3.3	"Fluctuation adjusted" population measures of well-	
			being	62
	2.4	Compa	arison with Alternative Approaches	64
		2.4.1	Ex-post measures: transient and chronic poverty	64
		2.4.2	Ex-ante measures: risk and vulnerability	66
	2.5	Empir	ical Implementation and Application to Argentina	69
		2.5.1	Empirical implementation: alternative evaluation	
			functions	69
		2.5.2	Application to Argentina	73
	2.6	Conclu	usion	80
3	Risk	c Adjus	ted Poverty in Argentina	81
	3.1	Introd	uction \ldots	81
	3.2	Metho	dology	83
		3.2.1	Risk adjusted income	83
		3.2.2	Identifying assumption and risk adjusted poverty	
			measures	84
		3.2.3	Determinants of risk and risk adjusted income	88
	3.3	Risk A	djusted Poverty: Aggregates and Determinants	89
		3.3.1	Descriptive results	89
		3.3.2	Regression analysis	94
	3.4	Conclu	usion	103
4	Cha		d Tasa si an t Desserter in Tashadan t Timos	105
4		onic and	a Transfent Poverty in Turbulent Times	105
	4.1	Introa		105
	4.2	Metho	$\mathbf{aology} \dots \dots \dots \dots \dots \dots \dots \dots \dots $	106
		4.2.1	Definitions of transient and chronic poverty	106
		4.2.2	Determinants of poverty	109
	4.3	Empir	1cal Kesults	111
		4.3.1	Population groups and poverty decompositions	111
		4.3.2	Kegression analysis	115
	4.4	Conclu	ision	120

II Fertility and Women's Labour Supply in Argentina: Identification of Causal Effects Through Sex Preferences 122

5	Effects of Fertility on Women's Labour Supply: Methodology, Es-	-
	timation Strategy and Data for Argentina	123

CONTENTS

	5.1	Introd	luction to Part II: Poverty, Fertility and Women's La-	
		bour S	Supply	123
	5.2	Theor	y and Methodology	126
		5.2.1	Theoretical framework: fertility and labour supply	126
		5.2.2	The potential outcomes framework	130
		5.2.3	Identification by instrumental variables: LATE and	
			the Wald estimator	132
	5.3	Fertili	ty, Women's Labour Supply and the "Same Sex" Esti-	
		matio	n Strategy	138
		5.3.1	Fertility and women's labour supply: endogeneity	
			and selection issues	138
		5.3.2	Identification through natural experiments	139
		5.3.3	The "same sex" strategy: sex mix as a natural experi-	
			ment	140
	5.4	Data a	and Summary Statistics	142
		5.4.1	Description of the main dataset	142
		5.4.2	Variables and descriptive statistics	144
	5.5	Concl	usion	146
6	Sex	Prefere	ences and Fertility in Argentina	148
	6.1	Introd	luction	148
	6.2	Validi	ity of the Identification Strategy in Developing Countries	149
		6.2.1	First stages: nature of sex preferences and the same-	
			sex effect	149
		6.2.2	Sex preferences and the identifying assumption	151
		6.2.3	Direct effects of same sex	154
	6.3	Sex Pr	references and Identification Assumptions in Argentina .	155
		6.3.1	Independence of the instrument and fertility effects of	
			sex preferences	155
		6.3.2	Institutional setting and effects of son preferences	161
		6.3.3	Identification concerns beyond sex preferences: direct	
			effects of "same sex"	167
	6.4	Concl	usion	169
	East	·1:		150
1	ren		la women's Labour Supply in Argentina	170
	7.1		and True Steve Level Courses Estimates	170
	1.2		and Two-Stage Least Squares Estimates	172
		7.2.1	First stars and transition has the stars and the stars and	1/2
		7.2.2	First stages and two-stage least squares: benchmark	100
	-	4 1 1.	results	176
	7.3	Addit		183
		7.3.1		183
		7.3.2	Heterogeneous effects: results by education level	186
	7.4	Concl	usion	188

CON	TEN	JTS
-----	-----	-----

8	Pote 8.1 8.2	IntroductionIntroductionIntroduction"Same sex" Compliers and the Generality of LATE Results.8.2.1"Same sex" compliers and extrapolation.8.2.2Proportion of compliers in different samples.8.2.3Identification of outcomes for compliers.	190 190 191 191 194 195
	8.3	 8.3.1 Decomposition of LATE estimates by potential out- comes 	198 198
	8.4	8.3.2 Are compliers different? Extrapolation and homo- geneity assumptions	201 203
Сс	Cone Cone Cone	usions clusion to Part I	205 205 212
A	218		
-	-		
A	Abb	previations and Notation	218
A B	Abb Arge B.1 B.2	ereviations and Notation entina's Statistical Regions and Urban Areas Regions and Urban Areas Regional Heterogeneity	218 221 221 223
A B C	Abb Arge B.1 B.2 Deta	ereviations and Notation entina's Statistical Regions and Urban Areas Regions and Urban Areas	 218 221 221 223 225
A B C D	Abb Arge B.1 B.2 Deta Labo D.1 D.2	previations and Notation entina's Statistical Regions and Urban Areas Regions and Urban Areas Regional Heterogeneity ailed Poverty Figures, Argentina 1995-2002 our Supply: Robustness Checks Robustness Checks with Different Covariates, Samples and Variable Definitions Estimation with an Alternative Dataset	 218 221 221 223 225 232 232 236
A B C D	Abb Arge B.1 B.2 Deta Labo D.1 D.2 Com	ereviations and Notation entina's Statistical Regions and Urban Areas Regions and Urban Areas Regional Heterogeneity ailed Poverty Figures, Argentina 1995-2002 our Supply: Robustness Checks Robustness Checks with Different Covariates, Samples and Variable Definitions Estimation with an Alternative Dataset Automation of Results for Compliers	 218 221 221 223 225 232 232 236 241
A B C D	Abb Arge B.1 B.2 Deta Labo D.1 D.2 Com E.1 E.2	oreviations and Notation entina's Statistical Regions and Urban Areas Regions and Urban Areas Regional Heterogeneity ailed Poverty Figures, Argentina 1995-2002 our Supply: Robustness Checks Robustness Checks with Different Covariates, Samples and Variable Definitions Estimation with an Alternative Dataset number of Results for Compliers Computation of Outcomes for Compliers by Regression Methods Computation of the Auxiliary Test for Compliers by Regres-	 218 221 223 225 232 232 236 241
A B C D	Abb Arge B.1 B.2 Deta Labo D.1 D.2 Com E.1 E.2	Previations and Notation entina's Statistical Regions and Urban Areas Regions and Urban Areas Regional Heterogeneity ailed Poverty Figures, Argentina 1995-2002 our Supply: Robustness Checks Robustness Checks with Different Covariates, Samples and Variable Definitions Estimation with an Alternative Dataset number of Results for Compliers Computation of Outcomes for Compliers by Regression Methods Computation of the Auxiliary Test for Compliers by Regression	 218 221 221 223 225 232 236 241 241 242

LIST OF TABLES

1.1 1.2 1.3	Unweighted Sample Sizes for National Cross Sections, EPH23Rotating Sample: Cohorts and Waves in the GBA Panel24INDEC Equivalence Scale: Caloric Needs by Age and Gender26and Equivalent Adults26
2.1	Relative Variability Premium by Quintile of Mean Income, Isoelastic Evaluation Function with Aversion Parameter=2, Greater Buenos Aires, 1995-2002
2.2	Variability Adjusted Income for Different Values of the Dis- count Factor, Isoelastic Evaluation Function with Aversion Parameter=2
3.1	Summary Statistics for the Dependent and Independent Vari- ables, Risk Adjusted Income Regressions
3.2	Determinants of Log Average Income and Log Risk Adjusted Income
3.3	Difference in Coefficients of Determinants of Log Average In- come and Log Risk Adjusted Income for Different Values of Risk Aversion
4.1	Population Groups by Mean Income and Persistence of Po- verty Status by Cohort, Greater Buenos Aires, 1995-2002 112
4.2	Decomposition of Squared Poverty Gap, Greater Buenos Aires, 1995-2002
4.3	Censored Quantile Regressions for Total, Chronic and Tran- sient Poverty, Greater Buenos Aires, 1995-2002
5.1	Descriptive Statistics of Variables of Interest
6.1	Differences in Selected Characteristics by "Same Sex" Indicator 156
6.2	Parity Progression Ratios by Parity and Sex Mix of Children, Married Women
6.3	School Enrolment, Children Aged 6 to 12
6.4	Difference in Budget Shares, Income and Expenditure by Sex Composition of Children

LIST OF TABLES

7.1	Wald Estimates of the Effect of Fertility on Women's Labour
7.2 7.3	First Stages: Effect of Sex Composition on Further Childbearing179 OLS and Two-Stage Least Squares Estimates of Fertility and
7.4	Women's Labour Supply
7.5	Heterogeneous Effects: Results by Education Level 187
8.1	Effect of "More than two children" on "Worked for pay": Expected and Potential Outcomes by Education Level, Married
8.2	Effect of Training on Earnings: Expected and Potential Out- comes, JTPA Participants
A.1 A.2 A.3	List of Abbreviations
B.1	Poverty, Income and Labour Market Indicators by Region, Argentina, October 1998
C.1	Detailed Prices, Poverty Lines and Income Figures, Urban Argentina, 1995-2002
C.2	Detailed Poverty Figures, Urban Argentina, 1995-2002 227
C.3	Poverty Profile, Urban Argentina, October 1995
C.5	Poverty Profile, Urban Argentina, October 2001
C.6	Poverty Profile, Urban Argentina, May 2002
D.1	Robustness Checks: OLS and Two-Stage Least Squares Re- sults with Geographic Controls
D.2	Robustness Checks: OLS and Two-Stage Least Squares Re- sults with Additional Covariates
D.3	Robustness Checks: Alternative Dependent Variable and Different Sample Definitions
D.4	OLS and Two-Stage Least Squares Results, 1987 Expenditure Survey

LIST OF FIGURES

1.1	Consumer Price Index, Argentina, 1990-2003	19
1.2	Yearly Change in Real GDP and Key Economic Events, Ar-	
	gentina, 1990-2003	21
1.3	Unemployment Rate, Real and Nominal Equivalised House-	
	hold Income, Urban Argentina, 1995-2002	30
1.4	Regional Poverty Lines, Urban Argentina, 1995-2002	32
1.5	Equivalised Household Income in Real Terms (Sept. 2001 pe-	
	sos) for Individuals, Regions, Urban Argentina, 1995-2002	34
1.6	Poverty and Extreme Poverty Headcounts, Individuals and	
	Households, Urban Argentina, 1995-2002	37
1.7	Poverty Headcount by Region, Individuals, Urban Ar-	
	gentina, 1995-2002	39
1.8	Proportion of Individuals as a Function of Previous Poverty	
	Status, Urban Argentina, 1995-2002	41
•		
2.1	Ex-Ante Risk and Ex-Post Variability: States of the World and	
	Realised Incomes	55
2.2	Stability Equivalent Income and Variability Premium	62
2.3	Transient, Chronic and Variability Adjusted Measures of Po-	
	verty	65
2.4	Poverty, Vulnerability and Income Fluctuations – Cardinal	
	and Money Metric Measures	68
2.5	Evaluation Function Contours for Different Degrees of Vari-	
	ability Aversion	72
2.6	Variability Adjusted Measures of Income for Different Evalu-	
	ation Functions, Greater Buenos Aires, 1995-2002	74
2.7	Variability Adjusted Squared Poverty Gap with Different	
	Evaluation Functions, Greater Buenos Aires, 1995-2002	77
3.1	Risk Adjusted Normalised Income by Cohort. Greater	
	Buenos Aires, 1995-2002	91
3.2	Risk Adjusted Measures of Poverty by Cohort. Greater	
	Buenos Aires, 1995-2002	93
		-

LIST OF FIGURES

4.1	Population Groups by Mean Income and Persistence of Po- verty Status
4.2	Decomposition of Total Poverty in Chronic and Transient Components, Greater Buenos Aires, 1995-2002
5.1	Female Labour Force Participation (Women Aged 14 and Older) and Fertility (Children per Woman), Argentina 1960-2010
6.1	Sex Ratios – Number of Boys / Number of Girls, 0 to 4 Years Old, Selected Countries, 1990
B.1	Map of Argentina's Statistical Regions, INDEC

INTRODUCTION

This thesis presents methodological developments and empirical findings on the economics of poverty and labour markets, using data from Argentina in the 1991-2002 period for empirical applications. It is divided into two Parts, which deal with related issues.

The Chapters in Part I focus on poverty as a dynamic phenomenon, motivated by the recurring economic crises that affect developing countries and the incidence of income fluctuations on household welfare. In the Introduction to a pioneering volume on panel data on incomes, Atkinson and Cowell (1982) noted that these datasets "provide a unique opportunity" for modelling the instability of family well-being and the incidence of poverty, among other economic phenomena. While the increasing availability of household panel data has been exploited in theoretical analysis and empirical applications, the methodological and applied literatures still lack a unified framework. Echoing Atkinson (1987), Part I addresses the question of how poverty should be measured *over time* – or, in more general terms, how to measure well-being based on repeated observations of household income. The aim is to develop and illustrate a set of tools for empirical work based on theoretically sound extensions of the existing methodology for static distributional analysis. Since the proposed family of measures does not rely on a specific functional form, the framework developed in the following pages encompasses some of the existing approaches as special cases.

These tools are illustrated with longitudinal data for Argentina in the 1995-2002 period, which is well suited for this type of analysis given the large fluctuations in household income. As described in Chapter 1, during the 1990s the country's economy underwent a process of market-oriented structural reforms. The resulting openness of the economy and the hard peg of the local currency to the US dollar contributed to a high degree of

INTRODUCTION

vulnerability to the succession of international financial crises of the second half of the decade, which was characterised as a period of "boom and bust." This series of external macroeconomic shocks and the weaknesses of the Argentine economy led to a severe economic and social crisis that started at the end of 2001 and continued well into 2002. Chapter 1 describes the trends in poverty and its dynamics during this period, and sets up the basis for the analysis of the following three Chapters: it discusses the methodological issues of poverty measurement in Argentina, introduces the main dataset employed in Part I (a rotating panel from the Greater Buenos Aires region), and covers the existing literature on poverty in Argentina.

Chapter 2 presents a general framework for the evaluation of well-being based on panel data on incomes. The methodology relies on a formal analogy with the theory of choice under uncertainty, and on the existing literature on distributional analysis. It defines a family of indicators that account for the negative impact of income fluctuations on household welfare. The methodology is illustrated with the Greater Buenos Aires panel. The following Chapters present two examples of more detailed empirical analysis, which can be interpreted as special cases of the evaluation framework. Chapter 3 accounts for risk and the effect of fluctuations on household welfare by defining risk adjusted measures of income. The evolution of these indicators is complemented by a regression analysis, which sheds light into the household characteristics associated with the impact of risk on wellbeing. Finally, Chapter 4 provides a decomposition of poverty into its transient and chronic components, and studies their correlates by means of regression analysis.

The research presented in Part II is motivated by some of the stylised facts in distributional analysis that emerge from regressions of this type. While most poverty profiles find strong links between poverty, childbearing and labour market status, these links cannot be interpreted as causal relationships given the problems of endogeneity and simultaneity that are pervasive in applied economic research.

The Chapters in Part II establish the determinants of the income-generating process of women as primary and secondary earners in the household. Specifically, these Chapters make a series of methodological and empirical contributions by studying the causal effect of fertility on female labour sup-

INTRODUCTION

ply from an empirical perspective, based on data from the 1991 Argentine Census.

Chapter 5 presents the theoretical and econometric framework employed to deal with the endogeneity of the fertility decision. The simultaneity problems in the economic analysis of fertility and labour supply arise from the fact that the two are joint decisions: while children provide utility to their parents, they enter the household's budget constraint through the costs in terms of goods and time. Moreover, fertility and labour supply decisions might be driven by unobservable factors, such as preferences and ability, which imply selection problems in the estimation. The identification strategy in Part II exploits parental sex preferences as instrumental variables for childbearing, based on the observation that parents of two children of the same sex exhibit a higher propensity to have another child to obtain a gender-balanced offspring composition.

However, the nature of sex preferences in developing countries poses some challenges to the application of this strategy to the Argentine case: in particular, a stronger preference for boys might have implications in terms of lifetime income and labour supply, invalidating the postulated exogeneity of the instrument. Chapter 6 discusses these problems in the context of developing countries. It provides original evidence in support of the validity of the identification strategy for Argentina, studying the plausibility of the untestable identifying assumptions by means of auxiliary evidence. Based on these results, Chapter 7 presents the instrumental variable estimates of the causal effect of fertility on labour supply. Finally, Chapter 8 proposes a new test for the generality of instrumental variable results, which is illustrated with the same dataset.

The Conclusion reviews the main contributions of both Parts, highlighting their implications for theoretical analysis, empirical application and policy formulation, and outlining possible avenues for further research.

Part I

Poverty and Income Fluctuations, Argentina 1995-2002

CHAPTER 1

ARGENTINA'S CRISES AND THE POOR

1.1 INTRODUCTION TO PART I: Key Economic Events, Argentina 1991-2002

The objective of this introductory Chapter is to set up the context for the rest of Part I, which presents original developments in the measurement of poverty and well-being over time. The main motivation is the effect of the recurring economic crises that affect developing countries and the incidence of income fluctuations on welfare. For this reason, the empirical applications of this Part exploit the longitudinal dimension of household survey data for Argentina in the 1995-2002 period. Since the empirical work of Part II is based on Argentine data from 1991, the following pages present the main economic events of the 1991-2002 period.¹

The 1980s represented a "lost decade" for most of Latin America. In the case of Argentina, the decade ended in political instability and a series of hyperinflation episodes that extended into 1990 and 1991, as can be appreciated in the evolution of the Consumer Price Index (CPI), presented in Figure 1.1. In March 1991, the country adopted the "Convertibility Plan," a currency board where the Argentine peso was pegged to the US dollar. This plan was accompanied by a series of market oriented structural reforms that included privatisations of public utilities and the opening of the economy to flows of goods and capital. These reforms and the liquidity of international credit markets prompted a steady inflow of capital, which sustained growth between 1991 and 1994. This is depicted in Figure 1.2, which presents the

¹This summary draws on Lewis (2002), Bonvecchi (2003) and Gerchunoff and Llach (2004a), which are recommended for further reference on the economic history and the political economy of Argentina.

CHAPTER 1. ARGENTINA'S CRISES AND THE POOR



Figure 1.1: Consumer Price Index, Argentina, 1990-2003

evolution of Gross Domestic Product (GDP) annotated with the main economic events of the 1990-2003 period for Argentina.

The Convertibility's currency board, however, made the economy highly vulnerable to external shocks. At the end of 1994, Mexico devalued its currency and triggered an international financial crisis (the "Tequila crisis") that affected most emerging markets. Argentina rapidly suffered from contagion, with runs against the peso and significant capital flights, but the currency board resisted this large external shock. As shown in Figure 1.2, a brief recovery followed during 1996 and 1997, but the economy was hit again by the economic crisis in South-East Asia, which started in Thailand in early 1997, and its aftermath. Russia, in turn, entered a severe financial crisis in August 1998, and its sovereign debt default prompted yet another contagion to Argentina, which was – like most emerging markets – badly hit by capital outflows and interest rate rises. In January 1999, Brazil, Argentina's main trading partner, was forced to devalue its currency, worsening a recession that started around 1998.

After three years of negative growth, the Argentine government was

forced to impose restrictions on bank accounts in late 2001. This precipitated events, which converted the recession into an economic meltdown. The decision to freeze bank deposits was followed by social unrest and political instability, and lead to the resignation of President De La Rua. The currency board could not be sustained in this context, and in early 2002 the new government put an end to the parity between the peso and the US dollar and announced a default on the country's sovereign debt. The subsequent fall in confidence and the disruption of productive activity resulted in a fall in GDP of 10.9 per cent during 2002.

The sources of the crisis can be traced back to, among other factors, the exchange rate parity with the US dollar and the resulting over-valuation of the local currency, the vulnerability of the country to external shocks, and the economy's own structural weaknesses (Galiani et al., 2003).

As discussed in detail in this Chapter, the crisis was reflected in the poverty rates, which reached 53 percent of the population in May 2002, 15 percentage points higher than in October 2001. This increase in poverty mirrored a large fall in household income, caused by two factors (World Bank, 2003). On the one hand, labour market conditions deteriorated sharply: in May 2002, the unemployment rate exceeded 21 percent, more than 3 percentage points higher than in October 2001 (see the discussion of Figure 1.3 below). On the other hand, real incomes fell because of the large increase in consumer prices induced by the devaluation (Figure 1.1).

The rest of this Chapter is organized as follows. Section 1.2 introduces the survey data and the income aggregate. It also discusses methodological issues on poverty measurement in Argentina, and proposes an extension to the official methodology that allows the construction of consistent national poverty figures for the 1995-2002 period. The Section also describes a rotating panel dataset from the Greater Buenos Aires region, which is used extensively in the following Chapters. Section 1.3 builds on the data and income aggregate to present the main trends of poverty during the period at the national and regional levels. It also provides a presentation of the short term poverty dynamics. Finally, Section 1.4 briefly covers the existing literature on poverty in Argentina for this period. Conclusions follow.



Figure 1.2: Yearly Change in Real GDP and Key Economic Events, Argentina, 1990-2003

21

1.2 HOUSEHOLD DATA AND MEASUREMENT OF POVERTY IN ARGENTINA

1.2.1 National cross sections and the Greater Buenos Aires rotating panel

The empirical analysis conducted in Chapters 1 to 4 (Part I) is based on data from the Argentine Permanent Household Survey ("Encuesta Permanente de Hogares," EPH). This is a labour market and living conditions survey that has been collected since 1975 in the Greater Buenos Aires region, which covers the country's capital and adjacent municipalities, and constitutes the country's largest urban centre. The EPH is one of the longest running household surveys in Latin America, and is considered to be of relatively high quality (World Bank, 2000a). The data is collected by the national statistical agency, the Instituto Nacional de Estadísticas y Censos (INDEC), which is responsible for household, expenditure and manufacturing surveys, the national Census, price indices and the national accounts.

During the 1980s and 1990s, other cities were added to the EPH's sampling frame, reaching a maximum of 28 urban centres in 1995 (INDEC, 1996). This made the EPH representative of the urban population of the country – about 80 percent of the total. While some of the estimates in this Part will be referred to as "national," it should be stressed that the results correspond to the urban areas covered by the EPH and are representative at the urban level only.² Relatively little is known about the small but not negligible fraction of the population residing in rural areas. Fiszbein et al. (2002), discussed in Section 1.4, provide some limited evidence on living conditions in these areas. The main urban centres covered by the EPH are divided by INDEC into six statistical regions, and it is thus assumed that the cities and provinces within each region share some structural characteristics. The regions are Greater Buenos Aires, Pampeana, Noreste, Noroeste, Cuyo and Patagonia. Appendix B lists the main urban centres covered by the EPH, and places the regions in a map of the country.

During the 1995-2002 period, the survey was collected every year in two waves, in May and October (denoted waves 1 and 2 for each year), and all estimates in this Part are based on the fifteen waves available between May

²The empirical application of Part II is based on the 1991 Census and thus covers urban and rural areas.

Wave	Households	Individuals
May 1995	22,023	82,721
October 1995	29,509	112,439
May 1996	29,861	113,209
October 1996	27,200	103,847
May 1997	29,506	110,487
October 1997	29,362	109,307
May 1998	28,511	105,399
October 1998	26,810	99,035
May 1999	24,918	92,371
October 1999	24,715	91,512
May 2000	22,834	83,571
October 2000	22,763	83,583
May 2001	22,833	83,264
October 2001	22,998	83,988
May 2002	22,814	83,349
Source: EPH household	survey data (INDEC).	

Table 1.1: Unweighted Sample Sizes for National Cross Sections, EPH

1995 and May 2002. The EPH is structured as a rotating sample, where 25 percent of households surveyed are replaced in each wave, and the data is treated as a series of repeated cross sections (INDEC, 2002). The INDEC provides household weights, which are used in all the estimates presented in this Chapter. Table 1.1 presents the unweighted sample sizes for the fifteen waves, both in terms of the number of individuals and the number of households surveyed.

The rotating structure of the EPH's sample implies that households stay in the sample for four consecutive waves, a period of about a year and a half. A consistent series of panels, however, can only be constructed for the Greater Buenos Aires region (GBA), since for other urban centres INDEC did not release all the matching codes, and changes were made to the sample rotations. The Greater Buenos Aires region represents around 60 percent of the total population and 70 percent of the urban population of the country.

The fifteen waves between May 1995 and May 2002 contain data for twelve "cohorts" of households observed in the same four consecutive waves. Table 1.2 illustrates the structure of the panels and clarifies the distinction between waves and cohorts. Only households observed four times and with complete information on income for every member of the household in the four waves are kept in the sample, which results in an average of 453

	Wave:	95-1	95-2	96-1	96-2	97-1	97-2	98-1	98-2	99-1	99-2	00-1	00-2	01-1	01-2	02-1
	Cohort:															
1	95-1 to 96-2	I	1	1	1											
2	95-2 to 97-1		2	2	2	2	\wedge									
3	96-1 to 97-2			3	3	3	[3]									
4	96-2 to 98-1				4	4	4	4								
5	97-1 to 98-2					5	5	5	5							
6	97-2 to 99-1						6	6	6	6						
7	98-1 to 99-2							7	7	7	7					
8	98-2 to 00-1						Wave		8	8	8	8	Coho	rt		
9	99-1 to 00-2									9	9	9	9			
10	99-2 to 01-1										10	10	10	10		
11	00-1 to 01-2											11	11	11	11	
12	00-2 to 02-1												12	12	12	12

Table 1.2: Rotating Sample: Cohorts and Waves in the GBA Panel

retained households per cohort – about 60 percent of the theoretical total for the GBA region. Cruces and Wodon (2003c) argue that the attrition from the panel is compensated by the INDEC's weighting structure, and does not bias income and poverty measures in a significant way. Moreover, given the relatively short span of the panels, the problems identified by Cowell (1982) with respect to changes in family structure do not affect the results. Cruces et al. (2004) discuss alternatives for dealing with these issues in a long panel.

1.2.2 Income aggregate and equivalence scale

The EPH collects information on the income and labour market status of every member of a household, as well as some dwelling and individual characteristics. To obtain results which are comparable to official figures and to the literature on poverty and labour economics in Argentina, this Chapter employs INDEC's methodology for the computation of household income aggregates, which is critically assessed below.

The aggregation of income at the household level is not a trivial task, as witnessed by the long discussion in the poverty literature (Ravallion, 1994; Deaton, 1997, cover these debates and most of the issues raised in this Section). While some authors base their estimates on per capita income, this is problematic as an indicator of well-being because it does not allow for economies of scale in the household, nor for differences in needs between members of different age and gender. Ignoring these aspects may result in an over-estimation of the negative impact of household size on poverty (Coulter et al., 1992; Lanjouw and Ravallion, 1995) and inequality (Cowell and Mercader Prats, 1999).

Following Cowell (2000, Section 2), total family income ψ_i must be adjusted by the characteristics of the household and its members by means of an equivalence scale. This process results in equivalised income y_i^e , which represents a suitable money metric of utility. The procedure of equivalisation is defined by a function χ such that:

$$y_i^e = \chi(\mathbf{a}_i, \psi_i)$$

where a_i represents a list of demographic and other attributes of the household and its members. Equivalised income is often obtained as

$$y_i^e = \frac{\psi_i}{\nu(\mathbf{a}_i)} \tag{1.1}$$

where $v(\mathbf{a}_i)$ is a function that defines the number of equivalent adults in the household. INDEC follows this methodology in the construction of its income aggregate, and accounts for differences in needs between household members by adopting an equivalence scale based on calory intake by age and gender (INDEC, 2002). Specifically, the number of equivalent adults for each household *i* with k_i members is defined as $v(\mathbf{a}_i) = \sum_{j=1}^{k_i} q_j$, where each q_j is determined by member *j*'s age and gender. These adult equivalent coefficients q_j are given in Table 1.3 (reproduced from Morales, 1988) for different categories. The coefficients are defined as the ratio of each category's caloric intake requirements to those of men aged 30 to 59 (2,700 kcal per day), which are defined as an equivalent adult. For instance, a three year old girl requires 1,500 kcal per day, and thus represents 0.56 of an equivalent adult. INDEC's adjustments for differential needs are within the range of those employed in the distributional literature, and they are robust to adjustments for economies of scale in the household.³

³While INDEC choses not to adjust household income for economies of scale, this is relatively straightforward to implement by means of a parameter *s*, $0 \le s \le 1$, with each extreme representing full and no economies of scale respectively. The number of adult equivalents is then computed as $[\sum_{j=1}^{k_i} q_j]^s$, where q_j is the coefficient in Table 1.3 and *k* is the size of the household. This is not implemented in this Chapter to maintain compatibility with official statistics and existing academic work. Gasparini (2003a) conducts a sensitivity analysis with EPH data and finds that most income and poverty measures are robust to

Age	Gender	Calories	Units per equi-
Up to 1 year old	<u> </u>	1,170	0.43
2	Girls	1,360	0.50
3	and	1,500	0.56
4 to 6	Boys	1,710	0.63
7 to 9		1,950	0.72
10 to 12		2,230	0.83
13 to 15	Men	2,580	0.96
16 to 17		2,840	1.05
10 to 12		1,980	0.73
13 to 15	Women	2,140	0.79
16 to 17		2,140	0.79
18-29		2,860	1.06
30-59	Men	2,700	1.00
60 and +		2,210	0.82
18-29		2,000	0.74
30-59	Women	2,000	0.74
60 and +		1,730	_0.64
Source: Morales (1988). The adult equivalent units are defined as the			
ratio of the age-gender category caloric needs with respect to those of an			
adult male aged 30 to 59.			

Table 1.3: INDEC Equivalence Scale: Caloric Needs by Age and Gender and Equivalent Adults

To complete the definition of y_i^e in Equation 1.1, total household income is computed by INDEC for each household *i* with k_i members as $\psi_i = \sum_{j=1}^{k_i} y_i^j$, where y_i^j represents each individual member's total monetary income. Most individuals have only one source of income which consists of salaries for the active population and pensions for those who are retired. Combining this figure and the number of equivalent adults in the household given by the function ν , total household equivalent income is defined by the following expression:

$$y_{i}^{e} = \frac{\sum_{j=1}^{k_{i}} y_{i}^{j}}{\sum_{j=1}^{k_{i}} q_{j}}$$
(1.2)

This aggregate is attributed to every member of the household, which is why the text refers interchangeably to households and individuals. By adjusting for differences in household composition, y_i^e represents a better measure of well-being than per capita income (Deaton, 1997).

reasonable deviations from INDEC's implicit choice of s = 1.

As an index of income, INDEC's y_i^e satisfies the basic criteria of measurability and comparability among different individuals (Cowell, 1995, Chapter 1). However, the Chapters in this Part deal with observations spanning the period 1995 to 2002, and y_i^e as defined in Equation 1.2 is not comparable across regions or in different periods if prices differ geographically or over time. While it is possible to deflate y_i^e with respect to prices at a given period to express it in constant units (i.e., in terms of real income), this Chapter and the rest of Part I adapt INDEC's (2002) methodology to time and geographical variations by normalising the adult equivalent income normalised by the contemporaneous poverty line z_t . This income aggregate is defined as:

$$y_{it} = \frac{y_{it}^e}{z_t} = \left[\frac{\sum\limits_{j=1}^{k_i} y_{it}^j}{\sum\limits_{j=1}^{k_i} q_j} \right] / z_t$$
(1.3)

This formulation is known as the "welfare ratio" in the literature and has a number of advantages (Blackorby and Donaldson, 1987; Ravallion, 1998). In addition to making equivalised incomes comparable over time and space,⁴ Equation 1.3 can be given an interpretation in terms of poverty measurement: $y_{it} < 1$ indicates that a household's income is below the poverty line, and thus its members can be classified as poor.⁵ For these reasons, the choice of the poverty line as the unit of measurement is preferable to deflating incomes with respect to the CPI.

Finally, the quality of the income aggregate y_{it} is given by INDEC's validation process, which checks each of the components of individual income for consistency and discards households for which a complete total income cannot be computed, adjusting the sample weights accordingly. The small fraction of households reporting zero total income are kept in the final sample since they are considered valid by INDEC.

⁴As discussed in the following pages, poverty lines vary between regions, and Equation 1.3 should refer to the income of a household living in region r and the regional poverty lines z_{rt} . The notation in the text is preferred for being more compact.

⁵The denominator of the right hand side of Equation 1.3, $\tilde{z}_{it} = z_t \sum_{j=1}^{k_i} q_j$, can be interpreted as a household specific poverty line.

1.2.3 *Poverty lines and regional heterogeneity*

The rest of this Section covers the construction of poverty lines in Argentina, which establish the fundamental partition of the population between the poor and the non-poor (Cowell, 2003). For each wave of the EPH, INDEC reports the cost of an extreme (or "indigent") poverty line, z_t^I , which is based on a basic food basket (INDEC, 2002). The components of this basket are constructed from a household expenditure survey,⁶ and its cost is updated with changes in prices between each wave of the EPH.

The poverty line z_t is derived from the basic food basket z_t^l using the inverse of the Engel coefficient to incorporate the cost of basic non-food goods.⁷ The idea behind this approach is that the extreme poor cannot afford a minimum food basket, whereas the moderately poor, while able to cover their basic nutritional needs, cannot afford other essential goods and services. While other alternatives for the definition of poverty lines exist, this Chapter follows the literature on distributional analysis in Argentina and adopts INDEC's methodology to ensure comparability with previous results.

As argued in the discussion of Equation 1.3, normalising incomes by the poverty lines facilitates comparisons over time. Poverty lines are also used in distributional analysis to reflect geographic heterogeneity in prices and in the purchasing power of income, by means of region and time specific lines z_{rt} .

INDEC's geographic coverage, however, is incomplete for most of the 1995-2002 period. While the EPH is collected in the major urban areas of the country, INDEC only constructed poverty lines for the Greater Buenos Aires region. From that wave onwards, INDEC provides official poverty lines for the six statistical regions based on the purchasing power parity of income. The lack of regional poverty lines implies that INDEC only computed official poverty statistics for the GBA region before 2001, even though the regional survey data is available for the 1995-2000 period. These statistics are presented in Appendix C (Table C.1, page 226).

This Chapter proposes a simple method to fill in the gaps in regional esti-

⁶This survey, the "Encuesta Nacional de Gasto de los Hogares," is carried out only every ten years by INDEC. Part II presents some results based on it.

⁷Ravallion (1998) and INDEC (2002) discuss this procedure in detail.

mates. The aim is to complement INDEC's methodology to provide regional and nationwide poverty measures for the period 1995-2000 accounting for the geographic heterogeneity in costs of living. The proposed method relies on an updating rule. The rate of change of the GBA poverty line over the period May 1995-May 2001 represents an implicit GBA poverty line deflator. This deflator can be applied retrospectively to the five remaining official poverty lines of May 2001, applying the GBA rate of change to obtain regional values for the previous waves. The official poverty and extreme poverty lines, and those obtained with the updating procedure, are presented in Appendix C (Table C.1).

A drawback is that this simple rule implicitly assumes that the change in poverty lines over time in all regions is the same as that of the GBA region. While this appears to be a strong assumption, the changes in official poverty lines from May 2001 to May 2002 are strongly correlated across regions, as shown in Table C.1 and depicted in Figure 1.4. The main advantage of this simplifying assumption is that it provides consistency and continuity with INDEC's estimates for 2001 and onwards.⁸

The income aggregate and poverty lines described in this Section are used in the following pages to estimate poverty figures for each region and the country as a whole over the period 1995-2002, complementing the official statistics for 2001-2002.

1.3 POVERTY TRENDS AND SHORT TERM DYNAMICS

1.3.1 Income, prices and poverty lines

This Section presents the main trends in poverty for the period 1995-2002. It first discusses the evolution of the income aggregate and the poverty lines, and then describes the evolution of regional and national poverty figures. The final subsection deals with the short term dynamics of poverty over this period.

The impact of the series of crises and recoveries described in the Introduction to this Chapter can be appreciated in the evolution of the unem-

⁸Cruces and Wodon (2003a) discuss this procedure in more detail, and compare it to the approach developed by Lee (2000). While the latter is more sophisticated, it does not ensure a continuity between the pre- and post-May 2001 measures.



Figure 1.3: Unemployment Rate, Real and Nominal Equivalised Household Income, Urban Argentina, 1995-2002

Source: Author's estimations based on EPH household survey data (INDEC).

30

ployment rate, which mirrors the changes in household income and GDP, as depicted in Figure 1.3. In the aftermath of the Mexican crisis, the unemployment rate reached 18.8 percent in May 1995, and remained high until October 1996. It fell to 12.4 percent in October 1998 during the recovery, but from that wave onwards it increased again, reaching the highest recorded rate of 21.5 percent in May 2002, with the largest rise between waves of over 3 percentage points between October 2001 and May 2002.

Figure 1.3 also depicts the evolution of the sample average of y_{it}^e , the adult equivalent income defined in Equation 1.2, in nominal (current pesos) and real terms, for all the large urban areas covered by the EPH. Real values are adjusted by the CPI (Figure 1.1) and correspond to September 2001 prices.⁹

The y_{it}^e aggregate decreased almost 4 percent from May 1995 until October 1996, as a consequence of the contraction that followed the contagion from the Mexican crisis (Figure 1.2). Household income recovered briefly until May 1998, but from then on it fell almost continuously in both nominal and real terms, with the exception of a brief recovery between May and October 2000. The sharpest decrease corresponds to the crisis of 2001-2002, as captured by the May 2002 wave of the EPH (the last in the Figure).

Between October 2001 and May 2002 nominal income fell 10 percent, but it is interesting to note in Figure 1.3 that the sudden surge in consumer prices caused by the crisis of 2001-2002 and depicted in Figure 1.1 was reflected in a much larger fall in real income of 25 percent. This increase in prices is also reflected in Figure 1.4, which shows the evolution of the official and constructed regional poverty lines along with the Consumer Price Index.

The poverty lines followed the trend in the CPI for most of the period: for instance, from October 1998 to October 2001 the poverty lines and the CPI fell in a similar manner, reflecting the deflationary pressures of the period's recession, which also explains why nominal income is above real income in Figure 1.3. Between the last two waves, corresponding to the 2001-2002 crisis, both the CPI and the nominal poverty lines grew sharply. The two effects do not cancel out completely: the Figure also portrays the evo-

⁹The CPI, poverty lines, equivalent income and poverty estimates are available in Tables C.1 and C.2 in Appendix C.





CHAPTER 1. ARGENTINA'S CRISES AND THE POOR

32

Source: Author's estimations and official figures by INDEC (2002).

lution of the GBA poverty line in real terms (bold line), which increased by 7.5 percent in real terms from October 2001 to May 2002. This reflects the fact that the rise in the cost of goods consumed by the poor was larger than the overall increase in the CPI, hitting them more than the rest of the society. The Figure also illustrates the regional differences in living costs: the GBA and Patagonia regions have higher poverty lines than the rest of the country for the whole period, while the Noroeste region always has the lowest value.¹⁰

While Figure 1.4 indicates some geographical heterogeneity in living costs, Figure 1.5 depicts the trend of real adult equivalent income y_{it}^e in September 2001 pesos by region, showing the substantial regional differences in living standards. The urban centres in the GBA and Patagonia regions have the highest levels of income, staying above the national average over the whole period. Those below the average are, in descending order, the Pampeana, Cuyo, Noroeste, and Noreste regions. While there are large regional differences in levels of income – for instance, the average over time for the Noreste region is just above half of that of the Patagonia region – the time trends are nevertheless similar for all the regions. The regions with higher and lower incomes are also those with higher and lower poverty lines, and it is not clear a priori which of the two effects prevails in the resulting poverty measures. This is covered in the following pages.

1.3.2 Poverty trends

Before discussing the poverty figures for 1995-2002, this Section presents the methodology followed by INDEC to compute its official statistics, which is adopted in this Chapter to provide comparable results.

According to Cowell (2003), the measurement of poverty requires three components: an income estimate, a poverty line, and a measure or index, which represents a device for aggregating the poverty evaluations obtained at the household level into a population figure. The first two components were covered from the methodological and empirical points of view in the preceding pages. Regarding the third element, a household *i*'s poverty evaluation is given by a function defined over its equivalised income y_i^e and a

¹⁰Appendix B presents a set of regional socio-economic indicators and provides some background on the heterogeneity of Argentina's regions (Table B.1, page 223).




poverty line *z* as (Cowell, 2003):

$$\begin{cases} p\left(y_{i}^{e}, z\right) & \text{if } y_{i}^{e} < z \\ 0 & \text{otherwise} \end{cases}$$
(1.4)

The Chapters in this Part rely on the decomposable poverty measures proposed by Foster et al. (1984), which belong to the general class defined by Atkinson (1987). These FGT measures imply the following functional form for the poverty evaluation:

$$p(y_i^e, z) = \left[\frac{\max(z - y_i^e, 0)}{z}\right]^{\alpha}$$
(1.5)

where α is a sensitivity parameter ($\alpha \ge 0$).¹¹ The resulting poverty measure is given by the sample average of p, which can be represented as:

$$FGT(y^{e}, z, \alpha) = \frac{1}{N} \sum_{i=1}^{N} \left[\frac{\max(z - y_{i}^{e}, 0)}{z} \right]^{\alpha}$$
(1.6)

where *N* denotes the total number of households or individuals in the population.¹² With the parameter set to $\alpha = 0$, Equation 1.6 represents the poverty headcount. With $\alpha = 1$ and $\alpha = 2$, the resulting measures are the poverty gap and the squared poverty gap, which take into account not only the proportion of the poor in the population (as the headcount does) but also the intensity of poverty.

The following Figures were constructed using the equivalent income y_i^e in the *FGT* measures in Equation 1.6. Both the poverty lines z_t and extreme poverty lines z_t^I were allowed to vary by region.

Figure 1.6 presents the headcounts of poverty and extreme poverty (indigence) for the urban areas covered by the EPH, corresponding to *FGT* with $\alpha = 0$ (Equation 1.6).¹³ As in the official statistics provided by INDEC, both

¹¹If the income aggregate y_i^e is normalised by the poverty line as in Equation 1.3, p simplifies to $p(y_{it}, z) = [\max(1 - y_{it}, 0)]^{\alpha}$.

¹²The resulting *FGT* measure can refer to the number of households or the number of individuals in poverty. The latter is derived from computing Equation 1.6 with weights reflecting household size. This procedure is equivalent to defining *FGT* in terms of N_I , the number of individuals in the population, since the poverty status and the income aggregate are defined at the household level and then assigned to every member.

¹³This Section discusses only the evolution of the poverty headcounts. The poverty gap and squared poverty gap measures for each region are reported in Appendix C (Table C.2).

headcounts are calculated as fractions of the total number of households (share of households in poverty or extreme poverty) and the total number of individuals (share of individuals in poverty or extreme poverty).

As is usually the case, the proportion of individuals under the poverty and extreme poverty lines is always higher than the proportion of households, reflecting the fact that poor households tend to be larger than nonpoor households.¹⁴

Since the income aggregate and the poverty lines were affected by the repeated crises and recoveries of the 1995-2002 period, the measures of poverty based on those numbers were also sensitive to the evolution of the economy. During the contagion from the Mexican crisis, the poverty and extreme poverty headcounts increased significantly from May 1995 to October 1996, then fell slightly from October 1996 until May 1998.

The May 1998 wave of the EPH represents a turning point in the data: the individual-based poverty headcount grew steadily from 28.6 percent to 38.3 percent in October 2001, with a similar trend for extreme poverty and for household-based measures. This rise in poverty of almost 10 percentage points in a little over three years reflects the worsening of the labour market conditions and economic activity during the recession, depicted in Figures 1.2 and 1.3. Over the same period, the proportion of the population in extreme poverty doubled from 6.8 percent to 13.6 percent.

These increases, however, are relatively minor when compared with the changes occurring between the October 2001 and May 2002 waves of the EPH. At the national level, the individual-based poverty headcount jumped from 38.3 percent to 53 percent (13.6 to 24.8 percent for extreme poverty), with household measures following the same upward trend. This jump in poverty rates is the result of the sharp increase in prices and hence poverty lines (Figure 1.4), coupled with the fall in real and nominal income of the households (Figure 1.3).

Finally, with respect to the geographical heterogeneity previously ob-

Their trends are very similar to those observed for the headcounts.

¹⁴A full poverty profile is beyond the scope of this Chapter, which focuses on the construction of consistent poverty figures at the national level. Tables C.3, C.4, C.5 and C.6 in Appendix C present some characteristics of the samples in general and for the non-poor, the poor and the extremely poor for four of the fifteen waves under study. These descriptive statistics are for illustration only – for more detailed poverty profiles, see World Bank (2000a), World Bank (2003) and Cruces and Wodon (2003a).





sehold survey data (INDEC).

an

s) s) ndivie

ousel

37

CHAPTER 1.

ARGENTINA'S CRISES AND THE POOR

served, Figure 1.7 presents a consistent series of individual-based regional headcounts.¹⁵ While the areas with lower incomes were also those with lower poverty lines, the former prevails: the regional ranking of adult equivalent income of Figure 1.5 is reversed for poverty measures. There are significant differences in poverty rates within the country, with the GBA and Patagonia regions faring systematically better, and the North (Noreste and Noroeste regions) being consistently poorer than the rest of the country. Through most of the period under review, the ranking of the regions in terms of poverty estimates did not change, with fairly similar regional trends. However, the increase in poverty in the GBA area from October 2001 to May 2002 was larger than the rise observed in any of the other regions.

1.3.3 Poverty transitions and short term dynamics

The analysis of cross section data usually results in discussions of changes in poverty rates between two periods, as in Figure 1.6. The rotating sample structure of the EPH, however, allows for a deeper analysis in terms of the poverty transitions of households between two periods. The evidence for Argentina is depicted in Figure 1.8, which is based on a special version of the EPH dataset where households were paired in two consecutive waves, resulting in a series of two-period panels. This dataset is representative at the national level, and because of the sample rotation it contains about 70 percent of the observations listed in Table 1.1.¹⁶

Figure 1.8 presents the basic poverty transition between two waves (a six-month period). Starting from the October 1995, it decomposes the population into four transition categories, according to their current and past poverty status: the non-poor who stayed non-poor, the poor who stayed poor, the poor who escaped poverty, and the non-poor who became poor in the following period. The Figure represents a decomposition of the change

¹⁵These regional measures match the official INDEC statistics from 2001 onwards. For 1995-2000, the regional and national poverty rates were constructed using the methodology described in Section 1.2.

¹⁶This paired dataset was prepared and kindly provided by Juan Martín Moreno, from the Argentine Ministry of Labour. It should be noted that the resulting poverty rates are slightly different from the ones presented in Figure 1.6. This is due to the nature of the rotating sample: since the dataset consists of observations paired accross two waves, only a maximum of 75% of the total observations for each wave is available, as illustrated in Table 1.2.



Figure 1.7: Poverty Headcount by Region, Individuals, Urban Argentina, 1995-2002

Source: Author's estimations based on EPH household survey data (INDEC).

39

in poverty between two waves of the EPH, which is equal to those who entered poverty minus those who escaped poverty – for this reason, the latter appears as negative in the Figure.

This evidence complements the description of the main poverty trends in Figure 1.6. Excluding the change between the last two waves, which cover the unusual circumstances of the 2001-2002 crisis, the proportion of individuals switching poverty status is fairly stable throughout the 1995-2001 period. In each wave (representing a period of about six months), an average of 7 to 8 percent of the population manages to escape poverty, while an average of 8 to 9 percent of the population enters poverty. These fluctuations are relatively large when compared to the changes in the crosssectional headcounts, which were never higher than 2.4 percentage points for the same period (Figure 1.6). These large movements in and out of poverty compensate each other and result in relatively low net changes in the static cross-sectional poverty rates. Moreover, it should be noted that these large fluctuations occur even when aggregate rates were falling, as shown in Figure 1.8 for the years 1996-1998.

These relatively high and stable levels of switching in poverty status were affected by the worsening of the economic conditions over the 1995-2001 period. This is manifested in the almost continuous increase in the fraction of the population in poverty that stayed in poverty in the following wave of the survey.

Figure 1.8 also presents interesting results on the effects of the 2001-2002 crisis, which are captured by the changes in poverty status between October 2001 and May 2002. The Figure indicates that the recession and the subsequent crisis affected the persistence of poverty and not only its level. The proportion of individuals who were poor and remained poor increased from 26.4 percent between the waves of October 2000 and May 2001 to 36.6 percent between October 2001 and May 2002. Moreover, 18.3 percent of the population was non-poor in October 2001 but was recorded as poor in May 2002, a major increase when compared to the average of about 8 to 9 percent for the 1995-2001 period. This is also reflected in the proportion of the population that was above the poverty line and remained at that level, which fell from a fairly stable 60 to 65 percent to a low 42.6 percent between the last two waves.

May 97 May 98 May 99 96 Oct. 96 Oct. 97 Oct. 98 May 00 Oct. 00 May 01 Oct. 01 Oct. 99 56.5% 60.3% 59.8% 62.6% 62.6% 63.1% 61.0% 61.4% 60.3% 58.4% 54.3% 26 9.7% 7.9% 8.2% 7.9% 7.8% 8.8% 7.2% 9.1% 7.6% 9.9% 9.0% -7.7% -7.4% -9.0% -7.3% -7.8% -6.6% -7.8% -7.1% -6.7% 16 -8.9% -7.2% 22.3% 23.5% 21.9% 20.6% 21.8% 22.9% 23.6% 24.1% 26.2% 26.4% 30.1% % % 32.0% 31.4% 30.1% 28.5% 29.6% 31.7% 30.8% 33.1% 33.9% 36.4% 39.0% 54.9%

poor." The legend refers to the transition from the state in t-1 to the state in t. Note also that (5)=(2)+(4)=100-(1)+(3)

I household survey data (INDEC).



Individuals as a Function of Previous Poverty Status, Urban Argentina, 1995-2002

ARGENTINA'S CRISES AND THE POOR

CHAPTER 1.

41

Finally, Cruces and Wodon (2003a) provide evidence on movements within the poor using the same paired dataset. An average of around 3 percent of the population was found to switch between extreme poverty and poverty, and vice versa, in every wave of the EPH in the 1995-2001 period. From May 1998 onward however, there was a continuous increase in the share of individuals who were moderately poor and became indigent.

1.4 GUIDE TO THE LITERATURE ON POVERTY IN ARGENTINA

1.4.1 Income distribution and poverty in Argentina

This Section provides a background for the analysis presented in the following Chapters by covering a selection of the literature on poverty in Argentina during the 1995-2002 period. The aim is threefold. Firstly, this Section introduces some of the key contributions on distributional analysis in Argentina for further reference. It also covers some of the related issues that are not explored in the following Chapters, notably the social assistance programmes and active strategies adopted by households to face the repeated economic crises of the period. Finally, it focuses on the few existing studies based on panel data for Argentina, to complement and provide a benchmark for the empirical findings of the following Chapters. Rather than providing a full summary of each article, the Section reviews only the results that are connected to the analysis of Part I.

The availability of household survey data since the mid 1970s has resulted in a substantial literature on income distribution and poverty in Argentina. Nonetheless, most of it focuses on the Greater Buenos Aires region since the EPH was not systematically extended to the rest of the country until the 1980s. Among these studies, Gasparini et al. (2001) cover the whole extension of data available from the EPH, from the mid 1970s until the end of the 1990s. The book details trends in poverty, inequality and social welfare measures. It describes the available data and the methodology for the construction of these measures, and studies the robustness of the identified movements with respect to changes in poverty lines, economies of scale and equivalence scales at the household level. Gerchunoff and Llach (2004b) goes further back in time, providing a long term perspective on equality and growth, from 1880 until the present, in the context of the economic history of Argentina. Given its historical coverage, the book builds on national accounts, tax and a multitude of other data sources.

With respect to the recent past, the articles presented in FIEL (1999) cover different aspects of distributional issues during the 1990s. The volume focuses on the impact of the structural changes in the Argentine economy in this period, and presents an in-depth analysis of poverty, social expenditure and the incidence of the tax system on the income distribution.

Another good source of data and analysis on household welfare can be found in the labour economics literature. Galiani and Hopenhayn (2003) provide a thorough analysis of unemployment risk and duration in the 1990s based on the EPH, while Galiani and Gerchunoff (2003) present a historical account of the evolution of labour market institutions and their role in the economy during the twentieth century in Argentina. Finally, Mackenzie (2004) presents an account of the aggregate economic shocks and labour market responses during the 1995-2002 period.

These volumes and articles provide exhaustive accounts of distributional and household welfare issues in Argentina. For a comparative perspective, Thorp (1998) and World Bank (2004) constitute comprehensive studies of the evolution of poverty, inequality, and their causes and consequences in the context of Latin America, the former in a historical perspective and the latter concentrating on the more recent experience. Finally, Heckman and Pages (2003) and IADB (2004) focus on labour market and employment issues in the countries of the region.

1.4.2 Economic crises, household welfare and coping strategies

The studies cited above are mainly descriptive, identifying trends and correlates of poverty. The analysis in World Bank (2000b) assesses the management of social risks in Argentina, studying private and public interventions to derive policy recommendations. This article highlights the fact that although Argentina had the highest GDP per capita in Latin America and some of the region's highest levels of social expenditure during the 1990s, poverty was still widespread as a consequence of the gaps in the provision of basic infrastructure and in the social insurance system. Based mainly on data from the EPH, the article identifies the main risks faced by different age groups, and analyses the social protection programmes designed to prevent and mitigate them.

The study finds that infants (0 to 5 years old) have higher rates of poverty and extreme poverty than any other group in the population, and points to the relative lack of early child development programmes in comparison with other Latin American countries. For the groups ranging from 6 to 14 years and 15 to 24 years, World Bank (2000b) focuses on educational outcomes and how these might lead to low levels of human capital accumulation. While the coverage of primary education is almost universal (98.9 percent), the study questions its quality and efficiency in the poorer regions of the country. The figures for secondary level enrolment and completion are much lower, with the lowest corresponding to the poorest areas.

Moreover, the article identifies employment issues as the main risk for the population aged 25 to 64 years old. While unemployment is one of the most important determinants of poverty in Argentina, World Bank (2000b) reports that a large proportion of the poor and extreme poor are employed, so besides joblessness, this group also faces low levels of wages and underemployment. While some social insurance programmes exist (for instance, unemployment insurance and family income supplements), access is granted mainly through formal work, which limits their impact given the high rate of informality among the population in general and the poor in particular. For the group of those aged 65 years old and more, the main risk is the low level of state pensions.

Finally, for the population in general, social programmes concentrate mainly on housing, with very low coverage, and on the provision of public health. While access to the latter is generally good, the article argues that its quality is relatively low. World Bank (2000b) recommends a social protection strategy based on early child development, retention in the education system and the promotion of formality in the labour force, concentrating the assistance in fewer programmes with better coverage.

While this study concentrates on the years before the economic crisis of 2001-2002, Fiszbein et al. (2002) present the results of a special survey carried out in June and July of 2002, which focuses on the impact of the crisis on household welfare and the strategies adopted to cope with its consequences.

The changes in employment patterns are striking: Fiszbein et al. (2002) find that more than half of those who entered employment during the crisis

were secondary workers within the household, implying a change in family roles. They also find evidence of deterioration in the quality of jobs, manifested in an increase in the number of temporary and informal workers, and a reduction of work-related benefits for those in permanent jobs.

An original feature of the survey is that it collects information about subjective welfare and social unrest. Fiszbein et al. (2002) find rising levels of discouragement and pessimism, an increase in participation in various forms of social protest, and high levels of violent crime, which are all attributed to the economic crisis. Another original characteristic of the survey is the collection of data on coping strategies. Most of the households in the sample changed their consumption patterns to cope with the effects of the crisis, resorting to less expensive products. Other active strategies included the addition of a new worker to the family group, although this was not very effective given the fall in labour demand and the high rate of unemployment at the time. The poorest households were those who resorted the most to social networks, relying on assistance from friends, family, nongovernmental organizations and the government.

An innovative feature of the data is that, unlike the EPH and most other surveys for Argentina, the study carried out by Fiszbein et al. (2002) covers smaller towns in rural areas. A notable result is the high inequality between rural and urban areas, with the average income in the former only 60 percent of urban average. However, this appears to have only a marginal effect on national poverty and income figures, given the relatively small proportion of the population living in the countryside.

Finally, two articles evaluate the two major public intervention programmes carried out in Argentina during this period. Ravallion and Jalan (2003) review the *Plan Trabajar*, a public works programme introduced in the wake of high unemployment levels after the Mexican financial crisis of 1995. *Trabajar* provided short-term jobs at low wage in infrastructure projects. The central government allocated the programme's budget across provinces based on the unemployment levels among the poor. For practical purposes, *Trabajar* operated as a poverty alleviation programme targeted through unemployment (de Ferranti et al., 2000b).

Ravallion and Jalan (2003) present an evaluation, based on propensity score matching methods, of the programme's efficacy in reducing poverty.

Their results indicate that the *Plan Trabajar* generated income gains among the poor, and that its targeting was relatively good, with 80 percent of beneficiaries in the poorest quintile of the population. Ravallion and Jalan (2003) find that the average gain is one half of the average wage, allowing for the foregone income in the form of jobs displaced by the programme. In terms of the distribution of gains from the programme, the gains are similar for men and women, but they are higher for younger workers.

Galasso and Ravallion (2003) review the *Plan Jefes y Jefas de Hogar Desempleados,* implemented in January 2002 as a safety net for those most affected by the crisis of 2001-2002. The programme consisted of direct income support for unemployed heads of households with dependent children. It was designed as a universal programme, and it eventually reached more than 2 million beneficiaries (INDEC, 2003b).

Galasso and Ravallion (2003) evaluate this programme with counterfactual samples created with matching techniques. The *Plan Jefes y Jefas de Hogar Desempleados* was successful in reducing unemployment, although half of its beneficiaries – mostly women – would have been inactive (and not unemployed) without the programme. The authors found a substantial number of beneficiaries who were ineligible in principle, but its targeting performance, while not as good as that of the *Plan Trabajar*, was still better than the average for social spending in Argentina. Galasso and Ravallion (2003) find that half of the participants in the *Plan Jefes y Jefas* belonged to the lowest quintile of the income distribution. While its design was not completely respected in practice, and it covered more than those unemployed, the programme was successful in providing income support to most of those affected by the crisis, reducing the aggregate levels of extreme poverty.

1.4.3 Longitudinal studies of poverty and well-being in Argentina

The longitudinal dimension given by the rotating structure of the EPH sample, described in Section 1.2, has only recently started to be exploited in the literature on poverty analysis in Argentina.¹⁷ This Section focuses on stud-

¹⁷Studies based on the EPH panels concentrate mostly on labour markets and employment (Galiani and Hopenhayn, 2003; Mackenzie, 2004). Some exceptions are Lavergne et al. (1999), who discuss the variability of income at the individual level along a cohort of the

ies of EPH panels and deals mainly with poverty spells and dynamics: they can be considered complementary to the results presented in this thesis.

Paz (2002) covers the dynamics of poverty within one cohort of the EPH, using the four waves of the period October 1998 to May of 2000 at the national level. The article provides an analysis of simple transitions similar to the one provided in the previous Section, reaching comparable conclusions about the magnitude of movements into and out of poverty.

The most interesting aspect of the article is a regression analysis of the determinants of poverty duration and entry and exit rates. Paz (2002) finds that persistently poor households tend to be headed by younger individuals with low education and higher unemployment levels, and tend to have fewer income earners. The Noreste and Noroeste regions are significantly correlated with higher levels of persistency. In terms of serial correlation, the poverty status at one point in time is highly correlated with that of previous periods, and that the effect is stronger for periods closer in time. Besides its effect on the duration of poverty, the unemployment of the household head is also found to substantially increase the probability that a household falls into poverty. Finally, older heads with higher education levels significantly increase the probability of leaving poverty (for the poor) and not entering poverty (for the non-poor).

Gasparini (2003b) also uses data from one cohort of the EPH at the national level, although the article concentrates on the yearly change between the October 2000 and October 2001 waves of the survey. Its aim is to identify and describe the characteristics of the households that entered poverty between these two waves, and to study the effects of the deepening of the recession during that period.

An interesting finding of the analysis is that the structure of the household does not have a major impact on the switch in poverty status between the two periods after controlling for other characteristics, with the exception of single-parent households headed by women with low education levels, for which the probability of becoming poor is significantly greater than for others. As in Paz (2002), Gasparini (2003b) finds that households headed by better-educated individuals have a lower probability of falling into poverty,

EPH, and Albornoz and Menéndez (2004), who analyse income mobility and its relationship with inequality.

and also finds that the probability of being poor is highly correlated with the previous poverty status.

Finally, Corbacho et al. (2003) use a series of two-period panels from May 1999 to May 2002, covering the recession of 1999-2001 and the economic crisis of 2001-2002 to study which groups bore the burden of the crisis. They also consider the transmission channels from macro shocks to individuals and the smoothing mechanisms adopted by the households to cope with the crisis.

Their measure of the effect of shocks on household welfare is the change in household income between two waves of the EPH. This proxy for vulnerability is regressed on household characteristics and geographic and time controls. The results show again that households headed by better-educated individuals are less vulnerable, as are those with heads employed in the public sector. Geographically, the inhabitants of the Noreste and Pampeana regions experienced the largest declines in income.

Corbacho et al. (2003) also identify employment status as the main transmission channel between macroeconomic shocks and individual welfare, and provide some evidence on labour market outcomes. They find that unemployment was higher among individuals with lower levels of education, and that younger workers were more likely to lose their jobs over the period. Public employees were those with higher job stability, whereas within the private sector unemployment was higher in the construction sector. Finally, the authors also study whether other income sources allow households to smooth the shocks to the labour income of the household head, finding that almost a third were not able to do any smoothing, with the extremely poor faring worse in this respect than the rest of the population.

1.5 CONCLUSION

This Chapter has discussed the methodology of poverty measurement in Argentina, presented the main datasets to be employed in this Part and documented the evolution of poverty measures and its components over the 1995-2002 period. Its aim was to provide a framework for the analysis of Chapters 2 to 4, in terms of the characteristics of the data and the economic context. The Chapter has described the large fluctuations of household income over this turbulent period, and reported evidence of high volatility in movements in and out of poverty. The 2001-2002 crisis was found to have a large impact on the persistence of poverty. An important finding was that relatively modest changes in poverty rates between two periods were the result of large but offsetting movements into and out of poverty. Moreover, these large fluctuations were not restricted to periods of crisis: a substantial fraction of the population was found to enter poverty even when aggregate rates were falling. The use of panels revealed some features of the data that could not have been captured with the cross-sectional datasets usually employed in poverty analysis.

These large fluctuations in household income and poverty status are the main motivation for the following three Chapters, which explore the theoretical basis for the incorporation of income variability in poverty measurement and the definition of well-being, and illustrate different methodologies with empirical implementations based on the Argentine case.

CHAPTER 2

INCOME FLUCTUATIONS, POVERTY AND Well-Being Over Time

2.1 INTRODUCTION

The previous Chapter covered the movements into and out of poverty in two consecutive periods in urban Argentina (Section 1.3, page 29). This Chapter, and the following two, address a related but different question: echoing Atkinson (1987), they deal with the problem of how poverty should be measured *over time* – or, in more general terms, how to measure wellbeing based on repeated observations of household income. The framework developed in the following pages accounts explicitly for the negative effects of income variability. This welfare criterion is based on the intuition, derived from the risk aversion literature, that households will prefer a steady stream of income to a variable one with the same mean, at least in a second-best world with incomplete insurance and capital markets (Cowell, 1989).

The evaluation of well-being with panel data can be thought of as an extension of the standard model of distributional analysis. Cowell (2000) describes the welfare theory of income distribution in terms of \mathcal{F} , "the space of all univariate probability distributions" F of income y_i , and defines a "welfare ordering" $W : \mathcal{F} \to \mathbb{R}$ as a function that maps income distributions into the real line. The analysis of repeated observations is based on the distribution of N vectors of T observations y_{it} over the period t = 1 to T, defined as $\mathbf{y}_i = [y_{i1}, ..., y_{iT}]$, in a population with N households. Accordingly, let \mathcal{F}^T be the space of distributions F^T of vectors \mathbf{y} . The methodology focuses on a mapping from \mathcal{F}^T into the real line, with a transformation of the form $W^T : \mathcal{F}^T \to \mathbb{R}$.

The contribution of this Chapter is to define a transformation W^T in two steps, exploiting analogies with well-established results in economics and distributional analysis theory in each stage. The first step is the definition of an aggregate of the observations of income over time for household *i* that maps each vector \mathbf{y}_i into the real line. As discussed below, the average \bar{y}_i does not account for the welfare effects of income variability: the insight is to exploit the formal analogy between states of the world in the expected utility model and past incomes in a multi-period setting, in a procedure that echoes the social welfare function approach in distributional analysis (Atkinson, 1970). Building on the concept of the certainty equivalent of income, the first step reduces a distribution F^T of N vectors \mathbf{y}_i to F, a distribution ution of N scalars \tilde{y}_i . The second stage of the proposed W^T transformation involves an additional analogy: by showing that these scalars are appropriate money metrics of well-being, all the available tools of distributional analysis can be directly applied to the distribution F. The W^T transformation is done first from each vector \mathbf{y}_i to a scalar \tilde{y}_i , and then from $F(\tilde{y}_i)$ into some distributional index.

This methodology owes a great deal to the standard model of risk (Pratt, 1964; Arrow, 1970) and to its reinterpretation in the social welfare context (Atkinson, 1970), as well as to the literature on lifetime income (Cowell, 1979). In terms of recent work in the poverty literature, the methodology is related to (and draws from) the concept of expected poverty (Ravallion, 1988), the transient-chronic decomposition (Jalan and Ravallion, 1998) and the recent body of work on economic vulnerability (Ligon and Schechter, 2003; Calvo and Dercon, 2003). The approach proposed below is discussed in the light of this literature, and it attempts to unify some of the existing methodologies under a general framework.

The discussion starts in Section 2.1 with a simple expected utility model to establish the main intuitions behind the methodology. Section 2.3 then presents the proposed framework for the evaluation of past incomes. Section 2.4 compares this method with existing approaches, which are interpreted as special cases of the evaluation framework. Finally, Section 2.5 specifies functional forms for the general evaluation function of Section 2.3, and illustrates the uses of the methodology with data from the Greater Buenos Aires region in the 1995-2002 period. Conclusions follow.

2.2 EX-ANTE AND EX-POST INCOME VARIABILITY

2.2.1 Prospective evaluation of well-being: ex-ante utility and income risk

The objective of this Chapter is to develop tools for the analysis of wellbeing based on panel data on household income. To reduce this problem to a tractable form, the proposed methodology aggregates these repeated observations into a single indicator for each household by means of an evaluation function. This Section presents a simple model of choice under uncertainty to introduce the intuition and some formal results for deriving such an indicator, and to clarify the difference between ex-ante and ex-post evaluations of well-being.

The standard expected utility framework, due to von Neumann and Morgenstern (1944), specifies a household's utility function u(y), where y is income, consumption or wealth (henceforth denoted income).¹ In this formulation, utility is defined over a single argument, in the tradition of Friedman and Savage (1948), Pratt (1964) and Arrow (1970).² The function u represents a reduced form that encapsulates the utility level resulting from behavioural responses in savings, labour supply and other choice variables (Cowell, 2005, Chapter 9): the focus is placed on the outcome and not on the process through which it is reached.³

Following the standard risk literature, the function u is assumed to be differentiable, strictly increasing and strictly concave. Uncertainty enters this model as a countable set Ω of possible future states of the world. The household evaluates its future prospects in time t = 0. These uncertain prospects are defined as state-contingent incomes y_w that materialise in t = 1. Each of these possible states of the world $w \in \Omega$ has an associated probability τ_w .

The function *u* is an ex-post utility function, since it evaluates the utility

¹The discussion could be based on income or consumption. As money metrics of welfare, the two are used interchangeably in this Chapter. See Ravallion (1994) and Deaton (1997) for discussions of the advantages and disadvantages of using either for evaluating well-being.

²A concept lost in this formulation is the function of assets to link consumption in different periods. Kimball (1990), however, demonstrates the formal equivalence of the model of precautionary savings under income risk (Leland, 1968; Sandmo, 1970; Drèze and Modigliani, 1972) and models of choice under uncertainty.

³See Besley (1995) and Browning and Lusardi (1996) for reviews on responses to risk. de Ferranti et al. (2000c) covers the empirical evidence for Latin America in recent years.

of a determinate income y (Felli, 2003). The household's ex-ante utility is defined as the expectation over the ex-post outcomes: u is defined on certain incomes, but ex-ante utility is defined on lotteries over incomes. By means of the expected utility theorem (first due to von Neumann and Morgenstern, 1944), ex-ante utility can be expressed as:

$$U = E[u(\hat{y})] = \sum_{\omega \in \Omega} \tau_{\omega} u(y_{\omega})$$
(2.1)

where *E* is the expectations operator, \hat{y} the uncertain income prospect and y_{ω} the contingent income associated with state of the world ω .⁴

A key result in the theory of choice under uncertainty is that expected utility depends not only on $E[\hat{y}]$, the expectation of future outcomes, but also on their distribution. A simple example clarifies this assertion. The random variable \hat{y} is equal to either $y_H = \bar{y} + h$ or $y_L = \bar{y} - h$ with equal probability, and h > 0, so that a change in h represents a mean-preserving spread in future income.⁵ Then U can be written as $U = \frac{1}{2}u(y_H) + \frac{1}{2}u(y_L)$, which depends on the value of h in the following way:

$$\frac{\partial U}{\partial h} = \frac{1}{2}u'(y_H) - \frac{1}{2}u'(y_L)$$
(2.2)

where u' represents the first derivative of u. Risk aversion follows from the concavity of u, which implies that u' is decreasing, and results in a negative value of $\partial U/\partial h$. The intuition is that greater uncertainty about future income makes a risk averse household worse-off, since by the concavity of u the possibility of a gain is outweighed by the prospect of a loss.

An important implication of risk aversion is that the expected value of income $-E[\hat{y}] = \bar{y}$ in the example above – is not a suitable money metric measure of well-being since it does not reflect the effects of risk on utility (Cowell, 1979). As shown in the example, an increase in *h* reduces ex-

⁴Alternatively, the set of outcomes can be continuous. In that case, the expectation is defined over the integral of the distribution of \hat{y} . It is assumed throughout that this distribution is well-behaved, that $\sum_{w \in \Omega} p_w = 1$, and that all functions have the regularity conditions of the standard model of choice under uncertainty (Newbery and Stiglitz, 1981b). See Mas-Colell et al. (1995, Chapter 6) for a detailed presentation of the expected utility model and its underlying axioms.

⁵These results hold under the much more general conditions of second order stochastic dominance (Rothschild and Stiglitz, 1969), of which this example is a special case.

pected utility, yet leaves $E[\hat{y}]$ unchanged. However, the concept of certainty equivalence – a fundamental notion in the theory of choice under uncertainty – provides an intuitively appealing indicator of well-being. The (nonrandom) certainty equivalent income \tilde{y}_{ce} is implicitly defined by:

$$U = E[u(\hat{y})] = u(\tilde{y}_{ce}) \tag{2.3}$$

An expected utility maximising household with preferences defined by u would be indifferent between receiving \tilde{y}_{ce} with certainty in the future or facing the uncertain prospect \hat{y} . For a risk-averse household, Equation 2.3 implies, by the concavity of u and Jensen's inequality, that $\tilde{y}_{ce} < E[\hat{y}]$. Moreover, since $E[u(\hat{y})]$ is decreasing in the level of risk, it follows from Equation 2.3 that \tilde{y}_{ce} is also negatively affected by greater uncertainty. The certainty equivalent is thus better suited than $E[\hat{y}]$ as a money metric indicator of well-being, since it captures the disutility arising from uncertainty.

Risk aversion and the certainty equivalent, however, are ex-ante concepts, and this Chapter's aim is to measure well-being based on (ex-post) panel data: the following pages deal with the distinction between ex-ante prospects and ex-post outcomes.

2.2.2 Retrospective evaluation of well-being: ex-post utility and income fluctuations

The expected utility U in Equation 2.1 is defined over events that have not occurred, while the underlying utility function u is defined over certain ex-post outcomes. This is an aspect of the theory that is often overlooked (Hammond, 1981; Milne and Shefrin, 1988; Ravallion, 1988, constitute some exceptions). It is necessary to distinguish between "income risk," an ex-ante concept based on variability in future prospects, and "income fluctuations," defined as experienced variability in the past. Panel data on incomes is inherently linked with both concepts: it reflects income fluctuations and thus it is ex-post by definition, but these fluctuations result from the presence of ex-ante income risk, as discussed below.

The setting described by Equation 2.1 corresponds to the case of income risk: once the state of the world ω materialises in t = 1, the random variable \hat{y} takes a value and ex-post utility becomes $u(y_{\omega})$, either $u(y_H)$ or $u(y_L)$

CHAPTER 2. INCOME FLUCTUATIONS AND POVERTY





in the example above. To introduce the idea of income fluctuations, it is necessary to consider multiple past periods. It is assumed that a household at T + 1 has faced a series of consecutive independent realisations of states of the world drawn from the set Ω , which results in a past stream of income $\mathbf{y} = [y_1, ..., y_T]$. While every \hat{y}_t is a random variable before its realisation, from the retrospective point of view of t + 1, every y_t is just a determinate quantity given by the materialised state of the world ω_t .

Well-being over time is determined by the experienced income stream **y**. The average of experienced utilities, while not accounting for time preferences, provides a simple aggregate of utility over the *T* periods:

$$\bar{u} = \frac{1}{T} \sum_{t=1}^{T} u(y_t)$$
(2.4)

Equations 2.1 and 2.4 are formally similar. However, the two represent the distinct but related concepts of income risk and income fluctuations.

The example in Figure 2.1 illustrates this point. The Figure depicts the past outcomes and the future prospects for a household from the point of view of the present (t = 0) according to the formulations of Equations 2.1 and 2.4. The income variable \hat{y}_t is assumed to have the same distribution in each period $t = \{-3, -2, -1, 1\}$, being either $y_H = \bar{y} + h$ or $y_L = \bar{y} - h$ with equal probability.

While both U and \bar{u} are determinate quantities, Equation 2.1 reflects the expected utility of the household in t = 1 from the point of view of t = 0, while Equation 2.4 is the evaluation of a series of past outcomes at $t = \{-3, -2, -1\}$ from the same point of view. To stress this difference, \bar{u} in Figure 2.1 is the result of two y_H and one y_L realisations, which differs from the expected utility U = E[u], represented by the average of $u(y_H)$ and $u(y_L)$.

While income risk as summarised in Equation 2.1 and income fluctuations in Equation 2.4 are different in their nature, the connection between the two is that (ex-ante) risk is the source of (ex-post) fluctuations: with no risk, the distribution of \hat{y}_t would be a fixed value at every point in time, and the resulting stream of past income would be flat.

Moreover, while a risk averse household would prefer lower variability for a given value of expected future prospects (Equation 2.2), a similar intuition applies to the simple aggregate of past utilities given by Equation 2.4: a risk averse household would trade off a reduction in the average for lower variability in past incomes, at least in a second-best world with incomplete insurance and capital markets (Cowell, 1989).

The following Section develops a framework for the evaluation of past incomes based on this analogue of risk aversion in an ex-post setting and on the formal similarity between the formulations of expected and average experienced utilities.

2.3 A FRAMEWORK FOR THE EVALUATION OF INCOME FLUCTUATIONS

2.3.1 The structure of the evaluation function

A general formulation for an aggregate of household income over time, $\mathbf{y} = [y_1, ..., y_T]$, is given by an evaluation function V that maps a vector of T

observations into the real line:

$$V(\mathbf{y}) = V(y_1, ..., y_T)$$
 (2.5)

In terms of the terminology used in the Introduction, *V* defines a transformation $W : \mathcal{F} \to \mathbb{R}$, from the distribution of past incomes for a household into the real line.

The problem remains in defining a functional form for V, which determines the normative criteria associated with the evaluation of y. The presence of the time dimension introduces a higher degree of complexity with respect to the analysis of an income distribution at one point in time.

The framework proposed here concentrates on a series of intuitive criteria. As a starting point, it is reasonable to assume that V should be nondecreasing in its arguments. Moreover, the aggregate level of welfare over the T periods should depend not only on the level of \mathbf{y} , but also on its variability. The idea, pervasive in economic theory, is that risk averse agents are willing to trade off a reduction in expected income for certainty. In an ex-post setting, the concept of risk aversion translates into a "dislike" of fluctuations, or variability aversion (to be formally defined below).

These two basic normative principles can be incorporated into the evaluation function V based on the results and intuitions of the previous Section. The evaluation framework, however, does not rely on utility functions u: the function V is interpreted within a social welfare context as a judgement on the welfare value of the experienced income stream. This approach, followed by Cruces and Wodon (2003b) and Ligon and Schechter (2003) among others, implies that it is not necessary to impute a utility function and assume homogeneous preferences in the population.

In the evaluation framework, the stream of past income $\mathbf{y} = [y_1, ..., y_T]$ is assessed retrospectively from the point of view of period T + 1. The parallelism of V with the expected utility formulation in Equation 2.1 means that each past income y_t is evaluated by an instantaneous evaluation $v(y_t)$, assumed to be continuous, strictly increasing and twice-differentiable.⁶ The result is the following characterisation of V:

⁶While the emphasis is on panel data and thus on the evaluation of *past* incomes, the methodology can be applied to a certain but fluctuating future income stream.

Definition 2.1 Additive, time-separable evaluation function. The evaluation of the observed stream of past income $\mathbf{y} = [y_1, ..., y_T]$, $V(\mathbf{y})$, is the discounted average of the instantaneous evaluation function v for each period from t = 1 to T:

$$V(\mathbf{y}) = \sum_{t=1}^{T} \Delta(t) v(y_t)$$
(2.6)

The weights are given by a discounting function $\Delta(t)$, with $0 < \Delta(t) \le 1$, and normalised (without loss of generality) so that $\sum_{t=1}^{T} \Delta(t) = 1$.

The structure imposed by Equation 2.6 implies the following analogy: the model of choice under uncertainty in a single period (Equation 2.1) and the evaluation of past incomes based on an additive, time-separable evaluation function as in Equation 2.6 are formally equivalent. The results from the former can be applied to the latter by: a) replacing the function *u* by its analogue *v*, b) replacing state-contingent incomes y_w by observed incomes y_t , and c) replacing probabilities τ_w by $\Delta(t)$.

This analogy is established by inspection of Equations 2.1 and 2.6: ranking vectors of past incomes **y** according to *V* is formally identical to ranking probability distributions according to the expected utility criterion. From a formal point of view, Equation 2.6 can be treated as a special case of Equation 2.1 with u = v, $\omega = t$, $y_{\omega} = y_t$ and $\tau_{\omega} = \Delta(t)$, i.e. "as if" past incomes were drawn from *T* events with outcomes y_t . The formulation for *V* in Equation 2.6 implies that the formal results from risk theory can be applied directly to the evaluation framework, although the interpretation of these results differs. The connection between the theory of risk and the evaluation framework stems from the discussion of income risk and income fluctuations in the previous Section.⁷

The main difference between risk and the formulation of Equation 2.6 is the presence of the discounting function $\Delta(t)$, which accounts explicitly for the time dimension of the problem of evaluating past incomes. The motivation for the incorporation of $\Delta(t)$ into *V* is the presence of pure time preferences: in the example of Figure 2.1, a household would not be indif-

⁷The idea of borrowing results from risk theory is at the basis of Atkinson's (1970) reinterpretation of choice under uncertainty in a social welfare context. However, a social welfare function aggregates the distribution of income at a point in time for a population, while V is a social evaluation of *household* welfare as defined by Equation 2.5.

ferent to the ordering of past incomes, giving more weight to more recent events.⁸ The function $\Delta(t)$ is thus required to increase as t approaches T. Since the discount factors are normalised to sum 1, they can be interpreted as "discounting weights." In the simplest form of aggregation, every period of time is given an equal weight so that $\Delta(t) = 1/T$. Section 2.5 below discusses specific functional forms for v(y) and $\Delta(t)$.

The parallel with the theory of risk is completed by the following Proposition, which is the evaluation framework's analogue of risk aversion, and specifies a key condition for v:

Proposition 2.1 Variability aversion. The function v is assumed to be strictly concave, which implies that $V(\mathbf{y})$ is strictly decreasing in the dispersion of $\mathbf{y} = [y_1, ..., y_T]$ weighted by the discounting function $\Delta(t)$. The dispersion is defined in the sense of Rothschild and Stiglitz's (1969) second order stochastic dominance.

This result derives from the concavity of v, and it is equivalent to the risk aversion result ($\partial U/\partial h < 0$) in Equation 2.2. The curvature of v determines the degree of variability aversion, and its magnitude can be quantified by defining measures of absolute and relative aversion by analogy with the canonical model of risk (Pratt, 1964; Arrow, 1970).

Proposition 2.1 implies that for a given average discounted income over time, $\bar{y}_{\Delta} = \sum_{t=1}^{T} \Delta(t) y_t$, a higher variability in the underlying stream reduces welfare as captured by *V*. The properties of *V* and *v* given by Definition 2.1 and Proposition 2.1 adapt the concept of risk aversion to the intertemporal setting, incorporating in the evaluation framework the principle that past fluctuations reduce welfare, and should be penalised by an evaluation function. While not all fluctuations might be considered bad, for instance when income grows over time (Cowell, 1989), the variability aversion is based on the discussion of the effects of riskiness on household utility in Section 2.2. Moreover, the presence of the discounting function $\Delta(t)$ in *V* ensures that the evaluation of past incomes is not invariant with respect to the ordering of the components of **y**, except for the special case in which $\Delta(t) = 1/T$. For instance, in a setting with T = 2, if $y_2 > y_1$

⁸Alternatively, $\Delta(t)$ can also be motivated by the presence of imperfect storage technologies, in which only a limited amount of income can be kept for future use.

then $V(y_1, y_2) \ge V(y_2, y_1)$: an increasing income stream results in a higher evaluation of well-being than a decreasing stream.

The following pages build on the additive structure of the evaluation function to specify measures of well-being and its variability over time.

2.3.2 Evaluation of well-being and variability over time

The concept of variability aversion and the structure of V given by Definition 2.1 imply that another important notion from the theory of choice under uncertainty can be adapted to the evaluation of past incomes. The analogue of the certainty equivalent income (Equation 2.3) is given by:

Definition 2.2 The stability equivalent income \tilde{y}_{se} is a real number such that

$$V(\mathbf{y}) = v\left(\tilde{y}_{se}\right) \tag{2.7}$$

 \tilde{y}_{se} is the level of income that, if received in every past period t = 1 to T, as $\tilde{\mathbf{y}} = [\tilde{y}_{se}, ..., \tilde{y}_{se}]$, would result in the same level $V(\mathbf{y})$ of the evaluation function as the observed stream $\mathbf{y} = [y_1, ..., y_T]$.

The continuity of v guarantees that \tilde{y}_{se} exists, and its concavity implies that it is decreasing in the dispersion of **y**. Both results are formally analogous to those for the certainty-equivalent in risk theory (Pratt, 1964).⁹

The counterfactual stability equivalent \tilde{y}_{se} is a function of the shape of v and the level and distribution of y_t in **y**. Under the assumption that the variability of past income reduces well-being, the \tilde{y}_{se} can be interpreted as a "variability adjusted" income. It constitutes a welfare-based counterpart to the statistical measure \bar{y}_{Δ} , and it is thus superior to the discounted average income as an indicator of well-being, just as the certainty equivalent \tilde{y}_{ce} was deemed superior to $E[\hat{y}]$ in the expected utility model of Section 2.2.¹⁰

Finally, another concept that can be adapted from the theory of choice under uncertainty is the risk premium. Since \tilde{y}_{se} is lower than the average income \bar{y}_{Δ} because of the concavity of v, the difference between the two

⁹This stability equivalent income is formally equivalent to Atkinson's (1970) "equally distributed equivalent level of income," and it is closely related to Ravallion's (1988) notion of "stabilised income."

¹⁰This argument is derived from Cowell's (1979) discussion of a "lifetime welfareequivalent current income" for an income stream (Definition 2, page 12).

provides a money metric of the loss in household welfare attributable to income fluctuations, as described in the following Definition:

Definition 2.3 *The* variability premium π_v and the relative variability premium Π are real numbers such that

$$\pi_v(\mathbf{y}) = \bar{y}_{\Delta} - \tilde{y}_{se} \tag{2.8}$$

$$\Pi(\mathbf{y}) = \frac{\pi_v}{\bar{y}_\Delta} \tag{2.9}$$

where \bar{y}_{Δ} is the weighted average income over time given by $\bar{y}_{\Delta} = \sum_{t=1}^{T} \Delta(t) y_t$, and \tilde{y}_{se} is the stability equivalent income defined above.

Since \tilde{y}_{se} is decreasing in the dispersion of \mathbf{y} , π_v and Π are increasing in the same parameter. Moreover, the curvature of v also affects these quantities through its effect on \tilde{y}_{se} : for a given \mathbf{y} , the stability equivalent income falls and the premium increases with v's degree of variability aversion. The premium π_v can be considered to be a welfare-based measure of the variability of past incomes, while the relative premium Π shares the same property and has the advantage of being unit-free.¹¹

Figure 2.2 depicts \tilde{y}_{se} and π_v for T = 2 in the evaluation-income space.¹² As in risk theory, the stability equivalent falls and the variability premium increases with a higher dispersion in past incomes due to the concavity of v. For a fixed level of dispersion, the effect of an increase in the curvature of v is the same. While the Figure is identical to any textbook example of risk aversion, Figure 2.1 presents the setting in the utility-time (or evaluation-time, setting u = v) space, stressing the difference between future states of the world (Equation 2.1) and materialised outcomes (Equations 2.4 and 2.6). From this example, it can be seen that the certainty equivalent \tilde{y}_{ce} depends solely on the distribution of one future outcome, whereas only actual realised incomes matter for the stability equivalent \tilde{y}_{se} , even if these realised incomes originate in repeated draws from the same distribution as \tilde{y}_{ce} .

¹¹Some manipulation shows that $\Pi(\mathbf{y}) = 1 - \tilde{y}_{se}/\tilde{y}_{\Delta}$, which is formally equivalent to Atkinson's (1970) simple inequality index.

¹²For expositional convenience, all the diagrams in this Chapter are based on the no discounting case, in which $\Delta(t) = 1/T$. This implies that $V(\mathbf{y})$ represents the simple average of $v(y_t)$ and that $\bar{y}_{\Delta} = \bar{y} = (1/T) \sum_{t=1}^{T} y_t$.



Figure 2.2: Stability Equivalent Income and Variability Premium

The following pages complete the discussion of the evaluation framework by studying the properties of the stability equivalent \tilde{y}_{se} as a variability adjusted income.

2.3.3 "Fluctuation adjusted" population measures of well-being

In terms of the Introduction's terminology, both *V* and \tilde{y}_{se} define transformations $W : \mathcal{F} \to \mathbb{R}$ that result in scalar measures of well-being based on a household's past incomes. While $V(\mathbf{y})$ and \tilde{y}_{se} provide equivalent mea**Proposition 2.2** The stability equivalent income \tilde{y}_{se} , given by Definition 2.2, is a sufficient money metric statistic of household welfare defined by the evaluation functions v and V.

The proof of this Proposition relies on the uniqueness of the certainty equivalent in risk theory (Pratt, 1964). With a well-behaved, standard function v, Equation 2.7 implicitly defines V and \tilde{y}_{se} as monotonic transformations of each other, because v is strictly increasing and continuous. Since v takes income as its argument, the scalar measure of well-being, \tilde{y}_{se} , is money metric.

This implies that the stability equivalents \tilde{y}_{se} and \tilde{y}'_{se} , corresponding to two functions $V(\mathbf{y})$ and $V'(\mathbf{y})$, are directly comparable because they are both measured in money terms. Besides this property, Proposition 2.2 and the nature of the stability equivalent imply that \tilde{y}_{se} satisfies Cowell's (1995, Chapter 1) requirements for a measure of income, which must be "measurable [...] and comparable among different persons."¹³ For this reason, Proposition 2.2 ensures that all the tools of univariate distributional analysis can be applied to the distribution of \tilde{y}_{se} .

This procedure constitutes a second $W : \mathcal{F} \to \mathbb{R}$ transformation. The problem of studying the distribution of vectors \mathbf{y} in the population is reduced, by means of the evaluation function V, to the study of $F(\tilde{y}_{se})$, the univariate distribution of the stability equivalent income. This means that any poverty measure P, inequality measure I, and social welfare function W defined over the distribution of incomes y at one point in time can also be applied to the distribution of \tilde{y}_{se} (Atkinson and Bourguignon, 2000). Moreover, since \tilde{y}_{se} is money metric, its distribution can be compared to that of the average over time for each household, \bar{y} . This exercise is akin to the comparison of distributions before and after tax or transfers, for which there exists an extensive literature and a standard set of tools (Cowell, 1995).

The two-step methodology described in this Section is similar in spirit to the process of equivalisation in distributional analysis. Survey data usually contains information about a number of income-earners in a household. The equivalisation process converts a vector of incomes from different members

¹³Incomes y_t are adjusted for differences in household size and composition by an equivalence scale, and normalised so that they can be compared over time. See Section 1.2 (page 22) for details of these adjustments.

of a household into a single measure, according to some welfare criteria – usually taking into account the gender and age composition of the household (see Section 1.2, page 22, for details on the Argentine case). The analysis is then carried out on the distribution of the scalar equivalised aggregate.

The following Section compares the evaluation framework with the related methodologies in the poverty and distributional literature.

2.4 COMPARISON WITH ALTERNATIVE APPROACHES

2.4.1 *Ex-post measures: transient and chronic poverty*

The evaluation framework has a series of advantages over the existing approaches for the analysis of panel data on incomes. This Section reviews the results from the two main alternatives in the literature.

The first approach, widely used in empirical applications, is the transient-chronic poverty decomposition. This methodology originates in Ravallion's (1988) contribution on poverty and welfare variability, on which Jalan and Ravallion (1998; 2000) base their definitions of transient and chronic poverty. Chapter 4 builds on these categories to study the Argentine case.

The approach applies Atkinson's (1987) family of additive poverty measures to a multi-period setting. A household's poverty in time t is given by the evaluation function $p(y_t)$, where p is required to be additive, strictly convex and decreasing up to the poverty line, and taking a value of zero thereafter. Intertemporal poverty P_i , chronic poverty C_i and transient poverty T_i are defined as:

$$P_{i} = \frac{1}{T} \sum_{t=1}^{T} p(y_{it}),$$

$$C_{i} = p(\bar{y}_{i}) \text{ and}$$

$$T_{i} = P_{i} - C_{i}$$

$$(2.10)$$

Intertemporal poverty is the undiscounted average of the poverty evaluations over time for a household, while chronic poverty reflects the poverty evaluation at the average income over time for *i*, \bar{y}_i . Finally, transient poverty is calculated as the difference between the two. Jalan and Ravallion (1998) compute these measures for every household and then aggregate



Figure 2.3: Transient, Chronic and Variability Adjusted Measures of Poverty

them into population averages, using the squared poverty gap function for p (Equation 1.5).

In terms of empirical applications, the main difference with the evaluation framework is that Jalan and Ravallion (1998) work with poverty evaluations, whereas the methodology presented in Section 2.3 first derives variability adjusted measures of income with an evaluation function, and then computes poverty indices based on them (Section 2.5 below presents an example of this procedure).

Despite this difference, the transient-chronic decomposition represents a special case of the evaluation framework. The poverty evaluation function p can be interpreted as an evaluation function by setting v = -p, which reflects an assessment of *i*'s well-being that gives zero weight to income above the poverty line. This is illustrated in Figure 2.3, which presents an example for T = 2, with no discounting ($\Delta(t) = 1/T$) and with y_1 and y_2 below the poverty line. In the Figure, the poverty evaluation p is mirrored by the

evaluation function v = -p. This representation highlights the connection between the two methodologies: the money metric indicator \tilde{y}_{se} based on v = -p represents the fixed level of income that would result in the same intertemporal poverty *P* as the observed stream **y**.

A disadvantage of the Jalan and Ravallion (1998) approach is that the aversion to variability is implicitly built into the poverty evaluation function p, which amalgamates the poverty and time dimensions. This function, however, may not be appropriate for evaluating income over time. For instance, most of the transient-chronic applications are based on the squared poverty gap, which is akin to a quadratic utility function and thus implies the possibly undesirable property of increasing relative risk aversion (Kurosaki, 2003).¹⁴ On the other hand, the two-step procedure proposed here ensures that these two facets are accounted for by a separate set of principles. The stability equivalent is derived from a set of principles specific to the time dimension, summarised by v, and the measure of poverty is then obtained by applying a function p, specific to the income dimension, to this household aggregate.

Finally, the evaluation framework has two additional advantages. On the one hand, it allows the computation of variability adjusted measures of income for the whole population, while the transient-chronic decomposition by definition applies only to the poor. On the other hand, the incorporation of a discount factor in Equation 2.6 accounts for the trajectory of income, whereas the measures in Equation 2.10 are invariant to changes in the ordering of incomes y_t in **y**.

Some of the advantages of the evaluation framework over the transientchronic decomposition are also present when compared with the vulnerability approach, analysed in the following pages.

2.4.2 Ex-ante measures: risk and vulnerability

The vulnerability approach, as defined by Ligon and Schechter (2003), attempts to capture the ex-ante risk faced by households.¹⁵ They rely on a

¹⁴The properties of the quadratic utility function in terms of risk aversion are analysed by Deaton and Muellbauer (1980, page 400).

¹⁵Thorbecke (2003) and Ligon and Schechter (2004) provide extensive overviews of the literature, including its relationship with Ravallion's (1988) concept of "expected poverty."

"welfare function" U_i^{LS} defined over household income y_i . The vulnerability of a household *i*, V_i^{LS} , is given by the difference between U_i^{LS} evaluated at the poverty line *z* and the expectation of $U_i^{LS}(y_i)$:

$$V_i^{LS} = U_i^{LS}(z) - E[U_i^{LS}(y_i)]$$
(2.11)

which is decomposed into "poverty" and "risk" components:

$$V_{i}^{LS} = \underbrace{\{U_{i}^{LS}(z) - U_{i}^{LS}(E[y_{i}])\}}_{Poverty} + \underbrace{\{U_{i}^{LS}(E[y_{i}]) - E[U_{i}^{LS}(y_{i})]\}}_{Risk}$$
(2.12)

The expectation operator in Equations 2.11 and 2.12 refers to the distribution of future income: V_i^{LS} is meant to capture ex-ante risk and is thus "inherently forward-looking" (Ligon and Schechter, 2004). This is the main difference between the vulnerability approach and the evaluation framework: the former attempts to capture ex-ante income risk, while the latter evaluates ex-post fluctuations, as illustrated in Figure 2.1 and discussed in Section 2.2.

Since observed data is ex-post by definition, this approach requires an identifying assumption to use past realisations "to estimate the probability of possible future outcomes" (Ligon and Schechter, 2003). The assumption made by these authors is stationarity, which imposes the restriction that "the probability distribution of income in one period is identical to the probability distribution of income in any other period" (Ligon and Schechter, 2004). This implies that the last term in Equation 2.11, $E[U_i^{LS}(y_i)]$, becomes $(1/T) \sum_{t=1}^{T} U_i^{LS}(y_{it})$.

However, whether trying to capture past variability or future risk, from an applied point of view only realisations of income **y** are available to the researcher. The vulnerability approach and the evaluation framework methodologies differ conceptually, but the identifying assumption made by the former implies that the two result in similar empirical applications. This means that, as the transient-chronic decomposition, Ligon and Schechter's (2003) vulnerability measures can be interpreted as a special case of the evaluation framework. This is illustrated in Figure 2.4 (based on Thorbecke, 2003), which presents an example with T = 2 and no discounting ($\Delta(t) = 1/T$). In this setting, the evaluation function defined in Equation 2.6 be-

CHAPTER 2. INCOME FLUCTUATIONS AND POVERTY





comes $V(\mathbf{y}) = (1/T) \sum_{t=1}^{T} v(y_{it})$. The connection between the two methodologies emerges from setting the evaluation and welfare functions to coincide: assuming $U_i^{LS} = v$ results in $V(\mathbf{y}) = E[v(y_t)] = E[U_i^{LS}(y_i)]$, the last term in Equation 2.11. As can be appreciated in Figure 2.4, Ligon and Schechter's (2003) measure of vulnerability is equivalent, in the evaluation framework, to the difference between the evaluation of the poverty line, v(z), and that of the observed income stream, $V(\mathbf{y})$.

The Figure also illustrates, in its vertical axis, the decomposition of vulnerability given by Equation 2.12. This example shows that the same exercise can be carried out within the evaluation framework: along its horizontal axis, the Figure presents a monotone transformation of the "poverty" and "risk" components of Equation 2.12 in money metric terms, $z - \bar{y}$ and $\bar{y} - \bar{y}_{se}$ respectively. The latter corresponds to the variability premium de-

fined in Equation 2.8.¹⁶

A disadvantage of Ligon and Schechter's (2003) vulnerability measure, similar to that of the transient-chronic decomposition, is that the function U_i^{LS} determines not only the value of $U_i^{LS}(E[y_i]) - E[U_i^{LS}(y_i)]$, the "risk" component in Equation 2.12, but also the functional form of the "poverty" component, $U_i^{LS}(z) - U_i^{LS}(E[y_i])$. In the evaluation framework, however, the stability equivalent \tilde{y}_{se} is derived from a function v, and the poverty measures are then based on \tilde{y}_{se} , which ensures that fluctuations and poverty are disentangled.

Moreover, V_i^{LS} in Equation 2.11 is derived in units of the cardinal welfare function U_i^{LS} ("utils" in Ligon and Schechter, 2003), which implies that measures of vulnerability based on two functions U_i^{LS} and $U_i^{LS'}$ are not directly comparable. As discussed in Section 2.3, a money metric indicator like \tilde{y}_{se} ensures the comparability of results for different evaluation functions.

Finally, by attempting to capture the ex-ante risk faced by the households, the stationarity assumption means that the measure of vulnerability in Equation 2.11 does not take into account the dynamic dimension of the observed stream **y**: V_i^{LS} is the same for the vectors $\mathbf{y} = [y_1, y_2]$ and $\mathbf{y}' = [y_2, y_1]$ with $y_1 \neq y_2$. While assuming stationarity is plausible in some contexts, the evaluation framework can account for the dynamic nature of **y** through the discounting function $\Delta(t)$. This is illustrated in the empirical applications presented in the following Section.

2.5 Empirical Implementation and Application to Argentina

2.5.1 Empirical implementation: alternative evaluation functions

This Section deals with the implementation of the evaluation framework. It adds structure to the formulation in Section 2.3 by stipulating a series of functional forms for v and studying the characteristics of the resulting stability equivalent incomes \tilde{y}_{se} .

The definition of V in Equation 2.6 relies on the functions v and Δ . The empirical applications presented below are based on an exponential dis-

¹⁶Moreover, the representation of V_i^{LS} in terms of income on the horizontal axis of Figure 2.4 reveals that this measure of vulnerability is a monotone transformation of the poverty gap ($\alpha = 1$ in Equation 1.5, page 35) evaluated at \tilde{y}_{se} .

counting function, although $\Delta(t)$ can in principle accommodate hyperbolic discounting or other suitable principles (O'Donoghue and Rabin, 1999). In what follows, $\Delta(t)$ is given by:

$$\Delta(t,T,\delta) = \frac{\delta^{T-t}}{\sum_{t=1}^{T} \delta^{T-t}}$$
(2.13)

with a bounded discount factor, $0 < \delta \leq 1$. The weighted or discounted average of income is then defined as:

$$\bar{y}_{\Delta} = \sum_{t=1}^{T} \left[\frac{\delta^{T-t}}{\sum_{t=1}^{T} \delta^{T-t}} y_t \right]$$
(2.14)

The formulation in Equation 2.13 and the bounds in the parameter δ ensure that $\sum_{t=1}^{T} \Delta(t) = 1$ and that the function is increasing in t.¹⁷ The parameter δ is the discount factor, which defines the relative weight given to the recent past with respect to events further away in time. As δ approaches 0, more weight is placed in the last period, *T*, and in the limit $\Delta_{\delta \to 0}(T) = 1$ and $\Delta_{\delta \to 0}(t \neq T) = 0$. The opposite case is that of no discounting, which corresponds to $\delta = 1$: this implies that the "discount weights" simplify to $\Delta(t) = 1/T$. In this case, the evaluation function *V* becomes the average of $v(y_t)$, and Equation 2.14 represents \bar{y} .

Regarding the functional form of v, the prominence of choice under uncertainty in the evaluation framework implies that intuitive functional forms for v are derived from the instantaneous utility functions used in the theory of risk.

A first alternative is the analogue of the isoelastic utility function,¹⁸ the Constant Relative Variability Aversion (CRVA). The following Equations describe this function and the implied stability equivalent income:

$$v(y) = \begin{cases} \frac{y^{1-\rho}}{1-\rho} & \text{if } \rho \neq 1\\ \ln y & \text{if } \rho = 1 \end{cases}$$
(2.15)

¹⁷The motivation for an increasing $\Delta(t)$ derives from pure time preferences, which give more weight to more recent events. However, the formulation in Equation 2.13 allows for a decreasing $\Delta(t)$ if $\delta \ge 1$. In that case, larger values of δ imply more weight for events further in the past. In the limit, $\Delta_{\delta \to +\infty}(1) = 1$ and $\Delta_{\delta \to +\infty}(t \ne 1) = 0$.

¹⁸This formulation is also known as the the Constant Relative Risk Aversion (CRRA) utility function.
which results in

$$\tilde{y}_{se} = \begin{cases} \left[\sum_{t=1}^{T} \Delta(t) y_t^{1-\rho} \right]^{\frac{1}{1-\rho}} & \text{if } \rho \neq 1 \\ \prod_{t=1}^{T} y_t^{\Delta(t)} & \text{if } \rho = 1 \end{cases}$$
(2.16)

This functional form allows for a sensitivity parameter ρ , the analogue of the relative risk aversion parameter in the Constant Relative Risk Aversion (CRRA) utility function. Since \tilde{y}_{se} is decreasing in ρ , it quantifies the effect of past variability on well-being: for a fixed dispersion of past incomes, higher values of ρ result in lower stability equivalent incomes.

The CRVA form implies that the degree of aversion to fluctuations is constant relative to the household's income, since the curvature of v is constant. This is consistent with the intuition that "the rich are more tolerant of risks than the poor" (Deaton and Muellbauer, 1980, Chapter 14), and it is reflected in the fact that the relative stability premium Π (Equation 2.9) based on Equation 2.16 remains constant when all incomes in **y** are multiplied by the same positive factor (in the case of no discounting).

An alternative to the CRVA functional form is given by the analogue of the Constant Absolute Risk Aversion (CARA) utility function, which is also widely used in the risk literature. The Constant Absolute Variability Aversion (CAVA) is given by:

$$v(y) = -\frac{1}{\eta}e^{-\eta y} \tag{2.17}$$

resulting in the stability equivalent:

$$\tilde{y}_{se} = -\frac{1}{\eta} \ln \left[\sum_{t=1}^{T} \Delta(t) e^{-\eta y_t} \right]$$
(2.18)

Equation 2.17 also allows for a sensitivity parameter, $\eta \neq 0$, which captures the degree of variability aversion, since larger values of η imply lower stability equivalents \tilde{y}_{se} . Moreover, this formulation is also consistent with the intuition mentioned above, namely that as income grows, households are willing to accept larger fluctuations. Compared to the CRVA, the CAVA functional form implies that the relative stability premium Π (Equation 2.9) falls when all incomes in **y** are multiplied by the same positive factor (again,

CHAPTER 2. INCOME FLUCTUATIONS AND POVERTY



Figure 2.5: Evaluation Function Contours for Different Degrees of Variability Aversion

in the case of $\delta = 1$).

Finally, two extreme cases are presented for illustration. The first case, in which *v* is not strictly concave, is given by a linear evaluation function:

$$v(y) = y \tag{2.19}$$

resulting in

$$\tilde{y}_{se} = \bar{y}_{\Delta} = \sum_{t=1}^{T} \Delta(t) y_t \tag{2.20}$$

This formulation can be interpreted as the limit case of the CRRA function with $\rho = 0$, in other words, with no variability aversion, the fluctuation adjusted income reduces to the discounted average over time.

The opposite case to Equation 2.19 is given by extreme variability aversion, corresponding to the limit case of the CRRA function with $\rho \rightarrow +\infty$.

In the case of no discounting, this formulation results in:

$$\tilde{y}_{se} = \min(y_t) \tag{2.21}$$

The implied evaluation function only takes into account the lowest of past incomes, and it is the analogue, in the evaluation context, of a "Rawlsian" social welfare function (Hammond, 1975).

Figure 2.5 highlights the difference between these different degrees of variability aversion, which are not readily apparent in the evaluation-income space of previous figures. With T = 2, the Figure represents the stability equivalent income in the y_1 , y_2 space for evaluation function contours with different degrees of variability aversion and no discounting. The CRVA and CAVA cases are represented by the "intermediate aversion" curve in the Figure, while the contour implied by Equation 2.19 is the "no aversion" solid straight line, which results in $\tilde{y}_{se} = \bar{y}$. Finally, the extreme aversion case is depicted by the kinked contour in Figure 2.5.

2.5.2 Application to Argentina

The following pages present alternatives for empirical analysis using the evaluation framework and the functional forms discussed above. The results in Chapters 3 and 4, which can be interpreted in the evaluation context (as discussed in Section 2.4), provide further examples of empirical work.

The data corresponds to the Greater Buenos Aires dataset described in Section 1.2 (page 22) – a series of rotating panels over the 1995-2002 period with twelve cohorts of households with four observations each (i.e. T = 4). The evaluation functions and stability equivalents defined above are applied to the equivalised and normalised income of these households, given by y_{it} in Equation 1.3 (page 27).

The simplest analysis can be carried out over the population average of \tilde{y}_{se} , depicted in Figure 2.6 for each of the twelve cohorts. The evaluation functions in this Figure are the CRVA (Equation 2.15), CAVA (Equation 2.17) and the extreme aversion (Equation 2.21) functions, while the average of income over time (Equation 2.19) is used as the benchmark case. For the CRVA and CAVA formulations, the parameters ρ and η are set to 2, a value adopted for empirical analysis in Chapter 3 and by Ligon and Schechter



Figure 2.6: Variability Adjusted Measures of Income for Different Evaluation Functions, Greater Buenos Aires, 1995-2002

Source: Author's estimations based on EPH household survey data (INDEC).

74

(2003), among others.¹⁹ This example concentrates on different functional forms, and thus the parameter δ in Equation 2.13 is set to 1, resulting in $\Delta(t) = 1/T$.

As described in Chapter 1, incomes are normalised by their contemporaneous poverty lines so their unit is the poverty line. The four variability adjusted measures and the average income in Figure 2.6 follow the basic trends described in Section 1.3 (page 29), confirming the highly pro-cyclical nature of household income. Notably, the difference between the average of income over the four periods in which households are observed (bold solid line) and its minimum (solid line) is quite sizeable at about three quarters of the poverty line. This indicates the presence of strong within-panel fluctuations in household income. Moreover, the difference between the two is large even in periods of income growth, for instance during the years 1996-1998.

This "minimum" stability equivalent can be interpreted as the result of an extreme aversion evaluation function, while the average income represents no aversion and the CRVA and CAVA constitute intermediate cases (see the diagram in Figure 2.5). This implies that in Figure 2.6 the stability equivalents based on these two formulations fluctuate between the average and the minimum. On average, the difference between the stability equivalent given by the CRVA function with $\rho = 2$ and the average income is around a quarter of the poverty line, while the difference between the latter and \tilde{y}_{se} based on the CAVA with $\eta = 2$ is about half of this unit. These differences represent the population averages of the absolute variability premium defined in Equation 2.8, and they are relatively large with respect to the average income, which fluctuates between around 3 and 3.25 times the poverty line. Finally, while the four measures tend to move similarly, the CRVA is more sensitive to increases and decreases in the average income over time, magnifying its fluctuations.

Another type of empirical analysis based on the evaluation framework is presented in Table 2.1, which depicts the evolution of the relative variability premium Π (defined in Equation 2.9) by quintile of average income, based on a CRVA with $\rho = 2$ and no discounting. The advantage of this for-

¹⁹Section 3.3 (page 89) discusses the range of plausible values and the sensitivity of measures of this type with respect to ρ .

Table 2.1: Relative Variability Premium by Quintile of Mean Income, Isoelastic Evaluation Function with Aversion Parameter=2, Greater Buenos Aires, 1995-2002

Cohort	Bottom Quintile	Second Quintile	Third Quintile	Fourth Quintile	Top Quintile	Overall
95-1 to 96-2	16.7%	11.3%	9.0%	10.4%	7.8%	11.0%
95-2 to 97-1	25.5%	9.4%	8.2%	11.2%	8.2%	12.4%
96-1 to 97-2	22.9%	14.3%	9.5%	10.0%	7.5%	12.7%
96-2 to 98-1	19.2%	11.9%	9.8%	9.4%	8.8%	11.8%
97-1 to 98-2	24.6%	8.6%	7.6%	6.9%	6.9%	11.0%
97-2 to 99-1	22.1%	10.4%	7.2%	7.4%	5.3%	10.4%
98-1 to 99-2	24.4%	11.0%	8.1%	7.9%	5.2%	11.2%
98-2 to 00-1	21.9%	12.8%	7.2%	5.6%	7.1%	10.9%
99-1 to 00-2	24.3%	10.1%	7.3%	5.2%	7.4%	10.8%
99-2 to 01-1	22.2%	10.0%	10.1%	7.5%	5.1%	10.9%
00-1 to 01-2	20.7%	15.2%	7.6%	5.8%	7.0%	11.2%
00-2 to 02-1	34.9%	19.7%	13.2%	12.2%	9.8%	17.9%
Overall	23.3%	12.1%	8.7%	8.3%	7.2%	11.8%
Source: Author	's estimatio	ns based on	EPH house	ehold surve	y data (IND	EC).

mulation is that the relative variability premium is constant with respect to proportional changes in the income vector **y** when $\delta = 1$, so that differences in its value at different points of the income distribution reflect the differential impact of income fluctuations as a proportion of total income. As can be appreciated from the Table, the poorest quintile bears the highest level of fluctuations in relative terms, with values of around 20 and 25 percent of the average income, and with a peak of almost 35 percent in the period corresponding to the 2001-2002 crisis (see Section 1.3, page 29, for details). The second quintile also has a relatively high level of the variability premium at around 12 percent, but for the three richest groups a pattern is not clearly discernible, remaining between 7 and 9 percent on average.

Figure 2.7 presents the population squared poverty gap for the different evaluation functions considered above, with $\rho = 2$, $\eta = 2$ and $\delta = 1$. These poverty measures are based on the stability equivalent incomes presented in Figure 2.6: as expected, the order of the series is reversed with respect to





e: Author's estimations based on EPH household survey data (INDEC).

77

		Observed Income					Stability Equivalent, CRVA α=2			
Income Profile	t=1	t=2	t=3	t=4	σ	δ=1	δ=0.9	δ=0.5	δ=0.1	
Flat	1	1	1	1	0	1.00	1.00	1.00	1.00	
"Early" MPS	1.5	0.5	1	1	0.35	0.86	0.85	0.90	0.99	
"Late" MPS	1	1	1.5	0.5	0.35	0.86	0.83	0.69	0.53	
Increasing	0.5	0.75	1.25	1.5	0.40	0.83	0.88	1.14	1.46	
Decreasing	1.5	1.25	0.75	0.5	0.40	0.83	0.79	0.64	0.52	

Table 2.2: Variability Adjusted Income for Different Values of the Discount Factor, Isoelastic Evaluation Function with Aversion Parameter=2

Note: MPS refers to a mean preserving spread of income over time. The constant relative variability aversion (CRVA) function and the stability equivalent are defined in the text.

that Figure, with the highest poverty when using the minimum income over the period and the lowest when using the average over time. The difference between these two series is again sizeable, but the most notable fact from the Figure is the evolution of the CRVA series. While the averaging of incomes over time smooths income and poverty measures – a fact discussed at length in Chapter 3 – the CRVA formulation is more sensitive than the CAVA to the variability of the underlying incomes. This can be appreciated in its higher curvature at the points where the poverty measure (based on average income) changes its trend.

Finally, while the previous examples concentrated on functional forms for v and thus considered cases with no discounting, Table 2.2 illustrates the effect of different values of δ in Equation 2.13 on the evaluation of past incomes.²⁰ The left hand side panel of the Table presents five benchmark cases of income trajectories with T = 4 ($\mathbf{y} = [y_1, y_2, y_3, y_4]$) and $\bar{y} = 1$ and their standard deviations σ , while the right hand side panel reports the resulting stability equivalent based on a CRVA function with $\rho = 2$ and four values of the discount factor: $\delta \in \{1, 0.9, 0.5, 0.1\}$.

With $\delta = 1$, the discounting weight is equal to $\Delta(t) = 1/T$ for every period, which is the case for the previous applications. The first line of the Table represents a "flat" income trajectory, which results in a stability

²⁰The introduction of $\Delta(t)$ with $\delta \neq 1$ in the GBA examples given by Figures 2.6 and 2.7 shifts the level of the stability equivalents and poverty measures. Table 2.2 is more informative since it illustrates the effect of δ on household income trajectories.

equivalent (given by Equation 2.16) equal to $\tilde{y}_{se} = \bar{y} = 1$. The following rows of the Table illustrate the effect of a mean-preserving spread in **y** on \tilde{y}_{se} : with a mean of $\bar{y} = 1$, the stability equivalent falls to 0.86 (second and third rows) and 0.83 (fourth and fifth rows) as the standard deviation increases.

The comparison between the second and third (and fourth and fifth) rows of the Table illustrates the invariance of the stability equivalent with respect to the ordering of incomes y_t in y when no discounting is applied. The second and third rows represent mean-preserving spreads (MPS) of the "flat" income trajectory, the difference being that for the former the spread occurs at t = 1 and t = 2, while for the latter it occurs closer to the present at t = 3 and t = 4. The stability equivalent \tilde{y}_{se} with $\delta = 1$ is the same in both cases, since this implies that $\Delta(t) = 1/T$ and thus $V(\mathbf{y}) = V(\mathbf{y}')$ when \mathbf{y}' is a permutation of y.²¹ However, the introduction of discounting changes this invariance result: with $\delta = 0.9$, the stability equivalents are 0.85 and 0.83, respectively, and the difference between the two reflects the fact that more weight is given to the low realisation for t = 4 in the "late MPS" trajectory. The \tilde{y}_{se} resulting from $\delta = 0.5$ and $\delta = 0.1$, in turn, reflects the trade-off between mean and dispersion: while $\bar{y} = 1$ for all trajectories in the Table, the discount weights imply that \bar{y}_{Δ} changes with δ . With $\delta = 0.1$, most of the weight is placed on t = 4 and very little on t = 1, which explains the large difference in \tilde{y}_{se} between the second and third rows.²²

Finally, the fourth and fifth rows of Table 2.2 present the same incomes as an increasing and a decreasing trajectory. As discussed above, the dynamic structure of **y** does not affect the stability equivalent for $\delta = 1$, but as this parameter increases, \tilde{y}_{se} is higher for the increasing case and lower for the decreasing trajectory. This example shows that the discounting function $\Delta(t)$ incorporates the dynamics of income streams into the evaluation framework.

²¹See Cowell and Cruces (2004) for a study of the effect of mean preserving spreads in individual perceptions of risk and inequality.

²²For $\delta = 1$ and T = 4, $\Delta(t) = 0.25$ for all periods. For $\delta = 0.9$, the discounting weights are 0.21, 0.24, 0.26 and 0.29 (for t = 1, 2, 3 and 4 respectively); for $\delta = 0.5$, the weights are 0.07, 0.13, 0.27 and 0.53; and for $\delta = 0.1$, the weights are 0.001, 0.01, 0.09 and 0.90.

2.6 CONCLUSION

This Chapter has presented a framework for the evaluation of well-being based on panel data on household income. The methodology relies on an analogy with choice under uncertainty and the expected utility model to define a family of welfare-based indicators of well-being and variability over time. This is achieved by means of a two-step procedure, which involves aggregating vectors of observations over time for a household into a scalar and then studying the distribution of this aggregate.

The methodology was discussed in the context of alternative approaches, such as the transient-chronic poverty and measures of vulnerability. The evaluation framework differs from the latter in that its measures are money metric, and in that it explicitly recognises the ex-post nature of observed data, allowing the incorporation of the dynamic nature of income processes through time preferences. Moreover, an advantage of the proposed approach is its flexibility: the first step – the derivation of a summary measure of well-being over time – does not depend on an *ad hoc* statistical procedure but on an explicit normative evaluation function of past incomes. Finally, the researcher can choose the appropriate measure (poverty, inequality, etc.) for the analysis in the second step.

The empirical results from this Chapter indicated a relatively large negative effect of income fluctuations on household welfare in Argentina, assuming only moderate levels of variability aversion in line with most estimates of risk aversion in the uncertainty literature. Moreover, the large fluctuations of income over time had a significant negative effect on household welfare even during periods of aggregate growth.

The Conclusion to this Part (page 205) discusses some possible extensions to the evaluation framework by incorporating other principles beyond variability aversion. The following two Chapters shift the focus from the measurement of well-being and income fluctuations to the study of its main correlates: Chapters 3 and 4 use regression analysis to study the determinants of variations of the measures proposed in this Chapter.

CHAPTER 3

RISK ADJUSTED POVERTY IN ARGENTINA

3.1 INTRODUCTION

In the introduction to a pioneering volume on panel data on incomes, Cowell (1982) noted that these datasets "provide a unique opportunity" for the modelling of the instability of family well-being and the incidence of poverty, among other economic phenomena. Since then, ample evidence has been gathered showing that poor households have highly variable living standards, and are prone to suffering from adverse shocks. As suggested by Blundell and Preston (1998), if insurance markets were complete, households would be able to fully offset the impact of any shocks. While these perfect markets do not exist in advanced economies, the situation in developing countries is even worse since credit markets and social safety nets tend to be underdeveloped. Jalan and Ravallion (1999), for instance, find that household consumption is never fully insured against income variability in rural China, and that poorer households are not as well insured as others. Despite the importance of income risk in these processes, much of the literature concentrates on the impact of specific shocks or on movements into and out of poverty – see for instance Glewwe and Hall (1998), Dercon and Krishnan (2000), and Scott (2000), among others.

This Chapter does not rely on the analysis of insurance, shocks or poverty dynamics. Instead, it shares with Chapter 2 the aim of accounting for income variability in general, but it particularly focuses on the effects of risk on well-being. As Morduch (1994) notes, "while the economics of poverty and the economics of uncertainty are well developed, the nexus of issues at their intersection has been left relatively unexplored." This Chapter at-

CHAPTER 3. RISK ADJUSTED POVERTY

tempts to fill the void between the theories of poverty and uncertainty by following Morduch's (1994) proposal of measuring poverty in terms of the certainty equivalent of income, and develops a measure of income that reflects the disutility induced by risk. This modus operandi is similar in spirit to the model described by Cowell (1979), who defines the concept of lifetime income under uncertainty in a similar way.

This Chapter describes the empirical implementation of the certainty equivalent as a basis for adjusting incomes for risk, under the assumption that insurance and capital markets are imperfect and that individuals are risk averse. This methodology can be interpreted as a special case of the evaluation framework presented in Chapter 2 (Section 2.4, page 64). The main difference is that the discussion in the following pages relies explicitly on choice under uncertainty in the expected utility model by assuming that past observations of income for a household are representative of its future prospects.

The estimates of risk adjusted income are aggregated into risk adjusted poverty measures for the population. The procedure also allows to assess the determinants of risk adjusted income by means of regression analysis. Variations of this methodology were applied to risk adjusted measures of inequality by Makdissi and Wodon (2003), and extended to comparisons of long term relative deprivation between groups by Cruces et al. (2004).

This framework is illustrated with panel data on household income from the Greater Buenos Aires region for the period 1995-2002. The use of Argentine data is especially appropriate given the repeated shocks that have affected the country during the period, which resulted in large fluctuations in household income as described in Chapter 1. These can be interpreted as increases in the risk faced by households.

The rest of the Chapter is organized as follows. Section 3.2 presents the conceptual framework for the estimation of risk adjusted measures of income and poverty, and for the analysis of their determinants. Section 3.3 presents the empirical results using the Greater Buenos Aires panel. A brief conclusion follows.

3.2 METHODOLOGY

3.2.1 Risk adjusted income

The starting point of this methodology is the theory of choice under uncertainty in the expected utility model, discussed in Section 2.2 (page 52). A household's utility is represented by the function u, which is assumed to be differentiable, strictly increasing and strictly concave. The household faces uncertainty with respect to its future income, represented by the random variable \hat{y} . The distribution of \hat{y} is given by S possible future states of the world, each characterised by a state-contingent income y_s and its probability τ_s . The household's expected utility is defined analogously to Equation 2.1 (page 53) as:

$$U = E[u(\hat{y})] = \sum_{s=1}^{S} \tau_s u(y_s)$$
(3.1)

where *E* is the expectations operator. To avoid the problems implied by the assumption of homogeneity of preferences in a population (see the discussion in Section 2.3, page 56), the functions *U* and *u* in this Chapter are defined within a social welfare context as representing the social judgment on the welfare value of the random variable \hat{y} (Makdissi and Wodon, 2003).

As discussed in Section 2.2, the certainty equivalent income, \tilde{y}_{ce} , is the amount of income that, if received with certainty, would provide the same level of utility as the expected utility of \hat{y} in Equation 3.1. With *S* possible states of nature, \tilde{y}_{ce} can be implicitly defined as in Equation 2.3 (page 54):

$$u(\tilde{y}_{ce}) = \sum_{s=1}^{S} \tau_s u(y_s) \tag{3.2}$$

The certainty equivalent \tilde{y}_{ce} depends on the shape of u, and it is a function of the random variable \hat{y} . Section 2.2 argued that \tilde{y}_{ce} was better suited than the expectation of \hat{y} as a measure of well-being, since the latter is by definition not affected by mean-preserving spreads in the distribution of \hat{y} , while the certainty equivalent is a decreasing function of \hat{y} 's dispersion. This desirable property for \tilde{y}_{ce} reflects the principle of risk aversion, given in this framework by the assumed concavity of u.

The certainty equivalent \tilde{y}_{ce} is determined by its underlying utility function u, and its properties as a measure of well-being depend entirely on

those of *u*. The methodology presented in this Chapter relies on the Constant Relative Risk Aversion (CRRA) utility function, which is widely used in the risk literature and was discussed in the evaluation framework context in Section 2.5 (page 69). This function is increasing in its argument (reflecting non-satiation) and concave (reflecting risk aversion):

$$u(y) = \begin{cases} \frac{y^{1-\rho}}{1-\rho} & \text{if } \rho \neq 1\\ \ln y & \text{if } \rho = 1 \end{cases}$$
(3.3)

The sensitivity parameter ρ represents the Arrow-Pratt measure of relative risk aversion for this function, which is constant for all values of y. This property is the main reason for the widespread use of the isoelastic formulation in the economics of risk: as discussed in Section 2.5, its implication in terms of the tolerable degree of risk at different levels of income is intuitively sound and empirically plausible (Browning and Lusardi, 1996). It can be argued that other formulations might better capture risk aversion at certain points in the distribution: a constant absolute risk aversion formulation, for instance, might be better suited for the limited range of income between zero and the poverty line. However, the purpose of this Chapter is to derive risk adjusted measures of income for all levels, and the CRRA formulation results in a parsimonious representation of utility along the income distribution.

3.2.2 Identifying assumption and risk adjusted poverty measures

The previous discussion highlighted the relevance of a CRRA utility function for the certainty equivalent income \tilde{y}_{ce} as a measure of income adjusted for the disutility of risk. The elements of Equations 3.1 and 3.2, however, are inherently unobservable, and thus neither cross section nor panel data can be used for their empirical implementation without further assumptions on the nature of the underlying data generating process.

Some studies based on cross section data rely on the assumption that a household's distribution of future states of nature can be inferred from the observed distribution for the whole population (Elbers and Gunning, 2003; Ligon and Schechter, 2004, discuss alternative assumptions employed in the literature). Following Makdissi and Wodon (2003), Cruces et al. (2004) and Ligon and Schechter (2003), the methodology presented in this Chapter relies on an operational version of Equation 3.1 derived from a stationarity identifying assumption, which is only possible to make when repeated observations are available for every unit. The stationarity assumption exploits the panel dimension of the data, allowing for heterogeneity between households. The identification relies on the assumption that the underlying income process for each household is stable: the probability distribution of income for a household – the distribution of \hat{y} – is identical at every point in time. Every past observation represents a typical outcome for the household, and its past experience can be extrapolated to represent the distribution of prospects at any future period (Ligon and Schechter, 2004).

Under stationarity, the series of observed past incomes for a household i, $\mathbf{y}_i = [y_{i1}, ..., y_{iT}]$, represents a set of equally probable draws from the distribution of possible incomes for i. It is thus an empirical distribution of \hat{y}_i , where each y_{it} in \mathbf{y}_i is a possible state of the world y_s in Equations 3.1 and 3.2, with associated probability $\tau_s = 1/T$.

The certainty equivalent income for a household *i* can then be represented in terms of observed quantities – *i*'s stream of past incomes y_i – as:

$$u(\tilde{y}_{ce_i}) = \frac{1}{T} \sum_{t=1}^{T} u(y_{it})$$
(3.4)

A common practice in the risk literature (see, for instance, Newbery and Stiglitz, 1981a) is to use a Taylor approximation for the certainty equivalent implicitly defined in Equation 3.4, given by:

$$\tilde{y}_{ce_i} \approx \bar{y}_i - \frac{1}{2} R_A(\rho) \sigma_{y_i}^2 \tag{3.5}$$

where \bar{y}_i is *i*'s mean income over the whole period under consideration, $\sigma_{y_i}^2$ its variance and $R_A(\rho)$ is the Arrow-Pratt measure of absolute risk aversion, equal to $\frac{\rho}{\bar{y}_i}$ for the CRRA formulation. The problem with the Taylor approximation is that it is only valid for small levels of risks (Pratt, 1964), and thus is likely to produce biased results for high values of ρ . Given the analytical convenience of the isoelastic CRRA formulation, this Chapter relies on the exact measure of risk adjusted income implied by Equations 3.3 and 3.4,

which is a function of ρ and is equal to:

$$\tilde{y}_{ce_i}(\rho) = \begin{cases} \left[\frac{1}{T} \sum_{t=1}^T y_{it}^{1-\rho}\right]^{\frac{1}{1-\rho}} & \text{if } \rho \neq 1\\ y_i = \prod_{t=1}^T y_{it}^{1/T} & \text{if } \rho = 1 \end{cases}$$
(3.6)

It is evident that the formulation of the certainty equivalent in Equation 3.6 is identical to the stability equivalent with a constant variability aversion evaluation function given in Equation 2.15 (page 70). This derives from the stationarity assumption, which expresses the general formulation of \tilde{y}_{ce} in Equation 3.2 in terms of observed quantities in Equation 3.4. The latter is observationally equivalent to Equation 2.7 (page 60) with no discounting ($\Delta(t) = 1/T$), despite the conceptual differences between the two discussed in Section 2.2: while one estimates the effects of ex-ante risk, the other captures the negative impact of ex-post fluctuations. As noted in Chapter 2 (Section 2.4), alternative approaches sometimes result in the same type of empirical implementations when they are based on the same observed incomes **y**.

The definition of the certainty equivalent income in Equation 3.6 provides a straightforward way of implementing risk adjusted poverty measures: \tilde{y}_{ce} is a money metric of well-being, and thus represents a risk adjusted measure of income. The components of \mathbf{y}_i are given by the equivalised and normalised incomes y_{it} of Equation 1.3 (page 27), so that households with $y_{it} < 1$ are considered poor at time t. Instead of focusing on this static condition, the methodology presented in this Chapter relies on the distribution of certainty equivalents \tilde{y}_{ce} to define risk adjusted measures of poverty.

These measures are modified versions of the additive poverty indices of the *FGT* class defined in Equation 1.6 (page 35), and rely on the *FGT* evaluated at the risk adjusted incomes \tilde{y}_{ce} . The new class of indices is thus implicitly a function of the distribution of the vectors **y** and of the risk aversion parameter ρ :

$$FGT_{RA}\left(\mathbf{y},\rho,\alpha\right) = \frac{1}{N}\sum_{i=1}^{N}\left[\max(1-\tilde{y}_{ce_{i}},0)\right]^{\alpha}$$
(3.7)

where N is the total number of households and α can be interpreted as a

sensitivity parameter to the depth of poverty: with $\alpha = 0$, Equation 3.7 defines a risk adjusted headcount, and $\alpha = 1$ and $\alpha = 2$ define the risk adjusted poverty gap and squared poverty gap respectively. The special case of $\rho = 0$ corresponds to the *FGT* measure evaluated at the mean of each household's income over time, *FGT* (\bar{y}, α) = $\frac{1}{N} \sum_{i=1}^{N} [\max(1 - \bar{y}_i, 0)]^{\alpha}$.

The measures defined in Equation 3.7 can be compared to Jalan and Ravallion's (1998) transient-chronic poverty decomposition, covered in Section 2.4 and applied to the Greater Buenos Aires panel in the next Chapter. A virtue of this decomposition is that it presents the contribution of income variability to an overall measure of poverty, while the risk adjusted measure of poverty based on the certainty equivalent of income seems to merge the level and variability of income into one measure. However, a comparison between risk adjusted poverty with $\rho = 0$ (corresponding to *FGT* (\bar{y} , α)) and $\rho > 0$ allows an analysis which is similar in spirit to Jalan and Ravallion's (1998). This comparison captures the degree of poverty that is due to income risk and, as argued in Section 2.4 for the stability adjusted measures, the parameter ρ in the risk adjusted methodology allows for an extra degree of freedom over the transient-chronic approach.

Another characteristic of the risk adjusted poverty measures is their sensitivity to fluctuations around the poverty line. With standard poverty measurement procedures, a household with income just above the poverty line during the *T* periods in which it is observed would not be considered poor. However, the risk aversion implicit in \tilde{y}_{ce} means that the same household may have a positive value of risk adjusted poverty if there is enough variation in income. Thus, even with an average just above the poverty line, the adjustment for risk might result in $\tilde{y}_{ce} < 1$ for some value of ρ .

Finally, a potential problem arising from the use of panel data is that a trend in income over time may be interpreted as risk, since it leads to fluctuations in y_{it} if incomes are not normalised (for instance, by the contemporaneous population mean) to neutralise the effect of a common trend. Moreover, this could invalidate the stationarity assumption, although it is possible to assume stationarity of normalised incomes. Cruces et al. (2004) discuss these issues in the context of a nine-year panel from the United Kingdom, where the use of relative poverty measures corrects the impact of growth in household income. In the present application to the GBA panel,

CHAPTER 3. RISK ADJUSTED POVERTY

however, this is not much of an issue since each cohort spans a relatively short period of 1.5 years, and as discussed in Chapter 1 there is no uniform trend in household income over the whole period.

3.2.3 Determinants of risk and risk adjusted income

Besides the estimation of risk adjusted measures of poverty, the derivation of the certainty equivalent from observed data allows to study its main determinants and the correlates of risk. Poverty profiles are usually based on the effect of different individual and household characteristics on the probability of being poor, obtained by linear regression with the adult equivalent income (or its logarithm) as the dependent variable. It is then possible to infer the probability of being poor while avoiding the specification problems that occur with probits and logits (Ravallion and Wodon, 1999).

The approach in this Chapter is similar, using the logarithm of the risk adjusted income as the dependent variable in the regressions. Denoting by X_i the vector of characteristics for a household *i*, and by \bar{y}_i its mean income over time, a comparison of the effect of the components of X_i on average and risk adjusted incomes can be estimated by means of the following system:

$$\log \bar{y}_i = X'_i \beta + \varepsilon_i$$

$$\log \bar{y}_{ce_i} = X'_i \beta_{RA} + \varepsilon_{RA_i}$$
(3.8)

where ε_i and ε_{RA_i} are regression error terms, and X_i contains a constant. Differences in the parameters in β and β_{RA} can be interpreted as differential impacts of the underlying variables on well-being as captured by \bar{y} and \tilde{y}_{ce} .

While these two regressions could be estimated independently by Ordinary Least Squares (OLS), the analysis presented in the following Section exploits the theoretical relation between expected income and the certainty equivalent, and estimates the two equations jointly by means of Seemingly Unrelated Regression (SUR) techniques. The insight arises from the definition of the risk premium in choice under uncertainty, given by $\pi_r = \bar{y} - \tilde{y}$, which represents a measure of the impact of risk on well-being – see the discussion of the variability premium π_v (Equation 2.8, page 61), the counterpart of π_r in the evaluation framework, in Section 2.3. The difference of the logs of average and risk adjusted incomes can be interpreted as a "logarithmic" version of the risk premium.

Since the same set of independent variables X_i are used in the two regressions in Equation 3.8, SUR provides the same coefficients β and β_{RA} as OLS. The advantage of using SUR to estimate the system of Equation 3.8 is that it allows for a formal test for the difference in coefficient estimates between the two regressions, $\beta - \beta_{RA}$. Moreover, since the Equations in 3.8 are linear, the estimation of the system is analogous to a regression with the "logarithmic" risk premium as the dependent variable:

$$\log \bar{y}_i - \log \tilde{y}_{ce_i} = X_i' \beta_{RP} + \varepsilon_{RP_i}$$
(3.9)

where $\beta_{RP} = \beta - \beta_{RA}$. Testing for differences in the coefficients estimated by SUR is equivalent to testing for the statistical significance of the β_{RP} parameters in the risk premium regression defined in Equation 3.9. The advantage over computing Equation 3.9 directly is that by definition $0 \le \tilde{y}/\bar{y} \le 1$, which implies that the error term in Equation 3.9 might not have a normally distributed error term.

The individual regressions in Equation 3.8 provide an assessment of the effect of each characteristic on average and risk adjusted incomes, while the SUR test of differences in coefficients reveals the impact of these characteristics on the risk faced by the household. A positive and statistically significant value of $\beta - \beta_{RA}$ implies that the related independent variable contributes to an increase in the "logarithmic" risk premium. This provides a straightforward interpretation of the impact of various variables on income and risk: characteristics with a positive value of $\beta - \beta_{RA}$ are associated with higher risk and with a lower level of utility.

3.3 RISK ADJUSTED POVERTY: AGGREGATES AND DETERMINANTS

3.3.1 Descriptive results

The risk adjusted framework described in the previous Section can be applied to the study of poverty in Argentina using the EPH data. This Section exploits the longitudinal dimension of the survey with the Greater Buenos Aires rotating panel, described at length in Section 1.2 (page 22). The applications are based on the fifteen waves corresponding to the period May

1995-May 2002, which result in twelve cohorts as illustrated in Table 1.2 (page 61). Each cohort represents a group of households that are observed over the same four waves: all observations in a cohort start at the same wave and span the same period, and they are indicated by the waves they follow (99-2 to 01-1, for instance, means that the cohort covers from the second (i.e., October) wave of the EPH in 1999 until the first wave (i.e., May) of 2001).

As discussed in Section 1.2, some aggregates of household income are equal to zero but still considered valid by INDEC. Of almost 5,500 households in the final panel sample, only two reported zero incomes in all four waves, and just 1 percent of the 21,788 observations of income were equal to zero. To minimize the loss of potentially important observations, the cases with valid zeroes were assigned a token value of 1 percent of the poverty line in the analysis. This correction is necessary for computational reasons, since \tilde{y}_{ce} is not defined for $y_{it} = 0$ with $\rho > 1$ (Equation 3.6). The poverty measures presented below are virtually unaffected by this procedure since the focus in this Section is on the headcount index.¹

Figure 3.1 presents the sample average by cohort of the risk adjusted income \tilde{y}_{ce} of Equation 3.6 for $\rho = 0, 1, 2, \text{ and } 4$. The value corresponding to $\rho = 0$ represents the average of the normalised adult equivalent income, defined in Equation 1.3 – the unit of the Figure is thus the contemporaneous poverty line.

The evolution of risk adjusted income by cohort is similar to that of equivalent real income by wave (Figure 1.3, page 30) and of variability adjusted measures for different evaluation functions (Figure 2.6, page 74). The downward trend between the first two cohorts reflects the negative effect of the contagion from the Mexican crisis of 1995, while the recovery from this crisis explains the upward tendency for the following four cohorts. From the peak of the 97-2/99-1 cohort, the trend is almost continuously negative due to the prolonged recession of 1999-2001, and devaluation and subsequent crisis of 2002. The increase between the 99-2/01-1 and 00-1/01-2 corresponds to the slight slowdown of the downward trend in GDP and employment between 1999 and 2000 depicted in Figures 1.2 and 1.3 – see Section 1.1 (page 74) for

¹The results of this Section are robust to alternative token values in the 0.25-1.5 percent range. Since the lowest value of y_{it} for $y_{it} > 0$ is 1.5 percent of the poverty line, the token value of 1 percent ensures the presence of strictly positive incomes without affecting the ranking of households in the income distribution.



Figure 3.1: Risk Adjusted Normalised Income by Cohort, Greater Buenos Aires, 1995-2002

Source: Author's estimations based on EPH household survey data (INDEC).

91

CHAPTER 3. RISK ADJUSTED POVERTY

a summary of the key economic events of the period.

The differences between the risk adjusted incomes with $\rho = 0, 1, 2$, and 4 in Figure 3.1 contain valuable information about the risk faced by households in this period. The ordering of the series is due to the fact that \tilde{y}_{ce} is decreasing in ρ (Equation 3.6). It can be noted that each extra "unit" of ρ implies a reduction in the average of \tilde{y}_{ce} of 10 to 20 percent of the poverty line. These differences are not constant, which indicates the varying level of risk for different cohorts. For instance, the difference between incomes with $\rho = 0$ and $\rho = 4$ are constant at around 40 percent of the poverty line for most cohorts, but in the last one – corresponding to the 2001-2002 crisis and arguably the one with largest fluctuations in income – this difference increases to more than 50 percent of the poverty line.

Figure 3.2 presents the share of the population with risk adjusted incomes below the poverty line, corresponding to the FGT_{RA} measure with $\alpha = 0$ (Equation 3.7) based on the incomes presented in Figure 3.1.² The risk adjusted headcount is computed for $\rho = 0$, 1, 2, and 4, with $\rho = 0$ representing the measure based on average income over time. Moreover, Figure 3.2 presents the headcount computed for each cohort by pooling – but not averaging – the four observations available for each household, which can be combined since incomes y_{it} are normalised by their contemporaneous poverty lines.³

The first fact that stands out from Figure 3.2 is the difference between the cross-sectional measure (bold line) and the risk adjusted panel-based estimates: the use of the mean income ($\rho = 0$) reduces the headcount by about 5 percentage points with respect to the pooled sample estimate. This is due to the "smoothing" effect: poverty measures based on average income over time are lower than the cross-sectional figures because the averaging reduces the effect of transitory income shocks, which is not apparent

²The analysis in this Section concentrates on the headcount to emphasize the importance of changes in income around the poverty line. For the sensitivity of the "depth" of poverty based on the certainty equivalent, it should be noted that Figure 2.7 (page 77) can be interpreted as presenting the risk adjusted squared poverty gap ($\alpha = 2$ in Equation 3.7).

³A simple example with two households and two periods of time may clarify the setting. The two households have observed streams of y_{it} given by $\mathbf{y}_1 = [y_{11}, y_{12}]$ and $\mathbf{y}_2 = [y_{21}, y_{21}]$. These observations are used to compute the averages \bar{y}_1 and \bar{y}_2 , and the risk adjusted measures $\tilde{y}_{ce_1}(\rho)$ and $\tilde{y}_{ce_2}(\rho)$. Figure 3.2 presents poverty measures based on \bar{y} and \bar{y}_{ce} , but also on the cross-sectional use of the panel, which implies computing the *FGT* measure of Equation 1.6 over the pooled incomes $[y_{11}, y_{12}, y_{21}, y_{21}]$.



Figure 3.2: Risk Adjusted Measures of Poverty by Cohort, Greater Buenos Aires, 1995-2002

Author's estimations based on EPH household survey data (INDEC).

93

in Figure 3.1.⁴ While this "smoothing" effect is well known in the distributional literature (Shorrocks, 1978), the interesting feature in the Figure is the subsequent increase in the poverty measures from $\rho = 0$ (based on average income) to $\rho > 0$: the introduction of risk aversion increases poverty substantially, with the "smoothing" effect practically neutralised with $\rho = 1$. Risk adjusted poverty headcounts with $\rho = 2$ and $\rho = 4$ are consistently greater than those based on the pooled sample.

In terms of general trends, the evolution of the poverty headcounts in Figure 3.2 mirrors that of the risk adjusted incomes in Figure 3.1, with rates varying between 12-23 percent (for $\rho = 0$ and 4, respectively) for the first cohort up to a high 29-37 percent (for $\rho = 0$ and 4) for the last one. As for income, however, the impact of changes in ρ differs markedly between cohorts. For instance, for the 96-2/98-1 cohort the difference between the headcount with $\rho = 1$ and the headcount with $\rho = 2$ is relatively small, while the difference between measures with $\rho = 2$ and $\rho = 4$ is large. For the 98-2/01-1 cohort, in turn, the latter is much smaller than the period's average. The sensitivity of the headcounts to the values of ρ reflects the concentration of incomes around the poverty line: some values of ρ imply large falls in risk adjusted income, which makes a substantial number of households become poor. Finally, it can be noted that the large fall in income for the last cohort implies that the "smoothing" effect almost disappears, with the difference between the pooled sample and the headcount based on average income reduced to less than one percentage point, compared to 3 to 5 points over the period.

3.3.2 Regression analysis

This Section complements the analysis of trends in risk adjusted incomes and poverty by implementing the regressions defined in Equation 3.8. The aim is to establish the determinants of average and risk adjusted income, and to study which household characteristics are associated with larger or smaller effects of risk on well-being.

⁴By definition, the average of the pooled observations of y_{it} in a cohort and the average of \bar{y}_i (the average of y_{it} for households in the same cohort) are equal, given the linearity of the expectation operator. In the example of the previous footnote, the average of $[y_{11}, y_{12}, y_{21}, y_{21}]$ is equal to the average of $[\bar{y}_1, \bar{y}_2]$. This is not necessarily true for other functions of y, for instance for the *FGT* measures.

As discussed in the previous Section, the system of Equation 3.8 is estimated by SUR, with the logs of \bar{y}_i and \tilde{y}_{ce_i} as the dependent variables of each regression. The twelve cohorts are pooled for the regressions, resulting in a dataset of 5,446 observations (each derived from four values of y_{it} per household).

The independent variables X_i are: (a) household level variables, including the number of infants, children, youths, adults, and elderly household members, and the square of their values; whether the household head has a spouse; (b) characteristics of the household head, including gender; the age of the head and its square; migration status in the last five years; level of education (where the omitted category is incomplete primary or no formal education); whether he/she is unemployed or inactive (with employed as the omitted category); whether he/she is an employer, a self-employed worker, an informal worker,⁵ or a wage earner (omitted); the type of qualification (omitting the no qualification category);⁶ and whether he/she works in the public sector; and (c) the same set of characteristics for the spouse of the household head, when there is one. All these variables correspond to the initial conditions, that is, the values at the first observation in the cohort for each household. In addition, the regressions include controls for each of the cohorts, excluding the first one: given the degree of macroeconomic volatility over the 1995-2002 period, each cohort has been affected by different aggregate shocks, and these controls aim to capture their effects.

Table 3.1 provides summary statistics of the variables used in the estimation. The sample mean for the logarithm of average income over time is 0.754 (corresponding to 2.13 times the poverty line), while the average logarithm of risk adjusted income with $\rho = 2$ is 0.549 (corresponding to 1.73 times the poverty line). This suggests that assuming $\rho = 2$ as the coefficient for risk aversion results in a fall with respect to average income of almost 20

⁵There are several characterizations of informality in the literature – see Moreno and Roca (2002) for the discussion of alternatives in the Argentine case. The definition employed in this Chapter corresponds broadly to the International Labour Organization's recommendations: a worker is "informal" if he or she is non-professional and either self employed or a wage earner in a small firm (less than five employees), or working without a wage.

⁶Education and qualification measures might appear to be highly collinear. The correlation matrix, however, indicates a relatively low level of 0.46 between professional qualifications and higher education, with all other coefficients being lower. The results are not substantially affected by removing either group of variables from the regression.

Table 3.1: Summary	Statistics	for	the	Dependent	and	Independent	Vari-
ables, Risk	Adjusted	Inc	ome	Regressions			

Variable	Mean	Std. Dev.	Min	Max
Log of Average Income (p=0)	0.754	0.844	-4.610	3.951
Log of Risk Adjusted Income, p=0.5	0.720	0.866	-4.605	3.946
Log of Risk Adjusted Income, ρ=1	0.669	0.921	-4.605	3.942
Log of Risk Adjusted Income, 0=2	0.549	1.129	-4.605	3.933
Log of Risk Adjusted Income, $\rho=4$	0.441	1.251	-4.605	3.915
Number of infants (ages 0-5)	0.349	0.693	0	6
Number of children (ages 6-14)	0.543	0.922	0	8
Number of youths (ages 15-24)	0.586	0.907	0	6
Number of adults (ages 25-64)	1.552	0.940	0	6
Number of elderly (ages 65+)	0.395	0.677	0	4
Age of the head	51.002	16.061	15	96
Share of female headed households	0.242	0.428	0	1
Head is recent migrant	0.026	0.159	0	1
Head inactive	0.279	0.449	0	1
Head unemployed	0.069	0.254	0	1
Head: employer	0.032	0.177	0	1
Head: self-employed	0.143	0.351	0	1
Head: informal worker	0.232	0.422	0	1
Head in public sector	0.082	0.274	0	1
Head: operative (qualification)	0.350	0.477	0	1
Head: technical worker (qualification)	0.112	0.315	0	1
Head: professional worker (qualification)	0.068	0.252	0	1
Head with primary education - Complete	0.345	0.475	0	1
Head with secondary education - Incomplete	0.177	0.382	0	1
Head with secondary education - Complete	0.143	0.350	0	1
Head with further education - Incomplete	0.008	0.090	0	1
Head with further education - Complete	0.026	0.158	0	1
Head with university education	0.146	0.353	0	1
No spouse in the household	0.306	0.461	0	1
Spouse inactive	0.418	0.493	0	1
Spouse unemployed	0.048	0.215	0	1
Spouse: employer	0.007	0.081	0	1
Spouse: self-employed	0.056	0.229	0	1
Spouse: informal worker	0.103	0.303	0	1
Spouse: operative (qualification)	0.066	0.249	0	1
Spouse: technical worker (qualification)	0.050	0.218	0	1
Spouse: professional worker (qualification)	0.026	0.159	0	1
Spouse in public sector	0.044	0.206	0	1
Spouse with primary education - Complete	0.252	0.434	0	1
Spouse with secondary education - Incomplete	0.115	0.319	0	1
Spouse with secondary education - Complete	0.119	0.324	0	1
Spouse with superior education - Incomplete	0.010	0.097	0	1
Spouse with superior education - Complete	0.034	0.182	0	1
Spouse with university education	0.071	0.257	0	1
Source: Author's estimations based on EPH household su	rvev data (1	NDFC)		

percent. The interpretation of the other variables presented in Table 3.1 is straightforward: most of them are indicators, and so their means represent the share of the sample with these characteristics.

The dependent variable \tilde{y}_{ce} is a function of the constant risk aversion parameter ρ . Arrow (1970) argues on theoretical grounds that ρ should be around 1, which is consistent with the results of Chetty (2003) based on labour supply. However, Friend and Blume (1975) present empirical evidence based on portfolio holdings that the coefficient may be around 2, Hildreth and Knowles (1982) obtain estimates between 1 and 2, and in a review of the literature Cowell and Gardiner (2000) indicate a range of values from 0.5 to 4 depending on the type of evidence. To allow for a sizeable effect of risk, the regressions presented in Table 3.2 are based on \tilde{y}_{ce} computed with $\rho = 2$, which is within the boundaries in the literature. This value is used in other empirical applications based on the CRRA utility function (Ligon and Schechter, 2003). A robustness check of the regression results for alternative values of ρ is discussed below.

Table 3.2 presents the results of the regression analysis of the system of Equation 3.8 by SUR in the first two columns, and the tests for statistically significant differences in the parameter estimates of the two models in the third column, which corresponds to Equation 3.9.

The first column of the table corresponds to the logarithm of mean income over time ($\rho = 0$). The results are fairly intuitive. Larger households (whether having more infants, children, or adults) tend to have lower values of average income, but the impact is decreasing at the margin. In the case of elderly members, the impact on average income is not statistically significant, although part of its effect may be captured by the inactivity indicators for the household head or spouse, which have negative coefficients. Households with older and/or female heads tend to have higher incomes, although the coefficient on female heads is only significant at the 10 percent level. There seems to be no statistically significant impact from the migrant status of the head.

The fact that the head or its spouse are inactive or unemployed is associated with lower average income. The status of either of them as an employer or a self-employed worker does not have a statistically significant impact on average income, although the signs of the coefficients are posi-

Table 3.2: Determinants o	of Log A	Average	Income	and l	Log Risk 1	Adjusted	In-
come							

	Log average	Log risk- adjusted	Difference [1]-[2]
	meome [1]	income [2]	and P-value
Household characteristics			
Number of infants (ages 0-5)	-0.272	-0.297	0.024
	[0.02661]***	[0.04225]***	0.378
Number of infants squared	0.028	0.030	-0.002
	[0.01007]***	[0.01600]*	0.819
Number of children (ages 6-14)	-0.374	-0.365	-0.009
	[0.01826]***	[0.02899]***	0.640
Number of children squared	0.040	0.034	0.006
	[0.00539]***	[0.00855]***	0.258
Number of youths (ages 15-24)	-0.209	-0.187	-0.023
	[0.01913]***	[0.03038]***	0.251
Number of youths squared	0.038	0.042	-0.004
	[0.00630]***	[0.01000]***	0.521
Number of adults (ages 25-64)	-0.096	-0.085	-0.011
	[0.02706]***	[0.04295]**	0.695
Number of adults squared	0.023	0.027	-0.004
	[0.00642]***	[0.01019]***	0.559
Number of elderly members (ages 65+) 0.016	0.115	-0.099
	[0.03901]	[0.06194]*	0.014 **
Number of elderly members, squared	-0.021	-0.039	0.018
	[0.01667]	[0.02646]	0.293
No spouse in the household	0.103	0.059	0.044
	[0.05138]**	[0.08157]	0.402
Characteristics of the head			
Age	0.007	0.009	-0.002
	[0.00355]**	[0.00563]*	0.570
Age squared	0.000	0.000	0.000
	[0.00004]	[0.00006]	0.998
Female head	0.051	0.079	-0.028
	[0.02658]*	[0.04219]*	0.312
Recent migrant	-0.010	-0.098	0.089
	[0.04660]	[0.07398]	0.065 *
Inactive	-0.134	-0.157	0.022
	[0.04082]***	[0.05242]***	0.512
Unemployed	-0.440	-0.866	0.426
	[0.05073]***	[0.05790]***	0.000 ***
Employer	0.071	0.020	0.051
	[0.04619]	[0.07333]	0.287
Self-employed	-0.051	0.047	-0.098
	[0.02762]	[0.04386]	0.401
Informal Worker	-0.147	-0.224	0.077
	[0.02540]***	[0.04032]***	0.003 ***
Public Sector Worker	-0.063	-0.035	-0.028
	[0.02927]**	[0.04646]	0.350
Job Qualification: Operative	0.121	0.158	-0.037
	[0.03997]	[0.03914]***	0.143
Job Qualification: Technical	0.336	0.372	-0.036
	[0.05132]	[0.05480]***	0.315
Job Qualification: Professional	0.244	0.698	-0.454
	[0.04329]***	[0.06873]***	0.424
Primary Education – Complete	0.156	0.196	-0.039
	[0.02288]***	[0.03632]***	0.096 *
Secondary Education - Incomplete	0.319	0.345	-0.027
	[0.02705]***	[0.04294]***	0.341
Secondary Education - Complete	0.534	0.604	-0.070
	[0.03460]***	[0.04670]***	0.021 **
Further Education – Incomplete	0.217	0.976	-0.759
······	[0.08275]***	[0.13137]***	0.009 ***
Further Education – Complete	0.731	0.833	-0.103
Complete	[0.05243]***	[0.08285]***	0.056 *
University Education	0.423	0.847	-0.423
,	[0.04477]***	[0.05398]***	0.049 **

Intervent Intervent Intervent Characteristics of the spouse .0.244 .0.331 0.087 Inactive .0.244 .0.331 0.087 Unemployed .0.355 .0.543 0.187 Employer .0.06471*** [0.08033]*** 0.000 Employed .0.244 .0.126 0.082 [0.09697] [0.15395] 0.413 .0.663 Self-employed .0.027 .0.122 .0.025 [0.04175] [0.06629]** [0.06347] .0.496 Job Qualification: Operative .0.052 .0.024 .0.015 [0.02465]*** [0.0647]** [0.06347] .0.496 Job Qualification: Technical .0.075 .0.990 -0.015 [0.04452]*** [0.0642]*** [0.06437]** [0.06437]** Job Qualification: Technical .0.077 .0.076 .0.077 Job Qualification: Complete .0.152 .0.207 .0.011 [0.04416] [0.0711] .0.816 Primary Education - Complete	(continued)	Log Income	Log risk- adjusted	Difference [1]-[2]	
Characteristics of the spouse 0.244 0.331 0.087 Inactive -0.244 -0.331 0.040 ••• Unemployed -0.355 -0.543 0.187 0.000 Employer -0.044 -0.126 0.082 0.081 Employer -0.044 -0.126 0.082 0.024 -0.071 0.095 Informal Worker -0.097 -0.122 0.025 0.024 0.024 0.028 Job Qualification: Operative 0.052 0.024 0.028 0.024 0.028 Job Qualification: Technical 0.075 0.090 -0.015 0.06627 0.015 Job Qualification: Professional 0.062 0.244 0.418 0.06642 0.02637*** 0.0996 Public Sector Worker -0.017 -0.027 0.011 0.012 0.036 -0.076 0.02657*** 0.0660 * 0.229 0.306 -0.076 0.02657*** 0.0028*** 0.011 0.02652 **** ***** 1.026271 -0.026 <t< th=""><th>(commute)</th><th>[1]</th><th>income [2]</th><th>and P-value</th></t<>	(commute)	[1]	income [2]	and P-value	
Inactive -0.244 -0.331 0.087 (Domployed -0.355 -0.543 0.040 Employer -0.044 -0.126 0.0082 [0.09697] [0.13355] 0.413 0.036 Self-employed 0.024 -0.071 0.095 [0.04717] [0.06629] 0.636 Informal Worker -0.097 -0.122 0.025 [0.04289]*** [0.06810]* 0.568 Job Qualification: Operative 0.052 0.024 0.028 [0.04263]*** [0.06447]** [0.06417] 0.496 Job Qualification: Technical 0.075 0.024 0.028 [0.0445]*** [0.0642]*** [0.06437]* 0.496 Public Sector Worker -0.017 -0.027 0.011 [0.04416] [0.07011] 0.816 *** Secondary Education - Incomplete 0.229 0.306 -0.076 [0.03400]*** [0.03351]*** 0.028 *** Secondary Education - Complete 0.313 0.425	Characteristics of the spouse	···-·			
[0.03302]*** [0.0640]*** 0.040 ** [0.03647]*** [0.06053]*** 0.040 ** Employer -0.044 -0.126 0.082 [0.047]*** [0.0662]** 0.0413 SelF-employed 0.024 -0.071 0.095 [0.04175] [0.06629] 0.636 Informal Worker -0.07 -0.122 0.025 [0.04289]** [0.06347] 0.496 Job Qualification: Operative 0.052 0.024 0.028 Job Qualification: Technical 0.075 0.090 -0.015 Job Qualification: Professional 0.6622 0.244 0.418 [0.0642]*** [0.03451]** 0.996 Public Sector Worker -0.017 -0.027 0.011 Primary Education - Complete 0.152 0.207 -0.056 0.02857*** 0.060 ** Secondary Education - Complete 0.131 0.425 -0.112 0.028 *** Secondary Education - Complete 0.371 0.477 -0.166 0.025 Further Education - Complete	Inactive	-0.244	-0.331	0.087	
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		[0.03302]***	[0.06480]***	0.040 **	
Image: Barbon Stress of the stress	Unemployed	-0.355	-0.543	0.187	
Employer -0.044 -0.126 0.082 Enerployed 0.024 -0.071 0.095 [0.04175] [0.06629] 0.636 Informal Worker -0.097 -0.122 0.025 Job Qualification: Operative [0.02489]** [0.06310]* 0.568 Job Qualification: Technical 0.075 0.090 -0.015 [0.03452]*** [0.06317] 0.478 [0.0462]*** [0.06417]* 0.770 Job Qualification: Professional 0.662 0.244 0.418 [0.0662]*** [0.06317]* 0.996 Public Sector Worker -0.017 -0.027 0.011 [0.04535]*** [0.04535]*** 0.066 * Secondary Education - Incomplete 0.229 0.306 -0.076 [0.0350]*** [0.04535]*** 0.028 ** Secondary Education - Complete 0.313 0.425 -0.112 [0.025219*** [0.03531]*** 0.028 ** Secondary Education - Complete 0.371 0.477	•	[0.03647]***	[0.08053]***	0.000 ***	
$[0.09697]$ $[0.15395]$ 0.413 SelF-employed 0.024 -0.071 0.095 Informal Worker -0.097 -0.122 0.025 Job Qualification: Operative 0.0522 0.0244 0.023 Job Qualification: Operative 0.0552 0.0244 0.023 Job Qualification: Technical 0.075 0.090 -0.015 Job Qualification: Professional 0.6622 0.2444 0.413 Job Qualification: Professional 0.6622 0.2444 0.418 $[0.03452]^{***}$ $[0.0345]^{***}$ 0.996 Public Sector Worker -0.017 -0.027 0.0111 $[0.02857]^{***}$ $[0.0453]^{***}$ 0.0660^{**} Secondary Education - Complete 0.313 0.425 -0.112 $[0.02857]^{***}$ $[0.03531]^{***}$ 0.028^{***} Secondary Education - Incomplete 0.754 0.177 0.517 $[0.03521]^{***}$ 0.0282^{***} 0.0282^{***} 0.0262^{***} Further Education - Comple	Employer	-0.044	-0.126	0.082	
Self-employed 0.024 -0.071 0.095 Informal Worker -0.077 -0.122 0.025 Informal Worker 0.052 0.063101° 0.568 Job Qualification: Operative 0.052 0.024 0.023 Job Qualification: Technical 0.075 0.090 -0.015 Job Qualification: Professional 0.6622 0.244 0.418 [0.04645]*** $[0.03147]^{\circ}$ 0.996 Public Sector Worker -0.017 -0.027 0.011 [0.04416] $[0.07011]$ 0.816 Primary Education - Complete 0.152 0.207 -0.056 [0.02847]*** $[0.03351]^{***}$ 0.060^{***} Secondary Education - Complete 0.313 0.425° -0.112 [0.02942]**** $[0.03532]^{***}$ 0.060^{***} Further Education - Complete 0.774 0.177 0.578 [0.07953]**** $[0.03652]$ 0.625^{**} Further Education - Complete 0.777 0.511^{***} <		[0.09697]	[0.15395]	0.413	
	Self-employed	0.024	-0.071	0.095	
Informal Worker -0.097 -0.122 0.025 Job Qualification: Operative [0.04289]** [0.06310]* 0.568 Job Qualification: Technical 0.075 0.090 -0.015 [0.04453]*** [0.06437] 0.496 Job Qualification: Technical 0.075 0.090 -0.015 [0.0452]*** [0.06437] 0.770 0.011 Job Qualification: Professional 0.662 0.244 0.418 [0.0454]*** [0.0545]*** 0.996 Public Sector Worker -0.017 -0.027 0.011 [0.0454]*** [0.0454]*** 0.060 * Secondary Education - Incomplete 0.152 0.207 -0.012 [0.02857]*** [0.05492]*** 0.002 *** Further Education - Incomplete 0.754 0.177 -0.12 [0.02942]*** [0.03542]*** 0.060 * Secondary Education - Complete 0.313 0.425 -0.112 [0.02942]*** [0.03543]*** 0.166 Further Education - Incomplete 0.774 0.106 [0.03501]*** <t< td=""><td></td><td>[0.04175]</td><td>[0.06629]</td><td>0.636</td></t<>		[0.04175]	[0.06629]	0.636	
	Informal Worker	-0.097	-0.122	0.025	
Job Qualification: Operative 0.052 0.024 0.028 [0.02465]*** [0.0647] 0.496 Job Qualification: Technical 0.075 0.090 -0.015 [0.03452]*** [0.08147] 0.770 Job Qualification: Professional 0.662 0.244 0.418 [0.0642]*** [0.0645]** 0.996 Public Sector Worker -0.017 -0.027 0.011 [0.04416] [0.07011] 0.816 Primary Education - Complete 0.152 0.207 -0.056 [0.02857]*** [0.04535]*** 0.060 * Secondary Education - Incomplete 0.229 0.306 -0.076 [0.02942]*** [0.05351]**** 0.002 *** Secondary Education - Incomplete 0.371 0.447 -0.112 [0.02942]*** [0.05492]*** 0.002 *** Further Education - Incomplete 0.371 0.477 0.062 [0.02942]*** [0.03523]*** 0.002 *** Cohort controls 0.0777 0.517 0.260 [0.03653] [0.05800] 0.226 *		[0.04289]**	[0.06810]*	0.568	
$[0.02463]^{\bullet\bullet\bullet}$ $[0.06347]$ 0.496 Job Qualification: Technical 0.075 0.090 -0.015 $[0.03452]^{\bullet\bullet\bullet}$ $[0.08147]$ 0.770 Job Qualification: Professional 0.662 0.244 0.418 $[0.0642]^{\bullet\bullet\bullet}$ $[0.0643]^{\bullet\bullet\bullet}$ 0.996 Public Sector Worker -0.017 -0.027 0.011 $[0.04416]$ $[0.07011]$ 0.816 Primary Education - Complete 0.152 0.207 -0.056 $[0.02857]^{\bullet\bullet\bullet}$ $0.0606^{\bullet\bullet}$ 0.229 $0.306^{\bullet\bullet}$ -0.076^{\bullet} Secondary Education - Incomplete 0.229 $0.306^{\bullet\bullet}$ -0.076^{\bullet} $[0.02942]^{\bullet\bullet\bullet}$ $[0.05492]^{\bullet\bullet\bullet}$ $0.002^{\bullet\bullet\bullet\bullet}$ Secondary Education - Complete 0.313 0.425^{\bullet} $-0.112^{\bullet\bullet}$ $[0.02942]^{\bullet\bullet\bullet}$ $[0.05492]^{\bullet\bullet\bullet\bullet}$ $0.002^{\bullet\bullet\bullet\bullet}$ Further Education - Incomplete 0.754^{\bullet} 0.177^{\bullet} 0.578^{\bullet} $[0.07933]^{\bullet\bullet\bullet\bullet}$ $[0.03632]^{\bullet\bullet\bullet\bullet}$ $0.051^{\bullet\bullet}$ \bullet University Education 0.777^{\bullet} 0.517^{\bullet} 0.625^{\bullet} $95-2$ to 97-1 -0.038^{\bullet} -0.083^{\bullet} 0.046^{\bullet} $95-2$ to 97-1 -0.038^{\bullet} -0.042^{\bullet} 0.039^{\bullet} $96-2$ to 98-1 -0.003^{\bullet} -0.042^{\bullet} 0.039^{\bullet} $96-2$ to 98-1 -0.033^{\bullet} 0.0226^{\bullet} -0.044^{\bullet} $97-2$ to 99-1 0.021^{\bullet} 0.05731^{\bullet} 0.297^{\bullet} $97-2$ to 99-1 0.021^{\bullet} 0.05651° 0.994^{\bullet} $99-2$	Job Qualification: Operative	0.052	0.024	0.028	
Job Qualification: Technical 0.075 0.090 -0.015 $[0.03452]^{***}$ $[0.08147]$ 0.770 Job Qualification: Professional 0.662 0.244 0.418 $[0.06642]^{***}$ $[0.10545]^{**}$ 0.996 Public Sector Worker -0.017 -0.027 0.011 $[0.04416]$ $[0.07011]$ 0.816 Primary Education - Complete 0.152 0.207 -0.056 $[0.0370]^{***}$ $[0.04353]^{***}$ 0.060^{*} Secondary Education - Incomplete 0.313 0.425^{*} -0.112^{*} $[0.0370]^{***}$ $[0.035492]^{***}$ 0.002^{***} 0.028^{**} Secondary Education - Complete 0.371 0.477 0.518 $[0.0753]^{***}$ $[0.0322]^{***}$ 0.022^{***} Further Education - Complete 0.371 0.477 0.166^{*} $[0.0753]^{***}$ $[0.03232]^{***}$ 0.062^{*} $[0.0460]^{***}$ 0.033^{*} 0.046^{*} $[0.03400]^{***}$ $[0.03707]^{*}$ 0.226^{*} $[0.0416]^{*}$ 0.071^{*} 0.042^{*}		[0.02465]***	[0.06347]	0.496	
	Job Qualification: Technical	0.075	0.090	-0.015	
Job Qualification: Professional 0.662 0.244 0.418 [0.06642]*** [0.10545]** 0.996 Public Sector Worker -0.017 -0.011 [0.04416] [0.07011] 0.816 Primary Education - Complete 0.152 0.207 -0.056 [0.02857]*** [0.04535]*** 0.060 * Secondary Education - Incomplete 0.313 0.425 -0.176 [0.0370]*** [0.05492]*** 0.002 *** Secondary Education - Complete 0.371 0.477 0.578 [0.07953]*** [0.12627] 0.625 Further Education - Complete 0.371 0.477 -0.166 [0.0310]*** [0.03219]*** [0.0323]*** 0.051 * University Education 0.777 0.517 0.260 [0.03400]*** [0.07107]*** 0.042 ** Cohort controls 95-2 to 97-1 -0.038 -0.083 0.046 [0.03653] [0.05820] 0.622 * 96-1 to 97-2 -0.013 -0.084 0.071 96-2 to 98-1 -0.003 -0.042 0.039 <t< td=""><td></td><td>[0.03452]***</td><td>[0.08147]</td><td>0.770</td></t<>		[0.03452]***	[0.08147]	0.770	
$[0.06642]^{\bullet\bullet\bullet}$ $[0.10545]^{\bullet\bullet}$ 0.996 Public Sector Worker-0.017-0.0270.011 $[0.04116]$ $[0.07011]$ 0.816 Primary Education - Complete 0.152 0.207 -0.056 $[0.02857]^{\bullet\bullet\bullet}$ $[0.04335]^{\bullet\bullet\bullet}$ $0.660^{\bullet\bullet}$ Secondary Education - Incomplete 0.229 0.306 -0.076 $[0.03370]^{\bullet\bullet\bullet}$ $[0.05351]^{\bullet\bullet\bullet\bullet}$ $0.028^{\bullet\bullet\bullet}$ Secondary Education - Complete 0.313 0.425 -0.112 $[0.02942]^{\bullet\bullet\bullet}$ $[0.05492]^{\bullet\bullet\bullet}$ $0.002^{\bullet\bullet\bullet\bullet}$ Further Education - Incomplete 0.754 0.177 0.578 $[0.07953]^{\bullet\bullet\bullet\bullet}$ $[0.12627]$ 0.625 Further Education - Complete 0.371 0.477 -0.106 $[0.032519]^{\bullet\bullet\bullet\bullet}$ $[0.0360]^{\bullet\bullet\bullet\bullet}$ $0.051^{\bullet\bullet}$ $[0.03600]^{\bullet\bullet\bullet\bullet}$ $[0.03600]^{\bullet\bullet\bullet\bullet}$ 0.046 University Education 0.777 0.517 0.260 $[0.03600]^{\bullet\bullet\bullet\bullet}$ $[0.03653]$ 0.046 95-2 to 97-1 -0.038 -0.083 0.046 96-1 to 97-2 -0.013 -0.084 0.071 $[0.03651]$ $[0.03653]$ 0.042 0.039 96-1 to 98-2 0.034 0.070 0.227 $[0.03665]$ $[0.03725]$ 0.494 $97-2$ to 98-1 -0.021 0.026 -0.004 $[0.03579]$ $[0.03579]$ 0.297 $97-1$ to 98-2 -0.031 -0.070 0.227 $[0.03672]$ $[0.03579]$ $[0.03579]$ 0.994 </td <td>Job Qualification: Professional</td> <td>0.662</td> <td>0.244</td> <td>0.418</td>	Job Qualification: Professional	0.662	0.244	0.418	
Public Sector Worker -0.017 -0.027 0.011 $[0.04416]$ $[0.07011]$ 0.816 Primary Education - Complete 0.152 0.207 -0.056 $[0.02857]^{\bullet \bullet \bullet}$ $[0.04535]^{\bullet \bullet \bullet}$ 0.060 * Secondary Education - Incomplete 0.229 0.306 -0.076 $[0.03370]^{\bullet \bullet \bullet}$ $[0.0351]^{\bullet \bullet \bullet}$ 0.002 *** Further Education - Complete 0.754 0.177 0.578 $[0.07953]^{\bullet \bullet \bullet}$ $[0.12627]$ 0.625 Further Education - Complete 0.371 0.477 -0.106 $[0.05219]^{\bullet \bullet \bullet}$ $[0.03632]^{\bullet \bullet \bullet}$ 0.051 * University Education 0.777 0.517 0.260 $[0.03633]$ $[0.03633]$ 0.042 ** Cohort controls 95-2 to 97-1 -0.038 -0.083 0.046 $[0.03672]$ $[0.03633]$ 0.052 9 96-1 to 97-2 -0.013 -0.084 0.071 $[0.03672]$ $[0.03673]$ 0.297 97-1 to 98-2 0.034 0.070		[0.06642]***	[0.10545]**	0.996	
0.04416 $[0.07011]$ 0.816 Primary Education - Complete 0.152 0.207 -0.056 $[0.02857]^{\bullet \bullet \bullet}$ $[0.04535]^{\bullet \bullet \bullet}$ 0.060^{\bullet} Secondary Education - Incomplete 0.229 0.306 -0.076 $[0.03370]^{\bullet \bullet \bullet}$ $[0.05351]^{\bullet \bullet \bullet}$ $0.028^{\bullet \bullet \bullet}$ Secondary Education - Complete 0.313 0.425 -0.112 $[0.02942]^{\bullet \bullet \bullet}$ $[0.05492]^{\bullet \bullet \bullet}$ $0.022^{\bullet \bullet \bullet}$ Further Education - Incomplete 0.754 0.177 0.578 $[0.07953]^{\bullet \bullet \bullet}$ $[0.16227]$ 0.625 Further Education - Complete 0.371 0.477 -0.106 $[0.05219]^{\bullet \bullet \bullet}$ $[0.08323]^{\bullet \bullet \bullet}$ 0.061^{\bullet} University Education 0.777 0.517 0.260 $[0.03400]^{\bullet \bullet \bullet}$ $[0.03653]$ $[0.042$ \bullet 95-2 to 97-1 -0.038 -0.083 0.046 $[0.03672]$ $[0.05800]$ 0.226 96-1 to 97-2 -0.013 -0.084 0.071 $[0.03672]$ $[0.05739]$ 0.297 97-1 to 98-2 0.034 0.008 0.025 $[0.036615]$ $[0.05735]$ 0.494 97-2 to 99-1 0.021 0.026 -0.004 $[0.03579]$ $[0.05531]$ 0.904 98-1 to 99-2 -0.043 -0.070 0.027 $[0.03579]$ $[0.05575]$ 0.819 98-2 to 00-1 $[0.03575]$ 0.819 99-1 to 00-2 -0.022 -0.031 0.008 $[0.03575]$ $[$	Public Sector Worker	-0.017	-0.027	0.011	
Primary Education - Complete 0.152 0.207 -0.056 Secondary Education - Incomplete 0.229 0.306 -0.076 Secondary Education - Complete 0.313 0.425 -0.112 $[0.03370]^{***}$ $[0.05351]^{***}$ 0.002 $***$ Secondary Education - Complete 0.313 0.425 -0.112 $[0.032942]^{***}$ $[0.05492]^{***}$ 0.002 $***$ Further Education - Complete 0.754 0.177 0.578 $[0.07953]^{***}$ $[0.12627]$ 0.625 Further Education 0.777 0.517 0.260 $[0.075219]^{***}$ $[0.08323]^{***}$ 0.051 * University Education 0.777 0.517 0.260 * 95-2 to 97-1 -0.038 -0.083 0.046 * 96-1 to 97-2 -0.013 -0.084 0.071 96-2 to 98-1 -0.003 -0.042 0.039 97-1 to 98-2 0.034 0.070 0.227 97-2 to 99-1 0.021 0.026 -0.004		[0.04416]	[0.07011]	0.816	
Secondary Education - Incomplete $[0.02857]^{\bullet \bullet \bullet}$ $[0.04535]^{\bullet \bullet \bullet}$ $0.060 \bullet$ Secondary Education - Complete 0.313 0.425 -0.112 $[0.03370]^{\bullet \bullet \bullet}$ $[0.05351]^{\bullet \bullet \bullet \bullet}$ $0.002 \bullet \bullet \bullet$ Further Education - Incomplete 0.754 0.177 0.578 $[0.07953]^{\bullet \bullet \bullet \bullet}$ $[0.12627]$ 0.625 Further Education - Complete 0.774 0.177 0.517 $[0.07953]^{\bullet \bullet \bullet \bullet}$ $[0.03219]^{\bullet \bullet \bullet \bullet}$ $0.002 \bullet \bullet \bullet \bullet$ $[0.07953]^{\bullet \bullet \bullet \bullet}$ $[0.08323]^{\bullet \bullet \bullet \bullet}$ $0.051 \bullet \bullet$ University Education 0.777 0.517 0.260 $[0.03400]^{\bullet \bullet \bullet \bullet}$ $[0.03400]^{\bullet \bullet \bullet \bullet}$ $0.042 \bullet \bullet$ Cohort controls 0.0777 0.013 0.042 95-2 to 97-1 -0.038 -0.083 0.046 $[0.03672]$ $[0.05829]$ $0.622 \bullet$ 96-1 to 97-2 -0.013 -0.084 0.071 $[0.03672]$ $[0.05725]$ 0.494 97-1 to 98-2 0.024 0.039 $[0.03615]$ $[0.05725]$ 0.494 97-2 to 99-1 0.021 0.026 -0.004 $[0.03579]$ $[0.05634]$ 0.459 98-2 to 00-1 -0.022 -0.031 0.008 $[0.03579]$ $[0.05775]$ 0.819 98-2 to 00-1 -0.022 -0.031 0.008 $[0.03579]$ $[0.05634]$ 0.459 98-2 to 00-1 -0.022 -0.031 0.008 $[0.03579]$ $[0.05675]$ 0.975 $0-11$ $-$	Primary Education - Complete	0.152	0.207	-0.056	
Secondary Education - Incomplete 0.229 0.306 -0.076 [0.03370]*** [0.05351]*** 0.028 ** Secondary Education - Complete 0.313 0.425 -0.112 [0.02942]*** [0.05492]*** 0.022 *** Further Education - Incomplete 0.754 0.177 0.578 [0.07953]*** [0.12627] 0.625 Further Education - Complete 0.371 0.477 -0.106 [0.05219]*** [0.08323]*** 0.051 * University Education 0.777 0.517 0.260 [0.03400]*** [0.07107]*** 0.042 ** Cohort controls 95-2 to 97-1 -0.038 -0.083 0.046 [0.03653] [0.05800] 0.226 96-1 to 97-2 -0.013 -0.084 0.071 [0.3672] [0.05829] 0.062 * 96-2 to 98-1 -0.003 -0.042 0.039 [0.03605] [0.05739] 0.297 97-1 to 98-2 0.034 0.007 0.025 [0.03569] [0.05631] 0.944 97-2		[0.02857]***	[0.04535]***	0.060 *	
Secondary Education - Complete $[0.03370]^{\bullet \bullet \bullet}$ $[0.05351]^{\bullet \bullet \bullet}$ $0.028 \bullet^{\bullet \bullet}$ Secondary Education - Incomplete 0.313 0.425 -0.112 Further Education - Incomplete 0.754 0.177 0.578 $[0.07953]^{\bullet \bullet \bullet}$ $[0.12627]$ 0.625 Further Education - Complete 0.371 0.477 -0.106 $[0.05219]^{\bullet \bullet \bullet}$ $[0.08323]^{\bullet \bullet \bullet}$ $0.051 \bullet$ University Education 0.777 0.517 0.260 $[0.05219]^{\bullet \bullet \bullet}$ $[0.07107]^{\bullet \bullet \bullet}$ $0.042 \bullet^{\bullet \bullet}$ Cohort controls95-2 to 97-1 -0.038 -0.083 0.046 96-1 to 97-2 -0.013 -0.084 0.071 $[0.03653]$ $[0.05800]$ 0.226 96-2 to 98-1 -0.003 -0.042 0.039 $[0.03615]$ $[0.05739]$ 0.297 97-1 to 98-2 0.034 0.008 0.025 $[0.03660]$ $[0.05725]$ 0.494 97-2 to 99-1 0.021 0.026 -0.004 $[0.03579]$ $[0.05631]$ 0.904 98-1 to 99-2 -0.032 -0.037 0.010 $[0.03579]$ $[0.05634]$ 0.459 98-2 to 01-1 -0.028 -0.031 0.008 $[0.03579]$ $[0.05635]$ 0.975 $99-1$ to 00-2 -0.022 -0.031 0.008 $99-2$ to 01-1 -0.107 -0.108 0.001 $[0.03579]$ $[0.05675]$ 0.819 $99-2$ to 01-1 -0.107 -0.127 0	Secondary Education - Incomplete	0.229	0.306	-0.076	
Secondary Education - Complete 0.313 0.425 -0.112 Further Education - Incomplete 0.754 0.177 0.578 Further Education - Complete 0.754 0.177 0.625 Further Education - Complete 0.371 0.477 -0.106 $[0.05219]^{\bullet \bullet \bullet}$ $[0.08323]^{\bullet \bullet \bullet}$ $0.051 \bullet$ University Education 0.777 0.517 0.260 $[0.03400]^{\bullet \bullet \bullet}$ $[0.07107]^{\bullet \bullet \bullet \bullet}$ $0.042 \bullet \bullet$ Cohort controls 95-2 to 97-1 -0.038 -0.083 0.046 $[0.03653]$ $[0.03653]$ $[0.05800]$ 0.226 96-1 to 97-2 -0.013 -0.084 0.071 $[0.03672]$ $[0.05829]$ $0.062 \bullet$ 96-2 to 98-1 -0.003 -0.042 0.039 $[0.03615]$ $[0.05739]$ 0.297 $97-2$ to 99-1 0.021 0.026 -0.004 $[0.03559]$ $[0.05651]$ 0.904 $98-2$ to $09-1$ -0.028 -0.037 0.010		[0.03370]***	[0.05351]***	0.028 **	
$[0.02942]^{\bullet\bullet\bullet}$ $[0.05492]^{\bullet\bullet\bullet}$ $0.002 ***$ Further Education - Incomplete 0.754 0.177 0.578 $[0.07953]^{\bullet\bullet\bullet}$ $[0.12627]$ 0.625 Further Education - Complete 0.371 0.477 -0.106 $[0.08323]^{\bullet\bullet\bullet}$ $0.051 *$ $0.051 *$ University Education 0.777 0.517 0.260 $[0.03400]^{\bullet\bullet\bullet}$ $[0.07107]^{\bullet\bullet\bullet}$ $0.042 **$ Cohort controls95-2 to 97-1 -0.038 -0.083 0.046 $[0.03653]$ $[0.05800]$ 0.226 96-1 to 97-2 -0.013 -0.084 0.071 $[0.03672]$ $[0.05829]$ $0.662 *$ 96-2 to 98-1 -0.003 -0.042 0.039 $[0.3615]$ $[0.05739]$ 0.297 97-1 to 98-2 0.034 0.008 0.025 $[0.3666]$ $[0.05725]$ 0.494 97-2 to 99-1 0.021 0.026 -0.004 $[0.3559]$ $[0.05651]$ 0.904 98-1 to 99-2 -0.043 -0.070 0.277 $[0.3579]$ $[0.05634]$ 0.459 98-2 to 00-1 $[0.03579]$ $[0.05632]$ 0.797 99-1 to 00-2 -0.022 -0.031 0.008 $[0.03575]$ $[0.05675]$ 0.819 99-2 to 01-1 -0.107 -0.108 0.001 $[0.03635]^{\bullet\bullet\bullet\bullet}$ 0.975 0.748 $0-2 to 02-1$ -0.164 -0.316 0.152 $(0.0285]^{\bullet\bullet\bullet\bullet}$ $(0.0591]^{\bullet\bullet\bullet\bullet\bullet$ $0.000 ***$ <t< td=""><td>Secondary Education - Complete</td><td>0.313</td><td>0.425</td><td>-0.112</td></t<>	Secondary Education - Complete	0.313	0.425	-0.112	
Further Education - Incomplete 0.754 0.177 0.578 Further Education - Complete 0.371 0.477 -0.106 [0.05219]*** [0.08323]*** 0.051 * University Education 0.777 0.517 0.260 [0.03400]*** [0.07107]*** 0.042 ** Cohort controls 9 9 0.777 0.517 96-1 to 97-2 -0.013 -0.083 0.046 [0.03653] [0.05800] 0.226 96-1 to 97-2 -0.013 -0.084 0.071 [0.03672] [0.05829] 0.062 * 96-2 to 98-1 -0.003 -0.042 0.039 [0.03615] [0.05739] 0.297 97-1 to 98-2 0.034 0.008 0.025 97-2 to 99-1 0.021 0.026 -0.004 [0.03559] [0.03651] 0.904 97-2 to 99-1 0.022 -0.031 0.008 98-2 to 00-1 -0.028 -0.037 0.010		[0.02942]***	[0.05492]***	0.002 ***	
$[0.07953]^{\bullet\bullet\bullet}$ $[0.12627]$ 0.625 Further Education - Complete 0.371 0.477 -0.106 $[0.05219]^{\bullet\bullet\bullet}$ $[0.08323]^{\bullet\bullet\bullet}$ 0.051 *University Education 0.777 0.517 0.260 $[0.03400]^{\bullet\bullet\bullet\bullet}$ $[0.07107]^{\bullet\bullet\bullet}$ 0.042 **Cohort controls95-2 to 97-1 -0.038 -0.083 0.046 $[0.03653]$ $[0.05800]$ 0.226 96-1 to 97-2 -0.013 -0.084 0.071 $[0.03672]$ $[0.05829]$ 0.062 *96-2 to 98-1 -0.003 -0.042 0.039 $[0.03615]$ $[0.05739]$ 0.297 97-1 to 98-2 0.034 0.008 0.025 $[0.3666]$ $[0.05725]$ 0.494 97-2 to 99-1 0.021 0.026 -0.004 $[0.3559]$ $[0.05631]$ 0.904 98-1 to 99-2 -0.043 -0.070 0.027 $[0.03579]$ $[0.05631]$ 0.904 98-2 to 00-1 -0.028 -0.031 0.008 $[0.03579]$ $[0.05631]$ 0.904 98-2 to 10-1 -0.028 -0.031 0.008 $[0.03575]$ $[0.05675]$ 0.819 99-2 to 01-1 -0.107 -0.108 0.001 $[0.03635]^{\bullet\bullet\bullet\bullet}$ $[0.0575]$ 0.819 99-2 to 01-1 -0.107 -0.127 0.012 $[0.0365]^{\bullet\bullet\bullet\bullet\bullet+$ $[0.03672]^{\bullet\bullet\bullet\bullet+$ 0.000 (0.012) $[0.0365]^{\bullet\bullet\bullet\bullet+$ 0.0001 $(0.0365]^{\bullet\bullet\bullet\bullet+$ $(0.05826]^{$	Further Education - Incomplete	0.754	0.177	0.578	
Further Education - Complete 0.371 0.477 -0.106 [0.05219]*** [0.08323]*** 0.051 * University Education 0.777 0.517 0.260 [0.03400]*** [0.07107]*** 0.042 ** Cohort controls 95-2 to 97-1 -0.038 -0.083 0.046 [0.03653] [0.05800] 0.226 96-1 to 97-2 -0.013 -0.084 0.071 [0.03672] [0.05829] 0.062 * 96-2 to 98-1 -0.003 -0.042 0.039 [0.03615] [0.05739] 0.297 97-1 to 98-2 0.034 0.008 0.025 [0.03606] [0.05725] 0.494 97-2 to 99-1 0.021 0.026 -0.004 [0.03559] [0.05631] 0.904 98-1 to 99-2 -0.043 -0.070 0.027 [0.03579] [0.05631] 0.459 98-2 to 00-1 -0.022 -0.031 0.008 [0.		[0.07953]***	[0.12627]	0.625	
University Education $[0.05219]^{\bullet\bullet\bullet}$ $[0.08323]^{\bullet\bullet\bullet}$ $0.051 \ ^{\bullet}$ University Education 0.777 0.517 0.260 $[0.03600]^{\bullet\bullet\bullet\bullet}$ $[0.07107]^{\bullet\bullet\bullet}$ $0.042 \ ^{\bullet\bullet\bullet}$ Solution 2000 0.026 0.026 0.026 96-1 to 97-2 -0.038 -0.083 0.046 $[0.03653]$ $[0.05800]$ 0.226 96-1 to 97-2 -0.013 -0.084 0.071 $[0.03672]$ $[0.05829]$ $0.62 \ ^{\bullet}$ 96-2 to 98-1 -0.003 -0.042 0.399 $[0.03615]$ $[0.05739]$ 0.297 97-1 to 98-2 0.034 0.008 0.025 $[0.03606]$ $[0.05725]$ 0.494 97-2 to 99-1 0.021 0.026 -0.004 $[0.03559]$ $[0.05651]$ 0.904 98-1 to 99-2 -0.043 -0.070 0.027 $[0.03549]$ $[0.05634]$ 0.459 98-2 to 00-1 -0.028 -0.037 0.010 $[0.03579]$ $[0.05632]$ 0.797 99-1 to 00-2 -0.022 -0.031 0.008 $[0.03575]$ $[0.05826]^{\bullet}$ 0.975 $0-1 to 01-2$ -0.115 -0.127 0.012 $[0.03635]^{\bullet\bullet\bullet\bullet}$ $[0.05770]^{\bullet\bullet\bullet}$ 0.748 $0-2 to 02-1$ -0.164 -0.316 0.152 $(0.0988)^{\bullet\bullet\bullet\bullet}$ $[0.0385]^{\bullet\bullet\bullet\bullet}$ 0.080 $(0.0988)^{\bullet\bullet\bullet\bullet\bullet}$ $(0.0980)^{\bullet\bullet\bullet\bullet\bullet$ $(0.0980)^{\bullet$	Further Education - Complete	0.371	0.477	-0.106	
University Education 0.777 0.517 0.260 [0.03400]*** [0.07107]*** 0.042 ** Cohort controls 95-2 to 97-1 -0.038 -0.083 0.046 [0.03653] [0.05800] 0.226 96-1 to 97-2 -0.013 -0.084 0.071 [0.03672] [0.05829] 0.062 * 96-2 to 98-1 -0.003 -0.042 0.039 [0.03615] [0.05739] 0.297 97-1 to 98-2 0.034 0.008 0.025 [0.03606] [0.05725] 0.494 97-2 to 99-1 0.021 0.026 -0.004 [0.03559] [0.05631] 0.904 98-1 to 99-2 -0.043 -0.070 0.027 [0.03579] [0.05634] 0.459 98-2 to 00-1 -0.028 -0.031 0.008 [0.03575] [0.05675] 0.819 99-2 to 01-1 -0.107 -0.108 0.001 [0.03670]*** [0.05826]* 0.975 <		[0.05219]***	[0.08323]***	0.051 *	
$ \begin{bmatrix} 0.03400 \end{bmatrix}^{***} & \begin{bmatrix} 0.07107 \end{bmatrix}^{***} & 0.042 \\ 0.03673 & \begin{bmatrix} 0.07107 \end{bmatrix}^{***} & 0.042 \\ *** \end{bmatrix} \\ \hline 0.046 & \begin{bmatrix} 0.03653 \end{bmatrix} & \begin{bmatrix} 0.05800 \end{bmatrix} & 0.226 \\ 0.03673 \end{bmatrix} & \begin{bmatrix} 0.05800 \end{bmatrix} & 0.226 \\ 0.03672 \end{bmatrix} & \begin{bmatrix} 0.05829 \end{bmatrix} & 0.062 \\ * \end{bmatrix} \\ \hline 0.03672 \end{bmatrix} & \begin{bmatrix} 0.03672 \end{bmatrix} & \begin{bmatrix} 0.05829 \end{bmatrix} & 0.062 \\ * \end{bmatrix} \\ \hline 0.03672 \end{bmatrix} & \begin{bmatrix} 0.03672 \end{bmatrix} \end{bmatrix} \begin{bmatrix} 0.05739 \end{bmatrix} & 0.297 \\ \hline 0.03605 \end{bmatrix} \begin{bmatrix} 0.05739 \end{bmatrix} & 0.297 \\ \hline 0.03606 \end{bmatrix} \begin{bmatrix} 0.05725 \end{bmatrix} & 0.494 \\ 0.025 \end{bmatrix} \\ \hline 0.03606 \end{bmatrix} \begin{bmatrix} 0.05725 \end{bmatrix} & 0.494 \\ 0.026 & -0.004 \\ \hline 0.03559 \end{bmatrix} \begin{bmatrix} 0.05651 \end{bmatrix} & 0.904 \\ 0.03579 \end{bmatrix} \\ \hline 0.03549 \end{bmatrix} \begin{bmatrix} 0.05634 \end{bmatrix} & 0.459 \\ 0.011 \end{bmatrix} \\ \hline 0.011 \end{bmatrix} \\ \hline 0.012 \end{bmatrix} \\ \hline 0.011 \end{bmatrix} \\ \hline 0.022 \end{bmatrix} \\ \hline 0.022 \end{bmatrix} \\ \hline 0.022 \end{bmatrix} \\ \hline 0.022 \end{bmatrix} \\ \hline 0.025 \end{bmatrix} \\ \hline 0.03579 \end{bmatrix} \\ \hline 0.05632 \end{bmatrix} \\ \hline 0.0797 \\ \hline 0.011 \end{bmatrix} \\ \hline 0.021 \end{bmatrix} \\ \hline 0.028 \end{bmatrix} \\ \hline 0.03579 \end{bmatrix} \\ \hline 0.05632 \end{bmatrix} \\ \hline 0.0797 \\ \hline 0.022 \end{bmatrix} \\ \hline 0.022 \end{bmatrix} \\ \hline 0.03579 \end{bmatrix} \\ \hline 0.05675 \end{bmatrix} \\ 0.819 \\ \hline 0.098575 \end{bmatrix} \\ \hline 0.011 \end{bmatrix} \\ \hline 0.03670 \end{bmatrix} \\ \hline 0.03670 \end{bmatrix} \\ \hline 0.03670 \end{bmatrix} \\ \hline 0.011 \end{bmatrix} \\ \hline 0.021 \end{bmatrix} \\ \hline 0.03675 \end{bmatrix} \\ \hline 0.05820 \end{bmatrix} \\ \hline 0.05820 \end{bmatrix} \\ \hline 0.0797 \\ \hline 0.022 \end{bmatrix} \\ \hline 0.03579 \end{bmatrix} \\ \hline 0.05820 \end{bmatrix} \\ \hline 0.05820 \end{bmatrix} \\ \hline 0.0797 \\ \hline 0.011 \end{bmatrix} \\ \hline 0.02677 \end{bmatrix} \\ \hline 0.03675 \end{bmatrix} \\ \hline 0.05820 \end{bmatrix} \\ \hline 0.0797 \\ \hline 0.011 \end{bmatrix} \\ \hline 0.020 \end{bmatrix} \\ \hline 0.03673 \end{bmatrix} \\ \hline 0.05820 \end{bmatrix} \\ \hline 0.05820 \end{bmatrix} \\ \hline 0.011 \end{bmatrix} \\ \hline 0.012 \\ \hline 0.03673 \end{bmatrix} \\ \hline 0.05820 \end{bmatrix} \\ \hline 0.001 \\ \hline 0.0088 \\ \hline 0.00770 \end{bmatrix} \\ \hline 0.012 \\ \hline 0.03723 \end{bmatrix} \\ \hline 0.05870 \end{bmatrix} \\ \hline 0.000 \\ \hline 0.0088 \\ \hline 0.0000 \\ \hline 0.0088 \\ \hline 0.0000 \\ $	University Education	0.777	0.517	0.260	
Cohort controls 95-2 to 97-1 -0.038 -0.083 0.046 [0.03653] [0.05800] 0.226 96-1 to 97-2 -0.013 -0.084 0.071 [0.03672] [0.05829] 0.062 * 96-2 to 98-1 -0.003 -0.042 0.039 [0.03615] [0.05739] 0.297 97-1 to 98-2 0.034 0.008 0.025 [0.03606] [0.05725] 0.494 97-2 to 99-1 0.021 0.026 -0.004 [0.03559] [0.05651] 0.904 97-2 to 99-1 0.021 0.026 -0.004 [0.03559] [0.05631] 0.904 98-1 to 99-2 -0.043 -0.070 0.027 [0.03579] [0.05634] 0.459 98-2 to 00-1 -0.028 -0.037 0.010 [0.03575] [0.05675] 0.819 99-2 to 01-1 -0.107 -0.108 0.001 [0.03635]*** [0.05770]** 0.748 00-2 -0.115 -0.127 0.012 [0.03635]*** <td< td=""><td></td><td>[0.03400]***</td><td>[0.07107]***</td><td>0.042 **</td></td<>		[0.03400]***	[0.07107]***	0.042 **	
95-2 to 97-1 -0.038 -0.083 0.046 [0.03653] [0.05800] 0.226 96-1 to 97-2 -0.013 -0.084 0.071 [0.03672] [0.05829] 0.062 * 96-2 to 98-1 -0.003 -0.042 0.039 [0.03615] [0.05739] 0.297 97-1 to 98-2 0.034 0.008 0.025 [0.03606] [0.05725] 0.494 97-2 to 99-1 0.021 0.026 -0.004 [0.03559] [0.05651] 0.904 97-2 to 99-1 0.021 0.026 -0.004 [0.03559] [0.05631] 0.904 98-1 to 99-2 -0.043 -0.070 0.027 [0.03579] [0.05634] 0.459 98-2 to 00-1 -0.028 -0.037 0.010 [0.03575] [0.05675] 0.819 99-2 to 01-1 -0.107 -0.108 0.001 [0.03635]*** [0.05770]** 0.748 00-2 to 02-1 -0.115 -0.127 0.012 [0.03635]*** [0.05770]** 0.748 <	Cohort controls				
$ \begin{bmatrix} [0.03653] & [0.05800] & 0.226 \\ 0.013 & -0.084 & 0.071 \\ [0.03672] & [0.05829] & 0.062 * \\ 0.039 & 0.042 & 0.039 \\ [0.03615] & [0.05739] & 0.297 \\ 0.034 & 0.008 & 0.025 \\ [0.03606] & [0.05725] & 0.494 \\ 0.026 & -0.004 \\ [0.03559] & [0.05651] & 0.904 \\ 0.026 & -0.004 \\ [0.03559] & [0.05651] & 0.904 \\ 0.027 & 0.021 & 0.026 & -0.004 \\ [0.03559] & [0.05631] & 0.904 \\ 0.03549] & [0.05634] & 0.459 \\ 0.027 & 0.011 & -0.028 & -0.037 & 0.010 \\ [0.03579] & [0.05682] & 0.797 \\ 0.011 & -0.028 & -0.031 & 0.008 \\ [0.03575] & [0.05675] & 0.819 \\ 0.03575] & [0.05675] & 0.819 \\ 0.022 & -0.011 & -0.107 & -0.108 & 0.001 \\ [0.03575] & [0.05675] & 0.819 \\ 0.011 & -0.107 & -0.108 & 0.001 \\ [0.03670]^{\bullet \bullet \bullet} & [0.05826]^{\bullet \bullet} & 0.975 \\ 0.01 to 01-2 & -0.115 & -0.127 & 0.012 \\ [0.03635]^{\bullet \bullet \bullet} & [0.05770]^{\bullet \bullet} & 0.748 \\ 0.02 to 02-1 & -0.164 & -0.316 & 0.152 \\ 0.03723]^{\bullet \bullet \bullet} & [0.05911]^{\bullet \bullet \bullet} & 0.000 \\ \hline $	95-2 to 97-1	-0.038	-0.083	0.046	
96-1 to 97-2 -0.013 -0.084 0.071 [0.03672] [0.05829] 0.062 * 96-2 to 98-1 -0.003 -0.042 0.039 [0.03615] [0.05739] 0.297 97-1 to 98-2 0.034 0.008 0.025 [0.03606] [0.05725] 0.494 97-2 to 99-1 0.021 0.026 -0.004 [0.03559] [0.05651] 0.904 98-1 to 99-2 -0.043 -0.070 0.027 [0.03549] [0.05634] 0.459 98-2 to 00-1 -0.028 -0.037 0.010 [0.03579] [0.05682] 0.797 99-1 to 00-2 -0.022 -0.031 0.008 [0.03575] [0.05675] 0.819 99-2 to 01-1 -0.107 -0.108 0.001 [0.03670]*** [0.05826]* 0.975 00-1 to 01-2 -0.115 -0.127 0.012 [0.03635]*** [0.05770]** 0.748 00-2 to 02-1 -0.164 -0.316 0.152 [0.03723]*** [0.05911]*** 0.000 <td< td=""><td></td><td>[0.03653]</td><td>[0.05800]</td><td>0.226</td></td<>		[0.03653]	[0.05800]	0.226	
$ \begin{bmatrix} [0.03672] & [0.05829] & 0.062 * \\ 96-2 to 98-1 & -0.003 & -0.042 & 0.039 \\ [0.03615] & [0.05739] & 0.297 \\ 97-1 to 98-2 & 0.034 & 0.008 & 0.025 \\ [0.03606] & [0.05725] & 0.494 \\ 97-2 to 99-1 & 0.021 & 0.026 & -0.004 \\ [0.03559] & [0.05651] & 0.904 \\ 98-1 to 99-2 & -0.043 & -0.070 & 0.027 \\ [0.03549] & [0.05634] & 0.459 \\ 98-2 to 00-1 & -0.028 & -0.037 & 0.010 \\ [0.03579] & [0.05682] & 0.797 \\ 99-1 to 00-2 & -0.022 & -0.031 & 0.008 \\ [0.03575] & [0.05675] & 0.819 \\ 99-2 to 01-1 & -0.107 & -0.108 & 0.001 \\ [0.03670]^{\bullet\bullet\bullet} & [0.05826]^{\bullet\bullet} & 0.975 \\ 00-1 to 01-2 & -0.115 & -0.127 & 0.012 \\ [0.03635]^{\bullet\bullet\bullet\bullet} & [0.05770]^{\bullet\bullet\bullet} & 0.748 \\ 00-2 to 02-1 & -0.164 & -0.316 & 0.152 \\ Constant & 0.424 & 0.080 \\ [0.09885]^{\bullet\bullet\bullet\bullet} & [0.15693] \\ \hline \hline \hline \hline \hline \\ \hline \hline \\ \hline \hline \hline \hline \\ \hline \hline \hline \\ \hline \hline \hline \\ \hline \hline \hline \\ \hline \hline \hline \hline \\ \hline \hline \hline \\ \hline \hline \hline \hline \\ \hline \hline \hline \\ \hline \hline \hline \hline \hline \hline \\ \hline \\ \hline \hline$	96-1 to 97-2	-0.013	-0.084	0.071	
96-2 to 98-1 -0.003 -0.042 0.039 [0.03615] [0.05739] 0.297 97-1 to 98-2 0.034 0.008 0.025 [0.03606] [0.05725] 0.494 97-2 to 99-1 0.021 0.026 -0.004 [0.03559] [0.05651] 0.904 98-1 to 99-2 -0.043 -0.070 0.027 [0.03549] [0.05634] 0.459 98-2 to 00-1 -0.028 -0.037 0.010 [0.03579] [0.05682] 0.797 98-2 to 00-1 -0.022 -0.031 0.008 [0.03575] [0.05675] 0.819 99-2 to 01-1 -0.107 -0.108 0.001 [0.03670]*** [0.05826]* 0.975 00-1 to 01-2 -0.115 -0.127 0.012 [0.03723]*** [0.05770]** 0.748 00-2 to 02-1 -0.164 -0.316 0.152 Constant 0.424 0.080		[0.03672]	[0.05829]	0.062 *	
$ \begin{bmatrix} [0.03615] & [0.05739] & 0.297 \\ 0.03615] & [0.05739] & 0.297 \\ 0.034 & 0.008 & 0.025 \\ [0.03606] & [0.05725] & 0.494 \\ 97-2 to 99-1 & 0.021 & 0.026 & -0.004 \\ [0.03559] & [0.05651] & 0.904 \\ 98-1 to 99-2 & -0.043 & -0.070 & 0.027 \\ [0.03549] & [0.05634] & 0.459 \\ 98-2 to 00-1 & -0.028 & -0.037 & 0.010 \\ [0.03579] & [0.05682] & 0.797 \\ 99-1 to 00-2 & -0.022 & -0.031 & 0.008 \\ [0.03575] & [0.05675] & 0.819 \\ 99-2 to 01-1 & -0.107 & -0.108 & 0.001 \\ [0.03670]^{\bullet\bullet\bullet} & [0.05826]^{\bullet\bullet} & 0.975 \\ 00-1 to 01-2 & -0.115 & -0.127 & 0.012 \\ [0.03635]^{\bullet\bullet\bullet} & [0.05770]^{\bullet\bullet} & 0.748 \\ 00-2 to 02-1 & -0.164 & -0.316 & 0.152 \\ 0.03723]^{\bullet\bullet\bullet} & [0.05911]^{\bullet\bullet\bullet\bullet} & 0.000 \\ \hline $	96-2 to 98-1	-0.003	-0.042	0.039	
$97-1$ to $98-2$ 0.034 0.008 0.025 $[0.03606]$ $[0.05725]$ 0.494 $97-2$ to $99-1$ 0.021 0.026 -0.004 $[0.03559]$ $[0.05651]$ 0.904 $98-1$ to $99-2$ -0.043 -0.070 0.027 $[0.03549]$ $[0.05634]$ 0.459 $98-2$ to $00-1$ -0.028 -0.037 0.010 $[0.03579]$ $[0.05682]$ 0.797 $99-1$ to $00-2$ -0.022 -0.031 0.008 $[0.03575]$ $[0.05675]$ 0.819 $99-2$ to $01-1$ -0.107 -0.108 0.001 $[0.03670]^{***}$ $[0.05826]^*$ 0.975 $00-1$ to $01-2$ -0.115 -0.127 0.012 $[0.03635]^{***}$ $[0.05770]^{**}$ 0.748 $00-2$ to $02-1$ -0.164 -0.316 0.152 $(0.0723]^{***}$ $[0.05911]^{***}$ 0.000 **** Constant 0.424 0.080 $(0.09885]^{***}$ $(0.593]$ Observations 5446 5446 5446 <td></td> <td>[0.03615]</td> <td>[0.05739]</td> <td>0.297</td>		[0.03615]	[0.05739]	0.297	
$ \begin{bmatrix} [0.03606] & [0.05725] & 0.494 \\ [0.035725] & [0.05725] & 0.004 \\ & [0.03559] & [0.05651] & 0.904 \\ 98-1 to 99-2 & -0.043 & -0.070 & 0.027 \\ & [0.03549] & [0.05634] & 0.459 \\ 98-2 to 00-1 & -0.028 & -0.037 & 0.010 \\ & [0.03579] & [0.05682] & 0.797 \\ 99-1 to 00-2 & -0.022 & -0.031 & 0.008 \\ & [0.03575] & [0.05675] & 0.819 \\ 99-2 to 01-1 & -0.107 & -0.108 & 0.001 \\ & [0.03670]^{\bullet\bullet\bullet} & [0.05826]^{\bullet\bullet} & 0.975 \\ 00-1 to 01-2 & -0.115 & -0.127 & 0.012 \\ & [0.03635]^{\bullet\bullet\bullet} & [0.05770]^{\bullet\bullet} & 0.748 \\ 00-2 to 02-1 & -0.164 & -0.316 & 0.152 \\ & [0.03723]^{\bullet\bullet\bullet} & [0.05911]^{\bullet\bullet\bullet\bullet} & 0.000 \\ \hline \\ Constant & 0.424 & 0.080 \\ & [0.09885]^{\bullet\bullet\bullet} & [0.15693] \\ \hline \\ $	97-1 to 98-2	0.034	0.008	0.025	
$97-2$ to $99-1$ 0.021 0.026 -0.004 $[0.03559]$ $[0.05651]$ 0.904 $98-1$ to $99-2$ -0.043 -0.070 0.027 $[0.03579]$ $[0.05634]$ 0.459 $98-2$ to $00-1$ -0.028 -0.037 0.010 $[0.03579]$ $[0.05682]$ 0.797 $99-1$ to $00-2$ -0.022 -0.031 0.008 $[0.03575]$ $[0.05675]$ 0.819 $99-2$ to $01-1$ -0.107 -0.108 0.001 $[0.03670]^{\bullet **}$ $[0.05826]^{*}$ 0.975 $00-1$ to $01-2$ -0.115 -0.127 0.012 $[0.03635]^{\bullet **}$ $[0.05770]^{\bullet **}$ 0.748 $00-2$ to $02-1$ -0.164 -0.316 0.152 $[0.03723]^{\bullet **}$ $[0.05911]^{\bullet ***}$ 0.000 **** Constant 0.424 0.080 $[0.09885]^{\bullet ***}$ $[0.15693]$ Observations 5446 5446 5446 5446		[0.03606]	[0.05725]	0.494	
$ \begin{bmatrix} [0.03559] & [0.05651] & 0.904 \\ 0.98-1 to 99-2 & -0.043 & -0.070 & 0.027 \\ [0.03549] & [0.05634] & 0.459 \\ 98-2 to 00-1 & -0.028 & -0.037 & 0.010 \\ [0.03579] & [0.05682] & 0.797 \\ 99-1 to 00-2 & -0.022 & -0.031 & 0.008 \\ [0.03575] & [0.05675] & 0.819 \\ 99-2 to 01-1 & -0.107 & -0.108 & 0.001 \\ [0.03670]^{\bullet+\bullet} & [0.05826]^{\bullet} & 0.975 \\ 00-1 to 01-2 & -0.115 & -0.127 & 0.012 \\ [0.03635]^{\bullet+\bullet} & [0.05770]^{\bullet+\bullet} & 0.748 \\ 00-2 to 02-1 & -0.164 & -0.316 & 0.152 \\ [0.03723]^{\bullet+\bullet} & [0.05911]^{\bullet+\bullet} & 0.000 \\ \bullet & 10.09885]^{\bullet+\bullet} & [0.15693] \\ \hline \begin{tabular}{lllllllllllllllllllllllllllllllllll$	97-2 to 99-1	0.021	0.026	-0.004	
98-1 to 99-2 -0.043 -0.070 0.027 [0.03549] [0.05634] 0.459 98-2 to 00-1 -0.028 -0.037 0.010 [0.03579] [0.05682] 0.797 99-1 to 00-2 -0.022 -0.031 0.008 [0.03575] [0.05675] 0.819 99-2 to 01-1 -0.107 -0.108 0.001 [0.03670]*** [0.05826]* 0.975 00-1 to 01-2 -0.115 -0.127 0.012 [0.03635]*** [0.05770]** 0.748 00-2 to 02-1 -0.164 -0.316 0.152 [0.03723]*** [0.05911]*** 0.000 *** Constant 0.424 0.080 0.000 *** Observations 5446 5446 5446	00.1 + 00.0	[0.03559]	[0.05651]	0.904	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	98-1 to 99-2	-0.043	-0.070	0.027	
98-2 to 00-1 -0.028 -0.037 0.010 $[0.03579]$ $[0.05682]$ 0.797 99-1 to 00-2 -0.022 -0.031 0.008 $[0.03575]$ $[0.05675]$ 0.819 99-2 to 01-1 -0.107 -0.108 0.001 $[0.03670]^{\bullet+\bullet}$ $[0.05826]^{\bullet}$ 0.975 00-1 to 01-2 -0.115 -0.127 0.012 $[0.03635]^{\bullet+\bullet}$ $[0.05770]^{\bullet+}$ 0.748 00-2 to 02-1 -0.164 -0.316 0.152 $[0.03723]^{\bullet+\bullet}$ $[0.05911]^{\bullet+\bullet}$ 0.000 *** Constant 0.424 0.080 [0.09885]^{\bullet+\bullet} $[0.15693]$	00.0 . 00.1	[0.03549]	[0.05634]	0.459	
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	98-2 to 00-1	-0.028	-0.037	0.010	
99-1 to 00-2 -0.022 -0.031 0.008 $[0.03575]$ $[0.05675]$ 0.819 99-2 to 01-1 -0.107 -0.108 0.001 $[0.03670]^{\bullet+\bullet}$ $[0.05826]^{\bullet}$ 0.975 00-1 to 01-2 -0.115 -0.127 0.012 $[0.03635]^{\bullet+\bullet}$ $[0.05770]^{\bullet+\bullet}$ 0.748 00-2 to 02-1 -0.164 -0.316 0.152 $[0.03723]^{\bullet+\bullet}$ $[0.05911]^{\bullet+\bullet}$ 0.000 *** Constant 0.424 0.080 $[0.09885]^{\bullet+\bullet}$ $[0.15693]$ Observations 5446 5446 Standard errors in brackets (significant: * at 10%: ** at 5%: *** at 1%) $[0.001100000000000000000000000000000000$	00.1 += 00.2	[0.035/9]	[0.05682]	0.797	
[0.03573] [0.03673] 0.819 99-2 to 01-1 -0.107 -0.108 0.001 [0.03670]*** [0.05826]* 0.975 00-1 to 01-2 -0.115 -0.127 0.012 [0.03635]*** [0.05770]** 0.748 00-2 to 02-1 -0.164 -0.316 0.152 [0.03723]*** [0.05911]*** 0.000 *** Constant 0.424 0.080 [0.09885]*** [0.15693] Observations 5446 5446 Standard errors in brackets (significant: * at 10%: ** at 5%: *** at 1%)	99-1 to 00-2	-0.022	-0.031	0.008	
99-2 to 01-1 -0.107 -0.108 0.001 [0.03670]*** [0.05826]* 0.975 00-1 to 01-2 -0.115 -0.127 0.012 [0.03635]*** [0.05770]** 0.748 00-2 to 02-1 -0.164 -0.316 0.152 [0.03723]*** [0.05911]*** 0.000 *** Constant 0.424 0.080 [0.09885]*** [0.15693] Observations 5446 5446 Standard errors in brackets (significant: * at 10%: ** at 5%: *** at 1%)	00.2 += 01.1	[0.03575]	[0.05075]	0.819	
[0.03670]*** [0.03626]** [0.03626]** [0.03626]** 00-1 to 01-2 -0.115 -0.127 0.012 [0.03635]*** [0.05770]** 0.748 00-2 to 02-1 -0.164 -0.316 0.152 [0.03723]*** [0.05911]*** 0.000 *** Constant 0.424 0.080 [0.09885]*** [0.15693] Observations 5446 5446 Standard errors in brackets (significant: * at 10%: ** at 5%: *** at 1%)	99-2 10 01-1	-0.107	-0.108	0.001	
00-1 to 01-2 -0.115 -0.127 0.012 [0.03635]*** [0.05770]** 0.748 00-2 to 02-1 -0.164 -0.316 0.152 [0.03723]*** [0.05911]*** 0.000 *** Constant 0.424 0.080 [0.09885]*** [0.15693] Observations 5446 5446 Standard errors in brackets (significant: * at 10%: ** at 5%: *** at 1%)	00 1 to 01 2	0.115	0 127	0.975	
[0.0305]*** [0.0570]** 0.748 00-2 to 02-1 -0.164 -0.316 0.152 [0.03723]*** [0.05911]*** 0.000 *** Constant 0.424 0.080 [0.09885]*** [0.15693] Observations 5446 5446 Standard errors in brackets (significant: * at 10%: ** at 5%: *** at 1%)	00-1 10 01-2	-0.113	-U.12/	0.012	
00-2 to 02-1 -0.164 -0.516 0.152 [0.03723]*** [0.05911]*** 0.000 *** Constant 0.424 0.080 [0.09885]*** [0.15693] Observations 5446 5446 Standard errors in brackets (significant: * at 10%: ** at 5%: *** at 1%) 5446	00.2 += 02.1	0.164	0.216	0.748	
[0.03723]*** [0.03713]*** 0.000 *** Constant 0.424 0.080 (0.09885)*** (0.15693) Observations 5446 5446 5446 Standard errors in brackets (significant: * at 10%: ** at 5%: *** at 1%) (0.000 ****) (0.000 ****)	00-2 10 02-1	-U.104	-0.310	0.134	
U.424 U.080 [0.09885]*** [0.15693] Observations 5446 5446 Standard errors in brackets (significant: * at 10%: ** at 5%: *** at 1%)	Constant	0 4 2 4	0.03911]***	0.000 ***	
Observations 5446 5446 Standard errors in brackets (significant: * at 10%: ** at 5%: *** at 1%)	Constant	U.424	0.000		
Standard errors in brackets (significant: * at 10%: ** at 5%: *** at 1%)	Observations	5446	5446	<u> </u>	
	Standard errors in brackets (significan	t: * at 10% ** at	5%: *** at 1%)		

ard errors in brackets (significant: * at 10%; ** at 5%; *

P-Value of the test below the difference for the fourth column.

Source: Author's estimations based on EPH household survey data (INDEC).

tive as expected since the excluded category is wage earner. However, part of the impact of being self-employed – for those with no professional qualifications – is reflected by the negative and statistically significant coefficient for the informal worker indicator for both the head and the spouse.

A higher job qualification of the head (at the professional level) or the spouse (at each of the three levels considered) have a positive impact on the average income of the household, as do the education levels of the head and the spouse, with higher levels of education implying progressively higher household income. Being in the public sector reduces expected income in the case of the head, although for the spouse the impact is also negative but not statistically significant. Finally, only the three cohorts covering the period October 1999-May 2002 are associated with statistically significant and negative coefficients, reflecting the progressive deepening of the recession and the subsequent crisis.

The data confirms the stylised facts that richer households are smaller, better-educated, and have employed heads or spouses in better jobs. The parameters obtained in the second column of Table 3.2 correspond to the regression with risk adjusted income as the dependent variable, and it is easy to notice that most of the results from the first column still apply. The most interesting results of Table 3.2, however, are related to the impact (or the lack of impact) of the same independent variables on the risk faced by the household, and thus the discussion focuses on the tests of differences obtained by the estimation of the system by SUR.

The third column of Table 3.2 presents the tests of statistical significance for the differences in the coefficients of the two regressions. If the difference is not significantly different from zero, it can be interpreted that the associated independent variable has the same effect (or lack of effect) in both mean and risk adjusted incomes, and thus no discernible correlation with risk. However, a significant difference implies that the independent variable has a differential effect on the two dependent variables, or, in other words, it has an effect on risk. As explained in the previous Section, the SUR test can also be interpreted as a regression with the "logarithmic" risk premium as the dependent variable (Equation 3.9). In that context, a negative value of $\beta - \beta_{RA}$ represents a negative effect of a variable on the risk premium, which is associated with lower risk, while a positive value implies a higher

risk premium.

Only a subset of the variables in X_i has a significant effect on risk. Regarding the structure of the household, only the presence of adults aged 65 and over significantly affects the household's level of risk in a negative way. This can be explained by the fact that the elderly often receive a steady stream of income from pensions (Fiszbein et al., 2002), and consequently experience less income variability – and face lower income risk – than other age groups.

While there is no statistically significant impact of the age or gender of the head on risk, other characteristics do have an effect. If the head has migrated to the GBA region (from other provinces in Argentina or from other countries) in the last five years, the household suffers more from risk. Interestingly, the same indicator was found to have no statistically significant impact on average income after controlling for human capital and other characteristics. This result implies that recent migrants are more exposed to income risk than non-migrants.

For both the head and the spouse, being unemployed contributes to risk over and above the negative effect of this characteristic on average income. The same is true for the informality of the head in the current job, or in the previous job if unemployed. Most importantly, a higher level of education has a negative impact on the risk faced by the household: the less-educated suffer not only from a lower average income, but also from higher levels of risk.

Finally, two cohort indicators signal a positive effect on risk. This is the case for the last cohort, which is not surprising given the large fall in real incomes induced by the crisis of 2001-2002 (see Figure 1.3). The cohort covering the period 1996-1997 also indicates a higher level of risk, which may be due to the variability in incomes induced by the aftermath of the 1994-1995 crisis and the subsequent recovery.

The regressions in Table 3.2 are based on risk adjusted income with $\rho = 2$ since, as discussed above, this value was found to be reasonable and within the range employed in the literature. The analysis of income and poverty aggregates, however, was done with different values of the risk aversion parameter, and Table 3.3 presents the correspondent robustness check of the regression analysis for $\rho \in \{0.5, 1, 2, 4\}$. It is not necessary to emulate

Table 3.3: Difference in Coefficients of Determinants of Log Average Income and Log Risk Adjusted Income for Different Values of Risk Aversion

	ρ=0.5	o=1	ρ=2	ρ=4
Household characteristics		F	F -	
Number of infants (ages 0-5)	0.0028	0.0072	0.0242	0.0355
Number of infants squared	-0.0001	-0.0003	-0.0024	-0.0033
Number of children (ages 6-14)	-0.0002	-0.0028	-0.0088	-0.0045
Number of children squared	0.0007	0.0023	0.0063	0.0076
Number of youths (ages 15-24)	0.0002	-0.0048	-0.0227	-0.0187
Number of youths squared	-0.0013 *	-0.0027	-0.0042	-0.0067
Number of adults (ages 25-64)	-0.0012	-0.0046	-0.0110	-0.0100
Number of adults squared	-0.0006	-0.0015	-0.0039	-0.0057
Number of elderly (ages 65+)	-0.0107 **	-0.0361 **	-0.0995 **	-0.1268 **
Number of elderly members, squared	0.0014	0.0057	0.0181	0.0224
Characteristics of the head				
Age	-0.0002	-0.0007	-0.0021	-0.0030
Age Squared	0.0000	0.0000	0.0000	0.0000
Female head	-0.0044	-0.0139	-0.0277	-0.0280
Recent migrant	0.0087	0.0308 *	0.0886	0.1091
Inactive	-0.1070	0.0090	0.0224	0.0228
Unemployed	0.1434	0.2572	0.4264	0.6395
Employer	0.0087	0.0208	-0.0635	0.0725
Self-employed	-0.0722	-0.0767	-0.0240	-0.0969
Informal Worker	0.0084	0.0267	0.0771	0.1024
Public Sector Worker	0.0424	-0.0107	-0.0282	0.0068
Job Qualification: Operative	-0.0041	-0.0128	-0.0373	-0.0536
Job Qualification: Technical	-0.2662	-0.2733	-0.0358	-0.0554
Job Qualification: Professional	-0.4224	-0.4310	-0.0358	-0.4057
Primary Education – Complete	-0.0106	-0.0212	-0.0393	-0.0500
Secondary Education - Incomplete	-0.0002	-0.2535 ***	-0.2011 **	-0.1209
Further Education – Incomplete	-0.5699 ***	-0.0907 ***	-0.221	-0.3198
Further Education – Complete	-0.3736 **	-0.0432 **	-0.2221	-0.4859 *
University Education	-0.0151 ***	-0 3901 ***	-0 4232 **	-0.4561 **
No Spouse	0.0079	0.0260	0.0444	0.0358
Characteristics of the spouse	0.0075	0.0200	0.0111	0.0000
Inactive	0.1203 **	0.0328 **	0.0867 **	0.1053 **
Unemployed	-0.0632 ***	-0.0181 ***	0.1873 ***	0.1533 ***
Employer	0.0182	0.0415	0.1963	0.1194
Self-employed	0.0781	0.0825	0.0204	0.1078
Informal Worker	0.0037	0.0110	0.0253	0.0279
Job Qualification: Operative	0.0039	0.0110	0.0281	0.0300
Job Qualification: Technical	0.2573	0.2529	-0.0155	-0.0260
Job Qualification: Professional	0.4185	0.4177	0.0004	0.4131
Public Sector Worker	-0.0451	0.0035	0.0106	-0.0395
Primary Education - Complete	-0.0020 *	-0.0141 *	-0.0555 *	-0.0724 **
Secondary Education - Incomplete	-0.0101 **	-0.0267 **	0.0132 **	-0.0211 **
Secondary Education - Complete	-0.0119 ***	0.1853 ***	0.1097 ***	0.0649 ***
Further Education - Incomplete	0.5362	0.0090	0.0401	0.5665
Further Education - Complete	0.3527	-0.0275	0.2539	0.2130
University Education	-0.0088	0.3290	0.2600	0.2231
Cohort controls				
95-2 to 97-1	0.0026	0.0117	0.0457	0.0544
96-1 to 97-2	0.0046	0.0191	0.0707 *	0.0913 *
96-2 to 98-1	0.0032	0.0123	0.0389	0.0449
97-1 to 98-2	-0.0001	0.0028	0.0255	0.0306
97-2 to 99-1	-0.0016	-0.0043	-0.0044	-0.0056
98-1 to 99-2	0.0017	0.0068	0.0271	0.0322
98-2 to 00-1	-0.0018	-0.0016	0.0095	0.0134
99-1 to 00-2	-0.0034	-0.0047	0.0084	0.0077
99-2 to 01-1	-0.0006	-0.0031	0.0012	0.0089
00-1 to 01-2	-0.0013	-0.0019	0.0121	0.0179
00-2 to 02-1	0.0240	0.0633	0.1521	0.2154
Observations	5446	5446	5446	5446
Superscripts indicate levels of significanc	e (* 10%; ** 5	%; *** 1%).		
Source: Author's estimations based on EP	H household su	rvey data (INI	DEC).	

102

the structure of Table 3.2, since the first column (the regression of the log average income) need not be repeated: Table 3.3 only presents the equivalent of the third column of Table 3.2, namely the difference in the parameters for average and risk adjusted incomes, for each value of ρ .

While higher values of ρ accentuate the disutility arising from income risk and lower values attenuate it, Table 3.3 is reassuring in terms of the robustness of Table 3.2's results. With the exception of the education variables of the spouse, all the variables that were found to be significantly correlated with the risk faced by the household, under the assumption that $\rho = 2$, are also relevant for all other values of ρ in the Table. While the magnitude of the effects is sometimes different, their statistical significance is markedly similar.

3.4 CONCLUSION

This Chapter presented a simple methodology for incorporating the disutility arising from income risk in the measurement of poverty. The methodology, based on the economics of choice under uncertainty, was applied to household panel data from the Greater Buenos Aires region for the period 1995-2002.

This application demonstrates that averaging income data over time at the household level reduces poverty measures by mitigating the impact of negative shocks, but this effect is more than offset when the disutility from income fluctuations is taken into account with plausible levels of risk aversion.

A regression analysis of the determinants of risk adjusted income reveals that risk is not uniform across households. The effect of a number of household characteristics on risk adjusted measures differs from their impact on average income. This implies that these characteristics are related to the risk faced by the household over and above their correlation with average income. Households with better-educated members not only have higher incomes, but also experience lower levels of income risk. Households with elderly members, as well as households with informal workers and/or unemployed or inactive members tend to suffer more from risk than other types of households. Being a recent migrant also increases risk, even though it does not affect average income. Finally, at the broader macroeconomic level, an economic crisis not only reduces income levels, but also increases risk, which magnifies its overall negative impact on poverty and well-being.

An alternative to the methodology presented in this Chapter consists of studying the mean of poverty evaluations for a household over time, instead of evaluating poverty at average or adjusted levels of income. This alternative approach is pursued in the following Chapter.

CHAPTER 4

CHRONIC AND TRANSIENT POVERTY IN TURBULENT TIMES

4.1 INTRODUCTION

The previous Chapters presented different approaches for the study of wellbeing based on panel data on incomes. Chapter 1 provided results on poverty transitions and short term dynamics by means of a series of twoperiod panels, which suggested that there were sizeable movements both within and into and out of poverty in the Argentine data. Chapter 2, in turn, proposed and illustrated a general framework for the welfare-based evaluation of income fluctuations and well-being over time, while Chapter 3 presented a related methodology that derives the impact of income risk from observed data.

The methodologies of the previous two Chapters concentrated on the derivation of measures of income adjusted for fluctuations and risk for the whole population, and used those measures to compute poverty and other distributional parameters for that population. The approach in this Chapter, which was briefly discussed in Section 2.4 in the context of the evaluation framework, focuses on poverty over time, providing a methodology for decomposing it into its chronic and transient components. This methodology was developed by Jalan and Ravallion (1998; 2000) to account for fluctuations of household income and poverty status over time, such as those observed during 1995 to 2002 in Argentina (Section 1.3, page 29). Moreover, given its prominence in the poverty and development literatures (Baulch and Hoddinott, 2000; Hulme and Shepherd, 2003, introduce two special journal editions related to the methodology), the results presented below constitute

an important benchmark for the applications of Chapters 2 and 3, and provide a complement to their results.

These two components of poverty constitute important inputs for the design of poverty alleviation and risk mitigation policies. Since the determinants and consequences of chronic and transient poverty are not necessarily the same, interventions to deal with each of them may differ. For instance, the reduction of chronic poverty may involve forms of asset redistribution and human capital investments in education and health, while the alleviation of transient poverty may be related to employment policies and the development of coping mechanisms and insurance devices. Given these crucial implications, this Chapter presents a regression analysis of the correlates of both components of poverty.

The Chapter is organised as follows. Section 4.2 introduces the methodology for decomposing poverty into its chronic and transient components, and presents the estimation strategy for the analysis of their determinants. Section 4.3 presents the empirical results. A brief conclusion follows.

4.2 METHODOLOGY

4.2.1 Definitions of transient and chronic poverty

The decomposition of poverty originates in Ravallion's (1988) discussion of expected poverty and welfare variability, on which Jalan and Ravallion (1998; 2000) elaborate to define the notions of transient and chronic poverty. The decomposition can be motivated by the division of the population into four mutually exclusive groups according to the evolution of their income over time and its average, as depicted in Figure 4.1 (adapted from Hulme and Shepherd, 2003). The four groups are the always poor (with income below the poverty line at all periods), the never poor, and types of transiently poor. The transiently poor are only poor for some intervals, and they can be further divided into those with mean income above the poverty line and those with mean income below that threshold.

This partition of the population is extremely informative about the nature of poverty and income processes over time. As found in Section 1.3 for Argentina, those who are poor at one point in time may have different past experiences and future prospects.
CHAPTER 4. CHRONIC AND TRANSIENT POVERTY



Figure 4.1: Population Groups by Mean Income and Persistence of Poverty Status

Jalan and Ravallion (1998; 2000) build their decomposition of poverty into chronic and transient components on the two parameters that define these four categories: average income over time and fluctuations around the poverty line. The basis of the decomposition is the vector of observed incomes over *T* periods for a household *i*, $\mathbf{y}_i = [y_{i1}, ..., y_{iT}]$, where y_{it} are equivalised and normalised by the contemporaneous poverty line as shown in Equation 1.3 (page 27). A household's poverty at a point in time is given by the evaluation function $p(y_{it})$, which is required to be additive, strictly convex and decreasing up to the poverty line (and taking a value of zero thereafter).

Jalan and Ravallion (1998) define the intertemporal poverty of a household i, P_i , as the average of the poverty evaluations over time:

$$P_{i} = \frac{1}{T} \sum_{t=1}^{T} p(y_{it})$$
(4.1)

 P_i represents the average of *i*'s poverty evaluation. This intertemporal mea-

sure is the total amount that Jalan and Ravallion (1998) decompose into chronic and transient components as $P_i = C_i + T_i$.

As the partition illustrated in Figure 4.1 shows, the decomposition does not depend only on fluctuations in income but also on its average over time, \bar{y}_i . Jalan and Ravallion's (1998) concept of chronic poverty is based on the poverty evaluation of this average level:

$$C_{i} = p(\bar{y}_{i}) = p(\frac{1}{T}\sum_{t=1}^{T} y_{it})$$
(4.2)

 C_i accounts for the depth of poverty over time.

Finally, transient poverty is obtained as the residual of total poverty and its chronic component:

$$T_i = P_i - C_i = \frac{1}{T} \sum_{t=1}^{T} \left[p(y_{it}) - p(\bar{y}_i) \right]$$
(4.3)

As illustrated by the term on the right of Equation 4.3, T_i can be interpreted as the average of the gaps between the poverty evaluation at each point in time and the evaluation of average income. Furthermore, the convexity of p ensures (by Jensen's inequality) that T_i is always greater than or equal to zero.

The intuition behind the definitions of P_i , C_i and T_i can be presented in terms of the differences between the four categories defined above and illustrated in Figure 4.1. The role of average income over time in Equation 4.2 implies that both the never poor and the transiently poor with mean income above the poverty line have a zero measure of chronic poverty, while this is positive for the other two categories. The difference between the never poor and the transiently poor with $\bar{y} > z$ is that the latter group has a positive value of the transient component, since it is poor for part of the period.

The two panels in the bottom of Figure 4.1 depict the difference between chronic and transient poverty: while in the Figure the groups in the two panels have the same chronic measure (since \bar{y} is the same), the always poor are below the poverty line in every period. This results in positive values of $p(y_{it})$ at every point in time, whereas the transiently poor have $p(y_{it}) = 0$ for the period or periods when they are above the poverty line. The always

poor, then, have higher values of both intertemporal and transient poverty than those in the bottom right panel.

The description of the methodology is completed by the specification of the aggregation procedure and the functional form for the poverty evaluation p. The aggregate measures of total, chronic and transient poverty are obtained by computing Equations 4.1, 4.2 and 4.3 for each household i, and averaging over the population.

Regarding the poverty evaluation function p, the strict convexity requirement rules out the use of the p functions implied by the headcount and poverty gap measures, defined in Equation 1.5 with $\alpha = 0$ and $\alpha = 1$ respectively. The practice in applied work has followed Jalan and Ravallion (1998; 2000) and used the squared poverty gap for this decomposition (Dercon and Krishnan, 2000; McCulloch and Baulch, 2000). This implies a functional form for p given by:

$$p(y_{it}) = [\max(1 - y_{it}, 0)]^2$$
(4.4)

which is a simplified version of Equation 1.5 (page 35), since it is based on income normalised by the poverty line. The chronic and transient decomposition is thus based on the *FGT* measures of Equation 1.6 (page 35).

The squared poverty gap, however, is not the only measure that fulfils the requirements set out for *p*. Kurosaki (2003) presents an in-depth study of the sensitivity of the transient-chronic approach to alternative underlying poverty evaluation functions, and makes an interesting parallel between the squared poverty gap and quadratic utility functions. He proposes using the Clark-Watts family of measures (Watts, 1968; Clark et al., 1981) given its affinity to the Constant Relative Risk Aversion utility function (see the discussions in Sections 2.4 and 3.2, pages 64 and 83).

4.2.2 Determinants of poverty

The study of the correlates of intertemporal, chronic and transient poverty relies on regression analysis with the corresponding measures defined in the Equations above as the dependent variables. This results in the following

three models:

$$T_{i} = X'_{i}\beta_{T} + \varepsilon_{Ti}$$

$$C_{i} = X'_{i}\beta_{C} + \varepsilon_{Ci}$$

$$P_{i} = X'_{i}\beta_{P} + \varepsilon_{Pi}$$

$$(4.5)$$

As in Equation 3.8 (page 88), the set of household characteristics used as explanatory variables is given by the vector X_i (which includes a constant term), and ε_{1i} , ε_{2i} and ε_{3i} are regression errors.

The three dependent variables are censored by definition, since they are bounded by 0 and 1 – the three take a value of zero for the non-poor, while $P_i = C_i = 1 - T_i = 1$ when $\bar{y}_i = 0$. The Tobit model is the usual choice of estimator in these situations. However, it may suffer from severe bias from non-normality and heteroskedastic errors, and there is no a priori reason to assume that the error ε in Equation 4.5 will be normally distributed. For this reason, the Chapter follows Jalan and Ravallion (2000) in estimating the three models by Censored Least Absolute Deviations (CLAD), which allows for non-normal, non-homoskedastic and non-symmetric errors by imposing the relatively less stringent condition of zero median error terms (Chay and Powell, 2001). The semiparametric estimation is based on quantile regression methods, and uses an iterative estimation process based on Buchinsky (1994).¹

However, the magnitude of the poverty figures for Argentina implies that the P_i , C_i , and T_i measures are heavily censored. For most of the 1995-2002 period, the poverty headcount in the GBA region fluctuated between 20 and 30 percent (see Section 1.3, page 29, and Appendix C), implying that between 70 and 80 percent of the population has a value of $p(y_{it}) = 0$ in Equation 4.4. With such a large fraction of the sample with censored values, the regression coefficients in Equation 4.5 must be estimated at very high quantiles – at least above 0.75. To reduce the level of censoring, the poverty line is scaled up by 50 percent of its value when conducting the regressions

¹The estimation was performed using the *qcenreg* Stata routine developed by Robert Vigfusson of Northwestern University. This is an implementation of Buchinsky's (1994) recensoring-regression algorithm: starting from an unrestricted quantile regression, the observations with a predicted value of the quantile below the censoring point are excluded, a procedure which is repeated until the estimate converges.

in Equation 4.5, since this reduces significantly the number of observations with a zero value of intertemporal, transient and chronic poverty, the dependent variables.² This is the procedure adopted by Jalan and Ravallion (2000) for their analysis of chronic and transient poverty in rural China. For consistency with the estimates reported in the previous Chapters, however, the Tables and Figures in the next Section rely on the normal poverty line.

4.3 EMPIRICAL RESULTS

4.3.1 Population groups and poverty decompositions

The applications presented in this Section are based on the Greater Buenos Aires (GBA) panel described in Section 1.2. Given the structure of the rotating panel, the fifteen waves between May 1995 and May 2002 contain data for twelve cohorts with four consecutive observations over a period of a year and a half, with an average of 453 households and 1812 observations per cohort.

Other studies based on the transient-chronic decomposition use panels that follow households for a longer period of time, but they typically present only one point-estimate of chronic and transient poverty for the whole period (Dercon and Krishnan, 2000; Cruces and Wodon, 2004). The advantage of the GBA dataset is that, as a series of twelve panels of four observations, it allows the construction of time series of transient and chronic poverty measures.

Table 4.1 presents the partition of the population according to the persistence of the poverty status of the household, as discussed in the previous Section and illustrated in Figure 4.1. As expected, the trends in these four categories follow the evolution of the economy in general. The proportion of those always poor is above the average for the period from the May 1999 to May 2001 cohort, reaching its highest point in the last cohort which corresponds to the economic crisis of 2001-2002. The proportion of those who are classified as never poor mirrors that of the previous group, reaching its lowest point of 42.7 percent for the last cohort, which reflects both the in-

²The change from 1 to 1.5 poverty lines reduces the level of censoring (observations with P_i , C_i , or $T_i = 0$) from 66 and 82 percent of the observations for the transient and chronic measures, to 48 and 66 percent respectively.

Cohort	Persistently Poor	Sometimes poor, mean income below pov. line	Sometimes poor, mean income above pov. line	Never Poor		
95-1 to 96-2	9.4%	23.8%	8.0%	58.8%		
95-2 to 97-1	16.0%	17.5%	10.5%	56.0%		
96-1 to 97-2	8.1%	18.7%	11.6%	61.6%		
96-2 to 98-1	12.0%	17.5%	13.1%	57.3%		
97-1 to 98-2	10.8%	15.1%	8.6%	65.5%		
97-2 to 99-1	9.9%	16.6%	12.1%	61.4%		
98-1 to 99-2	13.8%	15.4%	12.8%	58.1%		
98-2 to 00-1	13.9%	18.5%	11.6%	56.0%		
99-1 to 00-2	17.4%	15.8%	7.5%	59.4%		
99-2 to 01-1	14.5%	18.6%	15.0%	51.9%		
00-1 to 01-2	18.8%	14.9%	10.5%	55.9%		
00-2 to 02-1	23.3%	17.2%	16.8%	42.7%		
Average	14.0%	17.5%	11.5%	57.0%		
Source: Author's	Source: Author's estimations based on EPH household survey data (INDEC).					

Table 4.1: Po	pulation	Groups b	y Mean	Income a	and Per	rsistence o	of Poverty
Sta	tus by Co	ohort, Gre	ater Bue	nos Aire	s, 1995-	2002	

crease in cross-sectional poverty and the fall in households who stay out of poverty for two consecutive waves, as described in Section 1.3. These numbers imply that for the last cohort, covering the period October 2000 to May 2002, almost a quarter of individuals in the Greater Buenos Aires region were poor in the four waves of the EPH, while almost 60 percent were poor in at least one of the surveys.

Besides the evolution of these two "chronic" categories (the always poor and the never poor), Table 4.1 also reflects the considerable movements across the poverty line which were identified in Section 1.3 (Figure 1.8, page 41) and in the empirical applications of Chapters 2 and 3. The downward trend in average household income over the period (Figure 1.3, page 30) explains the fall in the proportion of the sometimes poor with mean income above the poverty line and the increase in the proportion of the sometimes poor with $\bar{y} > z$. Taken together, the evidence in Figure 1.3 and Table 4.1 means that those who escaped poverty in at least one of the four waves are more likely to have average incomes below the poverty line the generalised deterioration of household income over the period.

Figure 4.2 quantifies these effects by presenting the sample averages of

Cohort	Squared Poverty Gap	Chronic Poverty	Transient Poverty	% Chronic	% Transient
95-1 to 96-2	4.5%	2.5%	1.9%	56.7%	43.3%
95-2 to 97-1	6.5%	4.2%	2.2%	65.2%	34.8%
96-1 to 97-2	5.1%	3.1%	2.0%	60.6%	39.4%
96-2 to 98-1	5.7%	3.4%	2.4%	58.7%	41.3%
97-1 to 98-2	5.0%	3.0%	1.9%	60.7%	39.3%
97-2 to 99-1	5.4%	3.5%	2.0%	63.7%	36.3%
98-1 to 99-2	6.3%	4.0%	2.3%	63.2%	36.8%
98-2 to 00-1	6.1%	4.0%	2.1%	65.9%	34.1%
99-1 to 00-2	6.0%	4.0%	2.0%	67.1%	32.9%
99-2 to 01-1	6.9%	4.6%	2.3%	67.2%	32.8%
00-1 to 01-2	7.0%	4.9%	2.1%	69.7%	30.3%
00-2 to 02-1	11.6%	8.0%	3.6%	69.2%	30.8%
Average	6.3%	4.1%	2.2%	64.0%	36.0%
Source: Author	's estimations bas	ed on EPH ho	usehold survey	data (INDEC).	

Table 4.2: Decomposition of Squared Poverty Gap, Greater Buenos Aires, 1995-2002

the measures defined in Equations 4.1, 4.2 and 4.3. The Figure presents the squared poverty gap and its decomposition into its chronic and transient components, and it is complemented by Table 4.2, which also computes the proportion of total poverty attributable to each of these two components.

As expected from the discussion of the previous Chapters, there is a clear upward trend in total poverty, with an increase in the average of P_i from 0.045 for the first cohort to 0.116 for the last one. The results from the partition of the population presented in Table 4.1 are also reflected in Table 4.2: the increase in the always poor explains the large increase in chronic poverty from 0.025 to 0.080, which can also be traced to the fall in household income during the period.

The previous Chapters pointed out that the downward trend in income was accompanied by a marked increase in its variability. The results in Table 4.2 imply that the "level" effect is the most important in terms of this Chapter's decomposition. Despite a large increase in transient poverty from 0.019 to 0.036, its share of total poverty is lower and falls over time at the expense of the chronic measure (last two columns of Table 4.2). The share of chronic poverty increases from 57 percent of the intertemporal measure for the first cohort to almost 70 percent for the last two cohorts.





CHAPTER 4. CHRONIC AND TRANSIENT POVERTY

114

4.3.2 Regression analysis

Table 4.3 presents the result from the estimation of the three regressions in Equation 4.5 by CLAD. Following suggestions in Buchinsky (1994) and given the level of censoring in the data, the regressions are estimated at the 0.8 quantile.

The independent variables are similar to those used for the SUR system of Equation 3.8. They include the initial conditions for (a) household level variables, including the number of infants, children, young adults, adults, and elderly household members, and the square of their values; whether the household head has a spouse; (b) characteristics of the household head, including five age intervals; his/her employment status; his/her level of education; his/her gender; his/her migration status (in the last five years); whether he/she is an employer, a self-employed worker, or a wage earner; the type of his/her qualifications; his/her education level; and whether he/she works in the public sector or is an informal worker; and (c) a subset of these characteristics for the spouse of the household head, when there is one. In addition, the regressions include indicators for each of the cohorts, excluding the first one. The summary statistics of these independent variables are presented in Table 3.1 (page 96).

The results for total poverty (first column of Table 4.3, corresponding to the regression of P_i from Equation 4.1) are similar to those for chronic poverty (second column, with C_i from Equation 4.2 as the dependent variable), and are relatively standard for poverty profiles and regressions of this type (World Bank, 2000a; Cruces and Wodon, 2003a, present more detailed profiles for Argentina).

For P_i and C_i , the number of infants and children is associated with higher levels of poverty despite the adjustments made with the equivalence scale of Table 1.3 (page 1.4) according to Equation 1.3, although the magnitude of this result may be sensitive to the specific choice of scale (Lanjouw and Ravallion, 1995). While the gender of the head does not have a statistically significant effect, heads in the 40-49 age range (the excluded category) are associated with lower levels of poverty. With respect to labour market indicators, unemployment, inactivity and informality of the head or spouse have a very high positive impact on the poverty measures. Self employment of the household head and his/her status (and that of the spouse) as an employer have a positive effect on observed poverty. While surprising, this probably results from the nature of the sample, which consists only of poor people. Among them, the self-employed and those with employees tend to be part of the informal economy with small and precarious businesses (Moreno and Roca, 2002) – this is discussed below in terms of the effect of these indicators on transient poverty. Finally, as expected, higher education levels and professional qualifications of the head and the spouse are unambiguously associated with lower levels of poverty (the excluded categories are no education and no qualifications). These results are broadly consistent with those of Paz (2002) and Gasparini (2003b), discussed in Section 1.4 (page 42).

The most interesting feature of Table 4.3, however, is the presence of disparities in the determinants of chronic and transient poverty, which indicates a differential impact of a household characteristic on the two components.

While the results for the composition of the household are relatively homogeneous among the three columns, households with young heads (19 years or below) have lower levels of chronic poverty but higher levels of transient poverty. The first effect might be due to self-selection: younger individuals might chose to live with relatives if facing chronic poverty on their own. The positive coefficient on transient poverty, on the other hand, implies that younger individuals are subject to more fluctuations in the depth of poverty, probably due to lower job tenure and higher vulnerability to labour market shocks (Mackenzie, 2004). With respect to older heads of household, those aged 65 and more are not associated with higher total poverty (the coefficient is not statistically different from zero), although this characteristic has a negative effect on chronic and transient poverty, which probably reflects the fact that most of these individuals receive steady income streams in the form of pensions.

Another interesting feature of the Table is the public sector indicator, which has a positive and highly significant effect on total poverty for both the head and his/her spouse. This is probably due to the relatively low salaries of civil servants. However, the stability of this income source – and of public employment – is reflected in the lack of effect on transient poverty

	Total Poverty	Chronic	Transient
Household characteristics			
Number of infants (age 0-5)	0.02513	0.04728	0.01322
	[0.00495]***	[0.00656]***	[0.00163]***
Number of infants squared	0.01001	0.00589	-0.00214
-	[0.00140]***	[0.00230]**	[0.00048]***
Number of children (age 6-14)	0.10942	0.14871	0.02124
	[0.00407]***	[0.00332]***	[0.00111]***
Number of children squared	-0.00882	-0.01415	-0.00321
	[0.00088]***	[0.00062]***	[0.00023]***
Number of Youths (age 15-24)	-0.00633	0.02441	0.01252
······································	[0.00442]	[0.00395]***	[0.00135]***
Number of Youths squared	0.00465	-0.00683	-0.00333
	[0.00119]***	[0 00104]***	[0 00038]***
Number of adults (age 25-64)	0.02972	0.00383	0.00837
rumoer of adams (age 25 or)	[0 00744]***	[0 00748]	[0 00219]***
Number of adults squared	-0.01396	-0.00806	-0.00184
	[0 00169]***	10 001 501***	[0 00046]***
Number of elderly members (age 65+)	-0 10748	-0 15352	-0.01712
realiser of enderry memoers (age 05+)	[0 01150]***	[0.01153]***	[0.00413]***
Number of elderly members, squared	0.02211	0.04187	0.00272
Number of elderry members, squared	[0.00/72]***	[0 00/22]###	[0.00212]
No mouse in the household	0.01426	0.01925	0.01101
No spouse in the nousehold	-0.01430	-0.01033	-0.01101
	[0.01290]	[0.01226]	[0.00411]+++
Characteristics of the head	0.1144	0.17014	0.05602
Age - 19 and younger	0.1144	-0.1/914	0.03003
	[0.02546]***	0.013581***	[0.00724]+++
Age - 20-29	0.06319	0.03264	0.00745
4	[0.00804]***	0.00/81]***	[0.00267]***
Age - 30-39	0.02578	0.00103	0.00174
	[0.00596]***	[0.00526]	[0.00183]
Age - 50-59	0.02224	-0.01307	-0.00/25
	[0.00665]***	[0.00591]**	[0.00193]+++
Age - 60 and older	0.00291	-0.01523	-0.00956
F 1	[0.00909]	[0.00835]*	[0.00266]+++
Female	-0.00593	0.00/34	-0.00018
December 1	[0.00/25]	[0.00/64]	[0.00228]
Recent migrant	-0.12515	-0.02/18	-0.01861
Transfer	[0.01651]+++	[0.01188]**	[0.00356]***
Inactive	0.08131	0.12/93	0.01399
	[0.00907]***	[0.00813]***	[0.00269]***
Unemployed	0.20448	0.22963	0.0617
	[0.00825]***	[0.00725]***	[0.00248]+++
Employer	0.06007	0.02433	0.02785
	[0.02213]***	[0.02175]	[0.00489]***
Self-employed	0.029	0.03264	0.00644
	[0.00705]***	[0.00610]***	[0.00216]***
Informal sector worker	0.08725	0.11759	0.01316
	[0.00674]***	[0.00617]***	[0.00197]***
Public sector worker	0.00324	0.03523	-0.00067
	[0.01035]	[0.00956]***	[0.00280]
Level of qualification: Operative	-0.03878	-0.0421	-0.00697
	[0.00574]***	[0.00518]***	[0.00169]***
Level of qualification: Technical / Professional	-0.06805	-0.08	-0.01679
	[0.01307]***	[0.01622]***	[0.00363]***
Primary Education – Complete	-0.02666	-0.05328	0.0026
	[0.00549]***	[0.00480]***	[0.00169]
Secondary Education – Incomplete	-0.03212	-0.06173	-0.01483
	[0.00677]***	[0.00608]***	[0.00203]***
Secondary Education - Complete	-0.16102	-0.18975	-0.02457
	[0.01064]***	[0.01116]***	[0.00261]***
Further Education - Incomplete	-0.11621	-0.07938	-0.04006
-	[0.02494]***	[0.02436]***	[0.00705]***
Further Education - Complete	-0.14017	-0.16669	-0.04981
-	[0.02096]***	[0.01772]***	[0.00885]***
University Education	-0.17906	-0.14143	-0.05126
-	[0.01684]***	[0.01279]***	[0.00474]***

Table 4.3: Censored Quantile Regressions for Total, Chronic and Transient
Poverty, Greater Buenos Aires, 1995-2002

Total Poverty	Chronic	Transient
0.04692	0.0569	0.01223
[0.01155]***	[0.01017]***	[0.00368]***
0.13502	0.12663	0.02476
[0.01315]***	[0.01152]***	[0.00416]***
0.22281	0.07813	0.03661
[0.03064]***	[0.02825]***	[0.00991]***
-0.01644	-0.01411	0.00677
[0.01070]	[0.00957]	[0.00325]**
0.02914	0.01824	0.00537
[0.01276]**	[0.01122]	[0.00406]
-0.01686	-0.03378	-0.01042
[0.01071]	[0 01032]***	[0 00337]***
_0.08941	_0 12380	_0.01078
-0.00941	-0.12J07	-0.01778
0.02007]	0.021	0.01729
-U.UOJZ	-0.031	-0.01/20
[0.02880]***	[0.02557]	[0.01172]
-0.11384	0.03390	-0.01230
[0.02169]***		[0.0053/]**
-0.06353	-0.02894	-0.02344
[0.00621]***	[0.00532]***	[0.00196]***
-0.14017	-0.16398	-0.03052
[0.00757]***	[0.00714]***	[0.00229]***
-0.16016	-0.19071	-0.03345
[0.01019]***	[0.01065]***	[0.00273]***
-0.04291	-0.09544	-0.043
[0.02247]*	[0.01927]***	[0.00859]***
-0.0921	-0.33523	-0.02125
[0.02013]***	[0.01912]***	[0.00514]***
-0.33934	-0.13979	-0.06746
[0.02516]***	[0.01405]***	[0.01014]***
0.07824	0.15982	-0.01241
[0.01138]***	[0.01103]***	[0.00303]***
0.05245	0.0811	-0.00191
[0.01219]***	[0.01251]***	[0.00310]
0.04365	0.09046	-0.00526
[0.01135]***	[0.01128]***	[0.00291]*
0.05176	0.1155	-0.00235
[0.01181]***	[0.01159]***	[0.00310]
0.05642	0 11633	0.00237
[0 01160]***	[0.01135]***	[0.00298]
0 10109	0 16046	0 00464
[0.01121]***	[0 01117]***	[0 00290]
0.05722	0.12870	0.00256
0.03722	0.12079 [0.01000]***	0.00230
0.07760	0.14167	0.002003
0.07/09 [0.01150]###	0.1410/	-0.00007
[0.01138]***	0.19422	0.00302]**
0.09395	0.18423	0.00003
[0.01152]***	[0.0110/]***	[0.00292]**
0.09066	0.14889	0.00282
[0.01145]***	[0.01101]***	[0.00302]
0.17291	0.2458	0.02856
[0.01117]***	[0.01092]***	[0.00280]***
-0.05622	-0.20806	0.01912
[0.01877]***	[0.01884]***	[0.00572]***
2439	1409	3275
0.2000	0 2646	0 1090
	Iotal Poverty 0.04692 [0.01155]*** 0.13502 [0.01315]*** 0.22281 [0.03064]*** -0.01644 [0.01276]** -0.01686 [0.01071] -0.08941 [0.02607]*** -0.0852 [0.02886]*** -0.11584 [0.02169]*** -0.06353 [0.00621]*** -0.14017 [0.02757]*** -0.16016 [0.01019]*** -0.16016 [0.0213]*** -0.16016 [0.0213]*** -0.04291 [0.02247]* -0.0921 [0.0213]*** -0.3934 [0.02516]*** 0.07824 [0.01138]*** 0.05245 [0.01138]*** 0.05176 [0.01131]*** 0.05722 [0.0114]*** 0.05722 [0.01141]*** 0.05955 <td>Iotal Poverty Chronic 0.04692 0.0569 [0.01155]*** [0.0107]*** 0.13502 0.12663 [0.01315]*** [0.02825]*** -0.01644 -0.01411 [0.01070] [0.00957] 0.02914 0.01824 [0.01276]** [0.01122] -0.01686 -0.03378 [0.01071] [0.01032]*** -0.08941 -0.12389 [0.02607]*** [0.01913]*** -0.0852 -0.031 [0.02886]*** [0.02357] -0.11584 0.05596 [0.02169]*** [0.01616]*** -0.06353 -0.05894 [0.00621]*** [0.00532]*** -0.14017 -0.16398 [0.0075]*** [0.0065]*** -0.04291 -0.09544 [0.02247]* [0.01927]*** -0.0921 -0.33523 [0.02516]*** [0.01103]*** -0.03244 0.15982 [0.01138]*** [0.01103]*** -0.05245 0.0811</td>	Iotal Poverty Chronic 0.04692 0.0569 [0.01155]*** [0.0107]*** 0.13502 0.12663 [0.01315]*** [0.02825]*** -0.01644 -0.01411 [0.01070] [0.00957] 0.02914 0.01824 [0.01276]** [0.01122] -0.01686 -0.03378 [0.01071] [0.01032]*** -0.08941 -0.12389 [0.02607]*** [0.01913]*** -0.0852 -0.031 [0.02886]*** [0.02357] -0.11584 0.05596 [0.02169]*** [0.01616]*** -0.06353 -0.05894 [0.00621]*** [0.00532]*** -0.14017 -0.16398 [0.0075]*** [0.0065]*** -0.04291 -0.09544 [0.02247]* [0.01927]*** -0.0921 -0.33523 [0.02516]*** [0.01103]*** -0.03244 0.15982 [0.01138]*** [0.01103]*** -0.05245 0.0811

Source: Author's estimations based on EPH household survey data (INDEC).

•

for the head of household, and the negative and significant impact on T_i for the spouse. A similar effect is found with respect to changes in income by Corbacho et al. (2003), who use EPH data from the 1999-2002 period.

In the P_i regressions, all education and qualification indicators were associated with lower poverty. This is also true for the lowest level of education of the head (completed primary school) and its effect on chronic poverty. However, the lack of a significant effect of this variable on transient poverty (and its positive sign) indicates that little education may help in terms of reducing the level of poverty, but not in moderating its variability.

The positive and significant effect of the head's employer indicator on transient poverty, and its lack of effect on chronic poverty, may explain the puzzle discussed above: small scale entrepreneurs may not face higher levels of chronic poverty, although the risky nature of their businesses implies higher variability and thus higher transient measures. The same explanation probably applies to the self-employed indicator for the spouse, which has a (non-significant) negative effect on total and chronic poverty, but a positive and significant effect on its transient component.

The positive and significant values of the cohort indicators for the total and chronic measures have a clear upward trend, which reflects the fall in household income over the 1995-2002 period. Interestingly, these indicators only have a statistically significant effect on T_i for a few cohorts, indicating that while the increase in the depth of poverty was unambiguous, its transient dimension varied over the period. The excluded cohort is the first one (covering May 1995 to October 1996). The coefficients of the October 1995-May 1997 and the October 1996-May 1998 cohorts in the transient poverty regression are negative and significant, indicating that the period of recovery after the Mexican crisis was marked by a fall in the variability of well-being over time. The indicator corresponding to the May 1999 to October 2000 is also negative and significant in the T_i regression, due to the slight but short-lived recovery in income and unemployment for October 2000 (Figure 1.3). In contrast, the cohort indicators for these three cohorts are positive and strongly significant for the P_i and C_i regressions. The large fall in income during the recession of 1999-2001 and the 2001-2002 crisis, however, are reflected in the positive and highly significant values of the indicators for cohorts 10 and 12 in the three regressions.

Finally, the results on the determinants of chronic and transient poverty can be compared to those of the risk adjusted income regressions of Section 3.3 (page 89). As noted in the Introduction to this Chapter, the methodology presented in Chapter 3 concentrates on the effects of income risk at every point of the income distribution, while the transient-chronic decomposition deals only with the level and variability of well-being among the poor, so the two sets of results can be considered complementary.

The results for income and poverty levels are consistent: employment status, education and qualifications have the largest impacts on both. The comparison between the analysis of risk – the difference between average and risk adjusted income – and that of differences between transient and chronic poverty is certainly the most interesting. The regressions of Section 3.3 (Table 3.2, page 98) indicated that households with elderly members faced lower levels of risk, which is consistent with the negative effect of older household heads on transient poverty in Table 4.3. The informality indicators contributed to higher levels of risk and higher levels of poverty, although the employer and self-employed variables, which affect transient poverty, have no significant effect on average income or risk. Another co-incidence is that education and qualifications reduce both risk and the transient component of poverty. Finally, the results in Tables 3.2 and 4.3 capture the fall in income and the increase in its variability induced by the 2001-2002 crisis.

4.4 CONCLUSION

This Chapter presented a decomposition of intertemporal poverty that accounts for the large fluctuations of poverty and household income in Argentina for the period 1995-2002. This methodology was critically assessed in Chapter 2 in the context of the evaluation of income fluctuations over time.

The partition of the population into different groups, according to their average income over time and their exposure to poverty, complemented the decomposition of a poverty measure in its transient and chronic components. The analysis of the trend in these two elements revealed that the increasing deterioration of living standards over this period was not only reflected in its level, but also in its variability. However, given the large fall in household income due to the 2001-2002 crisis, the largest share of poverty was due to its chronic component.

The decomposition also allows for the study of the correlates of total, transient and chronic poverty. This type of regression analysis represents a step forward with respect to the simple poverty profiles, since the differences in the impact of variables on chronic and transient poverty reveals their effect not only on the level of well-being, but also on its variability. These results are broadly consistent with those of the risk adjusted income regressions presented in Chapter 3. They complement that Chapter's approach and the results from the longitudinal studies for Argentina reviewed in Section 1.4.

Part II

Fertility and Women's Labour Supply in Argentina: Identification of Causal Effects Through Sex Preferences

CHAPTER 5

EFFECTS OF FERTILITY ON WOMEN'S LABOUR SUPPLY: METHODOLOGY, ESTIMATION STRATEGY AND DATA FOR ARGENTINA

5.1 INTRODUCTION TO PART II: POVERTY, FERTILITY, AND WOMEN'S LABOUR SUPPLY

In the previous discussion of poverty, risk and income fluctuations in Argentina, larger households (and especially those with more children) were found to be poorer and more prone to suffering from income fluctuations. At the same time, the employment status of the head of household and the spouse was established as a strong correlate of chronic and transient poverty.

Studies of this type provide a valuable input for the analysis of poverty and its alleviation, and constitute a step towards understanding the mechanisms underlying deprivation and vulnerability. However, the recovered correlations do not imply causality, at least not without further assumptions: causal relations are blurred by the endogeneity and simultaneity of the different factors at stake. For instance, household size and fertility might be "causing" poverty by the simple mechanical effect of dividing income among a larger number of people,¹ but also by hampering the opportunities of income-generating activities of some household members. At the same time, poverty itself may induce a rise in fertility, because of a lower opportunity cost of a parent's time, or because of reduced access to family planning

¹In the case of poverty measurement in Argentina, this is manifested in the number of household members k_i in the denominator of Equation 1.2.

possibilities.² Similar concerns about the direction of causality also arise in the relationship between poverty and labour market outcomes.

The next step after the regression results of Chapters 3 and 4 consists in establishing the nature of the causal links underlying the identified correlations.

The results of Part I on the positive correlations between poverty and fertility, on the one hand, and poverty and joblessness, on the other, motivate the research presented in this Part, which establishes the determinants of the income-generating process of women as primary and secondary earners in the household. Specifically, the following Chapters study the effects of fertility on female labour supply, prompted not only by their correlation with poverty measures, but also by their evolution in Argentina: Figure 5.1 shows that female labour force participation increased by almost 60 percent in the 1980-2000 period, while fertility fell significantly by almost 25 percent during the same years and is expected to reach replacement levels by 2010-2015 (Pantelides, 2002; Binstock, 2004; CELADE, 2004).³ The following pages will attempt to establish whether the change in fertility can explain the large increase in female labour force participation.

Fertility transitions of this type, and an even higher incorporation of women into the labour force in developed countries during the twentieth century, prompted a continuous interest in the relationship between childbearing and women's labour supply in the labour economics and demography literatures.⁴

However, endogeneity issues are pervasive in applied empirical research. Concerns about the endogeneity of fertility and work decisions imply that much of this literature has been devoted to disentangle the causal mechanisms linking childbearing and women's labour supply. In a review of the literature, Willis (1987) wrote that "... it has proven difficult to find enough well-measured exogenous variables to permit cause and effect relationships to be extracted from correlations among factors such as [...] the decline of childbearing [...] and increased female labour participation." Since the

²See Anand and Morduch (1998) and Lipton (1998) for a discussion of these points.

³Replacement level fertility implies a total fertility rate usually between 2.1 and 2.2 children per woman, depending on a country's level of mortality.

⁴See, among others, Gronau (1973), Rosenzweig and Wolpin (1980), Killingsworth and Heckman (1986), Willis (1987), and Browning (1992).

CHAPTER 5. ESTIMATION STRATEGY AND DATA



Figure 5.1: Female Labour Force Participation (Women Aged 14 and Older) and Fertility (Children per Woman), Argentina 1960-2010

publication of that review, however, substantial progress has been made in labour economics by means of "natural experiments," which exploit the exogenous dimension of naturally occurring events as sources of variation (Angrist and Krueger, 1999, 2001).

The aim of Chapters 5 to 8 is to provide a baseline measure of the impact of fertility on women's labour supply in Argentina. The following Chap"cause and effect relationships."

It is argued, however, that whether the identification strategy can be applied to developing countries or not must first be established. The empirical results for Argentina constitute the first contribution of the following pages. The application to Argentina establishes whether the causal effect of childbearing on female labour supply holds in developing countries, where fertility is typically higher, and female education and labour force participation levels are lower than in developed countries (United Nations, 2002).

The second contribution, presented in this Chapter and the following, consists in the discussion of the "same sex" strategy in the context of developing countries, and in the proposed interpretation of a set of auxiliary evidence for the evaluation of the identifying assumptions. Chapter 6 maintains that since the exclusion restrictions of instrumental variables are inherently non-testable, their plausibility must be evaluated on a case by case basis by means of indirect evidence. In particular, it is argued that the type and degree of sex preferences have to be assessed to establish the validity of the exclusion restrictions. Finally, the third contribution, presented in Chapter 8, is the derivation of a new test to support the generality of instrumental variable results.

The present Chapter provides an econometric framework for the application of the "same sex" estimation strategy to Argentina. Section 5.2 illustrates the theoretical relationship between fertility and female labour supply. It then describes the potential outcomes framework and the conditions for identification of causal effects. Section 5.3 studies in detail the "same sex" strategy within this framework. Section 5.4 describes the data employed for the analysis and presents a set of summary statistics of the main variables. Some brief conclusions follow.

5.2 THEORY AND METHODOLOGY

5.2.1 Theoretical framework: fertility and labour supply

The analysis of women's labour supply is intrinsically connected to the economic models of fertility, time allocation and household decision making

household survey data from Korea, and Cruces and Galiani (2003) present additional data and estimates from Mexico.

(Becker, 1991, summarises the developments in this field).

The essence of the economic analysis of fertility and labour supply is that the two are usually considered joint decisions: while children provide utility to their parents, they also enter the household's budget constraint since they involve considerable costs, both in terms of goods (e.g., food and school materials) and time devoted to childcare.

These relations are captured in this Section by means of a stylised static model of a nuclear household adapted from Browning (1992). This model concentrates on the woman's side of the problem, taking other variables, such as the spouse's income, hours of work at home and in the market as fixed.

The woman's utility function is defined as $U = U(x_m, x_c, l, h_h, c)$, and is assumed to be increasing in all its arguments: x_m and x_c denote the consumption of the mother and the children, l is the time devoted to leisure, h_h is the time spent at home, and c is the number of children. The woman divides her total time T between work at home h_h , leisure l and work in the market h_m , for which she is compensated with a wage w. Finally, the budget constraint is completed by a fixed quantity I, representing the household's other sources of income, the price p_x of consumption goods and a cost per child given by p_c . The woman chooses the optimal quantities of her choice variables to solve the following maximisation problem:

$$\max_{x_m, x_c, l, h_h, c} U = U(x_m, x_c, l, h_h, c) \text{ subject to}$$

$$I + wh_m = p_x(x_m + x_c) + p_c c \quad \text{(budget constraint)}$$

$$T = h_h + h_m + l \qquad \text{(time constraint)}$$

The two constraints can be summarised as $I + wT = (wl + p_x(x_m + x_c)) + (p_cc + wh_h)$, which describes the allocation of the household's full income between the woman and the child, if work at home is considered mainly as childcare and household chores.

This stylised model highlights the main theoretical relationships needed for the discussion of this Part. The shape of the utility function U and the first and second order conditions of the optimisation problem in Equation 5.1 define the demand for children, the labour force participation and the labour supply of the woman. Killingsworth and Heckman (1986), Montgomery and Trussell (1986) and Browning (1992) present more elaborate models covering single-parent households, labour supply dynamics, private childcare, household production (Becker, 1965; Gronau, 1977), joint allocation of hours of man and woman (Gronau, 1973), and the quantityquality of children interaction (Becker and Lewis, 1973). Cigno (1991) and Ermisch (2003) provide full treatments of all these aspects of the economics of the family.

The simple setting of Equation 5.1 illustrates some important considerations. Firstly, the labour supply, childcare and fertility decisions are simultaneous, although this is partially due to the static nature of the model: dynamic models capture the irreversibility of fertility decisions and their subsequent effects on time allocation. Secondly, while the model of Equation 5.1 is too general for the derivation of demands for the different goods (and their interactions) without functional form assumptions, it still captures the essence of the causal relation between fertility and female labour supply: the utility function and the budget and time constraints imply a trade-off between "pure" utility from children, wage-income and the time and goods needs of children. Dynamic models often result in a negative causal effect of fertility on short-run labour supply through the time needs of children in the time constraint. Finally, the model can be used to illustrate the endogeneity that arises in the empirical estimation of labour supply models.

The objective of this Part is to obtain estimates of the direct effect of children c in the labour supply of women, represented by $h_m^* = T - l^* - h_h^*$. Following Browning (1992), the model from Equation 5.1 results in a conditional labour supply – either in terms of hours, or as a binary participation indicator – defined as $Y = f(\mathbf{K}, D)$, where **K** is a vector that contains the variables in the model of Equation 5.1 and some exogenous characteristics, and D is a measure of fertility – such as the number of children, or an indicator of more than c children in a sample of women with c or more children.

The parameter of interest is the labour supply response to changes in the fertility variable, f_D . This parameter, however, is difficult to recover by simple statistical methods as illustrated by the derivative of Y with respect to D. Ignoring the effects of fertility on all the other variables in Equation 5.1 (present in **K**), which requires a series of highly implausible assumptions (Browning, 1992), but considering the potential effects of fertility on wages,

for example, this derivative is equal to:

$$\frac{\partial Y}{\partial D} = \frac{\partial w}{\partial D} f_w + f_D \tag{5.2}$$

Childbearing might have an effect on wages, for instance because of the foregone appreciation in the woman's "stock of experience" during maternity leave (Cigno, 1991, Chapter 7). This would imply that $\partial w/\partial D \neq 0$. Moreover, the wage w is determined by ability and motivation factors that are unobservable and that may be correlated with fertility decisions through the childbearing and leisure preferences in the utility function U. Taking into account all the variables of the model in Equation 5.1 would add partial derivatives (in Equation 5.2) of the components of **K** with respect to D.

This discussion implies that a fertility indicator D would be endogenous in a labour supply model – a regression of Y on D is flawed because "variables that are explicitly or implicitly assumed to be fixed are [...] within the control of mothers" (Browning, 1992). An additional factor is that unobserved factors might be driving both decisions.

An alternative is to evaluate the structural parameters implied by the utility function and the constraints, but models of this type rely on a large number of assumptions, since preferences are not observable. As pointed out by Willis (1987), a solution to these endogeneity problems resides in finding a variable *Z* that induces variation in fertility but does not affect the labour supply decision directly, which allows the derivation of a reduced form relationship between fertility and labour supply.

Continuing the example presented in Equation 5.2, which only considers the endogenous effect of fertility on the wage rate, if Z is not related to the factors that account for $\partial w/\partial D$ then:

$$\frac{\partial Y}{\partial Z} = \frac{\partial w}{\partial Z} f_w + \frac{\partial D}{\partial Z} f_D, \text{ resulting in } f_D = \frac{\partial Y}{\partial Z} / \frac{\partial D}{\partial Z}$$
(5.3)

since the exogeneity of Z with respect to w implies that $\partial w/\partial Z = 0$. The parameter of interest, the response of the labour supply to changes in fertility, is thus identified. This intuitive idea is the basis of the statistical framework and identification strategy presented in the following pages.

CHAPTER 5. ESTIMATION STRATEGY AND DATA

5.2.2 The potential outcomes framework

Instrumental variables techniques, based on variables like Z in the previous Section, have been used since the 1920s to identify causal effects by exploiting exogenous variations in the variables of interest (Wright, 1928; Angrist and Krueger, 2001). This section presents the potential outcomes framework, a general setting for causal inference within which instrumental variables estimators can be given a causal interpretation without functional form assumptions.⁶

The objective is to evaluate the causal effect of some treatment over an outcome of interest: in this Part, the "treatment" denotes a measure of fertility D, and the outcome Y is related to the employment status of a woman. This Section focuses on the case of binary treatments and outcomes:⁷ Y_i is equal to 1 if the woman i is employed, and 0 if she is not, while D_i is an indicator of whether i has more than two children in a sample of women with at least two offspring, which corresponds to the Angrist and Evans (1998) setting discussed in the following Section.

As observed in the discussion of Equation 5.2, the direct effect of D on Y in empirical models of women's labour supply is difficult to obtain because of simultaneity and endogeneity concerns. The potential outcomes framework attempts to solve endogeneity problems of this type. Its starting point is the definition of a causal effect, which relies on the notion of potential – as opposed to observed – outcomes Y_d (Abadie, 2003a): Y_{0i} is the potential outcome without treatment for individual *i*, and represents the level of *Y* that *i* would attain if not exposed to the treatment. Y_{1i} is defined analogously: it represents the outcome that the same individual *i* would attain if exposed to the treatment. The key feature is that potential outcomes refer to unobserved counterfactuals. An individual *i* has either been exposed ($D_i = 1$) or not exposed to the treatment ($D_i = 0$), and Y_i represents *i*'s actual observed outcome that would have been observed in the two alternative situations, one of which is necessarily a counterfactual. In the case of fertility and wo-

⁶This Section draws mainly on Abadie's (2003a) presentation of results by Imbens and Angrist (1994), Angrist et al. (1996), Angrist (2001) and Abadie (2002, 2003b), among others.

⁷Most of the results hold in models with variable treatment intensity (Angrist and Imbens, 1995), as discussed below. The empirical results in Chapters 7 and 8 deal with both cases.

men's labour supply, the employment status of a woman *i* who has more than two children ($D_i = 1$) is Y_{1i} , which is equal to the observed Y_i , and Y_{0i} represents the same woman's counterfactual employment status which would have been observed if she had only two children ($D_i = 0$).⁸

The causal effect of the treatment *D* for individual *i* is defined in terms of counterfactuals as the difference between the two potential outcomes, $Y_{1i} - Y_{0i}$ (Abadie, 2003a). In the previous example, the causal effect of childbearing is the divergence in employment status between having two or more than two children – this difference is due only to childbearing, and thus it is not contaminated by other endogenous factors as in Equation 5.2.

Since causal effects are defined in terms of inherently unobserved counterfactuals (i.e., what labour supply would have been in another situation), the problem of identification is to find a way of expressing the parameter of interest in terms of observable quantities. Moreover, since Y_{0i} and Y_{1i} cannot both be observed at the same time for *i*, causal effects cannot be computed at the individual level: the identified parameter refers necessarily to the average causal effect $E[Y_1 - Y_0]$.

A first naive approach, corresponding to that of Equation 5.2, consists in comparing the average outcomes by treatment status. However, the simple comparison between the treated (E[Y|D = 1]) and the non treated (E[Y|D = 0]), as given by an ordinary least squares regression of *Y* on *D*, is unlikely to identify any meaningful causal effect. This can be shown by re-writing the difference of these expectations in terms of potential outcomes:

$$E[Y|D = 1] - E[Y|D = 0] =$$

$$= E[Y_1|D = 1] - E[Y_0|D = 0] + E[Y_1 - Y_0] - E[Y_1 - Y_0]$$

$$= E[Y_1 - Y_0] + \underbrace{\{E[Y_0] - E[Y_0|D = 0]\} - \{E[Y_1] - E[Y_1|D = 1]\}}_{bias}$$
(5.4)

where $E[Y_1 - Y_0]$ is the average treatment effect, and the last term represents the bias. The elements in the first line of Equation 5.4 are observable, but since the bias term is composed of non-observable counterfactuals the average treatment effect cannot be distinguished from the simple difference

⁸Formally, the observed outcome Y_i is a function of potential outcomes and the treatment indicator D_i : $Y_i = Y_{0i}(1 - D_i) + Y_{1i}D_i$.

in average outcomes.⁹

The identification of the causal effect $E[Y_1 - Y_0]$ in Equation 5.4 can still be attained in situations where the bias is equal to zero, although this is unlikely to be the case in most applications due to problems of selection and endogeneity like those described in the previous Section. For instance, individuals might choose whether to enter the treatment by estimating their own potential outcomes, resulting in a non-zero bias term in Equation 5.4 – Section 5.3.1 discusses this in detail in the context of fertility and female labour supply.

A well-known exception in which the left hand side of Equation 5.4 successfully estimates the average causal effect is the case of random assignment, which plays a central role in identification. When the population is randomly divided into treatment ($D_i = 1$) and control ($D_i = 0$) groups, the expectations of Y_1 and Y_0 are independent of D. Intuitively, the assignment ensures that individuals do not self-select into treatment, and thus the bias in Equation 5.4 is forced to be 0 by the experimental design.

While randomised experiments are difficult to carry out in economics, and infeasible in the study of fertility and women's labour supply, the problems of selection and endogeneity can be overcome by finding a setting akin to random assignment. The next Section discusses the case of instrumental variables, in which identification is attained with a variable, correlated with the treatment, which is "as good as randomly assigned" (Angrist, 2004).

5.2.3 Identification by instrumental variables: LATE and the Wald estimator

A solution to the endogeneity problem in the theoretical model described above was given in Equation 5.3 by means of exogenous variation in the causing variable. Imbens and Angrist (1994) translate the intuition of that

$$E[Y|D = 1] - E[Y|D = 0] = E[Y_1|D = 1] - E[Y_0|D = 0]$$

= $E[Y_1 - Y_0|D = 1]$
+ $\{E[Y_0|D = 1] - E[Y_0|D = 0]\}$

where $E[Y_1 - Y_0|D = 1]$ is defined as the average causal effect on the treated, and the term $E[Y_0|D = 1] - E[Y_0|D = 0]$ is the bias. Equation 5.4 seems better for presentational purposes since instrumental variables identify the average treatment effect and not necessarily the effect on the treated.

⁹Alternatively, the bias can be expressed in terms of the average treatment effect on the treated, as done by Abadie (2003b) and Angrist (2004):

Equation into the potential outcomes framework, showing that the identification of causal effects when the treatment is non-random can be attained with an instrumental variable (IV) Z which induces exogenous variation in the treatment. The condition, as in Equation 5.3, is that Z must be correlated with the treatment but only affect the outcome Y through its effect on D: in the Angrist and Evans (1998) example, the instrument Z is an indicator for women whose first two children are of the same sex, which, as discussed in the next Section, is correlated with having more than two children, but does not have a direct impact on the mother's labour supply.

Some additional notation and definitions are needed to derive the instrumental variables identification result. Given a binary instrument Z, D_z represents the potential treatment status given Z = z: $D_z = D_0$ if Z = 0and $D_z = D_1$ if Z = 1. Just as Y_1 and Y_0 represent the potential outcomes in terms of the treatment D, D_z represents the treatment status that would be observed for different values of the instrument. With this notation, the potential outcomes can be re-defined in terms of the treatment status and the instrument as Y_{zd} . As in the case of potential outcomes, just one treatment status is observable for each individual, either D_0 or D_1 , but both counterfactuals are defined.¹⁰

Following Angrist et al. (1996), this setting allows the partition of the population into four mutually exclusive categories according to the values of D_z :

- 1. Compliers: $D_0 = 0$ and $D_1 = 1$. These individuals receive the treatment when the instrument is 1, but do not receive it when it is 0.
- 2. Always takers: $D_0 = 1$ and $D_1 = 1$. These individuals are always exposed to the treatment, regardless of the value of the instrument.
- 3. *Never takers*: $D_0 = 0$ and $D_1 = 0$ These individuals are never exposed to the treatment, regardless of the value of the instrument.
- 4. *Defiers*: $D_0 = 1$ and $D_1 = 0$. These individuals receive the treatment when the instrument is 0, but do not receive it when it is 1.

¹⁰Formally, the observed treatment D_i is a function of potential treatment and the instrument Z_i : $D_i = D_{0i}(1 - Z_i) + D_{1i}Z_i$.

Compliers are those whose treatment status is changed by the instrument. In terms of the childbearing example, compliers are women who would have had an additional child if their first two children were of the same sex, but would not have had it if the first two were of different sex. Always takers always have more than two children irrespective of the sex of the first two, whereas never takers stop at two in any circumstance. Finally, defiers are simply those who behave in the opposite way of compliers: two children of the same sex induce them to stop having children, while two children of different sex encourage them to have more than two. Since these definitions are based on unobservable potential treatments, it is impossible to assign a single individual to any of these four groups from observable characteristics.

The following assumptions represent a set of nonparametric conditions¹¹ in terms of Y_{zd} , D_z and Z under which the instrumental variables estimators identify causal effects for the subpopulation of compliers (Abadie, 2003b):

Assumption 5.1 Independence: the vector $(Y_{00}, Y_{10}, Y_{01}, Y_{11}, D_0, D_1)$ is independent of the instrument Z.

Assumption 5.2 *Exclusion:* $Y_{1d} = Y_{0d}$ for d = 0 and d = 1.

Assumption 5.3 *First stage:* 0 < Pr(Z = 1) < 1 *and also* $Pr(D_1 = 1) > Pr(D_0 = 1)$.

Assumption 5.4 *Monotonicity:* $D_1 \ge D_0$.

The independence Assumption 5.1 refers to the treatment assignment mechanism with respect to Z, requiring that the instrument is randomly assigned.

Assumption 5.2 means that $Y_{00} = Y_{10} = Y_0$ and $Y_{01} = Y_{11} = Y_1$, implying that for a given treatment status *d*, the potential outcome Y_{zd} is always the same, irrespective of the value of the instrument. Potential outcomes can then be defined in terms of the treatment *D* alone, Y_d , as in Section 5.2.2. This exclusion restriction means that any influence of the instrument on the

¹¹Implicit in this notation is the "Stable Unit Treatment Value Assumption," which postulates that an individual *i* is not affected by the treatment received and instrument assigned to other individuals in the population (Angrist et al., 1996).

potential outcomes comes only through its effect on the treatment D: the instrument Z does not affect Y directly.

The first part of Assumption 5.3 rules out the possibility that either all or none of the individuals in the population have a positive value for the instrument indicator – any of the two extremes would render Z trivial. The second part of the Assumption states that the instrument affects the probability of treatment, which is required to be more likely to equal 1 for those with $Z_i = 1$ than for those with $Z_i = 0.12$

Finally, Assumption 5.4 requires that changing Z from 0 to 1 should not shift an individual from receiving the treatment to not receiving it. This effectively rules out the presence of defiers, which is not controversial in most applications.¹³

Given an instrument *Z* that satisfies Assumptions 5.1-5.4, Imbens and Angrist (1994) show that the causal effect of the treatment is identified for those whose treatment status is changed by the instrument ($D_0 = 0$ and $D_1 = 1$), the group of compliers:

Proposition 5.1 Local Average Treatment Effect (Imbens and Angrist, 1994): If Assumptions 5.1, 5.2, 5.3 and 5.4 hold, then:

$$E[Y_1 - Y_0|D_0 = 0, D_1 = 1] = \frac{E[Y|Z = 1] - E[Y|Z = 0]}{E[D|Z = 1] - E[D|Z = 0]}$$
(5.5)

This estimator is the Local Average Treatment Effect (LATE). It is an identification result in the sense that a difference in unobservable potential outcomes can be expressed as a function of observable quantities. In the presence of an instrumental variable that fulfils the conditions set above, the causal effect of the treatment D for the group of compliers is the quotient in Equation 5.5.¹⁴ This result is obtained without any functional form

¹²The "greater than" signs in Assumptions 5.3 and 5.4 can be reversed without loss of generality.

¹³This assumption is not specific to the potential outcomes framework: as noted by Angrist et al. (1996), it is implicit in standard instrumental variable models. The same authors provide a detailed discussion of these four assumptions and the consequences of their violations.

¹⁴While the discussion focuses on binary treatments, Angrist and Imbens (1995) show an equivalent identifying result in models with variable treatment intensity. When *D* is not binary, the parameter ϕ in Equation 5.5 captures the weighted average of causal responses to a unit change in treatment. In the fertility example, if the variable *D* is the number of

assumption.

Proposition 5.1 identifies the causal effect for the subpopulation of compliers, since for this group the treatment status *D* is determined by a random event *Z*. This rules out the issue of self-selection, making compliers akin to a group subject to an experiment with random assignment to treatment.

A limitation of the LATE parameter from Proposition 5.1 is that it can only be extrapolated as a causal effect for the whole population by assuming no heterogeneity in potential outcomes between compliers and other groups. Under that assumption, LATE is equal to the population average causal effect $E[Y_1 - Y_0]$. Chapter 8 addresses the plausibility of assumptions of this type in the context of fertility and women's labour supply.

The LATE parameter and its standard error can be recovered by means of an instrumental variable model with a dummy endogenous regressor. This coefficient is known as the "Wald" estimate, and is obtained by estimating the Equation:

$$Y_i = \gamma + \phi D_i + \varepsilon_i \tag{5.6}$$

where Y_i and D_i are defined as before, γ is a constant term, ε_i is an error term, and the treatment variable D_i is instrumented with Z_i .

In some cases, Assumptions 5.1, 5.2, 5.3 and 5.4 may only hold conditionally on some observable exogenous variables X_i . Alternatively, a researcher might be interested in controlling for these characteristics in the estimation of causal effects to extrapolate results to another group of the population with different values of X_i . In these cases, the simplest option is to estimate the causal effect is by means of a linear, constant-effects model of the form:

$$E[Y_{oi}|X_i] = X'_i\beta$$
 and
 $Y_{1i} = Y_{oi} + \phi$

where X_i is a vector of control variables that influence the outcome Y independently of the treatment status, as described in the first Equation (Angrist, 2001). The second Equation implies that the causal effect of the treatment represents a shift in the outcome, as in the LATE setting above. These as-

own children, Y_j represents the potential labour supply for an individual at each value j of D, and the LATE parameter represents the average effect of an increase in the number of children by one.

sumptions lead to the following linear causal model:

$$Y_i = X'_i \beta + \phi D_i + \varepsilon_i \tag{5.7}$$

This model can be estimated by two-stage least squares, with Z_i as an instrument for D_i , to obtain the causal effect parameter ϕ . The instrumental variable Z_i must satisfy an extended version of Assumptions 5.1-5.4 that makes them conditional on the controls X_i (Abadie, 2003b).¹⁵

The two-stage least squares (2SLS) model in Equation 5.7 is equivalent to the canonical version of the instrumental variable model with one endogenous regressor. Moreover, it has been shown that this setting can be considered as a special case of the latent index model.¹⁶ The advantages of the framework described in this Section are threefold. Firstly, it provides an alternative interpretation of these models in terms of causal inference. It also individualises the underlying assumptions, making conditions for identification explicit. Finally, these conditions do not depend on functional form assumptions. The following Section discusses the application of the framework to the case of fertility and women's labour supply, highlighting the endogeneity problems that arise in this context and the potential for identification by means of instrumental variables.

¹⁵Abadie (2003b) points out that the introduction of covariates X_i implies that the parameter ϕ in Equation 5.7 is not exactly a LATE as defined in Proposition 5.1, and he proposes a weighted "Casual IV estimator" for ϕ that has a LATE interpretation. However, his own results, as well as estimations from Angrist (2001) and Cruces and Galiani (2003), find the "Casual IV" estimates of ϕ in Equation 5.7 almost indistinguishable from those obtained by two-stage least squares. This is due to the fact that the covariates used in the estimation are discrete with finite support (Abadie, 2003b).

¹⁶The model in Equation 5.7 can have an alternative interpretation as a latent index model where $Y_i^* = X_i'\beta + \phi D_i + \varepsilon_i$, with $Y_i = 1$ if $Y_i^* > 0$, $Y_i = 0$ if $Y_i^* \le 0$, and $\varepsilon_i = \delta \theta_i + u_i$. Y_i^* is a latent outcome variable, and the error term is composed by a random component u_i and an unobservable parameter θ_i affecting the treatment and the outcome (Iacovou, 2001). Heckman et al. (2003) discuss the equivalence of the potential outcomes and latent index models, using a latent-variable framework "to unite the recent treatment-effect literature with the classical selection-bias literature." This issue is also explored at length by Angrist (2001, 2003).

5.3 FERTILITY, WOMEN'S LABOUR SUPPLY AND THE "SAME SEX" ESTIMATION STRATEGY

5.3.1 Fertility and women's labour supply: endogeneity and selection issues

A variety of studies from different disciplines as well as casual observation indicate the presence of a negative correlation between childbearing and women's labour supply (see references in Section 5.2.1). However, this correlation cannot be interpreted as a causal effect without further (and debatable) assumptions on the nature, size and sign of the selection bias term.

The bias term in Equation 5.4 provides a simple illustration of selection as an obstacle to identification. The bias arises because expected potential outcomes may affect self-selection into treatment: this term would be 0 in Equation 5.4 (and thus identification achieved) only if the decision to have a child is independent of a women's potential labour market outcomes when not having any children. In that case, the treatment D becomes irrelevant for the expectation of potential outcomes, and the bias disappears:

$$Bias = \{E[Y_0] - E[Y_0|D = 0]\} - \{E[Y_1] - E[Y_1|D = 1]\}$$
(5.8)
= $\{E[Y_0] - E[Y_0]\} - \{E[Y_1] - E[Y_1]\} = 0$

However, the theoretical results discussed in Section 5.2.1 and empirical evidence suggest that fertility and women's labour supply are jointly determined and that there is a causal link between the two, making this independence assumption implausible. Childbearing decisions take into account expected potential outcomes, career plans, comparative advantages, preferences and the division of labour within the household, implying that women with higher fertility are probably different in terms of potential outcomes from women with lower fertility.

For instance, women lacking opportunities for childcare arrangements, with stronger preferences for children, or who forecast relatively poor labour market outcomes (such as a low quality job, or low wages)¹⁷ might decide to work at home and have children. These women self-select into treatment because they expect a relatively low Y_1 , implying that their poten-

¹⁷This might be related to individual ability, but can also be affected by opportunities and by the extent of discrimination in labour markets, among other factors.

tial outcomes are lower than average: this results in $E[Y_1|D = 1] < E[Y_1]$. On the other hand, women who expect good labour market outcomes probably self-select into lower fertility (non treatment): the potential outcome of this group is above average, implying that $E[Y_0|D = 0] > E[Y_0]$.

A situation like this results in a negative bias term in Equation 5.4,¹⁸ but this is due to the nature of the example: in other cases, the bias might be positive. For instance, this is the case if $E[Y_1]$ is very low within women who self-select into D = 0, and relatively high among those who chose D = 1. In a situation like this, women who chose lower fertility do so because of low expected outcomes when having children (Y_1), and not because of good outcomes without children as in the previous example.

While the framework outlined in the previous Section proposed a solution to the endogeneity problems of this type through instrumental variables, the problem remains in finding a suitable variable *Z*, which is discussed in the following pages.

5.3.2 Identification through natural experiments

As shown by Proposition 5.1, the identification of the causal effect in Equation 5.4 can be attained through an instrument *Z* that meets the requirements of Assumptions 5.1-5.4. However, finding a valid instrument in socioeconomic data is not evident: most of the observed individual characteristics are plausibly correlated with unobservable factors driving the outcome, making them endogenous to the relationship of interest in violation of Assumption 5.2. For instance, there is strong evidence of a link between the level of education and women's fertility decisions, but it is unlikely that education could make a good instrument for childbearing since it certainly affects labour supply through channels other than the number of children. Moreover, while most meaningful socioeconomic characteristics might be considered to be jointly determined, other plausibly exogenous individual attributes might not have an impact in the variable of interest, as required by Assumption 5.3.

¹⁸Using the alternative definition of the bias in terms of the average treatment on the treated given in footnote 9 (page 132), the bias is $E[Y_0|D = 1] - E[Y_0|D = 0]$. Women who expect high outcomes without the treatment self-select into D = 0, which results in a negative bias since $E[Y_0|D = 0] > E[Y_0|D = 1]$.

These problems with socioeconomic characteristics and the difficulties of carrying out randomised experiments in the social sciences imply that researchers often rely on "natural experiments," situations equivalent to random treatments that arise by chance. Some examples of naturally occurring events exploited in labour economics are birth date, gender composition of offspring and twin births – Angrist and Krueger (1999) review these applications, and Rosenzweig and Wolpin (2000) provide a critical assessment of the literature.

The main advantage of the natural outcomes of the previous paragraph is that, in the words of Rosenzweig and Wolpin (2000), they are "plausibly random with respect to at least two of the major sources of heterogeneity in human population, tastes and abilities." The consequence is that these natural outcomes are exogenous, as required by Assumption 5.2. If they influence the variable whose effect is of interest, they comply with Assumption 5.3 and they can be exploited as instruments if Assumptions 5.1 and 5.4 are also met.

5.3.3 The "same sex" strategy: sex mix as a natural experiment

This is the approach proposed by Angrist and Evans (1998) for the study of fertility and female labour supply. The "same sex" estimation strategy relies on the sex of a women's first two children as an instrument for fertility in a model of female labour supply. Its suitability as an instrument is given by two hypotheses. On the one hand, the gender of children is random. On the other hand, the gender mix has an impact on fertility as a consequence of parental sex preferences, but does not affect directly the parent's labour supply. These premises are discussed in detail for Argentina in the following Chapter.

In general terms, sex preferences can be defined as the utility that parents derive from the gender of their children. This issue, and its relationship to fertility behaviour, has been dealt with from different angles, from indepth ethnographic studies to demographic statistical analysis (Williamson, 1983; Basu and Das Gupta, 2001) and models of rational utility maximising agents (Ben Porath and Welch, 1976; Leung, 1991; Ahn, 1995). While most studies concentrate on son or daughter preference, the "same sex" strategy exploits the parental predilection for a mixed sex composition of children, which is revealed as the desire to have at least one son and one daughter – a well-known stylised fact in the demography literature for developed countries (Williamson, 1983). When such preferences are present, parents of two children of the same sex exhibit a higher probability of having another child to attain their desired composition.

The "same sex" strategy relies on these preferences, and can be justified as follows. The sex composition of children, a naturally occurring random event,¹⁹ affects fertility through parental sex preferences, as required by Assumptions 5.3 and 5.4. Since the sex mix is random, making the identifying assumption that it affects labour supply only through its effect on the number of children satisfies the requirements of Assumptions 5.1 and 5.2. In these circumstances, the sex mix constitutes an instrumental variable for fertility which can be used to identify causal effects on women's labour supply through models like Equations 5.6 and 5.7. Because of its reliance in the sex mix of children, the "same sex" strategy studies women with at least two children.

Angrist and Evans (1998) analyze the case of the United States in detail, and discuss the plausibility of the exogeneity of sex preferences (and their effects) in labour supply decisions in their study. They estimate their model for the United States' 1980 and 1990 Censuses. For both years, they report strong first-stage results of the effects of sex mix on fertility, and compelling evidence of a negative causal effect of fertility on women's labour supply in the second stage.

Besides this original application, the "same sex" estimation strategy has been used in other contexts. Carrasco (2001) uses the Panel Survey of Income Dynamics from the United States, adapting the instrumental variables estimators for the longitudinal structure of the data. Her results are consistent with previous estimates from Census data. In an application to the United Kingdom, Iacovou (2001) employs the "same sex" strategy with data from the British Household Panel Survey and the National Child Development Study. While she still finds negative effects of fertility on women's

¹⁹Sex screening techniques might affect the randomness of the gender composition because of selective abortion. While these techniques have made substantial progress in the last decade, they were not widely available in the United States in 1980 and 1990, the years Angrist and Evans (1998) use in their estimation. Section 7.2.2 discusses the case of Argentina.

labour supply, the instrumental variables coefficients are not very precisely estimated, which might be attributed to the relatively small sample size of her datasets.

In developing countries, Chun and Oh (2002) applied the "same sex" strategy to household survey data from Korea (the National Survey of Family Income and Expenditure), finding significant first- and second-stage effects roughly consistent with the Angrist and Evans (1998) results for the United States. Rosenzweig and Wolpin (2000) find similar results with a dataset of rural households in India. Finally, Cruces and Galiani (2003) present additional data and estimates for Mexico, which show that the significant casual effect of fertility on female labour supply accounts for most of the increase in female labour force participation in the 1970-2000 period in that country.

While the fertility effects of the sex mix of children can be established from the data, identifying assumptions are not universally valid and must be evaluated on a case by case basis. After the description of the data in the following Section, Chapter 6 studies the plausibility of the strategy's premises for Argentina.

5.4 DATA AND SUMMARY STATISTICS

5.4.1 Description of the main dataset

The data used in this Chapter is from Argentina's 1991 Census of Population and Housing ("Censo Nacional de Población y Viviendas") conducted by the Instituto Nacional de Estadísticas y Censos (INDEC).

This Census is divided into two sections. A basic questionnaire was administered to the whole population (32,245,467 individuals and 8,927,289 households), while an extended questionnaire covered a large random sample representative of the whole country. The analysis in this Chapter is based on the dataset gathered by the latter questionnaire, since it contains detailed demographic and labour force participation data. The sample is composed of a total of 16,023,180 individuals and 4,287,580 households.

The study of fertility and labour supply needed some adjustments to the data. Women with two children were selected from the total sample, and their characteristics were linked to those of their children and spouses (if
present). As in Angrist and Evans (1998), the sample was limited to women between 21 and 35 years old whose oldest child was at most 18 years old at the time of the Census.²⁰

Since the relationship variable in the Census dataset indicates kinship with respect to the head of the household, it was only possible to match children with women who were heads of households or spouses of the head. The observations for women spouses of the head were checked so that the number of own surviving children (as asked for in a specific question) was the same as the number of matched children in the household.²¹ This ensured that children of a male head of household were not wrongly matched to a spouse who was not their mother. As shown in Appendix D, the empirical results of this Part are robust to all these adjustments.

Finally, in order to establish the generality of the results, the analysis was carried out on a series of variations of the total sample described in the previous paragraphs. The full sample consists of all women aged 21-35 with at least two children and whose oldest child was 18 or less (*Complete Sample*), whereas the *Married* sample corresponds to the subgroup of legally married women living with their spouses. Since Angrist and Evans (1998) exclude women whose second child was less than a year old from their estimations, this further adjustment was made to the two previous samples to obtain comparable datasets, resulting in the *AE* and *AE Married* categories. All variables, parameters and estimates of interest are reported separately for these four subgroups. The corresponding sample sizes are reported in the last row of Table $5.1.^{22}$

²⁰There were a total of 766,572 women between 21 and 35 with at least two children whose oldest child was 18 or less. The observations for which the age at first birth was less than 14 were discarded, taking this as an indicator of data entry errors or misallocated children, since most of the ages were far too low (8,745 women were discarded in total, of which 2,952 had imputed ages at first birth of 11 or less). A small fraction (5,718 observations) of married women for which the husband's age at first birth was less than 14 was also discarded.

²¹This is by far the most important adjustment of the sample. A total of 106,001 women were discarded (13.83 percent of the original 766,572), corresponding to those whose declared number of surviving children did not correspond to the number of children matched in the household.

²²These large sample sizes imply that the standard errors of the regression coefficients presented in Chapter 7 are virtually not affected by the use of Huber-White robust standard errors.

CHAPTER 5. ESTIMATION STRATEGY AND DATA

5.4.2 Variables and descriptive statistics

The outcome of interest for the estimations is the labour supply of women. In the Argentine Census data, individuals working for pay include employees (wage earners), the self-employed, owner-managers and civil and domestic servants, and exclude family workers without remuneration. The *Worked for pay* indicator is equal to 1 for this group, and 0 otherwise. Since income and hours of work were not collected in the Argentine Census, the analysis covers this employment status indicator only.

The fertility characteristics refer to the number and gender composition of children. The variable *Number of children* represents own children in the household, according to the allocation rules described above. The *More than two children* indicator is derived from this variable, and is defined as 1 for women with three or more children, and 0 otherwise. The *Same sex*, *Two boys* and *Two girls* variables refer to the sex of the first two children, and are equal to 1 if the first two were of the same sex, two males or two females respectively, and 0 otherwise in all cases.

Regarding individual characteristics, the *Age* and *Age at first birth* variables, measured in years, are self explanatory. The education indicators were constructed by dividing the sample into three homogeneous groups, composed of those with up to some primary schooling, those with some high school, and those with some higher education. Finally, all variables are defined analogously when referring to the spouses of married women.

As stated in the Introduction to this Chapter, demographic indicators in Argentina in 1991 were at a level consistent with an advanced fertility transition: the average number of children ever born for women in the 19-49 age group was 2.04 (and 2.83 for women with at least one child), with a mode of 2: these figures are in the 2-3 range, where the "same sex" strategy applies. Finally, for this age group, the labour force participation rose almost continuously during the twentieth century, reaching 35.4 percent in 1991 – Figure 5.1 presents these trends for women aged 14 and older.

These patterns are reflected in Table 5.1, which presents a set of summary statistics of the main variables used in the analysis for the four samples defined in Section 5.4.1 (*Complete Sample, Married, AE Sample, AE Married*), corresponding to women aged 21-35 with at least two children.

	Complete		A F	AE
Characteristics:	Complete	Married		Sample,
	Sample		Sample	married
Main variables:				
Worked for pay	0.314	0.305	0.315	0.305
(=1 if worked for pay, 0 otherwise)	(0.464)	(0.460)	(0.465)	(0.460)
More than 2 children	0.548	0.528	0.596	0.574
(=1 if more than two children, 0 otherwise)	(0.498)	(0.499)	(0.491)	(0.495)
Number of children	2.976	2.905	3.062	2.985
	(1.223)	(1.165)	(1.240)	(1.183)
Basic demographics:				
Age	29.410	29.679	29.660	29.928
C	(3.868)	(3.755)	(3.770)	(3.652)
Age at first birth	20.830	21.127	20.641	20.932
-	(3.434)	(3.436)	(3.337)	(3.340)
Fertility and sex mix:				
Same Sex	0.507	0.506	0.506	0.505
(=1 if first two children of same sex, 0 otherwise)	(0.500)	(0.500)	(0.500)	(0.500)
Two Boys	0.260	0.261	0.260	0.261
(=1 if first two children were boys, 0 otherwise)	(0.438)	(0.439)	(0.438)	(0.439)
Two Girls	0.246	0.245	0.246	0.244
(=1 if first two children were girls, 0 otherwise)	(0.431)	(0.430)	(0.431)	(0.430)
Boy 1st	0.508	0.510	0.508	0.510
(=1 if first child was a boy, 0 otherwise)	(0.500)	(0.500)	(0.500)	(0.500)
Boy 2nd	0.506	0.507	0.506	0.507
(=1 if second child was a boy, 0 otherwise)	(0.500)	(0.500)	(0.500)	(0.500)
Indicators of maximum education level:				
Some primary education	0.581	0.531	0.593	0.544
	(0.493)	(0.499)	(0.491)	(0.493)
Some secondary education	0.298	0.325	0.293	0.321
-	(0.457)	(0.468)	(0.455)	(0.457)
Some tertiary eduaction	0.121	0.144	0.114	0.135
	(0.326)	(0.351)	(0.318)	(0.326)
Observations	653,213	497,194	599,941	456,437

Table 5.1: Descriptive Statistics of Variables of Interest

Note: Means and standard deviations (in parentheses). The samples correspond to women aged 21-35 with two or more children aged 18 or younger from the extended questionnaire sample of the 1991 Census, as described in the text.

The labour force participation, as captured by the *Worked for pay* indicator, is in the 30.5-31.5 percent range, depending on the sample, with the lower rates corresponding to married women. The average fertility is substantially higher than for the 19-49 age group discussed above, because Table 5.1 refers to women with at least two children. The average number of children varies between 2.91 and 3.06, and it is slightly lower for married women. More than half of the women have more than two children, but unlike the case of the employment and number of children variables, there is a substantial variation between the four samples. This is due to the adjustment made to obtain the *AE* samples, which results in an average of 57.4-59.6 percent (*AE Sample* and *AE Married*, respectively) compared to a lower 52.8-54.8 percent for the *Complete* and *Married* samples.

The average age of the women in these samples is between 29 and 30 years old, and slightly higher for married women, while the age at first birth is around 21 years old for all samples. Regarding the sex composition of children, the proportion of males in first births is around 51 percent, and just above 50 percent of the women have two siblings of the same sex, with a slightly higher proportion of these cases corresponding to two boys (around 26 percent) than two girls (around 24 percent). These figures correspond to the stylised facts of the demography literature (Williamson, 1983).

Finally, the education level is relatively high in the four samples with respect to other Latin American countries, with more than 40 percent of the women with at least some secondary education. A noticeable difference between the samples is that married women have higher levels of secondary and further education than the corresponding overall samples.

5.5 CONCLUSION

This Chapter has discussed the theoretical and empirical problems for establishing a causal relationship between fertility and female labour supply. The discussion was carried out in terms of the potential outcomes framework, and it proposed instrumental variable techniques a possible solution to the endogeneity problems that invalidate simple estimation approaches.

The "same sex" strategy exploits the fertility effects of parental sex preferences as a natural experiment. While its necessary conditions were met in the original application to the United States (Angrist and Evans, 1998), the challenge for its application to another context is to show that the identification strategy is still valid. The data in Section 5.4 implies that Argentina in 1991 was substantially different from the United States in the two key variables: the average number of children ever born for women in the 19-49 age group was 2.04 (and 2.83 for women with at least one child), compared to 1.44 (and 2.23, respectively) for the equivalent United States sample of 1990, while the labour force participation for the same group was 52.8 percent in Argentina compared to 80.9 percent in the United States.²³

The next Chapter discusses the conditions for application of the "same sex" strategy and the problems of this identification strategy in the context of a developing country, where son preferences, among other institutional factors, might affect the plausibility of the identifying assumptions. Their validity is critically assessed in the Argentine case, using the data described in this Chapter and other complementary sources.

²³These figures were computed from Argentina's 1991 Census (as described in Section 5.4) and from the one percent Public Use Micro Sample from the United States 1990 Census, provided by Sobek et al. (2003).

CHAPTER 6

SEX PREFERENCES AND FERTILITY IN ARGENTINA

6.1 INTRODUCTION

The previous Chapter motivated the establishment of a link between fertility and female labour supply by noting that in Part I both variables were consistently associated with a household's poverty status in Argentina. While causal links are difficult to establish in applied empirical research, Chapter 5 discussed an instrumental variable estimation strategy for models of fertility and female labour supply based on the fertility effects of parental sex preferences.

This identification strategy was originally devised for the United States in the late twentieth century, and its justification is a priori valid only for that case. The analysis in this Chapter deals with the threats to the validity of the approach in its application to the Argentine census data of 1991, presenting evidence on the fertility effects of sex preferences and on their exogeneity with respect to labour market outcomes.

The Chapter is organised as follows. Section 6.2 discusses the threats to the validity of the strategy for developing countries. Section 6.3 presents evidence against these potential challenges, studying the nature and extent of sex preferences in Argentina, their fertility effects, and their institutional background. A brief conclusion follows.

6.2 VALIDITY OF THE IDENTIFICATION STRATEGY IN DEVELOPING COUNTRIES

6.2.1 First stages: nature of sex preferences and the same-sex effect

The "same sex" strategy is based on context-dependent evidence and assumptions that need to be considered on a case by case basis. This is especially relevant when attempting to apply the strategy to developing countries like Argentina, where the socioeconomic and cultural context differs substantially from the United States, where the original study was carried out.

The precondition for the application of the "same sex" strategy is the existence of a first-stage relationship between the sex mix of children and further childbearing in the context under study. A correlation between the treatment D and the instrument Z, as required by Assumption 5.3 (page 134), is easy to establish: it only requires to check if the sex mix of children has a significant effect on further childbearing. Angrist and Evans (1998) study the impact of a "same sex" indicator on further childbearing for women with at least two children, and Section 6.3.1 presents a detailed analysis of sex mix and fertility in Argentina by studying the probability of further childbearing conditional on the sex of previous children.

The basis of the "same sex" strategy is the fact that parents tend to have a preference for a balanced sibling sex composition, which implies that those with two children of the same sex are more likely to have another child than parents of a boy and a girl. These type of sex preferences, observed by Angrist and Evans (1998) in their United States data, are not uncommon. Based on multiple sources of attitudinal evidence on fertility, Basu and Das Gupta (2001) state that "in most regions of the world, parents express a preference for a gender-balanced family," usually mixed with "mild preferences for children of a particular gender." While stated preferences might not affect fertility behaviour in a significant manner, Williamson (1983) and Basu and Das Gupta (2001) report that there is compelling evidence that sex preferences result in differential fertility effects, either as a consequence of a gender-balanced preference or as a result of a preference for sons.

The evidence for developed countries is consistent with gender-balanced preferences: results from attitudinal surveys are confirmed by higher fertil-

ity for parents of children of the same sex. This is complemented with a minor son bias in some circumstances, which is characterised as a higher probability of further childbearing for parents of girls than for parents of boys. The United States provide a typical example for industrialised societies: Angrist and Evans (1998) report that for the 1980 and 1990 Census, parents of two children of the same sex had a significantly higher probability of having a third child than parents of a boy and a girl, and that mothers of two girls were slightly more likely to have a third child than were mothers of two boys. This is consistent with previous evidence from the United States (Ben Porath and Welch, 1976; Williamson, 1983). Iacovou (2001) reports similar results from two British datasets: in samples from the British Household Panel Survey, almost half of the mothers with two sons or two daughters had another child (with a difference between the two of only half a percentage point), compared to only 41.2 percent of mothers of a boy and a girl. These fertility patterns correspond to the effects of the conventional sex preferences observed by Basu and Das Gupta (2001).

In developing countries, sex preferences and their fertility effects are more heterogeneous, and can differ from the patterns of the previous paragraph. In general, the preference for a gender-balanced family coexists with stronger son preferences. For instance, Das (1987) finds for India that "at each parity a higher proportion of couples with no sons went on to have the next child than did those who already had one or more sons," but he also observes a positive effect on fertility when all living children are sons, "indicating that preference for sons is not to the exclusion of daughters." In a review of attitudinal evidence, family and population structures in Bangladesh, India and Pakistan, Nag (1991) concludes that the prevalent form is a "general preference of sons over daughters as well as a desire for at least one daughter."

In other developing countries, however, parental preferences indicate only a strong preference for sons. This implies that couples are more likely to stop having children when having at least one boy, and that parents of a certain number of girls are much more likely to have an additional child than parents of the same number of boys. Moreover, as reported by Das Gupta (2003) for China and Korea, extreme forms of son preference also result in abnormally high ratios of boys to girls in the population as a consequence of selective abortion, higher infant mortality for daughters, and even infanticide. In these cases, the sex composition of children cannot be considered to be purely random, and thus the independence Assumption 5.1 is not likely to hold.

In general, the fertility effects of sex preferences can be divided into three categories: gender-balanced, gender-balanced with moderate son preference, or strongly biased towards sons. It should be noted, however, that the presence of sex preferences other than the first case is not necessarily an impediment for the application of the "same sex" strategy in other contexts. If the gender composition of children can be considered as a random event and the sex preferences have some form of effect on childbearing, then the sex mix of children can still be exploited as a source of variation in fertility (i.e., as an instrument *Z* in Equations 5.6 or 5.7, pages 136 and 137).¹ However, this is only true if the exclusion and monotonicity Assumptions hold: a first-stage relationship is a necessary condition, but the crucial point is whether the variation in fertility induced by sex preferences can be considered to be exogenous. This is discussed in the following pages for developing countries.

6.2.2 Sex preferences and the identifying assumption

Establishing the case for the randomness of the instrument (Assumption 5.1, page 134) is relatively straightforward. The gender of a child is a naturally occurring random event, and the sex mix is thus "as good as randomly assigned" (Angrist, 2001). As discussed in the previous Chapter, a problem arises with extreme forms of son preference that lead to the neglect of daughters in basic healthcare, or when sex screening techniques are widely available and result in selective abortions. In those cases, the sex mix is manipulated and might be correlated with potential outcomes. The inspection of sex ratios by age, household consumption and school enrolment data provides evidence on extreme forms of son preferences that could cast doubts on the credibility of the independence assumption. This is accomplished in

¹Note, however, that different instruments imply different types of compliers and thus identify causal effects for different subgroups of the population. Section 8.2 (page 191) discusses whether differences in the "complier" subpopulations are relevant for the estimation results.

Section 6.3 for Argentina.

The independence of the sex mix with respect to potential outcomes cannot be established directly, but while the assignment mechanism itself is not testable, its implications are. A simple test is to compare characteristics of parents of same-sex and mixed sex siblings – if truly random, there should not be systematic differences in exogenous attributes between the two groups.

Identification is not attained solely through first stages and independence: the basis of instrumental variable estimation is that the instrument Z, while correlated with the treatment D, does not have a direct effect on the outcome Y and is thus correctly excluded from the second stage. However, as an exclusion restriction, Assumption 5.2 (page 134) is inherently non-testable and its plausibility has to be evaluated on a case by case basis. In the context of the "same sex" strategy in the United States, Angrist and Evans (1998) show that there is almost no association between the sex of children and labour supply. Moreover, they review the evidence on norms and behaviour and the institutional setting, concluding that the sex mix only affects labour supply through its influence on fertility. Finally, they argue that the inclusion of the sex of the first two children among the covariates X in the estimation of Equation 5.7 controls for any secular effects of these variables.

While this may or may not be the case in other developed countries, the presence of moderate and extreme son preferences raises additional concerns in developing countries. If the bias towards sons is a pure preference with no other behavioural consequence than its impact on fertility, it constitutes an idiosyncratic characteristic of the population under study and the plausibility of the exclusion restriction is not affected.² The "same sex" strategy can still be applied, accommodating for the preference for sons in the first stage by including in *X* the sex of the first two children.

However, the identification strategy may be compromised if the preference for sons has socioeconomic roots and consequences that are conceivably related to the potential outcomes, in violation of Assumption 5.2,

²In terms of the theoretical model of Secion 5.2.1 (page 126), this holds under the condition of separability of leisure and the utility derived from children and their gender (Rosenzweig and Wolpin, 2000).

which is more likely to happen in developing countries. Basu and Das Gupta (2001) argue that beyond cultural and religious factors, some societies exhibit a strong son preference because of the gap between sons' and daughters' "ability to contribute to the physical, emotional and financial well-being of their parental household." This gap is determined by kinship systems: if women's links with their parents are cut off after marriage, it becomes more attractive to rear sons that will be able to provide for their parents in old age.

This observation has important implications for the "same sex" strategy. For instance, it is possible to imagine a situation where having two daughters is a random event (Assumption 5.1) that induces a higher probability of further childbearing through a combination of son and gender-balanced preferences (Assumption 5.3). If in this hypothetical setting there is no social insurance and women have low employment rates, parents might rely on their male children for old age support, and this might be at the origin of the preference for sons. Thus parents of two girls are relatively worse off, and will have to work more in order to compensate for the foregone income. In terms of the theoretical model of Section 5.2.1 (page 126), taking *I* as the permanent income of the household, using sons for old age support implies that the instrument *Z* (the sex mix of children) cannot identify the direct effect of fertility on labour supply, f_D , because of the effect of the sex mix of children on *I*. The derivative of *Y* (labour supply) with respect to *Z* is given by:

$$\frac{\partial Y}{\partial Z} = \frac{\partial I}{\partial Z} f_I + f_D$$

and the use of sons for old age support implies that

$$\frac{\partial I}{\partial Z} \neq 0$$

which in turn implies that *Z* has a direct effect on labour supply, over and above its effect on fertility. In terms of potential outcomes, irrespective of the treatment *D*, the sex mix of children implies that $Y_{1d} > Y_{0d}$, that is, those who had two girls will work more than others. This is a direct effect of *Z* on potential outcomes over and above its influence on further childbearing *D*, and as such, it invalidates Assumption 5.2. In terms of the canonical

instrumental variables model, the preference for sons might induce a direct correlation of the sex mix with the labour supply or the unobservable factors that affect it, and in this case it is incorrectly excluded from the second stage.

There are other plausible problems for identification in contexts with non-extreme son preferences.³ In some societies, parents are expected to pay substantial dowries for their daughters, which affects their permanent income and, potentially, their labour supply. These effects on potential outcomes are not verifiable: it is not possible to determine if the sex mix of children or related factors, like dowries, have a direct effect on labour supply. However, an examination of the social and institutional setting of each application might reveal or rule out the presence of problems of this type. Section 6.3 conducts this analysis for Argentina.

6.2.3 Direct effects of same sex

The identification problems discussed above arise from the presence of son preferences. However, direct effects of the sex mix of children, unrelated to parental sex preferences, might also invalidate the application of the "same sex" strategy. As discussed above, Angrist and Evans (1998) rule out the possibility that the secular effects of the sex of children contaminate their instrumental variables estimates for the United States. Rosenzweig and Wolpin (2000) attempt to question this aspect of the estimation strategy using data from rural India. They show that having two children of the same sex, which is likely to be random in their data, implies a higher probability of further childbearing, establishing the presence of a "same sex" first-stage relationship.

Rosenzweig and Wolpin's (2000) caveats are based on the expenditure data for the same households. By studying outlays on child-related goods, they find that same-sex siblings are associated with lower levels of expenditure. They attribute this effect to "hand-me-down" savings, which are more likely to arise for items such as clothing and footwear when there are children of the same sex in the household. Based on this effect of sex mix on expenditure, Rosenzweig and Wolpin (2000) presume that "the sex compo-

³As discussed above, extreme forms of son preferences that result in selective abortions or neglect of girls invalidate the randomness of the instrument, in addition to their effects on the exclusion restriction.

sition of children plausibly alters labour supply through mechanisms other than through fertility change alone." According to this interpretation, parents of same-sex children in their Indian data may need to work less because of these savings, implying in terms of potential outcomes that $Y_{0d} > Y_{1d}$. While the authors provide evidence of a link of sex mix and expenditure, they do not establish a direct effect of the instrument *Z* on labour supply. Moreover, the critique only applies to their Indian dataset: they state that "it is not possible to infer from this evidence whether hand-me-down economies associated with the sex-mix of births are an important phenomenon" in other situations. However, these valid concerns can be addressed by testing the presence of savings of this type with expenditure data, and this is done for Argentina in Section 6.3.

Despite the context-dependent nature of the identifying assumptions, most of the applications of the "same sex" strategy mentioned in Section 5.3.3 (page 140) are justified on the basis of a first-stage relationship, without studying the potential impediments to the exclusion restriction and the other identifying assumptions. The exceptions are Rosenzweig and Wolpin (2000), although they only deal with the concern raised in the previous paragraph, and Chun and Oh (2002), who briefly discuss the strong son preference observed in their Korean data. The originality of this Chapter lies in the detailed analysis of different data sources to assess the plausibility of the "same sex" identifying assumptions in the Argentine case.

6.3 SEX PREFERENCES AND IDENTIFICATION ASSUMPTIONS IN ARGENTINA

6.3.1 Independence of the instrument and fertility effects of sex preferences

The purpose of this Section is to assess the plausibility of the identifying assumptions of the "same sex" strategy for Argentina, addressing the issues raised in the previous Section.

Assumption 5.1 requires the independence of the instrument with respect to the potential outcomes and treatments. While potential outcomes and treatments, by definition, cannot be observed, the randomness of the instrument can still be established by some of the implications of the independence assumption. One simple check is to compare the characteristics of

	Complete			AE	
Characteristics:	Complete	Married	AE Sample	Sample,	
	Sample			married	
Age	-0.0285	-0.0234	-0.0213	-0.0195	
	(0.0096)***	(0.0107)**	(0.0097)**	(0.0096)***	
Age at first birth	0.0090	0.0134	0.0057	0.0081	
	(0.0085)	(0.0097)	(0.0086)	(0.0085)	
Residence in rural area	-0.0015	-0.0016	-0.0011	-0.0014	
	(0.0012)	(0.0014)	(0.0013)	(0.0012)	
Indicators of maximum					
education level:					
Some primary education	-0.0002	-0.0015	-0.0002	-0.0014	
	(0.0012)	(0.0014)	(0.0013)	(0.0012)	
Some secondary education	0.0003	0.0012	0.0001	0.0009	
	(0.0011)	(0.0013)	(0.0012)	(0.0011)	
Some tertiary eduaction	-0.0001	0.0003	0.0001	0.0005	
	(0.0008)	(0.0010)	(0.0008)	(0.0008)	
Observations	653,213	497,194	599,941	456,437	

Table 6.1: Differences in Selected Characteristics by "Same Sex" Indicator

Note: Differences in means (mean of "same sex" mothers minus mean of mixed sex mothers) and their standard deviations (in parentheses). * significant at 10%; ** significant at 5%; *** significant at 1%. The samples correspond to women aged 21-35 with two or more children aged 18 or younger from the extended questionnaire sample of the 1991 Census, as described in the text.

women with same-sex and mixed-sex sibling compositions, i.e., according to the value of the instrument *Z*.

Table 6.1 presents the differences in selected variables between these two groups for the four samples of women with at least two children defined in Section 5.4 (page 142). With the exception of the *Age* variable, none of the differences is significant, not even at the 10 percent level, despite the extremely large sample sizes. Women whose first two children were of the same sex and mothers of mixed-sex children exhibit no differences in their ages at first birth, area of residence and education levels.

The only variable for which there is a significant difference is *Age*: the Table indicates that "same sex" mothers are about 0.02-0.03 years younger than mothers of mixed-sex siblings. While statistically significant, this effect is small, and it can be attributed to the high precision of the estimates for large sample sizes, since there does not seem to be a difference in the closely

related *Age at firth birth* variable. Moreover, women in the groups defined by the *Same sex* indicator cannot be distinguished in terms of their education level, which is an important determinant of earnings and labour supply. In any case, two-stage least squares models like Equation 5.7 can accommodate covariates like the *Age* variable to control for any effects that this might have on the outcome of interest.

While the evidence from Table 6.1 supports the hypothesis of random assignment of the *Same sex* instrument *Z*, another prerequisite of the estimation strategy is the existence of a first-stage relationship, embodied in Assumption 5.3. The analysis below serves two purposes: it first establishes the existence of a first-stage relationship between sex preferences and fertility in Argentina, and it also provides a detailed study of the nature of these preferences in the data.

A first-stage relationship requires a significant correlation between the instrument *Z* and the indicator *D*, but as discussed in the previous Section, the correlation between childbearing and an indicator of having two children of the same sex may be due to different types of preferences. The mixed-sex sibling preference consists in a desire of having at least one child of each sex, which induces a higher probability of having more children (D = 1) in couples whose children are all boys or all girls (Z = 1). The preference for sons, in turn, implies that parents of a certain number of girls are more likely to have an additional child than parents of the same number of boys. To discriminate between these two types of preferences, it is necessary to disaggregate the effect of the *Same sex* variable.

In the demography literature, sex preferences and their effect on fertility are discussed in terms of "parity progression ratios." The parity is defined as the number of children, while the progression ratio refers to the probability of having n + 1 children, given n children. Much of this literature is devoted to the presence of son preference and their effects, implicitly assuming that a positive correlation between the number of daughters and subsequent childbearing (or a negative correlation with the number of sons) is an unambiguous sign of son preference. Leung (1991), however, develops a dynamic model of sex preferences and fertility decisions to show that such tests of sex preferences are only valid under certain conditions. He derives the optimal fertility behaviour for a couple as a value function defined over the effective number of children given by:

$$a_t B_t + b_t G_t$$

where B_t represents the number of sons, G_t the number of daughters, and a_t and b_t the relative weight attached to children of each sex. Values of a_t or b_t greater (smaller) than zero indicate a positive (negative) valuation of children of the respective sex. Son preference, no sex preference, and daughter preference are defined as cases where $a_t > b_t$, $a_t = b_t$ and $a_t < b_t$ respectively.

Leung (1991) questions the commonly held view according to which a negative effect of the number of sons on subsequent fertility is an unambiguous sign of boy preference. The main conclusion from the model is that a negative relationship between the probability of further births and the number of boys, holding parity constant, can be explained by either son or daughter preference: the only definite conclusion is that sex preferences of some sort are present since the gender of children affects the probability of birth. For this negative correlation to be consistent with daughter preference, however, Leung (1991) shows that parents must value boys negatively at every single parity.⁴ In what follows, it is shown that this is unlikely to be the case in the Argentine data, so that this mild restriction allows the discussion of son and daughter preference in a way consistent with the previous literature.

Table 6.2 presents the progression ratios by parity and sex composition of children for the sample of married women – for completeness, the Table also includes those who had only one child. These results are similar for the other three samples previously defined.

The first row of the Table presents the results for parities one to two. More than 77 percent of the women went on to have another child, but those who had a girl in their first birth have a small but significantly higher probability of having a second child, while those who had a boy are less likely to have another child. It should be noted that, while significant, these effects are extremely small: the difference in the probability of further childbearing

⁴He shows that a negative effect of the number of sons on subsequent fertility can be observationally equivalent to cases of daughter preference, for instance if $a_t < 0$ and $a_t < b_t < 0$, but he notes that this requires $a_t < 0$ (a dislike of sons) at each parity.

		Fraction that had another child								
Douitre	Oha	Overall	Sama any Mi	Miradaar	PPR by number of boys:					
Farity	005.	PPR	Same sex	Mixeu sex	0	1	2	3	4	
1 to 2	644,084	0.7719	-	-	0.7737**	0.7703**				
2 to 3	497,194	0.5280	0.5460***	0.5096***	0.5568***	0.5096***	0.5359***			
3 to 4	262,525	0.4241	0.4422***	0.4176***	0.4499***	0.4229	0.4124***	0.4348***		
4 to 5	111,341	0.3941	0.4050***	0.3922	0.4031	0.3961	0.3909	0.3903	0.4070**	

Table 6.2: Parity Progression Ratios by Parity and Sex Mix of Children, Married Women

Note: *, ** and *** indicate that the number in the cell is different to the "Overall" category at the 10, 5 and 1% levels of significance respectively. The samples correspond to married women aged 21-35 with two or more children aged 18 or younger from the extended questionnaire sample of the 1991 Census, as described in the text (*Complete sample, married*).

between the two groups is only 0.34 percentage points. Although this is a sign of son preference, the size of the differences for higher parities (discussed below) indicates that this difference is not substantial and that its significance may be due to the precision granted by the large sample size.

The second row of Table 6.2 presents the transition probability between parities two and three, which is at the heart of the "same sex" strategy since it represents the first-stage relationship between the instrument Z (Same sex) and the treatment D (More than two children and Number of children). In the sample of married women with two children, 52.8 percent went on to have a third child, but as pointed by the two following columns, this group represented a larger fraction among women with two boys or two girls (54.6 percent) than among those with a son and a daughter (50.1 percent). This sizeable and significant difference of 4.5 percentage points indicates the presence of some form of sex preferences, which can be analysed with the help of the following columns. Mothers of two boys and of two girls both exhibit a significantly higher probability of further childbearing, but this probability is almost 2.1 percentage points higher for mothers of girls. The fact that mothers of two boys have a higher probability of having another child than mothers of a boy and a girl (53.6 percent compared to 50.1) suggests the presence of a strong gender-balanced preference with a moderate bias towards boys.

This conclusion is supported by the analysis of further parity progres-

sion ratios. The difference between same-sex and mixed-sex parents is only about 2.5 percentage points in the transition probabilities between three to four children, compared to the 4.5 for two to three, but the pattern is similar when observing the effect of boys and girls. This same-sex effect is still higher for parents of two girls than for parents of two boys.

In the transition from four to five children, the same-sex effect is still present, although the difference in the probability of further childbearing between parents of same-sex and mixed-sex children is reduced to 1.3 percentage points. An interesting point, however, is that at this parity the effect is driven by a preference for daughters, since the probability of further childbearing for those who have had no sons is not significantly different from the overall probability.⁵

Finally, it should be noted that the number of boys never has a negative effect on the probability of further childbearing, which is not consistent with a negative net value of sons at every parity. This addresses Leung's (1991) concern of observational equivalence between son and daughter preferences.

The results in Table 6.2 refer to the *Married* sample, but the magnitude and significance of the parity progression ratios are similar for the remaining three samples. However, it is possible that different groups in the population defined by some characteristics exhibit different types of sex preferences, which could bias the results with the pooled datasets if these characteristics are systematically related to labour market outcomes. While not presenting the full parity progression ratio analysis, Section 7.5 (page 187) discusses the first-stage relationship between the instruments *Same sex*, *Two boys* and *Two girls* and the treatments *More than two children* and *Number of children* for the groups defined by the women's education levels of Table 5.1 (page 145). This analysis indicates that the sex preferences identified in this Section are homogeneous across these three groups and among rural and urban populations, with only minor differences in the probabilities of

⁵A corollary of this analysis brings support to Assumption 5.4 (page 134). While monotonicity is not directly verifiable, because it is defined in terms of potential outcomes, Angrist and Imbens (1995) note that the assumption has a testable implication in the case of multivaried treatments such as the number of children: if the assumption holds, the cumulative density functions of *D* with Z = 0 and Z = 1 should not cross. Inspection of Table 6.2 indicates that this is the case for the Argentine data.

further childbearing.

This evidence implies that the first-stage Assumption 5.3 is valid, in the sense that the sex mix of children has an effect on fertility in Argentina. Moreover, sex preferences and their effect on fertility correspond to Section's 6.2.1 gender-balanced with moderate son preference category, and they are consistent with Basu and Das Gupta's (2001) widespread "preference for a gender-balanced family" with "mild preferences for children of a particular gender."

Given the presence of son preferences and the concerns raised in the previous Section about this phenomenon for the validity of the "same sex" strategy in developing countries, the following pages study the possibility of contamination of instrumental variable estimations by describing the institutional setting and presenting evidence on the exogeneity of the sex-mix for female labour supply decisions in Argentina.

6.3.2 Institutional setting and effects of son preferences

The analysis of the first-stage relationship revealed that the preference for a gender-balanced family is more intense for parents of girls, which was interpreted as a moderate bias for sons in Argentina. As discussed in Section 6.2, this might raise some concerns about the exclusion restriction in Assumption 5.2 in the context of a developing country: if son preferences are due to – or result in – economic factors that are correlated with the labour supply of the parents, the instrument *Z* has a direct effect on the outcome *Y*. In that case, instrumental variable results are biased and do not identify the causal effect in Proposition 5.1 (page 135). Since potential outcomes are not observable, exclusion restrictions are inherently non-testable, and their plausibility must be evaluated on a case by case basis by means of indirect evidence – the following Chapter presents direct evidence by means of Hausman-type overidentification tests.

As described in Section 6.2.2, Basu and Das Gupta (2001) attribute the extreme preference for sons in China and other developing countries to the gap between sons' and daughters' ability to contribute to the well-being of their parental household, which is determined by kinship systems. However, when reviewing the evidence for Latin America, they observe that "childbearing outside formal marriage or stable unions is common," a pattern that they find inconsistent with economic dependence of women on men. They conclude that despite a tradition of "male machismo [..] there are few rigid rules constraining women from participating actively in social and economic life, or from helping their parents" in the region.

This is consistent with the evidence from Argentina (INDEC, 2001). With respect to kinship systems, women stay in contact with their parents, and the lack of constraints pointed out by Basu and Das Gupta (2001) implies a relatively high labour force participation of women, as previously discussed. Moreover, dowries are unheard of, which implies that daughters are not necessarily more expensive to rear than sons, at least as far as marriage is concerned. Finally, the argument of children (and notably boys) as insurance for old age is weakened by the fact that since the middle of the twentieth century Argentina has had a fairly large social security system that supports retired workers but also destitute people in old age, regardless of their previous contributions,⁶ which was reflected in the results in Part I on the relative income stability in households with pensioners.

The institutional setting suggests that the roots of the moderate predilection for boys in Argentina is eminently cultural. As discussed in Section 6.2.2, however, even if due to purely cultural factors in their origin, son preference may have consequences that might call into question the validity of the "same sex" strategy in developing countries.

One of the most prominent manifestations of extreme son preferences is the phenomenon of "missing girls," attributed to discrimination in the form of gender-based stopping rules, selective abortion, neglect of daughters and even infanticide. The effect of stopping rules was considered in the discussion of Table 6.2, and selective abortion can be ruled out in 1991 Argentina since sex screening techniques were practically not available at the time of the Census.⁷

Figure 6.1 presents the infant sex ratios – representing the ratio of boys to girls aged zero to four years old – for selected countries around 1990. The Figure indicates that this ratio is slightly lower in Argentina than in

⁶See Arza (2004) for a thorough review of social security in Argentina.

⁷While screening techniques have made substantial progress in the last decade, access to them was limited (at best) in Argentina in 1991. Moreover, abortion was (and still is) illegal, further reducing the possibilities of selection of the sex of offspring. Das Gupta (2003) discusses selective abortion and their effects on gender mix in Korea.



Figure 6.1: Sex Ratios – Number of Boys / Number of Girls, 0 to 4 Years Old, Selected Countries, 1990

Source: Argentina and United States: author's calculations based on the respective censuses. China, Korea and India: Basu and Das Gupta (2003).

163

-	Bo	ys	Gi	rls	Ove	rall	Difference
Age	Population	Enrolment Rate	Population	Enrolment Rate	Population	Enrolment Rate	(percentage points)
6	319,150		311,510	96.51%	630,660	96.34%	-0.346
7	323,530	97.30%	314,370	96.95%	637,900	97.13%	0.348
8	332,694	98.01%	326,251	98.18%	658,945	98.10%	-0.173
9	337,167	98.08%	330,052	98.20%	667,219	98.14%	-0.127
10	336,681	97.80%	327,912	98.12%	664,593	97.96%	-0.319
11	345,115	97.54%	336,835	97.76%	681,950	97.65%	-0.221
12	340,912	96.11%	331,430	96.42%	672,342	96.26%	-0.307
Total	2,335,249	97.29%	2,278,360	97.46%	4,613,609	97.38%	-0.167
Note: the rat	Data from t e for girls.	he 1991 Ce	nsus. The dif	ference in a	ttendance is	the rate for	r boys minus

Table 6.3: School Enrolment, Children Aged 6 to 12

the United States, which is considered a case of very mild son preference (Williamson, 1983), and substantially lower than in extreme cases like China and Korea, identified as cases of extreme son preferences (Basu and Das Gupta, 2001).

The infant sex ratio in Argentina is consistent with the small but relatively higher proportion of boys in births, which is an almost universal feature of demographic data. Figure 6.1 can thus be taken as evidence of no discrimination against girls in the form of neglect or reduced healthcare that results in higher mortality among girls, and it shows the lack of an aggregate effect of the preference for sons in the Argentine data.

Another manifestation of extreme forms of son preference is the neglect of daughters in other aspects than nutrition and healthcare. In most countries where girls are systematically discriminated against, the school enrolment rates of younger females are significantly lower than that of their male counterparts. In Argentina, however, Pantelides (2002) reports that in 1991 women exhibited higher rates of course completion than men, which is consistent with the evidence in Table 6.3. This Table presents the school enrolment rates for children aged six to twelve years old from the 1991 Census. These rates are very close for boys and girls, but slightly higher for girls on average and for almost every age group. This pattern is not consistent with discrimination against girls in education.

As discussed in the previous Chapter, further evidence on the effect of sex preferences can be deducted from household's consumption patterns and the budget spent on goods for children of different sex.⁸ Table 6.4 presents evidence from the 1987 National Survey of Household Expenditure ("Encuesta Nacional de Gasto de los Hogares"–ENGH), carried out by INDEC.⁹ The sample is representative of the 1991 Census data (with the exception of isolated rural areas), and it was constructed following the criteria set out in Section 5.4 (page 142). It consists of 6,808 married women aged 18 to 45, with two or more children aged 18 or younger.

Table 6.4 presents data on differences in budget shares for a series of goods by sex composition of children. If girls were neglected or discriminated against, parents of boys would spend a higher proportion of their budget on food, health, clothing or education, among other goods. The first panel in the Table points out that parents of two children of the same sex, and parents of two boys and two girls have the same budget shares in seven categories ranging from marginal expenditures, like "Entertainment," to more substantial items like "Food and beverages," including also categories of children related goods. None of the differences for parents of two boys or two girls are different from zero at the normal levels of significance. Moreover, the last column presents the coefficient of a regression of the respective share on the ratio of boys to total children in the household, and none is significant either. Finally, the bottom panel of the Table presents the total and per capita measures of income and expenditure and their differences by the same variables: the results indicates that these aggregates are not affected by the sex mix of children in the household.

The consistent lack of effect of the sex composition of children on the budget shares of child-related goods can be taken as further evidence of no discrimination against girls and of the lack of direct effects of sex mix on relevant economic variables.

The evidence presented so far states that demographic and fertility patterns in Argentina are consistent with preferences for a gender-balanced

⁸Deaton (1997, Chapter 4) and Case and Deaton (2003) review these issues.

⁹Expenditure surveys are only carried out every ten years by INDEC. The 1987 survey is the closest to the 1991 Census year employed in the empirical analysis of fertility and labour supply.

• • • • • • • • • • • • • • • • • • •		First two children			Boys /
	Total	Same sex	Two boys	Two girls	Total
Budget shares of:					
Food and beverages	42.0%	0.0105	0.0117	0.0019	-0.0006
C C		(0.0140)	(0.0159)	(0.0164)	(0.0229)
Clothing and footwear	8.4%	0.0037	0.0030	0.0019	0.0030
-		(0.0022)*	(0.0025)	(0.0026)	(0.0037)
Clothing for children under 10	3.4%	0.0009	0.0003	0.0009	0.0002
		(0.0008)	(0.0009)	(0.0009)	(0.0013)
Health and related expenditures	4.6%	-0.0018	-0.0018	-0.0005	-0.0016
		(0.0021)	(0.0024)	(0.0025)	(0.0035)
Education, total	1.7%	-0.0031	-0.0017	-0.0025	0.0004
		(0.0022)	(0.0025)	(0.0026)	(0.0036)
School related materials	0.6%	-0.0002	-0.0007	0.0004	-0.0009
		(0.0005)	(0.0006)	(0.0006)	(0.0009)
Entertainment	6.1%	0.0002	0.0018	-0.0017	0.0043
		(0.0019)	(0.0022)	(0.0022)	(0.0031)
Income and expenditure measures:					
Total net income of the household	1085	9.0	27.3	-16.6	56.5
		(27.9)	(31.7)	(32.6)	(45.6)
Net income per capita	227	-1.8	4.1	-6.7	13.9
		(6.1)	(7.0)	(7.2)	(10.0)
Total expenditure	941	11.4	-2.7	18.4	-10.3
-		(20.1)	(22.9)	(23.5)	(33.0)
Per capita expenditure	196	-0.7	-0.9	0.1	1.4
		(4.5)	(5.1)	(5.3)	(7.4)

Table 6.4: Difference in Budget Shares, Income and Expenditure by Sex Composition of Children

Note: Differences in means (mean of the relevant group minus mean of the rest of the population) and their standard deviations (in parentheses), except for the last column which reports the regression coefficient of the ratio of boys (with the variable in column 1 as the dependent variable). * significant at 10%; ** significant at 5%; *** significant at 1%. The sample consists of 6,808 married women aged 18-45, with two or more children aged 18 or younger from the Encuesta Nacional de Gasto de los Hogares, INDEC, 1987. Income and expenditure in current Australes.

family with a moderate bias towards sons, and that these patterns are homogeneous across different groups of the population. Moreover, the institutional setting and the evidence presented in this Section supports the conclusion that these preferences are eminently cultural in their origins and limited in their effects. Data from sex ratios, school enrolment and household expenditure fail to reveal any significant form of discrimination against girls, supporting the hypothesis that the bias towards sons, observed in the parity progression ratios, is an idiosyncratic feature of the society under study which appears to be innocuous for the application of the "same sex" strategy in Argentina.

6.3.3 Identification concerns beyond sex preferences: direct effects of "same sex"

The indirect evidence presented above supports the hypothesis that son preferences do not invalidate the exclusion restriction in Assumption 5.2, since it seems unlikely that they induce a direct effect on the labour supply of women over and above their effect on fertility. It must still be established, however, that beyond the issue of son preference the sex mix of children (the instrument Z) does not have a direct effect on potential outcomes.

A possible threat to the validity of the "same sex" identification strategy is posed by Rosenzweig and Wolpin (2000), as discussed in Section 6.2.3. Studying outlays per children in rural India, they find that same sex siblings are related to substantially lower levels of expenditure on some childrelated goods. They attribute this effect to hand-me-down savings, which are more likely to arise for items such as clothing and footwear when there are children of the same sex in the household. Since these items represent a sizeable fraction of the household's expenditures, they note that the sex composition of children plausibly alters labour supply through mechanisms other than through fertility change alone.

Whether or not the effect identified by Rosenzweig and Wolpin's (2000) is present in other settings is a verifiable matter, and Table 6.4 presents evidence on the effects of sex mix on household consumption patterns in Argentina. Using expenditure data from the ENGH survey (described above), the Table reports the budget share of seven categories of goods (top panel) and the sample means of income and expenditure (bottom panel). It also presents the difference in those variables between the whole sample and

households with same sex children, two boys and two girls (columns 3, 4 and 5 respectively). In the last column, it reports the regression coefficient of the ratio of boys when the budget shares, income and expenditure variables are used as dependent variables.

The budget shares of clothing, footwear, and school-related goods, among other categories, are not significantly affected by the sex composition of children in the household. Only one of the figures in the Table is significant at the 10 percent level: households with two children of the same sex have a 0.3 percentage points higher budget share of clothing and footwear. Moreover, the sign of this difference contradicts the presence of hand-me-down savings, since those with same-sex children spend a significantly higher share of their budget in clothing and footwear. The evidence in Table 6.4 reflects the lack of hand-me-down savings or other effects of sex mix on the expenditure patterns of households in Argentina.

Finally, aggregate data from the ENGH suggests that hand-me-down effects not reflected in Table 6.4 are unlikely to have a noticeable effect on total expenditure. Argentine households in the full 1987 ENGH sample devoted 6.7 percent of their budget to the clothing and footwear (for all members), and only 2.8 percent on those items for children aged 10 or less (INDEC, 2000).

Meanwhile, Rosenzweig and Wolpin (2000) find in their Indian data that clothing expenditures on children under 18 represents 11 percent of household income, and their estimated hand-me-down savings for these goods amounts to 1.3 percent of average earnings. Even assuming that savings of this type exist in Argentina (and that they imply a direct effect on labour supply), their size would be too small to account for a meaningful reduced form relationship between a same sex indicator and parental labour supply. The results for India are specific to the nature of the setting, which consists of extremely poor households in rural areas that devote a large share of their resources to child-related goods.

The indirect evidence presented in this Section is thus consistent with the validity of the exclusion restriction of Assumption 5.2. The following Chapter presents further evidence on the exogeneity of the same sex variable in the labour supply decision based on Hausman-type overidentification tests of the instrumental variables regressions.

CHAPTER 6. SEX PREFERENCES AND FERTILITY

6.4 CONCLUSION

The discussion and the evidence presented in this Chapter indicates that the sex mix of children, as captured by the "same sex" instrument Z, has an effect on fertility D and probably no direct effect on the labour market outcome Y, and it is thus correctly excluded from the second-stage estimation in Argentina. While the preference for a gender-balanced family is complemented by a moderate bias for sons, this secular effect is not endogenous to the labour supply decision of the parents and can be controlled for by adding covariates in two-stage least squares estimations.

In the following Chapter, the significant and plausibly exogenous effect of the sex mix of children on further childbearing in Argentina is exploited to produce instrumental variables estimates of the causal effect of fertility on labour supply.

CHAPTER 7

FERTILITY AND WOMEN'S LABOUR SUPPLY IN ARGENTINA

7.1 INTRODUCTION

As discussed in the Introduction to Part II (page 123), labour market variables are strong correlates of poverty measures, but little is known about their determinants in Argentina. While the negative correlation between childbearing and labour force participation is an almost universal phenomenon, its interpretation in terms of causality is often marred by self-selection, endogeneity and simultaneity issues, as discussed with the help of a stylised model of female labour supply in Section 5.2.1 (page 126).

This Chapter establishes the causal nature of the link between fertility and female labour supply in Argentina by means of the "same sex" identification strategy. This strategy is based on the observation that parents of two children of the same sex exhibit a higher propensity to have another child to obtain a gender-balanced sibling composition. Under certain conditions, analysed in Chapters 5, the sex mix of children can thus be used as an instrument for further childbearing, and as shown in Chapter 6, the exogeneity requirements for the instrument are met in the Argentine case.

The data employed in this Chapter was described in Section 5.4 (page 142) and consists of a large sample of women from the 1991 Census, aged 21 to 35 and with two or more children. As discussed in Introduction to Part II, the period 1980-2000 was characterised by a large increase in female labour force participation and by a fall in fertility rates towards replacement levels. Moreover, as discussed in Section 1.1 (page 18), the year 1991 marked the beginning of a period of market oriented reforms that changed the structure

of Argentine labour markets and the economy as a whole. The post-Second World War period was characterised by relatively high levels of unionisation (about half of the labour force in 1990), job security and high employment protection (Galiani and Gerchunoff, 2003). The structural reforms of the 1990s affected the labour market directly, by modifying employment regulations and reducing the power of unions through labour laws, and indirectly, through the impact of trade liberalisation on unemployment and participation. The main effects were a deterioration of job stability and the growth of the informal sector¹ from 28.9 percent of workers in May 1991 EPH to almost 45 percent in May 2003,² which resulted in a dual market with relatively well paid workers in secure jobs, on the one hand, and self-employed and small-firm employees with no benefits and high de facto job flexibility on the other (Moreno and Roca, 2002).

The unemployment rate had an upward trend before and after the period under study. It raised from less than 3 percent in 1980 up to 6.9 percent in May 1991, with a relatively higher level for women than for men (7.3 and 6.9 percent, respectively). However, these figures are low when compared to the evolution of the 1990s, depicted in Figure 1.3 (page 30) – the unemployment rate eventually peaked at 21.5 percent in May 2002. This trend was simultaneous to a substantial increase in female labour force participation, from 35.4 in 1990 to 38.4 percent in 2000, compared to 66.5 to 67.7 for men (CELADE, 2004, based on Census data). The high levels of unemployment, however, imply that employment rates did not grow as much during this period for women, and decreased substantially for men and the overall population during the crisis of 2001-2002 (Cortés, 2003).

With respect to the distribution of hours, part-time work was relatively uncommon in Argentina. In May 1991, only 5.7 percent of employed men worked less than 30 hours. This modality was more common for women, with 10.8 percent worked less than 20 hours, and 26.1 percent less than 30 hours, but still relatively low when compared to the United Kingdom, for instance, where 40.5 percent of women worked part time in 2000 (ILO, 2003). The low participation of women and the low levels of part time work were

¹See the definition of informality in Section 3.3 (footnote 5, page 95).

²Unless stated otherwise, the figures in this Introduction correspond to EPH indicators presented by Cortés (2003). The May 1991 data is almost contemporaneous to the Census, but the EPH covers only major urban centres.

attributed, among other factors, to the limited supply of childcare facilities for children under school age and to the lack of free provision for poor households (Cortés, 2003). This issue is discussed later in this Chapter in the light of the results from the model of fertility and female labour supply estimated below.

This Chapter is organised as follows. Section 7.2 presents the benchmark results from the Wald and two-stage least squares instrumental variable estimations of fertility and female labour supply in Argentina. Section 7.3 extends these results to parents of at least three children, and provides an analysis of the effects of fertility for different groups defined by their education level. Conclusions follow.

7.2 WALD AND TWO-STAGE LEAST SQUARES ESTIMATES

7.2.1 Wald estimates

This Section studies the causal effect of fertility on female labour supply. The estimations are based on the utility maximisation problem presented in Equation 5.1 (page 127). This model results in a labour supply function $Y = f(\mathbf{K}, D)$, where \mathbf{K} is a vector of characteristics and choice variables, and D is a measure of fertility. The parameter of interest is the labour supply response to changes in the fertility variable, f_D , which may be difficult to recover due to the potential effects of D on the components of \mathbf{K} . As discussed in Section 5.2.1 (page 126), a solution to these endogeneity problems resides in finding an instrumental variable Z that induces variation in fertility (D) but does not affect the labour supply decision directly. Defining K as a generic component of \mathbf{K} , the derivative of Y (labour supply) with respect to Z is given by:

$$\frac{\partial Y}{\partial Z} = \frac{\partial K}{\partial Z} f_K + \frac{\partial D}{\partial Z} f_D$$

Since the exogeneity of *Z* ensures that $\partial K/\partial Z = 0$, the effect of fertility on labour supply is identified:

$$f_D = \frac{\partial Y}{\partial Z} / \frac{\partial D}{\partial Z} \tag{7.1}$$

Previous work on fertility and labour supply found that parental sex preferences provide valid instruments for models of this type in the United States (Angrist and Evans, 1998). The evidence presented in Chapter 6 for Argentina indicated that the sex mix of children, as captured by the "same sex" instrument Z, has an effect on further childbearing in women with at least two children (D), and no direct effect on the labour market outcome Y.

The simplest way to estimate the parameter in Equation 7.1 is by instrumental variables, in a model with a constant and one endogenous regressor like Equation 5.6:

$$Y_i = \gamma + \phi D_i + \varepsilon_i$$

This regression is estimated with Z as an instrument for D with a first-stage regression of the form:

$$D_i = \gamma_f + \theta Z_i + \epsilon_i \tag{7.2}$$

The parameter ϕ estimated in this way corresponds to the Local Average Treatment Effect of Proposition 5.1 (page 135), since by definition ϕ obtained by IV represents the difference in means of the outcome variable for the two values of the instrument *Z* divided by θ , which is equal to the difference in the expectation of the treatment *D* at the two values of *Z*.³

The Assumptions 5.1-5.4, required by Proposition 5.1, were the object of the previous Chapter, which presented evidence on their plausibility in the Argentine case. It can then be assumed that the parameter ϕ identifies a Local Average Treatment Effect (LATE) for the group of compliers.

Table 7.1 reports the results of the unconditional IV model of Equation 5.6, the LATE-Wald coefficient ϕ , for the four samples defined in Section 5.4 (page 142). For comparison purposes, the Table also presents the corresponding OLS results for ϕ , obtained by estimating the model in Equation 5.6 without instrumenting D.⁴

³While the discussion is in terms of a binary treatment D (*More than two children*), this Chapter also presents results obtained with D set to the *Number of children* variable. These two regressors correspond to the binary and variable treatments, respectively, as discussed in Section 5.2.3 (page 132). The case of variable treatment intensity requires only minor adjustments to the interpretation of the LATE parameter, as described in the same Section (footnote 14, page 135).

⁴While useful for comparison purposes, it should be noted that the OLS and instrumental variables coefficients are not necessarily comparable without further homogeneity assumptions. As discussed in Section 5.2.3 (page 132), instrumental variables estimates identify the causal effect only for the subpopulation of compliers, while OLS coefficients,

	Complete Sample Married					rried				
	Proportion of sample	Worked for pay	Number of children	More than two children	Proportion of sample	Worked for pay	Number of children	More than two children		
Overall mean		0.3138	2.9763	0.5480		0.3050	2.9050	0.5280		
Same sex (1) Mixed sex (2) Difference (1)-(2)	0.5066 0.4934	0.3124 0.3153 -0.0029	3.0047 2.9472 0.0574	0.5637 0.5318 0.0320	0.5056 0.4944	0.3034 0.3067 -0.0034	2.9365 2.8729 0.0636	0.5460 0.5096 0.0364		
Wald estimate Standard error			-0.0511 [0.0199]**	-0.0917 [0.0358] ^{**}			-0.0529 [0.0205] ^{***}	-0.0924 [0.0357] ^{***}		
OLS estimate Standard error			-0.0352 [0.0005] ^{***}	-0.0783 [0.0011]***			-0.0392 [0.0006] ^{***}	-0.0800 [0.0013] ^{***}		
Observations		653	3,213		497,194					
		AE S	Sample		AE Sample, married					
	Proportion of sample	Worked for pay	Number of children	More than two children	Proportion of sample	Worked for pay	Number of children	More than two children		
Overall mean		0.3155	3.0619	0.5955		0.3046	2.9848	0.5741		
Same sex (1) Mixed sex (2) Difference (1)-(2)	0.5062 0.4938	0.3139 0.3171 -0.0032	3.0935 3.0295 0.0640	0.6131 0.5775 0.0356	0.5053 0.4947	0.3025 0.3066 -0.0041	3.0196 2.9492 0.0704	0.5940 0.5538 0.0403		
Wald estimate Standard error			-0.0503 [0.0187] ^{***}	-0.0905 [0.0336] ^{***}			-0.0584 [0.0193] ^{***}	-0.1021 [0.0337]***		
OLS estimate Standard error		500	-0.0383 [0.0005]***	-0.0913 [0.0012]***		45	-0.0410 [0.0006]***	-0.0875 [0.0014]***		
Observations		395	7,941			450	5,4 <i>5</i> 7			

Table 7.1: Wald Estimates of the Effect of Fertility on Women's Labour Supply

Note: * significant at 10%; ** significant at 5%; *** significant at 1%. The Wald estimates represent the difference (1)-(2) of the *Worked* for pay indicator divided by the difference (1)-(2) of the *Number of children* and *More than two children* variables. The OLS estimates represent the coefficients of these two variables in the respective regressions with *Worked for pay* as the dependent variable. The samples correspond to women aged 21-35 with two or more children aged 18 or younger from the extended questionnaire sample of the 1991 Census, as described in the text.

The Wald estimates of the effect of the total number of children and having more than two children on the probability of working for pay is obtained with the following variables: the instrument *Z* is the binary indicator *Same sex*, the dependent variable *Y* is *Worked for pay*, and the endogenous regressor *D* is either the *Number of children* variable or the *More than two children* indicator.

Given the interpretation of OLS regressions with one binary independent variable as simple differences in the mean of the dependent variable, Table 7.1 presents the overall means of *Y* and *D*, and their values conditional on Z = 1 (same sex) and Z = 0 (mixed sex). The Table also reports the difference of the conditional means, which in the case of the variables *Number of children* and *More than two children* can be interpreted as the coefficient θ in the first-stage regression of Equation 7.2.

The Table has four panels, one for each of the samples on which the analysis is carried out. The first row within each panel contains the means of the variables of interest, as already presented in Table 5.1 (page 145). The first column indicates that among parents of at least two children there is a marginally higher probability of having the first two children of the same sex (about 50.5-50.6 percent) than one son and one daughter.

The reduced form effect of the *Same sex* instrument on the outcome variable *Worked for Pay*, denoted by the difference in the conditional means, is very small, as expected from the discussion of the previous Chapter. This difference, which represents the numerator of the LATE coefficient of Equation 5.5 (page 135), is of the expected sign: mothers of *Same sex* children have a lower probability of working for pay of about 0.3 to 0.4 percentage points, which can be attributed to the impact of the instrument on the endogenous fertility variables. The third and fourth columns of each panel report the means of *Number of children* and *More than two children* conditional on *Same sex*, and their differences. Women whose first two children were of the same sex had on average between 0.057 and 0.0584 more children, and a 0.032-0.0403 higher probability of having a third child, depending on the sample. These numbers represent the estimate of the parameter θ in the first-stage Equation 7.2, and correspond to the parametal preference for

as given by Equation 5.4 (page on 132), provide a (potentially biased) estimate of the average effect for the whole population.

a gender-balanced family in the transition from two to three children, presented in Table 6.2 and discussed in Section 6.3.1 (page 155).

The bottom part of each panel in Table 7.1 reports the OLS and Wald (IV) estimates of the effects of fertility on women's labour supply, represented by ϕ in Equation 5.6. The Wald coefficients indicate that an additional child reduces women's labour supply by about 5 percentage points, while having more than two children has a negative impact of about 9-10 percentage points. The results are very similar for the *Complete* and *AE Samples*, although the coefficients are slightly higher (in absolute value) for married women. Moreover, all estimates are strongly significant, with six of the eight Wald coefficients different from zero at the 1 percent level. Finally, for both instrumented variables, *More than two children* and *Number of children*, the Wald point estimates are higher than the OLS coefficients in absolute value, implying that OLS estimates might be underestimating the negative effect of additional children on women's labour supply if the compliers are similar in their characteristics to the rest of the population. The following Section presents a formal test of these differences.

The Wald estimates and their statistical significance represent strong evidence of a causal effect of fertility on labour supply in the Argentina data. However, as discussed in general in Section 6.2 and specifically for Argentina in Section 6.3 (page 155), the concerns raised by a mild son preference in the sample indicate the need for controlling for the secular effect of the sex of the first two children in the estimations. Moreover, other plausible exogenous demographic characteristics can be included as regressors in the second stage. The following Section addresses these issues, presenting the results of the two-stage least squares estimation of the fertility and labour supply model that accommodates a set of covariates.

7.2.2 First stages and two-stage least squares: benchmark results

The first-stage effect of the instrument was evident in the tabulations of Table 6.2, and it was confirmed in Table 7.1. However, the results from Table 6.2 point out the presence of a mixed-sex sibling preference complemented with a moderate bias for boys, so that the model could be improved by adding covariates to control for any secular independent of the sex of the first two children, and by distinguishing between the *Two boys* and *Two girls* components of the *Same sex* instrument. While the Wald setup of Equation 5.6 provides a compact model for causal effects, its drawbacks are the lack of controls for other exogenous characteristics and its reliance on a single instrument. This Section deals with the constant effect model of Equation 5.7:

$$Y_i = X'_i \beta + \phi D_i + \varepsilon_i$$

This model is estimated by two-stage least squares, with Z as an instrument for D, which implies a first-stage regression of the form:

$$D_i = X_i' \beta_f + \theta Z_i + \epsilon_i \tag{7.3}$$

The set of covariates X includes plausibly exogenous characteristics like the age of the woman and her age at first birth. Given the possibility of effects of the gender of children on omitted variables, the vector X also includes indicators for the sex of the first two children. As discussed in the robustness checks presented in Appendix D (page 232), the main results of this Section are not affected by the inclusion of other covariates in X, but the evidence suggests that some of them, like the indicators of the education level, might be endogenous. For this reason, this Section only reports results with a minimum of demographic variables in X.

Finally, with respect to the mild son preference found in the data, the estimation is also carried out by decomposing the *Same sex* indicator into the indicators *Two boys* and *Two girls*, since a further advantage of two-stage least squares is that it allows for more than one instrument to be included in the estimation: these two variables can be used as an instrument for *D* in Equation 5.7. The estimation with two separate instruments implies a first-stage model of the form:

$$D_i = X'_i \beta_f + \theta_1 Z_{1i} + \theta_2 Z_{2i} + \epsilon_i \tag{7.4}$$

where Z_1 and Z_2 represent the *Two boys* and *Two girls* indicators, and *D* and

⁵It should be noted that the regressions implicitly control for the age of the first child, equal to the difference of *Age* and *Age at first* birth. The ages of other children were not included because of the potential endogeneity of birth intervals, which are left for further research.

X are defined as before.⁶

Table 7.2 reports the coefficients θ from Equation 7.3 and θ_1 and θ_2 from Equation 7.4. These OLS first-stage regressions are estimated for the two instrumented variables *D*, *Number of children* and *More than two children*, and the Table reports only the coefficients and standard errors of the instruments.

Columns one to four contain the results for an OLS regression with *Num*ber of children as the dependent variable, and columns five to eight present the same information with *More than two children* as the dependent variable. The first row indicates that women in the complete sample with two children of the same sex have on average 0.0633 more children after conditioning on the characteristics *X*, and have a 0.0341 higher probability of having more than two children, with slightly higher numbers for the *AE* samples and for married women. It should be noted that all the coefficients presented in the Table are significant at the 1 percent level.

The results for the *Same sex* instrument in Table 7.2 are a weighted average of the effect of having two girls and two boys. These are presented in the second and third rows. For the whole sample, women with two girls have on average 0.0894 more children, and a 0.0439 higher probability of having more than two children. As observed previously in Table 6.2 (page 155), these figures are lower for women who had two boys (0.0372 and 0.0243 respectively). Finally, the panel at the bottom of the Table presents the results of the same first-stage regressions, but using the covariates corresponding to the women's spouse. The coefficients are similar to those for married women, which are higher than those for the overall sample.

Table 7.2 represents a multivariate version of the parity progression ratios of Table 6.2 for the transition from parity two to three, and the conclusions from that Table still hold conditional on the covariates X: the sex mix of the first two children has a strongly significant effect on fertility variables, and this first-stage relationship reveals a preference for a gender-balanced family with a moderate bias for sons.

Based on the effects of the instrument Z (and its components Z_1 and Z_2)

⁶When using these two instruments instead of *Same sex*, it is not possible to control for the sex of the first two children in the regression because of perfect multicollinearity. As in Angrist and Evans (1998), the results presented here control for the sex of the first child when using *Two boys* and *Two girls* as instruments.
Dependent variable:		Number of children				More than two children			
		Complete Sample	Married	AE Sample	AE Sample, married	Complete Sample	Married	AE Sample	AE Sample, married
Coefficient of:									
Same Sex Controls for sexes of 1st & 2nd children	Same Sex	0.0633 [0.0026]***	0.0695 [0.0029]***	0.0683 [0.0028]***	0.0750 [0.0031]***	0.0341 [0.0011]***	0.0387 [0.0013]***	0.0370 [0.0012]***	0.0419 [0.0014]***
Two Boys and Two Girls Control for sex of 1st child	Two Boys	0.0372 [0.0037]***	0.0436 [0.0040]***	0.0402 [0.0040]***	0.0466 [0.0043]***	0.0243 [0.0016]***	0.0286 [0.0018]***	0.0265 [0.0017]***	0.0306 [0.0019]***
	Two Girls	0.0894 [0.0038]***	0.0954 [0.0041]***	0.0964 [0.0040]***	0.1033 [0.0044]***	0.0439 [0.0016]***	0.0488 [0.0018]***	0.0475 [0.0017]***	0.0531 [0.0019]***
Spouse worked for pay						. <u>.</u>			
Same Sex Controls for sexes of 1st & 2nd children	Same Sex		0.0657 [0.0027]***		0.0753 [0.0031]***		0.0354 [0.0012]***		0.0420 [0.0014]***
Two Boys and Two Girls	Two Boys		0.0408		0.0475		0.0256		0.0310
Control for sex of 1st child	Two Girls		0.0906		[0.0044]*** 0.1032 [0.0045]***		0.0451		[0.0019]*** 0.0531 [0.0019]***
Observations		653,213	497,194	599,941	456,437	653,213	497.194	599,941	456,437

Table 7.2: First Stages: Effect of Sex Composition on Further Childbearing

birth of the woman or her spouse, in addition to the sex of the first and second children (where indicated). The samples correspond to women aged 21-35 with two or more children aged 18 or younger from the extended questionnaire sample of the 1991 Census, as described in the text.

on the endogenous regressors *D*, and given the plausibility of the identifying assumptions in the Argentine case as discussed in the previous Chapter, Table 7.3 presents the results of the estimation of Equation 5.7 by ordinary and two-stage least squares. The Table reports the coefficient ϕ by OLS and using *Same sex* or *Two boys* and *Two girls* as instruments for *Number of children* (columns 1-4) and *More than two children* (columns 5-8), with *Worked for pay* as the dependent variable.

The first panel of the Table presents the OLS results. Since the use of *Two boys* and *Two girls* as instruments implies that it is not possible to control for the sex of the first two children, due to multicollinearity, the panel presents the OLS estimations when controlling alternatively for the sex of the first child and the sex of the first two. These OLS results are unaffected by the exclusion of the second child, and are all strongly significant at the 1 percent level, indicating that an additional child reduces the mother's probability of working by 0.0415-0.0469, while the *More than two children* indicator has an impact of 0.0792-0.0984 on labour supply, depending on the samples. The impact is lower (in absolute terms) for married women, and for the *AE* samples.

The second panel of Table 7.3 presents the instrumental variables estimates of Equation 5.7. The first set of coefficients corresponds to the use of the *Same sex* variable as the sole instrument. Instrumental variables estimates for married women are higher (in absolute value) than the correspondent OLS coefficients for both *Number of children* and *More than two children* (around -0.04/-0.05 and -0.07/-0.09, respectively), but they are lower for the overall samples (*Complete* and *AE*). It should be noted that all these coefficients are significant at least at the 5 percent level, indicating that the instrument provides enough variation in the endogenous variables.

The next set of results in the second panel of Table 7.3 presents the estimation when the *Same sex* instrument is decomposed into its *Two boys* and *Two girls* components. The pattern of differences with OLS estimates is now much simpler: using the two instruments, IV coefficients are consistently smaller (in absolute value) than OLS, in contrast with the pattern of differences with the Wald estimates of Table 7.1. The differences are relatively large: *More than two children* OLS coefficients are in the -0.0793/-0.0985 range, but are restricted to -0.061/-0.0801 for the instrumental variables es-

	· · · · · · · · · · · · · · · · · · ·
ried AE Sample	AE Sample
	married
9 2 -0.0984 ·	-0.0826
4]*** [0.0013]***	[0.0015]***
93 -0.0985	-0.0826
4]*** [0.0013]***	[0.0015]***
	,
28 -0.0775	-0.0931
2]** [0.0319]**	[0.0320]***
5 0.5105	0.7425
22 -0.0621	-0.0801
2]** [0.0309]**	[0.0311]***
0 0.0653	0.0941
8 0.2391 (0.9371
<u></u>	
16 .	-0.0090
6]***	[0.0006]***
4	-0.0031
18]	[0.0134]
5 (0.6596
94 599,941 4	456,437
	22 -0.0621 22]** [0.0309]** 0 0.0653 8 0.2391 16

Table 7.3: OLS and Two-Stage Least Squares Estimates of Fertility and Women's Labour Supply

Note: Standard errors in brackets. * significant at 10%; ** significant at 5%; *** significant at 1%. All regressions include the age and the age at first birth of the woman or her spouse, in addition to the sex of the first and second children (where indicated). The samples correspond to women aged 21-35 with two or more children aged 18 or younger from the extended questionnaire sample of the 1991 Census, as described in the text.

181

timates; the pattern is similar for *Number of children*. For both variables, the coefficients for the whole samples are smaller than those for married women, and their level of significance is somewhat smaller than when using *Same sex* as an instrument, but still significant overall (three at the 10 percent level, four at 5 percent and one at 1 percent).

As noted by Angrist and Evans (1998), an advantage of decomposing the Same sex instrument is that the model becomes overidentified, with two instruments for one endogenous regressor, and thus a standard Sargan test of over-identifying restrictions can be applied. These results provide further evidence in favour of the exogeneity of the instrument, discussed in Section 6.3: the validity of the instruments is not rejected by the tests' results, as indicated by their p-values in Table 7.3. This implies that the instruments are uncorrelated with the error term and correctly excluded from the estimated equation. For only one case (Number of children - AE Sample), the test's null is rejected at the 5 percent level. Moreover, Angrist and Imbens (1995) show that this test also measures the difference in the coefficient ϕ when the estimation is carried out with the two instruments separately. The causal effect of childbearing for Two boys compliers is thus not significantly different from the effect for the *Two girls* compliers, providing further evidence that the moderate bias towards sons in the gender-balanced preferences is not a cause of concern, as discussed at length in Section 6.3.2 (page 155).

The last panel in Table 7.3 presents the results for *Spouse worked for pay* as the dependent variable, with the corresponding controls for the sex of the first two children, the spouse's age and age at first birth. While there seems to be a small but strongly significant effect in the OLS regressions (-0.0066/-0.0076 for *Number of children*, -0.0116/-0.009 for *More than two children*), none of the instrumental variables coefficients are significantly different from zero at the standard levels of confidence. This can be taken as evidence of no effect of the number of children on the spouses' labour supply, which might be explained by the small variation in the dependent variable: more than 95 percent of the male spouses work for pay. It should be noted, however, that the IV estimates are statistically indistinguishable from the OLS estimates, so that the instrument might not contain enough information to isolate the impact of children on this outcome.

Regarding the differences between IV and OLS, it is worth remember-

183

ing that the "same sex" strategy identifies the average effect of having more than two children on those whose fertility decisions are changed by the instrument, the group of compliers, while OLS is suspected to fail at identifying the same effect averaged over the whole population. While the two methods do not estimate results for the same groups without further assumptions (see the discussion of compliers in Chapter 8), they can still be compared and their differences can be tested for statistical significance: under homogeneity assumptions, this difference is the selection bias from OLS in Equation 5.4 (page 131). Table 7.3 reports the results from a Durbin-Wu-Hausman test for the two-stage least squares models. Most of the OLS coefficients in Table 7.3 are higher (in absolute value) than their IV counterparts, but the opposite is true for the Wald estimates of Table 7.1, pointing to upward and downward biases as discussed in Section 5.3.1 (page 155). However, while the instrumental variable coefficients are significant at the 5 percent level, their confidence intervals are still broad, and this is reflected in the failure to reject the Durbin-Wu-Hausman null hypothesis of significant differences between IV and OLS coefficients for all the models in the Table and in Table 7.1 (tests not reported).

7.3 ADDITIONAL RESULTS

7.3.1 More than three children

This Section provides additional findings, and some robustness checks of the benchmark results of this Chapter. A first extension to the "same sex" strategy is based on the evidence from the parity progression ratios in Table 6.2 (page 159), which indicates that the preference for a gender-balanced family in Argentina is not limited to the two to three children range: parents of three boys or three girls have a significantly higher probability of having a fourth child than parents of mixed sex siblings. The higher fertility implies that the decision to go from three to four children is more relevant in the Argentine case than in the other applications reviewed in Section 5.3.3 (page 140). In 1991, 33.8 percent of women aged 19-49 had three or more children, compared to only 20.5 percent in the United States in 1990. This allows the estimation of a 2SLS model like Equation 5.7 for the impact of childbearing at higher parities on women's labour supply by limiting the samples to women with three children or more, which none of previous studies has attempted.

184

The first-stage, OLS and IV results from this model are presented in Table 7.4. The instrument *Same sex* indicates that the first three children were of the same sex, and it is decomposed into its *Three boys* and *Three girls* indicators. The endogenous regressors are *Number of children* and *More than three children*, and the covariates X include controls for age, age at first birth and the sex of the first two or three children.

In accordance with the evidence in Table 6.2, the first stages are strongly significant. The *Same sex* indicator has an effect of about 0.041-0.046 more children and 0.025-0.029 higher probability of having a fourth child. While significant, these coefficients are substantially lower than the *Same sex* effect on the two to three children decision, again as expected from Table 6.2.

The OLS results in Table 7.4 suggest that the impact of childbearing on labour supply is relatively homogeneous, not varying widely with the number of children: the coefficients of *Number of children* are in the 0.036-0.038 range, lower but close to those in Table 7.3. The same is true for the coefficients of *More than three children*: at around 0.07-0.073, they are not far below than those for *More than two children*.

Although the sample size is smaller and the effect of the instruments is weaker, the instrumental variables coefficients of *Number of children* and *More than three children* in Table 7.4 are still significant at the 10 percent level for married women. Compared to the results of Table 7.3, these IV coefficients indicate that the negative causal effect of fertility on labour supply is higher for a fourth child than for a third child. When instrumenting *Number of children*, the estimate is around 0.07 for married women, above the 0.029-0.052 range for the two to three children effect, while the results for *More than three children*, at around 0.11, are higher than those of *More than two children* in Table 7.3 (0.061-0.093).

Finally, the Sargan null fails to be rejected in all the estimations, supporting the exogeneity of the *Three boys* and *Three girls* indicators, but as in Table 7.3, the Durbin-Wu-Hausman tests indicate that the differences between IV and OLS estimates are not significant.

<u> </u>	Instr	Instrumented: Number of children				Instrumented: More than three children			
Dependent variable:	Complete	Married	AE Sample	AE e Sample, married	Complete Sample	Married	AE Sample	AE	
Worked for pay	Sample							married	
First stages									
Coefficient of Same sex	0.0410	0.0463	0.0410	0.0462	0.0249	0.0295	0.0250	0.0296	
	[0.0040]***	[0.0044]***	[0.0040]***	[0.0044]***	[0.0018]***	[0.0020]***	[0.0018]***	[0.0021]***	
OLS estimates									
OLS	-0.0377	-0.0362	-0.0378	-0.0363	-0.0728	-0.0704	-0.0728	-0.0705	
	[0.0007]***	[0.0009]***	[0.0007]***	[0.0009]***	[0.0016]***	[0.0018]***	[0.0016]***	[0.0018]***	
IV Estimates			- · · · · · ·						
IV: Same Sex	-0.0538	-0.0708	-0.0540	-0.0712	-0.0886	-0.1111	-0.0885	-0.1112	
	[0.0409]	[0.0418]*	[0.0409]	[0.0419]*	[0.0673]	[0.0654]*	[0.0671]	[0.0653]*	
DWH p-value	0.6938	0.4063	0.6911	0.4022	0.8135	0.5340	0.8144	0.5321	
IV: Three Boys and Three Girls	-0.0493	-0.0695	-0.0492	-0.0698	-0.0846	-0.1094	-0.0843	-0.1094	
Controls for sexes of 1st & 2nd children	[0.0406]	[0.0418]*	[0.0406]	[0.0419]*	[0.0672]	[0.0655]*	[0.0669]	[0.0653]*	
Sargan p-value	0.3267	0.2921	0.3097	0.2757	0.3568	0.2967	0.3387	0.2792	
DWH p-value	0.7756	0.4251	0.7770	0.4216	0.8600	0.5509	0.8639	0.5503	
Observations	357,933	262,525	357,273	262,049	357,933	262,525	357,273	262,049	

Table 7.4: OLS and Two-Stage Least Squares Estimations for Three or More Children

Note: Standard errors in brackets. * significant at 10%; ** significant at 5%; *** significant at 1%. All regressions include controls for the age and the age at first birth of the women and the sex of the first two or three children. The samples correspond to women aged 21-35 with three or more children aged 18 or younger from the extended questionnaire sample of the 1991 Census, as described in the text.

7.3.2 Heterogeneous effects: results by education level

The results in the previous Sections of this Chapter captured the average causal effect of fertility on women's labour supply for the population as a whole. However, subgroups of the population can be expected to have different propensities to participate in the labour market, and the effect of childbearing might differ substantially for them.

186

In Table 7.5, the two-stage least squares model given by Equation 5.7 was estimated separately for three mutually exclusive subgroups of the population defined by the level of education of women: those with up to some primary education, those with some high school education, and those with some tertiary or university education. The first stages (not reported) for these three groups are remarkably close, and consistent with the genderbalanced preferences with a moderate bias for sons in all three cases: the fact that the same fertility effects are observed for very different groups of the population reinforces the conclusions of Section 6.3.1 (page 155) on the nature of the group of compliers.⁷

The results in Table 7.5 correspond to OLS and IV regressions of *Worked for pay* on the standard set of controls, with *Number of children* and *More than two children* instrumented by the *Same sex* indicator.

For the less educated group, the results are similar to those in Table 7.3 for the overall population, although the IV estimates are significant at the standard levels for the married samples only. For the group of women with some high school education, the effects of fertility in the IV regressions are significantly different from zero (for six of the coefficients at the 10 percent level, and at 5 percent for the other two), and they are higher than those in Table 7.3: for *Number of children*, the coefficients (in absolute value) stand around 0.07 (compared to 0.04-0.05), while the *More than two children* coef-

⁷The unconditional effects of *Two boys* and *Two girls* in a regression with *More than two children* as the dependent variable are 0.0223 and 0.0423, and the effect of the *Same sex* indicator is 0.0320 (as in Table 7.1) for the *Complete sample*. The same coefficients estimated for the primary education group are 0.0201, 0.0407 and 0.0301; for the high school sample 0.0254, 0.0440 and 0.0344; and for the further education group, 0.0284, 0.0422 and 0.0350. All these coefficients are similar, and only the *Two boys* coefficient for the further education sample is outside the 95 percent confidence intervals of the full sample coefficients, indicating that for this group the preference for boys is smaller than for the other two. However, this group still exhibits preferences for a gender-balanced family with a (more) moderate bias for sons. The same type of decomposition reveals that the first-stage effects are statistically equal between rural and urban households.

	Instrumented: Number of children			ildren	Instrumented: More than two childr			
Dependent variable: Worked for pay	Complete Sample	Married	AE Sample	AE Sample, married	Complete Sample	Married	AE Sample	AE Sample, married
			Educatio	on Level: Comp	oleted prima	ry or less		
IV: Same Sex	-0.0299 [0.0197]	-0.0508 [0.0197]**	-0.0293 [0.0194]	-0.0497 [0.0194]**	-0.0629 [0.0416]	-0.1028 [0.0399]**	-0.0613 [0.0407]	-0.1004 [0.0393]**
DWH p-value	0.7804	0.3624	0.7187	0.4101	0.7403	0.4094	0.5967	0.5150
OLS with same controls	-0.0354	-0.0329	-0.0362	-0.0338	-0.0767	-0.0699	-0.0828	-0.0749
	[0.0006]***	[0.0007]***	[0.0006]***	[0.0007]***	[0.0015]***	[0.0018]***	[0.0016]***	[0.0018]***
Observations	379,290	264,142	355,560	248,477	379,290	264,142	355,560	248,477
		E	ducation Le	vel: high schoo	ol, complete	or incomple	te	
IV: Same Sex	-0.0655	-0.0683	-0.0680	-0.0755	-0.1063	-0.1107	-0.1100	-0.1219
	[0.0353]*	[0.0363]*	[0.0341]**	[0.0351]**	[0.0573]*	[0.0588]*	[0.0551]**	[0.0567]**
DWH p-value	0.6976	0.5515	0.6442	0.4198	0.8062	0.5847	0.7803	0.4620
OLS with same controls	-0.0518	-0.0468	-0.0523	-0.0472	-0.0923	-0.0786	-0.0946	-0.0803
	[0.0013]***	[0.0014]***	[0.0013]***	[0.0014]***	[0.0023]***	[0.0025]***	[0.0023]***	[0.0025]***
Observations	194,683	161,631	176,000	146,492	194,683	161,631	176,000	146,492
			Educatio	n Level: some	university o	r tertiary		
IV: Same Sex	0.0196	0.0505	-0.0271	-0.0072	0.0280	0.0700	-0.0386	-0.0100
	[0.0663]	[0.0671]	[0.0613]	[0.0616]	[0.0945]	[0.0928]	[0.0875]	[0.0852]
DWH p-value	0.2515	0.1268	0.6386	0.4809	0.2985	0.1485	0.7141	0.5302
OLS with same controls	-0.0559	-0.0508	-0.0558	-0.0505	-0.0698	-0.0629	-0.0707	-0.0633
	[0.0025]***	[0.0026]***	[0.0025]***	[0.0027]***	[0.0037]***	[0.0039]***	[0.0038]***	[0.0040]***
Observations	79,240	71,421	68,381	61,468	79,240	71,421	68,381	61,468

Note: Standard errors in brackets. * significant at 10%; ** significant at 5%; *** significant at 1%. All regressions include the age and the age at first birth of the women or her spouse, in addition to the sex of the first and second children (where indicated). The samples correspond to women aged 21-35 with two or more children aged 18 or younger from the extended questionnaire sample of the 1991 Census, as described in the text.

187

ficients, at around 0.10-0.12, are higher than the previous 0.08-0.09. Given the wide IV confidence intervals, however, the equality of coefficients in the two Tables cannot be rejected.

As expected, the OLS coefficients are significantly different from zero for all samples and education groups. Despite this fact, none of the IV estimates for the higher education group are even marginally significant. This might be due to a lack of precision, since the smaller sample size may not provide enough variation to isolate a causal effect of children on labour supply (see the discussion of significance and sample sizes in Appendix D, page 236). Alternatively, there might be no effect at all for this group. A possible reason for this is that richer households (as proxied by the education level) may be able to afford market-provided childcare, which attenuates the fertility effect on labour supply. This hypothesis is supported by the data. The 1991 Census covered members of the household and their domestic employees living in the same dwelling, which can be taken as a lower bound for childcare services consumption. Overall, 1.03 percent of the women in the final sample had a domestic employee living in the house. The proportion for the lowest education group was only 0.19 percent, in contrast with 1.02 percent for those with some high school and a relatively high 4.51 percent for women with some higher education. While not conclusive, this result is consistent with the idea that access to childcare attenuates the negative effect of childbearing on women's labour supply. Moreover, it is also consistent with previous findings: as discussed in the Introduction to this Chapter, Cortés (2003) states that the lack of free childcare for children under school age was a major impediment for the labour force participation of women in poorer households in Argentina.

The heterogeneity in the impact of fertility on women's labour supply in Argentina is further analysed in terms of potential outcomes in the following Chapter.

7.4 CONCLUSION

Building on the results from Chapter 6, which revealed a significant and plausibly exogenous effect of the sex mix of children on further childbearing in Argentina, this Chapter exploited this source of variation to produce instrumental variables estimates of fertility on female labour supply.

The results of this Chapter indicate the presence of a strongly significant negative causal effect of fertility on the labour supply of women with at least two children. This result is robust to different specifications of the model, the variables and the underlying datasets, and it was shown to hold for women with three or more children. The strongest negative impact of fertility was on the labour supply of women with low education levels, while the effect on women with higher education was not significantly different from zero.

The findings of this Chapter indicate that childbearing leads to a reduction in female labour supply in Argentina – but how much of the increase in labour force participation in the last three decades can be explained by the large changes in fertility observed in Figure 5.1 (page 125)? According to CELADE (2004), fertility fell by 24 percent in Argentina between 1980 and 2000, while female labour force participation increased by 14.3 percentage points (just under 60 percent). Using the Wald estimates for the complete sample in Table 7.1, one fewer child implies a fall in female labour force participation of about 5.11 percentage points. The fall of 0.82 children in Argentina between 1980 and 2000 accounts for an increase of 4.2 percentage points in participation, which is 29.4 percent of the increase in female employment during the same period.

It should be noted, however, that the identified causal effect holds only for the group of compliers (defined in Chapter 5) without further assumptions: the calculation of the previous paragraph is only valid if the estimated elasticity can be generalised from compliers to the whole population of women with at least two children (and from this group to all women). The following Chapter discusses the generality of these Local Average Treatment Effects, and establishes whether the identified causal link can be extrapolated from compliers to the population. It also expands the results by education levels by applying additional identification results within the potential outcomes framework.

The robust causal link between fertility and female labour supply, the substantial magnitude of its effects and its stronger impact on the poorest group of the population have important policy implications, which are discussed in the Conclusion to Part II (page 212).

CHAPTER 8

POTENTIAL OUTCOMES AND EXTRAPOLATION OF RESULTS FOR COMPLIERS

8.1 INTRODUCTION

The previous three Chapters presented the "same sex" identification strategy, which consists of using the sex composition of children as instruments for childbearing in a model of fertility and female labour supply. The strategy exploits parental preferences for a mixed-sex sibling composition, which induce higher fertility in couples with two children of the same sex. The validity of the strategy was discussed in Chapter 6 by means of a detailed analysis of the identifying assumptions and their plausibility in the Argentine case, studying the institutional context and the testable implications of the assumptions set out in Chapter 5. The evidence supported the validity of the strategy for Argentina, and this allowed the estimation of the causal effect of fertility on women's labour supply in Chapter 7.

The aim of this Chapter is to extend and generalise these findings by specifying additional identification results in the context of instrumental variable estimation, and using the Argentine data on fertility and female labour supply to illustrate how these results can be applied. The motivation is twofold. On the one hand, without further assumptions, IV results like those obtained in the previous Chapter are only valid for the group of compliers (Imbens and Angrist, 1994). In the model of fertility and female labour supply, this group is composed of women who had an additional child because the first two were of the same sex. The issue is whether the results obtained for compliers are representative and can be extrapolated to the whole population, since the parameter of interest is how women in general (and not only compliers) adapt their labour supply to changes in fertility. On the other hand, the estimation of the labour supply model by education group, presented in Section 7.3 (page 183), revealed the presence of heterogeneous effects of fertility: childbearing only affected the labour supply of women with lower levels of education. The following pages present additional identification results within the potential outcomes framework to study the nature of heterogeneous effects beyond the simple estimation of causal effects for different subgroups of the population. While IV coefficients estimate the difference between outcomes (for instance, labour supply) in two counterfactual situations (for instance, low and high fertility), the analysis developed in this Chapter focuses on the outcome levels in these two situations, and not only in the difference between the two.

The Chapter is organised as follows. Section 8.2 discusses the nature of the group of compliers, presenting additional identification results and proposing an auxiliary test to verify to what extent the LATE estimates can be generalised to the whole population. Section 8.3 presents the main results on fertility and female labour supply broken down by education group, and carries out the discussion beyond causal effects by concentrating on potential outcomes. The issue of extrapolation of results for compliers is then discussed with respect to the three education subgroups, and the auxiliary test proposed in Section 8.2 is illustrated with the Argentine labour supply data and with a job training programme dataset from the United States. Conclusion follows.

8.2 "SAME SEX" COMPLIERS AND THE GENERALITY OF LATE RESULTS

8.2.1 "Same sex" compliers and extrapolation

The potential outcomes framework, presented in Section 5.2.2 (page 130), addresses problems of endogeneity in the estimation of causal effects. The definition of causal effects relies on the notion of potential – as opposed to observed – outcomes Y_d with respect to a binary treatment D: Y_{0i} and Y_{1i} represent the outcome that would have been observed for individual i in the two alternative situations $D_i = 0$ and $D_i = 1$, one of which is necessarily a counterfactual. In the case of fertility and women's labour supply, the employment status of a woman i who has more than two children ($D_i = 1$) is

 Y_{1i} , which is equal to the observed Y_i , and Y_{0i} represents the same woman's counterfactual employment status which would have been observed if she had only two children ($D_i = 0$). The causal effect of the treatment D for individual i is defined in terms of counterfactuals as the difference between the two potential outcomes, $Y_{1i} - Y_{0i}$ (Abadie, 2003a). Since Y_{0i} and Y_{1i} cannot both be observed at the same time for the same individual i, causal effects cannot be computed at the individual level: the identified parameter refers necessarily to the average causal effect $E[Y_1 - Y_0]$.

As discussed in Section 5.2.3, the LATE identification result of Proposition 5.1 (page 135) establishes that when an instrument *Z* is available, and satisfies Assumptions 5.1-5.4 (page 134), this average causal effect is identified (i.e., can be expressed in terms of observable quantities) for the group of compliers. The proposition results in Equation 5.5, which defines the Local Average Treatment Effect (LATE):

$$E[Y_1^c - Y_0^c] = \frac{E[Y|Z=1] - E[Y|Z=0]}{E[D|Z=1] - E[D|Z=0]}$$

This result, however, holds only for compliers, those whose treatment status is changed by the instrument. In the case of the "same sex" instrument, compliers consist of two groups: namely, women who had another child because their first two children were of the same sex, but who would have stopped at two if the children were of different sex, and the reciprocal group, women who did not have additional children because they had one boy and one girl, but would have had more children if the first two were of the same sex.

These two subgroups are akin to a randomised treatment-control setup, which is why compliers play such an important role in identification: their treatment status is solely determined by a random factor, the sex of the first two children. But even if the instrument *Z* plausibly complies with Assumptions 5.1-5.4, resulting in a valid instrumental variable estimation, the relevance and generality of the results depends on the nature of the group of compliers. The plausibility of homogeneity assumptions, the limits of extrapolation of results from compliers, and their relevance as a subgroup of the population are all matters that need to be considered with respect to specific applications.

The first point to raise is whether compliers are per se an interesting subset of the population. This is certainly true in some applications: for instance, in the JTPA job training programme in the United States, individuals were offered labour market oriented courses through a random mechanism. While not all of those assigned attended training sessions, the assignment itself provides a good instrument in randomised experiments with imperfect compliance (Abadie et al., 2002; Abadie, 2003b). Compliers in the JTPA case were those who attended training sessions because they were offered the opportunity of doing so, but would not have attended in the absence of an offer.

The identified causal effect for this group consists in the change in outcomes induced by the offer of training, which is an interesting result for policy and for the evaluation of programmes of this type.

In the "same sex" case, however, compliers only represent individuals with a type of sex preference that modifies their fertility behaviour in a specific way. "Same sex" compliers do not represent an especially interesting group in terms of economic theory or policy as far as their complier status is concerned, because their "experimental treatment" is driven by genetics and not an economic programme. For this reason, "same sex" compliers have an advantage for the extrapolation of results to the whole population. In the job training example, the identified effect is interesting per se, but it does not represent the average effect of the programme for the overall population because compliers represent a particular subset, i.e. those who reacted positively to the offer of training. On the other hand, generalising the effects for compliers in the "same sex" case seems less controversial: while exhibiting stronger effects of sex preferences in their fertility decisions, there is no a priori reason to argue that they are substantially different from the rest of the population in terms of labour supply behaviour.

This Section presents and derives additional identification results which can be used to test assertions of this type: while the extrapolation of causal effects for compliers remains intrinsically untestable, as in the case of the exclusion restriction (see Section 6.3, 155), its plausibility can be assessed with auxiliary evidence.

8.2.2 Proportion of compliers in different samples

The extrapolation of results from compliers to the whole population cannot be directly verified: since they are defined in terms of their counterfactual treatment status, it is not possible to assign any individual to this group, and thus their characteristics cannot be observed. For this reason, it is not possible to say whether compliers are representative of the whole population, and thus parameters like the LATE coefficient defined in Proposition 5.1 cannot be extrapolated without further assumptions.

However, if it is plausible to assume that compliers are not intrinsically different from the rest of the population except in their behaviour with respect to the instrument, this homogeneity warranties that estimates like LATE represent the average causal effect for the overall population (and not only for compliers). In the "same sex" case, compliers are women who had more children because the first two where of the same sex: if they are deemed similar to other women in other dimensions, then their labour supply parameters can be extrapolated to the rest of the population.

While homogeneity cannot be tested directly, some of its implications can be verified, as in the discussion of the exclusion restriction in Chapter 6. For instance, if homogeneity holds, then compliers should represent constant fractions of the population among subgroups distinguished by some exogenous characteristic *X*. This can be verified by means of an additional identification result due to Imbens and Rubin (1997) for the case of binary treatments:

Proposition 8.1 *Proportion of compliers (Imbens and Rubin, 1997): if Assumptions 5.1, 5.2, 5.3 and 5.4 hold, then the proportion of compliers in the population is identified and given by:*

$$Pr(D_0 = 0, D_1 = 1) = Pr(D_i = 1 | Z_i = 1) - Pr(D_i = 1 | Z_i = 0)$$

= $E[D|Z = 1] - E[D|Z = 0]$ (8.1)

This result follows from the definition of the different groups by potential treatment in Section 5.2.2 (page 130): the fraction of individuals with $D_i = 1$ and $Z_i = 1$ estimates the combined population of always-takers (those who would always be treated, independently of the instrument) and compliers. Moreover, the group with $D_i = 1$ and $Z_i = 0$ is composed only of always takers: the monotonicity Assumption 5.4, which rules out the presence of defiers, implies that the proportion of compliers given in Proposition 8.1 can be obtained as a residual between these two proportions of the sample when *D* is binary.

The difference in Equation 8.1 is the denominator of the LATE-Wald estimator (right hand side of Equation 5.5, page 135), and it is equivalent to the coefficient of Z in a first-stage regression of D on a constant and Z.¹

The unbiased estimator of the proportion of compliers given in Equation 8.1 can be calculated for different subsets of the population defined by exogenous characteristics: the rationale for this is that if the proportions vary significantly among groups, the status of complier would be associated with the discriminating characteristic, weakening the case for extrapolation of results to the whole population. Section 8.3.2 carries out this analysis with samples defined by education level in Argentina.

8.2.3 Identification of outcomes for compliers

The analysis of proportions of compliers in different subpopulations is useful as a check of homogeneity assumptions, but it does not provide a formal test to support claims about extrapolation. An auxiliary test can be derived from additional identification results for compliers.

As described in detail in Section 5.2 (page 126), causality and the LATE estimate are defined in terms of differences in potential outcomes of the form $E[Y_1 - Y_0]$. However, the nature of compliers² allows for more general identification results: Abadie (2002, Lemma 2.1) shows that the whole marginal distribution of potential outcomes is identified for this group. In particular, the average potential outcomes are given by Equations 8.2 and 8.3 in the following Proposition:

```
Y^c \equiv Y | D_0 = 0, D_1 = 1, and its expectation is given by:
```

¹Abadie (2003b, Lemma 2.1) provides an additional formal proof and extends this result to the case of covariates X.

²To ease the reading of the Equations in this Chapter, the superscritp "c" is used to indicate a variable for the groups of compliers only, and the conditioning of an expectation to this group: for instance, the outcome for compliers is written as:

 $E^{c}[Y] \equiv E[Y^{c}] \equiv E[Y|D_{0} = 0, D_{1} = 1].$

Proposition 8.2 *Potential outcomes for compliers (Abadie, 2002): If Assumptions 5.1, 5.2, 5.3 and 5.4 hold, then:*

$$E[Y_0^c] = \frac{E[Y(1-D)|Z=1] - E[Y(1-D)|Z=0]}{E[(1-D)|Z=1] - E[(1-D)|Z=0]}$$
(8.2)

$$E[Y_1^c] = \frac{E[YD|Z=1] - E[YD|Z=0]}{E[D|Z=1] - E[D|Z=0]}$$
(8.3)

Proposition 8.2 allows the decomposition of the causal effect for compliers, the LATE $E^c[Y_1 - Y_0]$, into its two components. By providing the baseline upon which the causal effect operates, this result facilitates the study of heterogeneous effects in the population. An analysis of this type for Argentina is presented in Section 7.3.2.³

Finally, the results from Propositions 8.1 and 8.2 allow the derivation of a new identification result, central to the discussion of the extrapolation of results for compliers.

The following proposition is the main result of the Chapter. It builds on Abadie's (2003b) proof that the expectation of functions of Y, D and Zare identified for compliers. Using this result and Equations 8.2 and 8.3, the Proposition shows that beyond the counterfactuals, the actual expected outcome is identified for compliers:

Proposition 8.3 Identification of average outcomes for compliers: if Assumptions 5.1, 5.2, 5.3 and 5.4 hold, then:

$$E[Y^{c}] = E[Y_{1}^{c}] \Pr(Z=1) + E[Y_{0}^{c}] \Pr(Z=0)$$

$$= \frac{E[YD|Z=1] - E[YD|Z=0]}{E[D|Z=1] - E[D|Z=0]} - \frac{E[Y|Z=1] - E[Y|Z=0]}{E[D|Z=1] - E[D|Z=0]} E[1-Z]$$
(8.4)

The proof of Proposition 8.3 is based on the fact that the average outcome for compliers is a weighted sum of the potential outcomes for this group:

Proof. The expected outcome can be expressed as a weighted sum according to the values of the instrument *Z*: $E[Y] = E[Y|Z = 0] \Pr(Z = 0) + E[Y|Z = 1] \Pr(Z = 1)$. This is also true for compliers:

$$E[Y^{c}] = E[Y^{c}|Z=0] \Pr(Z=0) + E[Y^{c}|Z=1] \Pr(Z=1)$$
(8.5)

³The details for the computation of the parameters in Equations 8.2 and 8.3 and their standard errors are provided in Appendix E.

Since for compliers D = Z, the independence of Z implies that:

$$E[Y^c|Z=0] = E[Y_0^c|Z=0] = E[Y_0^c]$$
(8.6)

A similar argument shows that:

$$E[Y^c|Z=1] = E[Y_1^c|Z=1] = E[Y_1^c]$$
(8.7)

Replacing 8.6 and 8.7 into 8.5 gives:

$$E[Y^{c}] = E[Y_{1}^{c}] \Pr(Z = 1) + E[Y_{0}^{c}] \Pr(Z = 0)$$

Using the identification results for Y_0 and Y_1 for compliers (Equations 8.2 and 8.3, Proposition 8.2) in the previous Equation results in:

$$E[Y^{c}] = \frac{E[YD|Z=1] - E[YD|Z=0]}{E[D|Z=1] - E[D|Z=0]} \Pr(Z=1) + \frac{E[Y(1-D)|Z=1] - E[Y(1-D)|Z=0]}{E[(1-D)|Z=1] - E[(1-D)|Z=0]} \Pr(Z=0)$$

Since Z is a binary variable, Pr(Z = 0) = E[1 - Z] = 1 - E[Z] and Pr(Z = 1) = E[Z]. Using this and the fact that E[Y(1 - D)] = E[Y] - E[YD] and E[1 - D] = 1 - E[D], some manipulation of the previous Equation results in:

$$E[Y^{c}] = \frac{E[YD|Z=1] - E[YD|Z=0]}{E[D|Z=1] - E[D|Z=0]} - \frac{E[Y|Z=1] - E[Y|Z=0]}{E[D|Z=1] - E[D|Z=0]} E[1-Z]$$

which is the result of the proposition.

An alternative and conceptually equivalent proof can be constructed using Theorem 3.1 in Abadie (2003b). This Theorem states that under the LATE identifying assumptions the expectation of any function g(Y, D, X)is identified for compliers. Proposition 8.3 can be thought of as the case in which g(Y, D, X) = Y. The alternative proof uses this Theorem and the results from Proposition 8.2.

Intuitively, the proof states that the expected outcome is a weighted sum of the potential outcomes for compliers with different values of the instrument, since this group is akin to a randomised experiment and thus there is no self-selection bias.

This Proposition can be used to assess the plausibility of the homogeneity of compliers with respect to the population. It is possible to verify if the average outcome for compliers differs from that of the overall population, with a test given by:

$$\lambda = E[Y] - E[Y^c] \tag{8.8}$$

The idea is that the extrapolation of LATE results from compliers to the overall population is less credible if $\lambda = 0$ is rejected.⁴ In that case, compliers would differ significantly from the overall sample in a crucial aspect, namely the expectation of the outcome variable.

Equation 8.8 is an auxiliary test for extrapolation, and it can be conducted for different subgroups defined by observable characteristics. The following Section provides examples of this test for the model of labour supply in Argentina, and for the JTPA training programme mentioned above, since the two were identified as opposed cases with respect to the extrapolation of results for compliers in the discussion of this Section.

8.3 POTENTIAL OUTCOMES AND HETEROGENEOUS EFFECTS

8.3.1 Decomposition of LATE estimates by potential outcomes

The definition of a causal effect in Chapter 5 referred to the difference between observed and counterfactual outcomes. For instance, in the labour supply example, the causal effect of fertility is obtained by comparing the labour supply of women with three or more children with the labour supply of the same women if they had only two children. While IV coefficients like LATE concentrate on this difference, represented by parameters like ϕ in Equation 5.7, the identification results in Proposition 8.2 show that, subject to Assumptions 5.1-5.4, it is possible to study the counterfactual levels themselves. The study of causal effects can be complemented by the analysis of the full set of potential outcomes for compliers: the LATE and IV coefficients, which present the difference $\phi = E^c[Y_1 - Y_0]$, may be masking important differences in the levels of $E[Y_1^c]$ and $E[Y_0^c]$. The results below concentrate on the levels of $E[Y_1^c]$ and $E[Y_0^c]$, and constitute a complement of the IV estimates of the previous Chapter because they measure the baseline upon which the causal effect operates.

Table 8.1 presents the decomposition of the LATE coefficients in potential

⁴The details for the computation of the test in Equation 8.4 with regression software are provided in Appendix E.

Table 8.1: Effect of "More than two children" on "Worked for pay": Expected and Potential Outcomes by Education Level, Married Women

	Outcome variable:	A 1	Some	Some	Some
	Worked for pay	Overall	primary education	secondary education	tertiary education
	Overall outcomes:				
(1)	Expected outcome	0.305	0.214	0.301	0.681
(2)	Expected outcome with D=0	0.355	0.237	0.330	0.703
(3)	Expected outcome with D=1	0.267	0.201	0.272	0.653
(4)	OLS Estimate (3)-(2)	-0.088	-0.035	-0.058	-0.050
	Potential outcomes for compliers:				
(5)	Y ₀ : Potential outcome with D=0	0.350	0.245	0.340	0.780
	Standard error	(0.023)	(0.025)	(0.043)	(0.064)
(6)	Y ₁ : Potential outcome with D=1	0.248	0.125	0.212	0.753
	Standard error	(0.025)	(0.033)	(0.041)	(0.059)
(7)	Wald-LATE estimate (6)-(5)	-0.102	-0.120	-0.128	-0.026
	Standard error	(0.034)***	(0.042)***	(0.059)**	(0.086)
	Outcomes for compliers:				
(8)	Proportion in the sample	0.040	0.040	0.040	0.044
(9)	Expected outcome	0.298	0.184	0.275	0.766
	Standard error	(0.017)	(0.021)	(0.030)	(0.044)
(10)	P-value of the test (1)=(9)	0.699	0.158	0.380	0.060
Ob	servations	456,437	248,477	146,492	61,468

Note: All potential outcomes and OLS coefficients are different from zero at the 1% level. For the Wald estimate, superscripts denote significance as in previous tables. The samples correspond to married women aged 21-35 with two or more children aged 18 or younger from the extended questionnaire sample of the 1991 Census, as described in the text (*AE sample, married*).

outcomes for the same education groups as in Table 7.5 (page 187) for the *AE sample, married* (the results for other samples are quantitatively similar and qualitatively the same).

The first panel of the Table presents the unconditional expectation of the outcome variable, E[Y] (row 1), and its value conditional on the instrument Z, E[Y|D = 0] (row 2) and E[Y|D = 1] (row 3). The average in the *Worked for pay* indicator, 0.305, conceals important differences in the population. In the lowest educated group only a fraction 0.214 of the women work for pay, while in the middle education group this proportion is 0.301, just below the population average. For the relatively small group with some higher education, however, the average of the *Worked for pay* indicator is much higher at 0.681.

The Table also presents the expected outcomes conditional on D = 0 and D = 1 and the difference between the two, which is the simple OLS coefficient of D when this variable is regressed on Y, as presented in Table 7.1 (page 174) for all samples. As in that Table, the coefficients are all negative and strongly significant.

The second panel of the Table breaks down the LATE-Wald estimate $E^{c}[Y_{1} - Y_{0}]$ (row 7) into its components $E[Y_{0}^{c}]$ (row 5) and $E[Y_{1}^{c}]$ (row 6), which are identified for compliers (Proposition 8.2). These potential outcomes represent the counterfactual probabilities of employment under the two values of the treatment. The LATE coefficients, which represent the difference between $E[Y_{0}^{c}]$ and $E[Y_{1}^{c}]$, are relatively close in the 0.10-0.13 range (in absolute value) for the overall sample and the low and middle education groups.

The compliers in the middle education group exhibit a counterfactual probability of working of 0.34 if not having a third child, while their average labour supply would be only 0.212 if they did have a third child. While resulting in a similar difference in absolute terms, the potential outcomes for the low education compliers are 0.245 and 0.125, respectively, substantially lower than those in the middle group. The 0.12-0.128 difference in potential outcomes given by the LATE estimates gives the impression of a relatively homogeneous impact of childbearing on labour supply in these two education groups: while this is true in absolute terms, the impact is much larger for the low education group when relative to the baseline potential outcomes.

Finally, as with the expected outcome, the potential outcomes for the higher education group are much higher than for the population as a whole, but the Wald estimate is not significantly different from zero, which might indicate a lack of precision given the smaller sample size.

Since regression coefficients of binary indicators represent differences in the expectation of variables, evidence like that of Table 7.5 might overlook important disparities in their levels. The breakdown presented in Table 8.1 is an important complement of the analysis of heterogeneity in the impact of childbearing on women's labour supply, and provides a clear picture of the differences between the three subgroups.

8.3.2 Are compliers different? Extrapolation and homogeneity assumptions

The division of the sample by education group can also be exploited to address the issue of extrapolation of results for compliers by means of the additional identification results discussed in Sections 8.2.2 and 8.2.3.

The bottom panel of Table 8.1 presents estimates of the proportion of compliers (row 8) for the different subpopulations (Proposition 8.1), which are given by the first-stage coefficients of *Z* when this variable is regressed on *D*. Compliers represent about 4 percent of the *AE married* sample, with about the same proportion in the low and middle education groups, and 4.4 percent of the higher education group.

The Table shows that the proportion of compliers is uniform across subpopulations with large differences in the outcome variable and other observable characteristics. Moreover, this also indicates that the first-stage effect of *Same sex* on *More than two children* is roughly constant in the population (as discussed in Section 7.3.2, page 186), which supports the hypothesis that compliers are not substantially different from the rest of the population in the "same sex" application to Argentina and suggests that the LATE results can be extrapolated.

Further evidence in this regard is provided by row 9 in the bottom panel of Table 8.1, which presents the computation of the expected outcome for compliers $E[Y^c]$, and its standard error, as defined in Proposition 8.3. For the three groups, these expectations seem to be close to the average outcomes for the respective complete populations. This is formally confirmed by the results in row 10, which presents the p-value of the test derived from Proposition 8.3 and given by Equation 8.8. The hypothesis of equality of mean outcomes between compliers and the overall population cannot be rejected at the standard levels of significance for the complete sample nor for any of the three groups.

The intuition of Section 8.2.1 is right: compliers are important in the "same sex" application because they are not substantially different from the rest, and thus the causal effects identified for this group may be considered as representative of the whole population.

For comparison purposes, the test in Equation 8.8 is also computed with participants in the JTPA programme, since the discussion in Section 8.2.1 hy-

Table 8.2: Effect of Trai	ning on Earnings:	Expected and I	Potential Outcomes,
JTPA Partici	pants		

	Outcome variable:	Mon	XX7	
	Annual earnings (US dollars)	wien	women	
	Overall outcomes:			
(1)	Expected outcome	19,147	13,029	
(2)	Expected outcome with D=0	17,485	12,078	
(3)	Expected outcome with D=1	21,456	14,211	
(4)	OLS Estimate (3)-(2)	3,970	2,133	
	Potential outcomes for compliers:			
(5)	Y ₀ : Potential outcome with D=0	19,830	12,311	
	Standard error	710	419	
(6)	Y ₁ : Potential outcome with D=1	21,655	14,253	
	Standard error	624	385	
7)	Wald-LATE estimate (6)-(5)	1,825	1,942	
	Standard error	(945)*	(569)***	
	Outcomes for compliers:			
8)	Proportion in the sample	0.612	0.640	
(9)	Expected outcome	21,046	13,612	
	Standard error	(479)	(293)	
(10)	P-value of the test (1)=(9)	0.000	0.046	
Ob	servations	5,102	6,102	

Note: All potential outcomes and OLS coefficients are different from zero at the 1% level. For the Wald estimate, superscripts denote significance as in previous tables. The treatment D is attending training sessions. The samples correspond to participants in the JTPA training programme, as described in Abadie et al. (2002).

pothesised that compliers in that application were fundamentally different from other groups.

Table 8.2 presents the same set of statistics as in Table 8.1 for men and women participating in the JTPA experiment, using the data from Abadie et al. (2002). Based on a random selection procedure, two thirds of the participants in the programme were assigned to training courses, while a further third was not allowed to attend them. The treatment in this case is the training received, and a random, exogenous instrument is provided by the offer of training.⁵ Abadie et al. (2002) find that the programme had a positive and

⁵The instrumentation is needed because even in randomised experiments the identification by simple differences in means only holds if all individuals assigned to treatment are indeed treated, and those in the control group are completely excluded – that is, if

significant causal effect on women's earnings, but only a marginally significant effect for men. These results correspond to the Wald-LATE estimates in Table 8.2 (row 7).

The decomposition of the LATE estimate $E^{c}[Y_{1} - Y_{0}]$ in row 7 into its components $E[Y_{0}^{c}]$ and $E[Y_{1}^{c}]$, presented in the second panel of the Table (rows 5 and 6), shows that the similar causal effects for men and for women (\$1,825 and \$1,942 respectively) operate on very different baselines, with potential outcomes substantially lower for women than for men.

Finally, as discussed in Section 8.2.1, compliers in this randomised experiment can be expected to differ substantially from the overall population. The evidence in Table 8.2 supports this hypothesis: for both men and women, compliers exhibit a significantly higher level of expected earnings than the overall population, as revealed by test of equality of E[Y] (row 1) and $E^c[Y]$ (row 9) given by Equation 8.8. The results in row 10 state that this hypothesis can be rejected at the 5 percent level of significance for women and at the 1 percent level for men in the JTPA experiment.

The discussion of the results in Tables 8.1 and 8.2 illustrates the use of the additional identification results derived in Section 8.2. Besides the useful information provided by the decomposition of causal effects into counterfactual potential outcomes, the computation of the test in Equation 8.8 for two different samples provides further support to the conjecture that "same sex" compliers are not significantly different from the overall population. This is because the result of the test in Table 8.2 captures the intuition that compliers in JTPA programme are fundamentally different from other groups.

8.4 CONCLUSION

This Chapter studied the issue of heterogeneity in the effects of fertility on female labour supply, and presented additional identification results to test the generality of the benchmark results of Chapter 7.

The analysis of LATE estimates broken down by education groups revealed the presence of some heterogeneity in the effect of childbearing on labour supply by education level. By studying the potential outcomes for

compliance with the experimental protocol is perfect.

each subgroup, the negative causal effect of fertility on female labour supply was found to be stronger among the lowest educated group. Since these women had the lowest levels of labour force participation, the reduction in labour supply induced by an increase in fertility was found to be larger, in proportional terms, than for women in the middle education category. These results also confirmed the finding (presented in Chapter 7) that women in the higher education group were not significantly affected by fertility in their labour supply decisions. The analysis in terms of potential outcomes, however, provided an explanation of this result: the labour force participation of women with some tertiary education was found to be very high regardless of the number of children, and thus the reduction in labour supply induced by a third child was not significant.

Finally, regarding the generality of the baseline findings of Chapter 7 in terms of the causal effect of fertility on female labour supply in Argentina, the evidence indicates that "same sex" compliers are not substantially different from the overall population in some key characteristics. This implies that the benchmark IV estimates of the previous Chapter can be generalised and extrapolated from the group of compliers to the whole population. It is then possible to say, with some confidence, that the significant negative effect of fertility on women's labour supply is not confined to a small subgroup but that it is representative of the population average causal effect for women with two or more children.

The following pages review the main findings and contributions of the previous Chapters. The discussion focuses on the practical lessons on some key questions about the welfare evaluation of income dynamics in a time of crisis, and labour supply decisions and income prospects in Argentina. The two Sections below also draw attention to the policy implications that can be derived from the empirical results in Parts I and II of this thesis.

CONCLUSION TO PART I: POVERTY AND INCOME FLUCTUATIONS

Methodological contributions: accounting for income fluctuations

The Chapters in Part I focused on poverty as a dynamic phenomenon. The motivation was the recurring economic crises that affect developing countries and the incidence of the resulting income fluctuations on household welfare. Indeed, a striking finding from Chapter 1 was that relatively modest changes in poverty rates between two periods were the result of large but offsetting movements into and out of poverty in Argentina. Moreover, the results from Chapters 2 and 3 indicate that these large fluctuations of income over time have a significant negative effect on household welfare even during periods of aggregate growth.

The use of panels revealed some features of the data that could not have been captured with the cross-sectional datasets usually employed in poverty analysis. Chapters 1 to 4 went beyond the simple analysis of poverty transitions, and explored the theoretical basis for the incorporation of income fluctuations into the measurement of poverty and well-being over time.

Chapter 2 presented a general framework for the welfare evaluation of income dynamics, based on extensions of the existing methodology for sta-

tic distributional analysis. An analogy with the concept of risk aversion provided a rationale for the incorporation of income fluctuations into the measurement of household well-being. This welfare criterion was rationalised using intuition derived from the risk literature: households prefer a steady stream of income to a variable one with the same mean, at least in a secondbest world with incomplete insurance and capital markets. The indicators derived from this evaluation framework accounted not only for the level of poverty, but also for the effects of income variability on welfare.

Moreover, existing alternative approaches for measuring well-being based on panel data, such as the literatures on transient-chronic poverty and on vulnerability, were interpreted as special cases of this evaluation framework. The following two Chapters presented variations of these special cases. Chapter 3 introduced a simple methodology to derive risk adjusted measures of income, and Chapter 4 analysed poverty over the same period based on a decomposition of a household's intertemporal poverty into its transient and chronic components. In both Chapters, regressions were used to identify a number of characteristics related to the income risk and the persistence of poverty faced by Argentine households.

These methodologies were illustrated with a panel dataset from the Greater Buenos Aires region for the 1995-2002 period, but the analysis carried out in Chapters 2, 3 and 4 can be applied to other contexts where panel data on household income or expenditure is available. The following pages deal with the main empirical findings and their policy implications for Argentina.

Empirical findings: fluctuations matter

The empirical findings of Chapters 1 to 4 imply that income fluctuations matter in at least four important dimensions.

The first dimension refers to the nature and extent of poverty analysis. As mentioned above, Chapter 1, which documented different aspects of household welfare over the 1995-2002 period in Argentina, found that changes in poverty rates between two periods were the result of large offsetting movements into and out of poverty. Moreover, these movements, which were not apparent in the simple analysis of changes in poverty rates between two periods, were not confined to economic crises: a substantial

fraction of the population was found to enter poverty even when rates were falling on aggregate.

The high proportion of individuals who changed poverty status in a relatively short period of time (about six months in the Argentine data), and the fact that these changes occurred in all stages of the business cycle, imply that traditional poverty studies based on cross-sectional data might be missing some fundamental information.

The second dimension refers to the relative importance of income fluctuations for household well-being. The framework developed in Chapter 2 provided a rationale, based on an analogy with the concept of risk aversion, for imputing a negative impact of fluctuations on welfare. However, the magnitude of this effect is an empirical question.

The evidence for Argentina in Chapters 2 and 3 demonstrated that income fluctuations substantially reduced household welfare under relatively mild assumptions. There is, however, a trade-off: when income observations over time are aggregated at the household level, welfare measures increase and poverty evaluations decrease when compared to indices based on punctual observations. This is because the averaging mitigates the impact of negative shocks. This smoothing effect, however, was more than offset once the disutility from income fluctuations was taken into account, assuming only moderate levels of risk aversion in line with most estimates of the uncertainty literature.

Most importantly, the sizeable effects of fluctuations on welfare and poverty were not limited to periods of crisis or downturns. The findings from Chapter 2 and 3 indicate that income fluctuations at the household level have substantial effects on well-being even during periods of aggregate growth, for instance during the 1996-1998 period in Argentina (see Figures 1.2 and 1.3, pages 21 and 30 respectively). This result reflects the finding that a substantial fraction of the population entered poverty even when aggregate rates were falling.

The third dimension refers to the effects of an economic crisis from a dynamic perspective. The empirical results in Part I indicate that major macroeconomic shocks, like the 2001-2002 crisis in Argentina, not only reduce income levels, but also increase income risk, which magnifies their overall negative impact on poverty and well-being. Moreover, the decomposi-

tion of a household's intertemporal poverty into its transient and chronic components, presented in Chapter 4, found that the chronic component accounted for most of the large increase in poverty over the period. While income levels fell and their variability increased, the deterioration in living standards had a larger effect than the rise in fluctuations during this turbulent period.

This evidence on the increase of chronic poverty is compatible with the results on short-term dynamics presented in Chapter 1: over the 1995-2002 period, the fraction of the poor population staying in poverty increased almost constantly, while the proportion of those escaping poverty between two periods fell significantly.

Finally, the fourth dimension refers to the heterogeneity of the effects of income fluctuations on household welfare. The increase in chronic and transient poverty and the fall in risk adjusted income was not constant among households: the regression analyses in Chapters 3 and 4 identified some of the characteristics associated with income risk and poverty variability in Argentina.

The first result that stands out from both Chapters is that the education levels of the head of household and of the spouse were systematically correlated to lower levels of income risk, and of transient and chronic poverty. Households with better-educated members were found to have higher incomes and to face lower income risk, and all education and qualification indicators were associated with lower measures of intertemporal poverty (with no education and no qualification as the excluded category).

The larger effects in the income and poverty regressions, however, were related to labour market variables. Households with inactive or unemployed members bore a larger share of income risk than any other group. The indicators for relatively stable sources of income, such as pensions and public sector jobs, had a negative impact on transient poverty, whereas the opposite was true for variables reflecting more precarious jobs: the informality indicators contributed to higher levels of risk and poverty, and being self-employed was found to have a positive effect on transient poverty. Finally, the demographic variables indicated that households with children were more prone to suffer from income risk, and experienced higher levels of intertemporal, chronic and transient poverty. The following Section

derives some policy lessons from these results.

Policy implications: safety nets for rainy (and sunny) days

These household characteristics allow us to derive some recommendations for public interventions in times of crisis. In broad terms, however, these results confirm the policy consensus in Latin America on the importance of labour market outcomes and family composition as strong predictors of poverty (de Ferranti et al., 2000a). This consensus was reflected in the two major public intervention programmes in Argentina, reviewed in Chapter 1: the *Plan Trabajar*, introduced in the wake of high unemployment levels after the Mexican financial crisis of 1995, and the *Plan Jefes y Jefas de Hogar Desempleados*, implemented during the crisis of 2001-2002.

The designs of both programmes took into account, implicitly or explicitly, the household characteristics identified in the previous Section. *Trabajar* provided short-term jobs at low wages, usually related to public works in poor areas (Ravallion and Jalan, 2003). For practical purposes, *Trabajar* operated as a poverty alleviation programme targeted through unemployment (de Ferranti et al., 2000b), and thus recognised labour market outcomes as proxies of poverty. In the *Plan Jefes y Jefas*, unemployed heads of households with dependent children received direct income support (Galasso and Ravallion, 2003). The programme design took into account the importance of labour market outcomes and also the demographic composition of the household as proxies for vulnerability.

At first glance, then, these results on the household characteristics associated with higher levels of income risk and poverty variability do not seem particularly original. The main point, however, is that the regression analyses in Chapters 3 and 4 included years of crisis but also periods of recovery. Labour market outcomes, education levels and family structure were correlated with the substantial income fluctuations observed during the whole 1995-2002 period in Argentina.

While the importance of dealing with the effects of aggregate shocks has long been recognised, the main conclusion from Chapters 1-4 is that safety nets and other social protection mechanisms, while vital during major crises, should also be implemented on a continuous basis, irrespective of the short term evolution of macroeconomic aggregates.

This conclusion is based on the empirical findings of Part I on the incidence of income fluctuations on household welfare. While the results indicate that major macroeconomic shocks substantially reduce income levels and increase income risk, the detrimental effect of income fluctuations at the household level was also found to be significant during periods of stability or recovery. Moreover, irrespective of changes in GDP, during the whole 1995-2002 period in Argentina a substantial proportion of the population entered poverty between two periods of time, even when poverty rates were falling, and a significant proportion of observed poverty was attributed to its transient component.

This conclusion is reinforced by recent figures that have uncovered some hysteresis of poverty in Argentina: the strong recovery in GDP growth observed in 2003 (see Figure 1.2, page 21) has reduced unemployment levels, but poverty rates are not falling as fast (INDEC, 2004). The implication for future interventions is that it is as important to insure households against income risk and to avoid entry into poverty on a continual basis as it is to provide coping mechanisms during future crises.

Given the importance of labour market outcomes for poverty dynamics and income risk, the design of a permanent social protection scheme in Argentina could draw on relatively successful welfare systems that manage to balance work incentives with income support in developed countries (Gregg and Wadsworth, 1999b, review the recent experience for the United Kingdom). However, the peculiarities of labour markets in developing countries introduce further complications. For instance, the existing unemployment insurance programme in Argentina only covers a fraction of the population in formal employment, the least vulnerable group: beneficiaries tend to be more numerous among middle and upper-middle income groups, as in most Latin American countries with programmes of this type (de Ferranti et al., 2000b). The challenge resides in designing unemployment insurance and other permanent social protection policies within economies with high levels of informality in the labour market. In the following pages, the Conclusion of Part II discusses the impact of fertility on female labour supply, and draws attention to the importance of childcare and early schooling programmes for the income prospects of poorer households.

Beyond the need for short term emergency interventions in times of crisis and for long term insurance mechanisms, the evidence in Chapters 3 and 4 confirms the importance of education and qualifications for poverty alleviation in the long run. Moreover, the results of this thesis suggest that human capital investments pay off in terms of increased returns but also in lower levels of risk. In this area, public interventions are also concentrated on periods of crises: for instance, the *Plan Jefes y Jefas* introduced some conditionality on school attendance for the children of beneficiaries. While primary school enrolment rates are relatively high in Argentina (see Table 6.3, page 164), the much needed long term initiatives on education (for instance, in terms of incentives for completion of secondary schooling) are scattered among different levels of government.

Finally, the results in Part I indicate that the design of long term policies for social protection must draw both on traditional static poverty profiles and on studies of income dynamics. Panel datasets, though not without their problems, provide vital information about the underlying movements that result in aggregate poverty changes, and thus their collection should be given a higher priority within statistical agencies in developing countries.

CONCLUSION TO PART II: FERTILITY AND WOMEN'S LABOUR SUPPLY

Methodological contributions: female labour supply, fertility and endogeneity in developing countries

The Chapters in Part II discussed the determinants of the income generating process at the household level in Argentina, focusing on the potential causal effect of fertility on female labour supply. This research was motivated by two main factors. The first consisted of the results of Part I on the correlations between poverty and fertility, on the one hand, and poverty and labour market indicators, on the other. The second factor was the significant fall of total fertility rates and the rise in female labour force participation in Argentina in the last twenty years, discussed in Chapter 5.

While the negative correlation between childbearing and labour force participation is an almost universal phenomenon, its interpretation in terms of causality is often marred by self-selection, endogeneity and simultaneity issues. The main results of Part II (presented in Chapter 7) were significant because they established the causal nature of this correlation in Argentina.

The methodological contributions of the Chapters in Part II were all related to the endogeneity problems that arise in models of fertility and female labour supply, and on how to deal with them with the "same sex" identification strategy in the context of developing countries. This strategy is based on the observation that parents of two children of the same sex exhibit a higher propensity to have another child to obtain a gender-balanced sibling composition. Under certain conditions, analysed in Chapters 5, the sex mix of children can thus be used as an instrument for further childbearing.

Chapter 6 provided a detailed discussion of the identification strategy and its application to developing countries, where female participation is lower and fertility is higher than in the developed economies. The Chapter presented evidence on the effects of parental sex preferences on fertility, the first of its type for Argentina,¹ and on the exogeneity of this relationship with respect to labour supply. Moreover, it presented a study of parental demographic and expenditure data, which failed to reveal any significant bias against girls in Argentine households. Based on these results, Chap-

¹Personal communication, Georgina Binstock and Alejandra Pantelides, Centro de Estudios de Población (CENEP), Buenos Aires, Argentina.

ter 7 presented the main instrumental variable estimates of the causal effect of fertility on female labour supply. Finally, Chapter 8 made two methodological contributions. On the one hand, it presented a test for the generality of instrumental variable estimates, establishing the conditions under which these estimates can be considered representative of the population under study. On the other hand, it presented an original decomposition of IV coefficients, which was applied to the study of the effects of fertility by education level in Argentina.

Empirical findings: childbearing matters

The main conclusion from Part II, presented in Chapter 7, is that in Argentina children significantly reduce the labour supply of their mothers. Using the estimated coefficients, the fall in fertility between 1980 and 2000 accounted for almost a third of the large increase in female labour force participation during the same period.

This finding was further qualified by the analysis conducted in Chapter 8, which divided mothers into three groups according to their education levels – those with some primary, some secondary or some higher education.

The participation rate among women in the higher education group was just below 70 percent, higher than the overall averages in developed countries with high female participation like the United Kingdom and the United States (ILO, 2003). For this group of women, the reduction in labour supply induced by childbearing was small and not significantly different from zero. Moreover, women in this group were found to have a much higher propensity towards employing live-in domestic help, which was interpreted as a proxy for childcare services consumption.

Meanwhile, the causal effect of childbearing was found to be stronger among women with lower education levels than in any of the other two groups. Women in poorer households (as proxied by the education level) experienced a large and significant reduction in labour supply: a third child implied a reduction of about 12 percentage points in participation. The analysis in Chapter 8 added a new dimension to this finding: it revealed that this effect operated at already low levels of participation. Those 12 percentage points represented a counterfactual reduction in participation from

24.5 to 12.5 percent, both lower than the average of 30.5 percent for women in the three groups.

These results provide a rationale for the correlations between household size and poverty, and poverty and labour market outcomes. The sizeable causal effect of children on labour supply and the fact that it was stronger for women with lower education levels support the hypothesis that childbearing hampers crucial income-generating opportunities, which contributes to poverty at the household level. This conclusion is backed by alternative data sources: for instance, the May 1991 EPH indicates that 39.9 percent of households with women aged 21-35 with two or more children were poor, compared to 31.1 percent for the whole population.² Imputing the average income of poor women who worked for pay to poor women who did not work for pay reduces the incidence of poverty to 23 percent, a decrease in the rate of more than 40 percent. While this is an upper bound of the effect of fertility on poverty, it highlights the importance of women's labour force participation for the welfare of the household: even if the majority of these women were second earners, their income would still be decisive in a society where one average full-time salary does not always cover the basic needs of a typical family. These back of the envelope calculations are consistent with other sources. For instance, Gasparini and Marchionni (2003) perform microsimulation exercises and find that the reduction of family size among poorer households in the Greater Buenos Aires area in the 1990s reduced poverty, a factor mainly driven by the "weakening of the relationship between the hours of work and the number of children in the case of female spouses."

Policy implications: childcare provision and social protection

The policy implications of this analysis are far-reaching. The robust causal link between fertility and female labour supply in Argentina rules out other explanations of the observed correlation. Most notably, the issue of selfselection, often invoked to prove the futility of policy interventions in this

²This dataset contains women aged 21-35 with two or more children, as described in Appendix D (footnote 2, page 238). The poverty figures, based on equivalised household income, were constructed with the Greater Buenos Aires poverty line, since official national poverty lines are not available for 1991. See the discussion in Section 1.2 (page 22).
area, can be discarded. In this context, self-selection implies that women with low expected labour market outcomes choose to specialise in housework and childbearing. If the observed correlation is due to self-selection, policies promoting work would fail because of the nature of the target group. The results of Part II, however, established that this is not the case in Argentina: the analysis of counterfactuals implies that a significant fraction of non-working mothers would work with lower fertility – or with better ways to accommodate childbearing and work. Since fertility levels are relatively low and falling, and family planning is freely available and well developed in Argentina, policy should concentrate on accommodating childbearing and work.

There is thus scope for policy interventions to promote work among women with children, especially considering the evidence on a stronger effect of fertility on the female labour supply of poorer women. In broad terms, the policy objective is to reduce the size of the coefficient linking fertility and female labour supply.

Recent policy developments in some industrialised countries, such as the working and family tax credits in the United States and the United Kingdom, focus specifically on improving work incentives for women and families with children. Initiatives such as the New Deal for Lone Parents have shown significant effects on employment levels among women in the UK (Dickens et al., 2004). While the scope for applying policies of this type is limited in countries lacking a comprehensive welfare system like Argentina, there still exist ways to ease the labour market participation of disadvantaged groups.

One option for reducing the negative link between fertility and female labour supply, especially for poorer women, is to provide free or low-cost childcare. In Argentina, the State only offers preschool facilities for children aged three to five, with no free childcare and no subsidies for newborns and children aged one or two (Cortés, 2003). The evidence presented in Part II suggests that a better supply of nurseries and day-care centres would improve the situation of the poorer sector of the population by easing the transition towards work.

The analysis presented in Chapters 5-8 also has important implications for market-oriented reforms in developing countries. Labour market re-

CONCLUSIONS

form in Latin America has been associated with the reduction of job security and of other forms of labour protection (Heckman and Pages, 2003), without substantial improvements so far: one of the main conclusions of an IADB (2004) report on the region is that "structural reforms did not produce the changes that were expected in labour markets." The wave of reforms of the 1990s concentrated on market deregulation, but neglected other aspects of labour market flexibility. For instance, the evidence presented in Part II suggests that poorer households could benefit from measures that increase the opportunities for part-time work, which is relatively infrequent in Argentina, and from the promotion of flexible work arrangements for parents.

Nevertheless, these policies are second order considerations with respect to macroeconomic stability and growth: an increase in labour supply might not be met by a labour demand in a context of high unemployment, as observed in Argentina over the 1999-2002 period. Appendices and Bibliography

APPENDIX A

ABBREVIATIONS AND NOTATION

Table A.1: List of Abbreviations

INDEC:	Instituto Nacional de Estadísticas y Censos, Argentina.
GBA:	Greater Buenos Aires region.
US:	United States.
EPH:	Encuesta Permanente de Hogares.
ENGH:	Encuesta Nacional de Gasto de los Hogares.
OLS:	Ordinary Least Squares.
2SLS:	Two-stage least squares.
IV:	Instrumental Variables.
LATE:	Local Average Treatment Effect (Proposition 5.1, page 135).
SUR:	Seemingly Unrelated Regressions (Equation 3.8, page 88).
FGT:	Foster et al.'s (1984) class of poverty measures (Equation 1.6, page 35).
CPI:	Consumer Price Index.

Table A.2: Main Variables: Definitions and Usual Notation, Part I

Y it	Total equivalised household's i income, normalised by the
-	Configuration 1.5, page 27).
<i>q_{ij}</i>	Coefficient of equivalent adult for member j in nousehold i
	(lable 1.3, page 26).
X_i	Household i's characteristics.
z_t, z_t^I	Poverty line and extreme ("indigent") poverty line at time <i>t</i> .
\mathbf{y}_i	Vector of observed incomes over time $[y_{i1},, y_{iT}]$ for <i>i</i> .
\bar{y}_i, σ_i	Average and standard deviation of income y_{it} over the period
	t = 1,, T.
Т	Total number of periods over which a household is observed
	(t = 1,, T).
P(y,z)	Poverty measure as a function of income y and poverty line z .
น,ับ ์	Household's ex-post and expected utility function defined
	over an income aggregate.
v, V	Evaluation functions defined over an income aggregate.
ρ	Arrow-Pratt coefficient of risk (or variability) aversion.
Ŷse	Stability equivalent income, defined by the evaluation func-
U	tion V and derived from observed incomes $[y_{i1},, y_{iT}]$.
π_v, Π	Variability and proportional variability premia for <i>i</i> , defined
	by Equations 2.8 and 2.9 respectively (page 61).
Note:	A simple variable (y) designates a fixed quantity, a variable with
	a hat (\hat{y}) indicates an ex-ante prospect (a random variable), and a
	variable with a wiggle ($ ilde{y}$) denotes a (fixed) counterfactual quantity.
	Finally, a bar as (\bar{y}) indicates the average of the underlying variable.
	When not required, the subscripts i and t are dropped for ease of
	exposition.

Table A.3: Usual notation, Part II

Υ: Outcome variable: labour supply. X: Set of individual characteristics. D: Endogenous treatment variable. Z: Instrument for the endogenous treatment variable *D*. *Y*₀, *Y*₁: Y_z represents the potential treatment status given Z = z: $Y_z =$ Y_0 if Z = 0 and $Y_z = Y_1$ if Z = 1. D_z represents the potential treatment status given Z = z: D_0, D_1 : $D_z = D_0$ if Z = 0 and $D_z = D_1$ if Z = 1. Note: To ease the reading of the Equations in this Part, the superscritp "c" is used to indicate a variable for the groups of compliers only, and the conditioning of an expectation to this group: for instance, the outcome for compliers is written as $Y^c \equiv Y | D_0 = 0, D_1 = 1$ and its expectation as $E^{c}[Y] \equiv E[Y^{c}] \equiv E[Y|D_{0} = 0, D_{1} = 1].$

APPENDIX B

ARGENTINA'S STATISTICAL REGIONS AND URBAN AREAS

B.1 REGIONS AND URBAN AREAS

The map in this Appendix (Figure B.1) depicts the 23 provinces and one autonomous city (Buenos Aires) of Argentina and their division into six statistical regions as defined by INDEC (1996). Each of these regions contains the following urban areas:

Greater Buenos Aires (GBA) region: Ciudad de Buenos Aires, Partidos del Conurbano.

Noroeste region: San Salvador de Jujuy-Palpalá, Gran Catamarca, Santiago del Estero-La Banda, La Rioja, Salta, Gran Tucumán.

Noreste region: Formosa, Gran Resistencia, Corrientes, Concordia, Posadas, Gran Paraná.

Pampeana region: Gran Córdoba, Gran La Plata, Mar del Plata, Río Cuarto, Bahía Blanca-Cerri, Gran Rosario, Gran Santa Fe, Santa Rosa-Toay.

Cuyo region: Gran Mendoza, San Luis, Gran San Juan.

Patagonia region: Río Gallegos, Ushuaia, Comodoro Rivadavia, Neuquén-Plottier.



Figure B.1: Map of Argentina's Statistical Regions, INDEC

222

Ponulation	Pov	erty and I	ncome	Labour	Market I	Qualifications		
Covered by EPH	Total Poverty	Extreme Poverty	Total Household Income	Active Popu- lation	Unem- ployment	Informa- lity	No Job Qualifi- cation	Illiteracy
2,225,209	25.9%	6.9%	1,124	59.9%	13.4%	34.4%	24.4%	1.4%
2,109,612	42.4%	11.9%	825	52.9%	12.3%	41.9%	29.6%	3.3%
1,140,157	50.3%	17.6%	725	51.7%	9.2%	42.3%	28.1%	3.7%
1,445,808	32.7%	7.7%	852	52.2%	6.1%	42.8%	29.6%	2.3%
5,255,024	29.3%	8.5%	843	54.0%	12.7%	37.9%	24.0%	2.1%
591,652	22.4%	8.0%	1,270	60.1%	10.6%	32.6%	24.7%	2.2%
2,767,462	29.9%	8.4%	998	57.1%	12.5%	36.5%	25.2%	1.9%
	Population Covered by EPH 2,225,209 2,109,612 1,140,157 1,445,808 5,255,024 591,652 2,767,462	Population Total Povered by EPH Total Poverty 2,225,209 25.9% 2,109,612 42.4% 1,140,157 50.3% 1,445,808 32.7% 5,255,024 29.3% 591,652 22.4% 2,767,462 29.9%	Population Total Poverty Extreme Poverty 2,225,209 25.9% 6.9% 2,109,612 42.4% 11.9% 1,140,157 50.3% 17.6% 1,445,808 32.7% 7.7% 5,255,024 29.3% 8.5% 591,652 22.4% 8.0% 2,767,462 29.9% 8.4%	Population Total Poverty Extreme Poverty Total Household Income 2,225,209 25.9% 6.9% 1,124 2,109,612 42.4% 11.9% 825 1,140,157 50.3% 17.6% 725 1,445,808 32.7% 7.7% 852 5,255,024 29.3% 8.5% 843 591,652 22.4% 8.0% 1,270 2,767,462 29.9% 8.4% 998	Population EPH Total Poverty Extreme Poverty Total Household Income Active Popu- lation 2,225,209 25.9% 6.9% 1,124 59.9% 2,109,612 42.4% 11.9% 825 52.9% 1,140,157 50.3% 17.6% 725 51.7% 1,445,808 32.7% 7.7% 852 52.2% 5,255,024 29.3% 8.5% 843 54.0% 591,652 22.4% 8.0% 1,270 60.1% 2,767,462 29.9% 8.4% 998 57.1%	Population Total Poverty Extreme Poverty Total Household Income Active Popu- lation Unem- ployment 2,225,209 25.9% 6.9% 1,124 59.9% 13.4% 2,109,612 42.4% 11.9% 825 52.9% 12.3% 1,140,157 50.3% 17.6% 725 51.7% 9.2% 1,445,808 32.7% 7.7% 852 52.2% 6.1% 5,255,024 29.3% 8.5% 843 54.0% 12.7% 591,652 22.4% 8.0% 1,270 60.1% 10.6% 2,767,462 29.9% 8.4% 998 57.1% 12.5%	Population EPH Total Poverty Extreme Poverty Total Household Income Active Popu- lation Unem- ployment Informa- lity 2,225,209 25.9% 6.9% 1,124 59.9% 13.4% 34.4% 2,109,612 42.4% 11.9% 825 52.9% 12.3% 41.9% 1,140,157 50.3% 17.6% 725 51.7% 9.2% 42.3% 1,445,808 32.7% 7.7% 852 52.2% 6.1% 42.8% 5,255,024 29.3% 8.5% 843 54.0% 12.7% 37.9% 591,652 22.4% 8.0% 1,270 60.1% 10.6% 32.6% 2,767,462 29.9% 8.4% 998 57.1% 12.5% 36.5%	Population EPH Total Poverty Extreme Poverty Total Household Income Active Popu- lation Unem- ployment Informa- lity No Job Qualifi- cation 2,225,209 25.9% 6.9% 1,124 59.9% 13.4% 34.4% 24.4% 2,109,612 42.4% 11.9% 825 52.9% 12.3% 41.9% 29.6% 1,140,157 50.3% 17.6% 725 51.7% 9.2% 42.3% 28.1% 1,445,808 32.7% 7.7% 852 52.2% 6.1% 42.8% 29.6% 5,255,024 29.3% 8.5% 843 54.0% 12.7% 37.9% 24.0% 591,652 22.4% 8.0% 1,270 60.1% 10.6% 32.6% 24.7% 2,767,462 29.9% 8.4% 998 57.1% 12.5% 36.5% 25.2%

Table B.1: Poverty, Income and Labour Market Indicators by Region, Argentina, October 1998

Source: Author's estimations based on EPH household survey data (INDEC). Household income is in current pesos. Active labour market status is defined as being 15 years or older, employed or actively looking for a job. The unemployment, informality and no job qualification indicators refer to active individuals. The illiteracy rate refers to individuals aged 25 and older.

B.2 REGIONAL HETEROGENEITY

INDEC (1996) describes the sampling frame for the EPH, and divides the 28 urban areas covered by the survey in the six regions depicted in Figure B.1. These six regions are highly heterogeneous from a socioeconomic perspective, as illustrated by indicators presented in Table B.1.

The Greater Buenos Aires is by far the largest, with more than 53 percent of the population covered by EPH. The following in size is the Pampeana region: its urban areas covered by EPH contain more than 5.2 million inhabitants. The Noroeste, Cuyo and Noreste regions cover between 1.1 and 2.1 millions each. Finally, the four urban areas of the Patagonia region included in the EPH sampling frame cover less than 600,000 inhabitants.

Most of the regional heterogeneity is captured by the differences in poverty rates. While almost 29.9 percent of the population in all urban areas covered by EPH in October 1998 lived below the poverty line, the rate was much higher in the northern regions (50.3 percent for the Noreste and 42.4 for the Noroeste), around the average in the Cuyo and Pampeana regions (32.7 and 29.3 percent, respectively), and below the national figure in the GBA (25.9 percent) and Patagonia regions (22.4 percent). The ranking in terms of extreme poverty (third column) is similar, and both partially reflect the large regional variation in average total household income (fourth col-

223

umn), from 725 pesos in the Noreste region up to 1,270 pesos in Patagonia.

In terms of labour market indicators, the three richer regions (GBA, Patagonia and Pampeana) have higher rates of activity and lower rates of informality,¹ but also the highest rates of unemployment. Of the three poorest regions (Noroeste, Noreste and Cuyo, only the Noroeste has unemployment levels similar to the national average of October 1998. In terms of qualifications, the three poorest regions have the higher levels of active workers with no job qualifications, at 28.1-29.6 percent of the working populations, compared to around 24-24.7 percent for the three regions with higher income. Finally, illiteracy is lower in the GBA region (1.4 percent of those aged 25 or older), with much higher levels in the Noroeste and Noreste regions (3.3 and 3.7 percent respectively).

For further references, SIEMPRO (2003) presents a broader set of indicators of human development, covering demographics, health and education, by region, province and urban areas. INDEC (2003a) also covers some of these aspects, and contains a thorough analysis of the different sectors of the economy in every province.

¹See the discussion of informality in footnote 5, page 95.

APPENDIX C

DETAILED POVERTY FIGURES, ARGENTINA 1995-2002

Tables C.1 and C.2 present detailed prices, poverty lines and poverty measures as described in Chapter 1. The price index and the official poverty lines are based on INDEC (2002; 2003a).

Tables C.3-C.6 present some characteristics of the population broken down for the non-poor, the moderately poor and the extremely poor for four of the fifteen periods under study. These descriptive statistics are for illustration only - for detailed poverty profiles, see Cruces and Wodon (2003a) and World Bank (2000a; 2001; 2003). The Tables correspond to the following four waves: October 1995 (corresponding to a recession period), October 1998 (peak of the recovery), October 2001 (after three years of recession), and May 2002 (a few months after the crisis of 2001-2002). The characteristics of the households presented in these tables include (a) household level variables, including the number of babies, children, adults, and elderly household members, and their square; whether the household head has a spouse; whether the household head is a woman; indicators for the age and the migration status of the head (in the last five years); (b) characteristics of the household head, including his/her level of education; whether he/she is unemployed or inactive; whether he/she is an employer, a self-employed worker, or a wage worker; the type of his/her qualification, and whether he/she works in the public/is an informal worker; and (c) the same set of characteristics for the spouse of the household head, when there is one. Geographic indicators for five of the six regions are reported, with the GBA area as reference category.

			,		J			0	,		0				
Consumer Price Index	May 95	Oct. 95	May 96	Oct. 96	May 97	Oct. 97	May 98	Oct. 98	May 99	Oct. 99	May 00	Oct. 00	May 01	Oct. 01	May 02
Base 9/2001=100	101.9	102.0	101.7	102.2	102.3	102.8	103.5	104.0	102.8	101.9	101.7	101.1	101.4	100.0	120.1
Inflation	-	0.14%	-0.35%	0.56%	0.09%	0.48%	0.68%	0.42%	-1.15%	-0.89%	-0.19%	-0.51%	0.28%	-1.41%	20.05%
Implicit poverty line deflactor, GBA	-	1.96%	0.08%	0.96%	-0.45%	1.30%	1.36%	0.89%	-3.00%	-0.89%	-1.32%	-1.19%	2.12%	-2.72%	29.09%
Description	CPI provi	ded by INI	DEC (2003)	. Values in	real terms	throughout	the paper a	are based o	n Sept. 200	1 prices. Tl	he implicit	poverty lin	e deflactor	is obtained	from
Nominal Poverty line	May 95	Oct. 95	May 96	Oct. 96	May 97	Oct. 97	May 98	Oct. 98	May 99	Oct. 99	May 00	Oct. 00	May 01	Oct. 01	May 02
GBA	151.7	154.7	154.8	156.3	155.6	157.6	159.8	161.2	156.4	155.0	152.9	151.1	154.3	150.1	193.8
NOROESTE	131.4	133.9	134.0	135.3	134.7	136.5	138.3	139.5	135.3	134.1	132.4	130.8	133.6	130.0	167.6
NORESTE	134.6	137.3	137.4	138.7	138.1	139.9	141.8	143.0	138.7	137.5	135.7	134.1	136.9	133.2	171.8
CUYO	132.6	135.2	135.3	136.6	136.0	137.8	139.6	140.9	136.6	135.4	133.6	132.1	134.9	130.8	168.7
PAMPEANA	137.1	139.8	139.9	141.3	140.6	142.4	144.4	145.7	141.3	140.0	138.2	136.5	139.4	135.9	175.2
PATAGONIA	144.5	147.3	147.4	148.9	148.2	150.1	152.1	153.5	148.9	147.6	145.6	143.9	146.9	142.7	183.6
GBA-Real	148.9	151.6	152.3	152.9	152.1	153.3	154.3	155.0	152.1	152.1	150.4	149.4	152.1	150.1	161.4
Description	Poverty li	nes in nom	inal terms (italics indi	cate officia	l values). N	lote: GBA	values only	are official	I (INDEC, 2	2002) for th	ne whole pe	riod.		
	Regional	values are	INDEC's fo	r 2001 and	2002, and	constructed	l for 1995-2	2000 apply	ing the GBA	A poverty l	ines rate of	change to	regional Ma	ay 2001 val	lues. See
Extrene Poverty Line	May 95	Oct. 95	May 96	Oct. 96	May 97	Oct. 97	May 98	Oct. 98	May 99	Oct. 99	May 00	Oct. 00	May 01	Oct. 01	May 02
GBA	64.8	66.1	65.9	67.4	65.4	67.4	68.3	69.8	66.0	64.6	62.9	62.4	63.2	61.0	81.8
NOROESTE	57.1	58.3	58.1	59.4 ·	57.6	59.4	60.2	61.5	58.1	56.9	55.5	55.0	55.7	53.7	72.0
NORESTE	58.3	59.5	59.3	60.6	58.8	60.6	61.4	62.8	59.4	58.1	56.6	56.2	56.9	54.8	73.4
CUYO	58.3	59.4	59.2	60.6	58.8	60.6	61.4	62.7	59.3	58.1	56.6	56.1	56.9	54.5	73.0
PAMPEANA	60.7	61.9	61.7	63.1	61.2	63.1	64.0	65.4	61.8	60.5	59.0	58.5	59.2	57.6	77.2
PATAGONIA	67.1	68.4	68.2	69.8	67.7	69.7	70.7	72.2	68.3	66.8	65.1	64.6	65.5	63.2	84.6
Description	Indigence	or extreme	poverty lii	ne, nomina	terms. See	text for de	tails.								
	Note: GB	A values of	nly are offic	cial (INDE	C's) for the	whole peri	od. Region	al values a	re INDEC's	for 2001 a	nd 2002, ar	nd construc	ted for 1993	5-2000.	
Equivalised income	May 95	Oct. 95	May 96	Oct. 96	May 97	Oct. 97	May 98	Oct. 98	May 99	Oct. 99	May 00	Oct. 00	May 01	Oct. 01	May 02
GBA	401.1	406.1	388.7	393.1	406.5	419.3	448.0	444.9	424.7	413.8	395.0	404.3	387.0	369.2	329.5
NOROESTE	236.9	236.8	223.5	215.5	242.6	237.7	256.6	257.9	239.3	239.5	229.5	227.8	223.2	214.3	194.0
NORESTE	227.2	225.0	221.5	211.2	220.1	226.2	238.4	230.3	230.1	212.9	200.1	201.1	200.8	185.8	167.9
CUYO	291.2	256.3	279.4	268.6	266.9	302.1	295.0	285.6	286.5	284.9	260.2	272.4	255.1	246.5	229.4
PAMPEANA	296.3	304.7	300.0	274.1	304.8	310.3	318.8	321.0	318.7	307.7	303.0	305.2	290.7	269.1	244.1
PATAGONIA	468.4	411.7	410.6	403.0	407.8	420.8	441.3	422.5	411.6	398.6	400.0	394.7	400.4	381.9	373.2
Total	354.6	346.6	336.7	336.4	349.0	358.9	378.5	375.6	361.8	351.6	337.3	343.2	329.4	311.6	280.7
Description	Equivalis	ed househo	ld income i	n nominal	terms. Adu	lt equivaler	ice scale us	ed by IND	EC - see M	oreles (198	8).				

Table C.1: Detailed Prices, Poverty L	ines and Income	Figures, Urban	Argentina.	1995-2002
---------------------------------------	-----------------	----------------	------------	-----------

Table C.2: Detailed Poverty Fig-	ures, Urban Argentina, 1995-2002
----------------------------------	----------------------------------

Indigent individuals headcount	May 95	Oct. 95	May 96	Oct. 96	May 97	Oct. 97	May 98	Oct. 98	May 99	Oct. 99	May 00	Oct. 00	May 01	Oct. 01	May 02
GBA	0.0567	0.0629	0.0694	0.0760	0.0569	0.0644	0.0534	0.0687	0.0759	0.0674	0.0755	0.0765	0.1032	0.1218	0.2267
NOROESTE	0.0875	0.1120	0.1160	0.1313	0.1096	0.1187	0.1181	0.1186	0.1298	0.1107	0.1252	0.1331	0.1454	0.1615	0.2947
NORESTE	0.1343	0.1261	0.1480	0.1717	0.1479	0.1494	0.1499	0.1761	0.1769	0.1776	0.1848	0.1968	0.2302	0.2693	0.3876
CUYO	0.0558	0.0858	0.0783	0.0883	0.0922	0.0878	0.0701	0.0765	0.0795	0.0739	0.0843	0.1063	0.1102	0.1231	0.2471
PAMPEANA	0.0905	0.0746	0.0856	0.1073	0.0751	0.0793	0.0768	0.0851	0.0800	0.0838	0.0881	0.0902	0.1132	0.1367	0.2510
PATAGONIA	0.0310	0.0721	0.0699	0.0756	0.0602	0.0621	0.0654	0.0799	0.0690	0.0779	0.0757	0.0714	0.0692	0.0734	0.1534
Total	0.0676	0.0755	0.0823	0.0931	0.0732	0.0790	0.0715	0.0838	0.0877	0.0822	0.0899	0.0938	0.1163	0.1363	0.2477
Description	Headcour	nt of indige	nt (extreme	ly poor) in	dividuals.										
Poor individuals headcount	May 95	Oct. 95	May 96	Oct. 96	May 97	Oct. 97	May 98	Oct. 98	May 99	Oct. 99	May 00	Oct. 00	May 01	Oct. 01	May 02
GBA	0.2217	0.2476	0.2668	0.2795	0.2629	0.2595	0.2427	0.2588	0.2711	0.2673	0.2973	0.2889	0.3267	0.3542	0.4965
NOROESTE	0.3783	0.4149	0.4419	0.4446	0.4066	0.4204	0.4023	0.4238	0.4558	0.4137	0.4455	0.4365	0.4742	0.4831	0.6355
NORESTE	0.4440	0.4531	0.4623	0.4889	0.4878	0.4837	0.4861	0.5026	0.5010	0.5061	0.5301	0.5314	0.5644	0.5730	0.6981
CUYO	0.2924	0.3599	0.3126	0.3496	0.3647	0.3204	0.3291	0.3274	0.3380	0.3342	0.3663	0.3917	0.3875	0.3966	0.5492
PAMPEANA	0.2848	0.2715	0.2916	0.3271	0.2921	0.2840	0.2863	0.2927	0.2936	0.2927	0.3092	0.3040	0.3387	0.3712	0.5271
PATAGONIA	0.1297	0.2203	0.2079	0.2269	0.2198	0.2183	0.2103	0.2242	0.2215	0.2476	0.2314	0.2417	0.2380	0.2325	0.3917
Total	0.2577	0.2870	0.3011	0.3200	0.3003	0.2954	0.2856	0.2991	0.3102	0.3048	0.3304	0.3262	0.3591	0.3831	0.5301
Description	Headcour	nt of poor i	ndividuals.												
Total poverty gap	May 95	Oct. 95	May 96	Oct. 96	May 97	Oct. 97	May 98	Oct. 98	May 99	Oct. 99	May 00	Oct. 00	May 01	Oct. 01	May 02
Total poverty gap GBA	May 95 0.0867	Oct. 95 0.0979	May 96 0.1059	Oct. 96 0.1147	May 97 0.1033	Oct. 97 0.1042	May 98 0.0930	Oct. 98 0.1081	May 99 0.1098	Oct. 99 0.1071	May 00 0.1235	Oct. 00 0.1219	May 01 0.1440	Oct. 01 0.1632	May 02 0.2656
Total poverty gap GBA NOROESTE	May 95 0.0867 0.1463	Oct. 95 0.0979 0.1637	May 96 0.1059 0.1830	Oct. 96 0.1147 0.1875	May 97 0.1033 0.1635	Oct. 97 0.1042 0.1732	May 98 0.0930 0.1645	Oct. 98 0.1081 0.1693	May 99 0.1098 0.1890	Oct. 99 0.1071 0.1699	May 00 0.1235 0.1901	Oct. 00 0.1219 0.1877	May 01 0.1440 0.2117	Oct. 01 0.1632 0.2211	May 02 0.2656 0.3315
Total poverty gap GBA NOROESTE NORESTE	May 95 0.0867 0.1463 0.1875	Oct. 95 0.0979 0.1637 0.1889	May 96 0.1059 0.1830 0.2008	Oct. 96 0.1147 0.1875 0.2231	May 97 0.1033 0.1635 0.2108	Oct. 97 0.1042 0.1732 0.2054	May 98 0.0930 0.1645 0.2120	Oct. 98 0.1081 0.1693 0.2289	May 99 0.1098 0.1890 0.2292	Oct. 99 0.1071 0.1699 0.2366	May 00 0.1235 0.1901 0.2501	Oct. 00 0.1219 0.1877 0.2638	May 01 0.1440 0.2117 0.2863	Oct. 01 0.1632 0.2211 0.3034	May 02 0.2656 0.3315 0.4139
Total poverty gap GBA NOROESTE NORESTE CUYO	May 95 0.0867 0.1463 0.1875 0.1010	Oct. 95 0.0979 0.1637 0.1889 0.1338	May 96 0.1059 0.1830 0.2008 0.1221	Oct. 96 0.1147 0.1875 0.2231 0.1337	May 97 0.1033 0.1635 0.2108 0.1429	Oct. 97 0.1042 0.1732 0.2054 0.1285	May 98 0.0930 0.1645 0.2120 0.1162	Oct. 98 0.1081 0.1693 0.2289 0.1213	May 99 0.1098 0.1890 0.2292 0.1313	Oct. 99 0.1071 0.1699 0.2366 0.1261	May 00 0.1235 0.1901 0.2501 0.1424	Oct. 00 0.1219 0.1877 0.2638 0.1618	May 01 0.1440 0.2117 0.2863 0.1607	Oct. 01 0.1632 0.2211 0.3034 0.1751	May 02 0.2656 0.3315 0.4139 0.2812
Total poverty gap GBA NOROESTE NORESTE CUYO PAMPEANA	May 95 0.0867 0.1463 0.1875 0.1010 0.1212	Oct. 95 0.0979 0.1637 0.1889 0.1338 0.1081	May 96 0.1059 0.1830 0.2008 0.1221 0.1229	Oct. 96 0.1147 0.1875 0.2231 0.1337 0.1410	May 97 0.1033 0.1635 0.2108 0.1429 0.1180	Oct. 97 0.1042 0.1732 0.2054 0.1285 0.1158	May 98 0.0930 0.1645 0.2120 0.1162 0.1123	Oct. 98 0.1081 0.1693 0.2289 0.1213 0.1171	May 99 0.1098 0.1890 0.2292 0.1313 0.1186	Oct. 99 0.1071 0.1699 0.2366 0.1261 0.1207	May 00 0.1235 0.1901 0.2501 0.1424 0.1294	Oct. 00 0.1219 0.1877 0.2638 0.1618 0.1322	May 01 0.1440 0.2117 0.2863 0.1607 0.1524	Oct. 01 0.1632 0.2211 0.3034 0.1751 0.1780	May 02 0.2656 0.3315 0.4139 0.2812 0.2772
Total poverty gap GBA NOROESTE NORESTE CUYO PAMPEANA PATAGONIA	May 95 0.0867 0.1463 0.1875 0.1010 0.1212 0.0474	Oct. 95 0.0979 0.1637 0.1889 0.1338 0.1081 0.0930	May 96 0.1059 0.1830 0.2008 0.1221 0.1229 0.0882	Oct. 96 0.1147 0.1875 0.2231 0.1337 0.1410 0.0928	May 97 0.1033 0.1635 0.2108 0.1429 0.1180 0.0849	Oct. 97 0.1042 0.1732 0.2054 0.1285 0.1158 0.0817	May 98 0.0930 0.1645 0.2120 0.1162 0.1123 0.0885	Oct. 98 0.1081 0.1693 0.2289 0.1213 0.1171 0.0943	May 99 0.1098 0.1890 0.2292 0.1313 0.1186 0.0889	Oct. 99 0.1071 0.1699 0.2366 0.1261 0.1207 0.1033	May 00 0.1235 0.1901 0.2501 0.1424 0.1294 0.0973	Oct. 00 0.1219 0.1877 0.2638 0.1618 0.1322 0.0988	May 01 0.1440 0.2117 0.2863 0.1607 0.1524 0.0964	Oct. 01 0.1632 0.2211 0.3034 0.1751 0.1780 0.1004	May 02 0.2656 0.3315 0.4139 0.2812 0.2772 0.1830
Total poverty gap GBA NOROESTE NORESTE CUYO PAMPEANA PATAGONIA Total	May 95 0.0867 0.1463 0.1875 0.1010 0.1212 0.0474 0.1019	Oct. 95 0.0979 0.1637 0.1889 0.1338 0.1081 0.0930 0.1137	May 96 0.1059 0.1830 0.2008 0.1221 0.1229 0.0882 0.1228	Oct. 96 0.1147 0.1875 0.2231 0.1337 0.1410 0.0928 0.1335	May 97 0.1033 0.1635 0.2108 0.1429 0.1180 0.0849 0.1200	Oct. 97 0.1042 0.1732 0.2054 0.1285 0.1158 0.0817 0.1198	May 98 0.0930 0.1645 0.2120 0.1162 0.1123 0.0885 0.1119	Oct. 98 0.1081 0.1693 0.2289 0.1213 0.1171 0.0943 0.1230	May 99 0.1098 0.1890 0.2292 0.1313 0.1186 0.0889 0.1270	Oct. 99 0.1071 0.1699 0.2366 0.1261 0.1207 0.1033 0.1247	May 00 0.1235 0.1901 0.2501 0.1424 0.1294 0.0973 0.1390	Oct. 00 0.1219 0.1877 0.2638 0.1618 0.1322 0.0988 0.1407	May 01 0.1440 0.2117 0.2863 0.1607 0.1524 0.0964 0.1604	Oct. 01 0.1632 0.2211 0.3034 0.1751 0.1780 0.1004 0.1794	May 02 0.2656 0.3315 0.4139 0.2812 0.2772 0.1830 0.2822
Total poverty gap GBA NOROESTE NORESTE CUYO PAMPEANA PATAGONIA Total Description	May 95 0.0867 0.1463 0.1875 0.1010 0.1212 0.0474 0.1019 Poverty g	Oct. 95 0.0979 0.1637 0.1889 0.1338 0.1081 0.0930 0.1137 rap - FGT n	May 96 0.1059 0.1830 0.2008 0.1221 0.1229 0.0882 0.1228 measure wit	Oct. 96 0.1147 0.1875 0.2231 0.1337 0.1410 0.0928 0.1335 h alpha=1.	May 97 0.1033 0.1635 0.2108 0.1429 0.1180 0.0849 0.1200 See text for	Oct. 97 0.1042 0.1732 0.2054 0.1285 0.1158 0.0817 0.1198 r details.	May 98 0.0930 0.1645 0.2120 0.1162 0.1123 0.0885 0.1119	Oct. 98 0.1081 0.1693 0.2289 0.1213 0.1171 0.0943 0.1230	May 99 0.1098 0.1890 0.2292 0.1313 0.1186 0.0889 0.1270	Oct. 99 0.1071 0.1699 0.2366 0.1261 0.1207 0.1033 0.1247	May 00 0.1235 0.1901 0.2501 0.1424 0.1294 0.0973 0.1390	Oct. 00 0.1219 0.1877 0.2638 0.1618 0.1322 0.0988 0.1407	May 01 0.1440 0.2117 0.2863 0.1607 0.1524 0.0964 0.1604	Oct. 01 0.1632 0.2211 0.3034 0.1751 0.1780 0.1004 0.1794	May 02 0.2656 0.3315 0.4139 0.2812 0.2772 0.1830 0.2822
Total poverty gap GBA NOROESTE NORESTE CUYO PAMPEANA PATAGONIA Total Description Total squared poverty gap	May 95 0.0867 0.1463 0.1875 0.1010 0.1212 0.0474 0.1019 Poverty g May 95	Oct. 95 0.0979 0.1637 0.1889 0.1338 0.1081 0.0930 0.1137 ap - FGT n Oct. 95	May 96 0.1059 0.1830 0.2008 0.1221 0.1229 0.0882 0.1228 measure wit May 96	Oct. 96 0.1147 0.1875 0.2231 0.1337 0.1410 0.0928 0.1335 h alpha=1. Oct. 96	May 97 0.1033 0.1635 0.2108 0.1429 0.1180 0.0849 0.1200 See text for May 97	Oct. 97 0.1042 0.1732 0.2054 0.1285 0.1158 0.0817 0.1198 r details. Oct. 97	May 98 0.0930 0.1645 0.2120 0.1162 0.1123 0.0885 0.1119 May 98	Oct. 98 0.1081 0.1693 0.2289 0.1213 0.1171 0.0943 0.1230 Oct. 98	May 99 0.1098 0.1890 0.2292 0.1313 0.1186 0.0889 0.1270 May 99	Oct. 99 0.1071 0.1699 0.2366 0.1261 0.1207 0.1033 0.1247 Oct. 99	May 00 0.1235 0.1901 0.2501 0.1424 0.1294 0.0973 0.1390 May 00	Oct. 00 0.1219 0.1877 0.2638 0.1618 0.1322 0.0988 0.1407 Oct. 00	May 01 0.1440 0.2117 0.2863 0.1607 0.1524 0.0964 0.1604 May 01	Oct. 01 0.1632 0.2211 0.3034 0.1751 0.1780 0.1004 0.1794 Oct. 01	May 02 0.2656 0.3315 0.4139 0.2812 0.2772 0.1830 0.2822 May 02
Total poverty gap GBA NOROESTE NORESTE CUYO PAMPEANA PATAGONIA Total Description Total squared poverty gap GBA	May 95 0.0867 0.1463 0.1875 0.1010 0.1212 0.0474 0.1019 Poverty g May 95 0.0520	Oct. 95 0.0979 0.1637 0.1889 0.1338 0.1081 0.0930 0.1137 (ap - FGT n Oct. 95 0.0578	May 96 0.1059 0.1830 0.2008 0.1221 0.1229 0.0882 0.1228 measure wite May 96 0.0641	Oct. 96 0.1147 0.1875 0.2231 0.1337 0.1410 0.0928 0.1335 h alpha=1. Oct. 96 0.0690	May 97 0.1033 0.1635 0.2108 0.1429 0.1180 0.0849 0.1200 See text for May 97 0.0576	Oct. 97 0.1042 0.1732 0.2054 0.1285 0.1158 0.0817 0.1198 r details. Oct. 97 0.0606	May 98 0.0930 0.1645 0.2120 0.1162 0.1123 0.0885 0.1119 May 98 0.0524	Oct. 98 0.1081 0.1693 0.2289 0.1213 0.1171 0.0943 0.1230 Oct. 98 0.0628	May 99 0.1098 0.1890 0.2292 0.1313 0.1186 0.0889 0.1270 May 99 0.0646	Oct. 99 0.1071 0.1699 0.2366 0.1261 0.1207 0.1033 0.1247 Oct. 99 0.0622	May 00 0.1235 0.1901 0.2501 0.1424 0.0973 0.1390 May 00 0.0714	Oct. 00 0.1219 0.1877 0.2638 0.1618 0.1322 0.0988 0.1407 Oct. 00 0.0712	May 01 0.1440 0.2117 0.2863 0.1607 0.1524 0.0964 0.1604 May 01 0.0889	Oct. 01 0.1632 0.2211 0.3034 0.1751 0.1780 0.1004 0.1794 Oct. 01 0.1041	May 02 0.2656 0.3315 0.4139 0.2812 0.2772 0.1830 0.2822 May 02 0.1826
Total poverty gap GBA NOROESTE NORESTE CUYO PAMPEANA PATAGONIA Total Description Total squared poverty gap GBA NOROESTE	May 95 0.0867 0.1463 0.1875 0.1010 0.1212 0.0474 0.1019 Poverty g May 95 0.0520 0.0785	Oct. 95 0.0979 0.1637 0.1889 0.1338 0.1081 0.0930 0.1137 (ap - FGT n Oct. 95 0.0578 0.0896	May 96 0.1059 0.1830 0.2008 0.1221 0.1229 0.0882 0.1228 measure wite May 96 0.0641 0.1031	Oct. 96 0.1147 0.1875 0.2231 0.1337 0.1410 0.0928 0.1335 h alpha=1. Oct. 96 0.0690 0.1080	May 97 0.1033 0.1635 0.2108 0.1429 0.1180 0.0849 0.1200 See text for May 97 0.0576 0.0908	Oct. 97 0.1042 0.1732 0.2054 0.1285 0.1158 0.0817 0.1198 r details. Oct. 97 0.0606 0.0970	May 98 0.0930 0.1645 0.2120 0.1162 0.1123 0.0885 0.1119 May 98 0.0524 0.0922	Oct. 98 0.1081 0.1693 0.2289 0.1213 0.1171 0.0943 0.1230 Oct. 98 0.0628 0.0939	May 99 0.1098 0.1890 0.2292 0.1313 0.1186 0.0889 0.1270 May 99 0.0646 0.1092	Oct. 99 0.1071 0.1699 0.2366 0.1261 0.1207 0.1033 0.1247 Oct. 99 0.0622 0.0963	May 00 0.1235 0.1901 0.2501 0.1424 0.0973 0.1390 May 00 0.0714 0.1097	Oct. 00 0.1219 0.1877 0.2638 0.1618 0.1322 0.0988 0.1407 Oct. 00 0.0712 0.1101	May 01 0.1440 0.2117 0.2863 0.1607 0.1524 0.0964 0.1604 May 01 0.0889 0.1256	Oct. 01 0.1632 0.2211 0.3034 0.1751 0.1780 0.1004 0.1794 Oct. 01 0.1041 0.1357	May 02 0.2656 0.3315 0.4139 0.2812 0.2772 0.1830 0.2822 May 02 0.1826 0.2182
Total poverty gap GBA NOROESTE NORESTE CUYO PAMPEANA PATAGONIA Total Description Total squared poverty gap GBA NOROESTE NORESTE	May 95 0.0867 0.1463 0.1875 0.1010 0.1212 0.0474 0.1019 Poverty g May 95 0.0520 0.0785 0.1085	Oct. 95 0.0979 0.1637 0.1889 0.1338 0.1081 0.0930 0.1137 cap - FGT r Oct. 95 0.0578 0.0896 0.1081	May 96 0.1059 0.1830 0.2008 0.1221 0.1229 0.0882 0.1228 measure wite May 96 0.0641 0.1031 0.1182	Oct. 96 0.1147 0.1875 0.2231 0.1337 0.1410 0.0928 0.1335 h alpha=1. Oct. 96 0.0690 0.1080 0.1341	May 97 0.1033 0.1635 0.2108 0.1429 0.1180 0.0849 0.1200 See text for May 97 0.0576 0.0908 0.1215	Oct. 97 0.1042 0.1732 0.2054 0.1285 0.1158 0.0817 0.1198 r details. Oct. 97 0.0606 0.0970 0.1161	May 98 0.0930 0.1645 0.2120 0.1162 0.1123 0.0885 0.1119 May 98 0.0524 0.0922 0.1224	Oct. 98 0.1081 0.1693 0.2289 0.1213 0.1171 0.0943 0.1230 Oct. 98 0.0628 0.0939 0.1379	May 99 0.1098 0.1890 0.2292 0.1313 0.1186 0.0889 0.1270 May 99 0.0646 0.1092 0.1382	Oct. 99 0.1071 0.1699 0.2366 0.1261 0.1207 0.1033 0.1247 Oct. 99 0.0622 0.0963 0.1452	May 00 0.1235 0.1901 0.2501 0.1424 0.0973 0.1390 May 00 0.0714 0.1097 0.1541	Oct. 00 0.1219 0.1877 0.2638 0.1618 0.1322 0.0988 0.1407 Oct. 00 0.0712 0.1101 0.1690	May 01 0.1440 0.2117 0.2863 0.1607 0.1524 0.0964 0.1604 May 01 0.0889 0.1256 0.1845	Oct. 01 0.1632 0.2211 0.3034 0.1751 0.1780 0.1004 0.1794 Oct. 01 0.1041 0.1357 0.2033	May 02 0.2656 0.3315 0.4139 0.2812 0.2772 0.1830 0.2822 May 02 0.1826 0.2182 0.2933
Total poverty gap GBA NOROESTE NORESTE CUYO PAMPEANA PATAGONIA Total Description Total squared poverty gap GBA NOROESTE NORESTE CUYO	May 95 0.0867 0.1463 0.1875 0.1010 0.1212 0.0474 0.1019 Poverty g May 95 0.0520 0.0785 0.1085 0.0502	Oct. 95 0.0979 0.1637 0.1889 0.1338 0.1081 0.0930 0.1137 cap - FGT r Oct. 95 0.0578 0.0896 0.1081 0.0718	May 96 0.1059 0.1830 0.2008 0.1221 0.1229 0.0882 0.1228 measure wite May 96 0.0641 0.1031 0.1182 0.0662	Oct. 96 0.1147 0.1875 0.2231 0.1337 0.1410 0.0928 0.1335 h alpha=1. Oct. 96 0.0690 0.1080 0.1341 0.0726	May 97 0.1033 0.1635 0.2108 0.1429 0.1180 0.0849 0.1200 See text for May 97 0.0576 0.0908 0.1215 0.0793	Oct. 97 0.1042 0.1732 0.2054 0.1285 0.1158 0.0817 0.1198 r details. Oct. 97 0.0606 0.0970 0.1161 0.0727	May 98 0.0930 0.1645 0.2120 0.1162 0.1123 0.0885 0.1119 May 98 0.0524 0.0922 0.1224 0.0590	Oct. 98 0.1081 0.1693 0.2289 0.1213 0.1171 0.0943 0.1230 Oct. 98 0.0628 0.0939 0.1379 0.0632	May 99 0.1098 0.1890 0.2292 0.1313 0.1186 0.0889 0.1270 May 99 0.0646 0.1092 0.1382 0.0711	Oct. 99 0.1071 0.1699 0.2366 0.1261 0.1207 0.1033 0.1247 Oct. 99 0.0622 0.0963 0.1452 0.0667	May 00 0.1235 0.1901 0.2501 0.1424 0.0973 0.1390 May 00 0.0714 0.1097 0.1541 0.0778	Oct. 00 0.1219 0.1877 0.2638 0.1618 0.1322 0.0988 0.1407 Oct. 00 0.0712 0.1101 0.1690 0.0913	May 01 0.1440 0.2117 0.2863 0.1607 0.1524 0.0964 0.1604 May 01 0.0889 0.1256 0.1845 0.0929	Oct. 01 0.1632 0.2211 0.3034 0.1751 0.1780 0.1004 0.1794 Oct. 01 0.1041 0.1357 0.2033 0.1060	May 02 0.2656 0.3315 0.4139 0.2812 0.2772 0.1830 0.2822 May 02 0.1826 0.2182 0.2933 0.1846
Total poverty gap GBA NOROESTE NORESTE CUYO PAMPEANA PATAGONIA Total Description Total squared poverty gap GBA NOROESTE NORESTE CUYO PAMPEANA	May 95 0.0867 0.1463 0.1875 0.1010 0.1212 0.0474 0.1019 Poverty g May 95 0.0520 0.0785 0.1085 0.0502 0.0751	Oct. 95 0.0979 0.1637 0.1889 0.1338 0.1081 0.0930 0.1137 cap - FGT r Oct. 95 0.0578 0.0896 0.1081 0.0718 0.0636	May 96 0.1059 0.1830 0.2008 0.1221 0.1229 0.0882 0.1228 measure wite May 96 0.0641 0.1031 0.1182 0.0662 0.0737	Oct. 96 0.1147 0.1875 0.2231 0.1337 0.1410 0.0928 0.1335 h alpha=1. Oct. 96 0.0690 0.1080 0.1341 0.0726 0.0881	May 97 0.1033 0.1635 0.2108 0.1429 0.1180 0.0849 0.1200 See text for May 97 0.0576 0.0908 0.1215 0.0793 0.0689	Oct. 97 0.1042 0.1732 0.2054 0.1285 0.1158 0.0817 0.1198 r details. Oct. 97 0.0606 0.0970 0.1161 0.0727 0.0695	May 98 0.0930 0.1645 0.2120 0.1162 0.1123 0.0885 0.1119 May 98 0.0524 0.0922 0.1224 0.0590 0.0638	Oct. 98 0.1081 0.1693 0.2289 0.1213 0.1171 0.0943 0.1230 Oct. 98 0.0628 0.0939 0.1379 0.0632 0.0685	May 99 0.1098 0.1890 0.2292 0.1313 0.1186 0.0889 0.1270 May 99 0.0646 0.1092 0.1382 0.0711 0.0686	Oct. 99 0.1071 0.1699 0.2366 0.1261 0.1207 0.1033 0.1247 Oct. 99 0.0622 0.0963 0.1452 0.0667 0.0727	May 00 0.1235 0.1901 0.2501 0.1424 0.0973 0.1390 May 00 0.0714 0.097 0.1541 0.0778 0.0776	Oct. 00 0.1219 0.1877 0.2638 0.1618 0.1322 0.0988 0.1407 Oct. 00 0.0712 0.1101 0.1690 0.0913 0.0804	May 01 0.1440 0.2117 0.2863 0.1607 0.1524 0.0964 0.1604 May 01 0.0889 0.1256 0.1845 0.0929 0.0948	Oct. 01 0.1632 0.2211 0.3034 0.1751 0.1780 0.1004 0.1794 Oct. 01 0.1041 0.1357 0.2033 0.1060 0.1164	May 02 0.2656 0.3315 0.4139 0.2812 0.2772 0.1830 0.2822 May 02 0.1826 0.2182 0.2933 0.1846 0.1894
Total poverty gap GBA NOROESTE NORESTE CUYO PAMPEANA PATAGONIA Total Description Total squared poverty gap GBA NOROESTE NORESTE CUYO PAMPEANA PATAGONIA	May 95 0.0867 0.1463 0.1875 0.1010 0.1212 0.0474 0.1019 Poverty g May 95 0.0520 0.0785 0.1085 0.0502 0.0751 0.0271	Oct. 95 0.0979 0.1637 0.1889 0.1338 0.1081 0.0930 0.1137 cap - FGT r Oct. 95 0.0578 0.0896 0.1081 0.0718 0.0636 0.0556	May 96 0.1059 0.1830 0.2008 0.1221 0.1229 0.0882 0.1228 measure wite May 96 0.0641 0.1031 0.1182 0.0662 0.0737 0.0550	Oct. 96 0.1147 0.1875 0.2231 0.1337 0.1410 0.0928 0.1335 h alpha=1. Oct. 96 0.0690 0.1080 0.1341 0.0726 0.0881 0.0558	May 97 0.1033 0.1635 0.2108 0.1429 0.1180 0.0849 0.1200 See text for May 97 0.0576 0.0908 0.1215 0.0793 0.0689 0.0506	Oct. 97 0.1042 0.1732 0.2054 0.1285 0.1158 0.0817 0.1198 r details. Oct. 97 0.0606 0.0970 0.1161 0.0727 0.0695 0.0454	May 98 0.0930 0.1645 0.2120 0.1162 0.1123 0.0885 0.1119 May 98 0.0524 0.0922 0.1224 0.0922 0.1224 0.0590 0.0638 0.0524	Oct. 98 0.1081 0.1693 0.2289 0.1213 0.1171 0.0943 0.1230 Oct. 98 0.0628 0.0939 0.1379 0.0632 0.0685 0.0580	May 99 0.1098 0.1890 0.2292 0.1313 0.1186 0.0889 0.1270 May 99 0.0646 0.1092 0.1382 0.0711 0.0686 0.0500	Oct. 99 0.1071 0.1699 0.2366 0.1261 0.1207 0.1033 0.1247 Oct. 99 0.0622 0.0963 0.1452 0.0667 0.0727 0.0631	May 00 0.1235 0.1901 0.2501 0.1424 0.0973 0.1390 May 00 0.0714 0.097 0.1541 0.0778 0.0776 0.0576	Oct. 00 0.1219 0.1877 0.2638 0.1618 0.1322 0.0988 0.1407 Oct. 00 0.0712 0.1101 0.1690 0.0913 0.0804 0.0590	May 01 0.1440 0.2117 0.2863 0.1607 0.1524 0.0964 0.1604 May 01 0.0889 0.1256 0.1845 0.0929 0.0948 0.0560	Oct. 01 0.1632 0.2211 0.3034 0.1751 0.1780 0.1004 0.1794 Oct. 01 0.1041 0.1357 0.2033 0.1060 0.1164 0.0627	May 02 0.2656 0.3315 0.4139 0.2812 0.2772 0.1830 0.2822 May 02 0.1826 0.2182 0.2933 0.1846 0.1894 0.1176
Total poverty gap GBA NOROESTE NORESTE CUYO PAMPEANA PATAGONIA Total Description Total squared poverty gap GBA NOROESTE NORESTE CUYO PAMPEANA PATAGONIA Total	May 95 0.0867 0.1463 0.1875 0.1010 0.1212 0.0474 0.1019 Poverty g May 95 0.0520 0.0785 0.1085 0.0502 0.0751 0.0271 0.1019	Oct. 95 0.0979 0.1637 0.1889 0.1338 0.1081 0.0930 0.1137 (ap - FGT n Oct. 95 0.0578 0.0896 0.1081 0.0718 0.0636 0.0556 0.1137	May 96 0.1059 0.1830 0.2008 0.1221 0.1229 0.0882 0.1228 measure wite May 96 0.0641 0.1031 0.1182 0.0662 0.0737 0.0550 0.1228	Oct. 96 0.1147 0.1875 0.2231 0.1337 0.1410 0.0928 0.1335 h alpha=1. Oct. 96 0.0690 0.1080 0.1341 0.0726 0.0881 0.0558 0.1335	May 97 0.1033 0.1635 0.2108 0.1429 0.1180 0.0849 0.1200 See text for May 97 0.0576 0.0908 0.1215 0.0793 0.0506 0.1200	Oct. 97 0.1042 0.1732 0.2054 0.1285 0.1158 0.0817 0.1198 r details. Oct. 97 0.0606 0.0970 0.1161 0.0727 0.0695 0.0454 0.1198	May 98 0.0930 0.1645 0.2120 0.1162 0.1123 0.0885 0.1119 May 98 0.0524 0.0922 0.1224 0.0590 0.0638 0.0524 0.0524 0.1119	Oct. 98 0.1081 0.1693 0.2289 0.1213 0.1171 0.0943 0.1230 Oct. 98 0.0628 0.0939 0.1379 0.0632 0.0685 0.0580 0.1230	May 99 0.1098 0.1890 0.2292 0.1313 0.1186 0.0889 0.1270 May 99 0.0646 0.1092 0.1382 0.0711 0.0686 0.0500 0.1270	Oct. 99 0.1071 0.1699 0.2366 0.1261 0.1207 0.1033 0.1247 Oct. 99 0.0622 0.0963 0.1452 0.0667 0.0727 0.0631 0.1247	May 00 0.1235 0.1901 0.2501 0.1424 0.0973 0.1390 May 00 0.0714 0.1097 0.1541 0.0778 0.0576 0.0576 0.1390	Oct. 00 0.1219 0.1877 0.2638 0.1618 0.1322 0.0988 0.1407 Oct. 00 0.0712 0.1101 0.1690 0.0913 0.0804 0.0590 0.1407	May 01 0.1440 0.2117 0.2863 0.1607 0.1524 0.0964 0.1604 May 01 0.0889 0.1256 0.1845 0.0929 0.0948 0.0560 0.1604	Oct. 01 0.1632 0.2211 0.3034 0.1751 0.1780 0.1004 0.1794 Oct. 01 0.1041 0.1041 0.1041 0.1041 0.2033 0.1060 0.1164 0.0627 0.1794	May 02 0.2656 0.3315 0.4139 0.2812 0.2772 0.1830 0.2822 May 02 0.1826 0.2182 0.2933 0.1846 0.1894 0.1176 0.2822

227

	Total	Non-poor	Poor	Extreme Poor
Log Normalised Income	0.599	0.978	-0.327	-2.273
Normalised Income	2.863	3.467	0.739	0.211
Demographic Composition				
Number of Infants (ages 0-5)	0.389	0.288	0.701	0.961
Number of Infants squared	0.708	0.468	1.391	2.248
Number of Children (ages 6-14)	0.587	0.418	1.150	1.409
Number of Children Squared	1.287	0.780	2.827	4.245
Number of Youths (ages 15-24)	0.632	0.569	0.869	0.858
Number of Youths Squared	1.275	1.120	1.872	1.806
Number of Adults (ages 25-64)	1.547	1.506	1.719	1.650
Number of Adults Squared	3.270	3.182	3.685	3.338
Number of Elderly (ages 65+)	0.357	0.392	0.255	0.125
Number of Elderly Squared	0.532	0.586	0.389	0.167
Household Head				
Age - 19 and younger	0.005	0.005	0.008	0.004
Age - 20-29	0.102	0.098	0.107	0.143
Age - 30-39	0.201	0.184	0.260	0.276
Age - 50-59	0.176	0.181	0.152	0.175
Age - 60 and older	0.299	0.330	0.199	0.125
Female	0.246	0.258	0.186	0.243
Recent migrant	0.061	0.061	0.055	0.068
Inactive	0.298	0.314	0.250	0.203
Unemployed	0.076	0.045	0.123	0.386
Primary - Complete	0.340	0.318	0.417	0.444
Secondary - Incomplete	0.158	0.158	0.163	0.150
Secondary - Complete	0.152	0.171	0.091	0.056
Further Education - Incomplete	0.008	0.009	0.003	0.001
Further Education - Complete	0.028	0.033	0.012	0.004
University Education	0.137	0.166	0.030	0.037
Boss	0.039	0.048	0.008	0.006
Self Employed	0.158	0.149	0.195	0.189
Informal Worker	0.242	0.223	0.320	0.292
Public Sector	0.101	0.112	0.070	0.032
Qualification: Operative	0.283	0.286	0.304	0.172
Qualification: Technician	0.142	0.156	0.098	0.063
Qualification: Professional	0.067	0.084	0.002	0.007
Spouse Not Present	0.326	0.350	0.212	0.306
Household Spouse				
Unemployed	0.044	0.034	0.070	0.109
Primary - Complete	0.238	0.211	0.346	0.322
Secondary - Complete	0.127	0.140	0.084	0.055
Secondary - Incomplete	0.099	0.092	0.137	0.092
Further - Complete	0.033	0.040	0.010	0.001
Further - Incomplete	0.008	0.008	0.006	0.005
University	0.062	0.076	0.012	0.006
Boss	0.007	0.008	0.005	0.000
Self Employed	0.055	0.056	0.052	0.044
Informal Worker	0.104	0.101	0.121	0.101
Public Sector	0.049	0.058	0.016	0.006
Qualification: Operative	0.055	0.061	0.032	0.021
Qualification: Technician	0.052	0.062	0.017	0.007
Qualification: Professional	0.023	0.028	0.001	0.000
Geographic location				
Noroeste	0.079	0.067	0.126	0.127
Noreste	0.044	0.036	0.073	0.079
Cuyo	0.061	0.056	0.082	0.072
Pampeana	0.239	0.242	0.220	0.237
Patagonia	0.025	0.026	0.018	0.026
Source: Author's estimations based on EPH	household s	urvey data (INDEC	C) .	

Table C.3: Poverty Profile, Urban Argentina, October 1995

Log Normalised Income 0.634 1.025 -0.338 -2.014 Normalised Income 3.033 3.712 0.731 0.232 Demographic Composition Number of Infants (ages 0-5) 0.368 0.262 0.673 0.948 Number of Infants (ages 6-14) 0.571 0.389 1.122 1.501 Number of Children Squared 1.264 1.062 1.944 2.117 Number of Youths (ages 15-24) 0.627 0.552 0.881 0.931 Number of Aduits (ages 2-64) 1.529 1.487 1.697 1.630 Number of Aduits (ages 2-54) 0.348 0.384 0.262 0.108 Number of Aduits (ages 2-64) 1.529 1.487 1.697 1.630 Number of Aduits (ages 2-64) 1.529 0.487 0.262 0.108 Number of Aduits (ages 2-74) 0.183 0.190 0.161 0.147 Age - 19 and younger 0.005 0.004 0.005 0.005 Age - 30.39 0.195 0.177 0.254 0.277		Total	Non-poor	Poor	Extreme Poor
Normalised Income 3.033 3.712 0.731 0.232 Demographic Composition	Log Normalised Income	0.634	1.028	-0.338	-2.014
Demographic Composition Unimber of Infants squared 0.368 0.262 0.673 0.948 Number of Infants squared 0.653 0.402 1.327 2.175 Number of Children (squared 1.233 0.705 2.691 4.344 Number of Youths (squared 1.264 1.062 1.944 2.117 Number of Adults (squared 5.5-64) 1.529 1.487 1.697 1.630 Number of Adults (squared 3.213 3.131 3.594 3.262 1.088 Number of Adults (squared 0.520 0.571 0.409 0.147 1.097 1.630 Number of Elderly Squared 0.250 0.571 0.409 0.147 1.93 3.213 3.313 3.594 3.226 1.018 1.014 1.014 1.014 1.014 0.015 0.005 0.004 0.005 0.005 0.005 0.005 0.005 0.005 0.017 0.254 0.277 Age - 30.39 0.115 0.170 0.163 0.189 Age - 60 and older 0.284 0.292 0.229	Normalised Income	3.033	3.712	0.731	0.232
Number of Infants (ages 0-5) 0.368 0.262 0.673 0.948 Number of Infants squared 0.653 0.402 1.327 2.175 Number of Children (ages 6-14) 0.571 0.389 1.122 1.501 Number of Children (ages 6-14) 0.627 0.552 0.881 0.931 Number of Youths (ages 15-24) 0.627 0.552 0.881 0.931 Number of Adults Squared 3.213 3.131 3.594 3.262 Number of Elderly Squared 0.220 0.711 0.409 0.040 0.005 0.005 Number of Elderly Squared 0.520 0.044 0.027 0.107 0.254 0.271 Age - 19 and younger 0.005 0.004 0.005 0.005 Age - 30-39 0.195 0.177 0.254 0.277 Female 0.266 0.276 0.209 0.292 Recent migrant 0.056 0.059 0.43 0.035 Inactive 0.284 0.296 0.241 0.2297 Primary - Complete	Demographic Composition				
Number of Infants squared 0.653 0.402 1.327 2.175 Number of Children Squared 1.233 0.705 2.691 4.344 Number of Youths Squared 1.264 1.062 1.944 2.117 Number of Youths Squared 1.264 1.062 1.944 2.117 Number of Adults Squared 3.213 3.131 3.594 3.262 Number of Elderly Squared 0.320 0.571 0.409 0.147 Household Head - - - - - Age - 19 and younger 0.005 0.004 0.005 0.005 0.004 0.147 Age - 30.39 0.195 0.177 0.224 0.277 Age - 30.30 0.185 0.190 0.163 0.183 Age - 60 and older 0.233 0.233 0.244 0.229 2292 Recent migrant 0.056 0.059 0.443 0.057 Inactive 0.284 0.283 0.432 0.386 Secondary - Incomplete 0.152	Number of Infants (ages 0-5)	0.368	0.262	0.673	0.948
Number of Children (ages 6-14) 0.571 0.389 1.122 1.501 Number of Youths (ages 15-24) 0.627 0.552 0.881 0.931 Number of Youths Squared 1.264 1.662 1.944 2.117 Number of Youths Squared 3.213 3.131 3.594 3.262 Number of Elderly Squared 0.220 0.571 0.489 0.484 0.462 0.108 Number of Elderly Squared 0.220 0.571 0.490 0.147 Household Head	Number of Infants squared	0.653	0.402	1.327	2.175
Number of Children Squared 1.23 0.705 2.691 4.344 Number of Youths Squared 1.264 1.062 1.944 2.117 Number of Youths Squared 1.264 1.062 1.944 2.117 Number of Adults Squared 3.213 3.131 3.594 3.262 Number of Elderly Gages 65+) 0.348 0.384 0.262 0.108 Number of Elderly Squared 0.200 0.571 0.409 0.147 Age - 19 and younger 0.005 0.004 0.005 0.004 Age - 30.39 0.195 0.177 0.254 0.277 Age - 60 and older 0.293 0.323 0.204 0.267 Inactive 0.284 0.296 0.241 0.229 Unemployed 0.060 0.035 0.102 0.277 Frinale 0.266 0.276 0.299 0.229 Unemployed 0.600 0.035 0.102 0.297 Inactive 0.152 0.175 0.800 0.498 </td <td>Number of Children (ages 6-14)</td> <td>0.571</td> <td>0.389</td> <td>1.122</td> <td>1.501</td>	Number of Children (ages 6-14)	0.571	0.389	1.122	1.501
Number of Youths (ages 15-24) 0.627 0.552 0.881 0.931 Number of Youths Squared 1.264 1.062 1.944 2.117 Number of Adults (ages 25-64) 1.529 1.487 1.630 Number of Edderly (ages 65+) 0.348 0.334 0.262 0.108 Number of Elderly Squared 0.520 0.571 0.409 0.147 Household Head 2 2 0.005 0.005 Age - 19 and younger 0.005 0.006 0.114 0.145 0.2277 Age - 50-59 0.185 0.190 0.163 0.189 0.266 0.276 0.209 0.2292 Recent migrant 0.056 0.059 0.043 0.057 1nactive 0.284 0.296 0.241 0.2297 Primary - Complete 0.313 0.283 0.432 0.386 Secondary - Incomplete 0.177 0.800 0.049 Purtmer Education - Incomplete 0.174 0.717 0.800 0.049 Purthere Edu	Number of Children Squared	1.233	0.705	2.691	4.344
Number of Youths Squared 1.264 1.062 1.944 2.117 Number of Adults (ages 25-64) 1.529 1.487 1.637 1.630 Number of Elderly Squared 3.213 3.131 3.594 3.262 Number of Elderly Squared 0.520 0.571 0.409 0.147 Household Head 0.005 0.004 0.005 0.005 Age - 19 and younger 0.005 0.004 0.0163 0.189 Age - 30-39 0.195 0.177 0.254 0.277 Age - 50-59 0.185 0.190 0.163 0.189 Age - 60 0.403 0.057 Inactive 0.284 0.296 0.241 0.229 0.229 0.432 0.386 Secondary - Incomplete 0.174 0.171 0.190 0.168 Secondary - Incomplete 0.174 0.171 0.190 0.168 Secondary - Complete 0.156 0.191 0.031 0.002 Furmary - Complete 0.174	Number of Youths (ages 15-24)	0.627	0.552	0.881	0.931
Number of Adults (ages 25-64) 1.529 1.487 1.697 1.630 Number of Adults Squared 3.213 3.131 3.594 3.262 Number of Elderly (ages 65+) 0.348 0.384 0.262 0.108 Number of Elderly Squared 0.520 0.571 0.409 0.147 Household Head 4 4 4 Age - 19 and younger 0.005 0.004 0.005 0.005 Age - 30.39 0.195 0.177 0.254 0.277 Age - 30.39 0.185 0.190 0.163 0.189 Age - 60 and older 0.2261 0.229 0.2292 1.020 0.222 Recent migrant 0.056 0.059 0.043 0.057 Inactive 0.284 0.296 0.241 0.229 Primary - Complete 0.131 0.283 0.432 0.386 Secondary - Incomplete 0.174 0.171 0.190 0.168 Secondary - Complete 0.152 0.175	Number of Youths Squared	1.264	1.062	1.944	2.117
Number of Adults Squared 3.213 3.131 3.594 3.262 Number of Elderly (squared 0.520 0.571 0.409 0.147 Household Head Age - 19 and younger 0.005 0.004 0.005 0.005 Age - 30-39 0.195 0.177 0.254 0.277 Age - 50-59 0.185 0.177 0.254 0.277 Age - 60 and older 0.293 0.323 0.204 0.127 Female 0.266 0.276 0.209 0.292 Recent migrant 0.056 0.057 1.022 0.102 0.297 Primary - Complete 0.313 0.283 0.432 0.386 Secondary - Incomplete 0.174 0.171 0.190 0.168 Secondary - Complete 0.152 0.175 0.080 0.002 Further Education - Incomplete 0.027 0.033 0.005 0.002 1.017 0.168 Secondary - Complete 0.276 0.231 0.344 0.326 D.026 D.026 D.	Number of Adults (ages 25-64)	1.529	1.487	1.697	1.630
Number of Elderly (ages 65+) 0.348 0.384 0.262 0.108 Number of Elderly Squared 0.520 0.571 0.409 0.147 Household Head 0.005 0.004 0.005 0.005 Age - 19 and younger 0.005 0.004 0.0145 0.127 Age - 30-39 0.1185 0.190 0.163 0.189 Age - 60 and older 0.293 0.323 0.204 0.127 Fernale 0.266 0.276 0.209 0.292 Recent migrant 0.056 0.059 0.043 0.057 Inactive 0.284 0.296 0.241 0.229 Unemployed 0.060 0.035 0.102 0.297 Primary - Complete 0.174 0.171 0.808 0.049 Further Education - Incomplete 0.090 0.010 0.005 0.002 Unemployed 0.061 0.033 0.005 0.003 0.005 Unemployed 0.026 <td>Number of Adults Squared</td> <td>3.213</td> <td>3.131</td> <td>3.594</td> <td>3.262</td>	Number of Adults Squared	3.213	3.131	3.594	3.262
Number of Elderly Squared 0.520 0.571 0.409 0.147 Household Head	Number of Elderly (ages 65+)	0.348	0.384	0.262	0.108
Household Head Age - 19 and younger 0.005 0.006 0.005 0.005 Age - 20-29 0.109 0.106 0.114 0.145 Age - 30-39 0.195 0.177 0.254 0.277 Age - 50-59 0.185 0.190 0.163 0.189 Age - 60 and older 0.292 0.323 0.204 0.127 Female 0.266 0.276 0.209 0.232 Recent migrant 0.056 0.059 0.043 0.057 Inactive 0.284 0.296 0.241 0.229 Unemployed 0.060 0.035 0.102 0.386 Secondary - Incomplete 0.152 0.175 0.080 0.049 Further Education - Incomplete 0.007 0.033 0.005 0.003 Secondary - Complete 0.156 0.191 0.031 0.026 Further Education - Complete 0.027 0.033 0.005 0.003 Self Employed 0.161 0.153 0.182 <td>Number of Elderly Squared</td> <td>0.520</td> <td>0.571</td> <td>0.409</td> <td>0.147</td>	Number of Elderly Squared	0.520	0.571	0.409	0.147
Age - 19 and younger 0.005 0.004 0.005 0.005 Age - 30-29 0.109 0.106 0.114 0.145 Age - 30-39 0.195 0.177 0.254 0.277 Age - 50-59 0.185 0.190 0.163 0.189 Age - 60 and older 0.293 0.323 0.204 0.127 Fernale 0.266 0.276 0.209 0.292 Recent migrant 0.056 0.059 0.043 0.057 Inactive 0.284 0.296 0.241 0.229 Primary - Complete 0.174 0.171 0.190 0.168 Secondary - Incomplete 0.174 0.171 0.190 0.002 Further Education - Incomplete 0.027 0.033 0.005 0.002 Further Education - Complete 0.027 0.033 0.005 0.003 Secondary - Complete 0.256 0.231 0.348 0.328 Pasos 0.036 0.045 0.005 0.003	Household Head				
Age - 20-29 0.109 0.106 0.114 0.145 Age - 30-39 0.195 0.177 0.254 0.277 Age - 50-59 0.185 0.190 0.163 0.189 Age - 60 and older 0.293 0.323 0.204 0.127 Female 0.266 0.276 0.209 0.292 Recent migrant 0.056 0.059 0.043 0.057 Inactive 0.284 0.296 0.241 0.229 Unemployed 0.060 0.035 0.102 0.297 Primary - Complete 0.152 0.175 0.080 0.049 Secondary - Incomplete 0.152 0.175 0.080 0.002 Further Education - Complete 0.027 0.033 0.005 0.002 University Education - Complete 0.256 0.231 0.348 0.328 Public Sector 0.107 0.116 0.053 0.012 Qualification: Operative 0.354 0.376 0.242 0.369	Age - 19 and younger	0.005	0.004	0.005	0.005
Age - 30-39 0.195 0.177 0.254 0.277 Age - 50-59 0.185 0.190 0.163 0.189 Age - 60 and older 0.293 0.323 0.204 0.127 Female 0.266 0.276 0.209 0.292 Recent migrant 0.056 0.059 0.043 0.057 Inactive 0.284 0.296 0.241 0.2297 Primary - Complete 0.313 0.283 0.432 0.386 Secondary - Complete 0.152 0.175 0.080 0.049 Further Education - Incomplete 0.009 0.010 0.005 0.002 University Education 0.156 0.191 0.031 0.026 Boss 0.036 0.045 0.005 0.003 Self Employed 0.161 0.153 0.182 0.212 Informal Worker 0.256 0.231 0.348 0.328 Public Sector 0.107 0.116 0.084 0.044 Qualification:	Age - 20-29	0.109	0.106	0.114	0.145
Age - 50-59 0.185 0.190 0.163 0.187 Age - 60 and older 0.293 0.323 0.204 0.127 Female 0.266 0.276 0.209 0.292 Recent migrant 0.056 0.059 0.043 0.057 Inactive 0.284 0.296 0.241 0.229 Unemployed 0.060 0.035 0.102 0.297 Primary - Complete 0.174 0.171 0.190 0.168 Secondary - Incomplete 0.174 0.175 0.080 0.049 Further Education - Incomplete 0.007 0.033 0.005 0.002 Further Education - Complete 0.027 0.033 0.005 0.003 Self Employed 0.161 0.153 0.182 0.212 Informal Worker 0.256 0.231 0.348 0.328 Public Sector 0.107 0.116 0.84 0.044 Qualification: Technician 0.111 0.135 0.322 0.012	Age - 30-39	0.195	0.177	0.254	0.277
Age - 60 and older 0.293 0.323 0.204 0.127 Fernale 0.266 0.276 0.209 0.292 Recent migrant 0.056 0.059 0.043 0.057 Inactive 0.284 0.296 0.241 0.229 Unemployed 0.060 0.035 0.102 0.297 Primary - Complete 0.174 0.171 0.190 0.168 Secondary - Incomplete 0.009 0.010 0.005 0.005 Further Education - Incomplete 0.027 0.033 0.005 0.005 University Education 0.156 0.191 0.031 0.026 Boss 0.036 0.045 0.005 0.003 Self Employed 0.161 0.153 0.182 0.212 Informal Worker 0.236 0.326 0.406 0.262 Qualification: Operative 0.336 0.326 0.404 Qualification: Professional 0.069 0.088 0.004 0.001 Spouse Not Present	Age - 50-59	0.185	0.190	0.163	0.189
Female 0.266 0.276 0.209 0.292 Recent migrant 0.056 0.059 0.043 0.057 Inactive 0.284 0.296 0.241 0.2297 Unemployed 0.060 0.035 0.102 0.297 Primary - Complete 0.174 0.171 0.190 0.168 Secondary - Incomplete 0.052 0.175 0.080 0.009 Further Education - Incomplete 0.027 0.033 0.005 0.005 University Education 0.156 0.191 0.031 0.026 Boss 0.036 0.045 0.005 0.003 Sector 0.107 0.116 0.153 0.182 0.212 Informal Worker 0.256 0.231 0.348 0.328 Public Sector 0.107 0.116 0.084 0.044 Qualification: Professional 0.069 0.88 0.004 0.012 Qualification: Professional 0.029 0.021 0.054 0.074 Primary - Complete 0.219	Age - 60 and older	0.293	0.323	0.204	0.127
Recent migrant 0.056 0.059 0.043 0.057 Inactive 0.284 0.296 0.241 0.229 Unemployed 0.060 0.035 0.102 0.297 Primary - Complete 0.313 0.283 0.432 0.386 Secondary - Incomplete 0.174 0.171 0.190 0.168 Secondary - Complete 0.027 0.033 0.005 0.002 Further Education - Incomplete 0.027 0.033 0.005 0.003 University Education 0.156 0.191 0.031 0.026 Boss 0.036 0.045 0.005 0.003 Self Employed 0.161 0.153 0.182 0.212 Informal Worker 0.256 0.231 0.348 0.328 Public Sector 0.107 0.116 0.084 0.044 Qualification: Technician 0.111 0.135 0.032 0.012 Qualification: Technician 0.111 0.136 0.034 0.074	Female	0.266	0.276	0.209	0.292
Inactive 0.284 0.296 0.241 0.229 Unemployed 0.060 0.035 0.102 0.297 Primary - Complete 0.313 0.283 0.432 0.386 Secondary - Incomplete 0.174 0.171 0.190 0.168 Secondary - Complete 0.122 0.175 0.080 0.049 Further Education - Incomplete 0.007 0.033 0.005 0.002 University Education 0.156 0.191 0.031 0.026 Boss 0.036 0.045 0.005 0.003 Self Employed 0.161 0.153 0.182 0.212 Informal Worker 0.256 0.231 0.348 0.324 Public Sector 0.107 0.116 0.084 0.044 Qualification: Professional 0.069 0.088 0.004 0.001 Spouse Not Present 0.354 0.376 0.242 0.369 Household Spouse Unemployed 0.029 0.021 0.054 <	Recent migrant	0.056	0.059	0.043	0.057
Unemployed 0.060 0.035 0.102 0.297 Primary - Complete 0.313 0.283 0.432 0.386 Secondary - Incomplete 0.174 0.171 0.190 0.168 Secondary - Complete 0.152 0.175 0.080 0.049 Further Education - Incomplete 0.009 0.010 0.005 0.002 Further Education - Complete 0.027 0.033 0.005 0.003 Boss 0.036 0.045 0.005 0.003 Self Employed 0.161 0.153 0.182 0.212 Informal Worker 0.256 0.231 0.348 0.328 Public Sector 0.107 0.116 0.084 0.044 Qualification: Operative 0.336 0.326 0.012 Qualification: Technician 0.111 0.135 0.032 0.012 Qualification: Technician 0.111 0.135 0.032 0.012 Qualification: Professional 0.069 0.088 0.004 0.001 Spou	Inactive	0.284	0.296	0.241	0.229
Primary - Complete 0.313 0.283 0.432 0.386 Secondary - Incomplete 0.174 0.171 0.190 0.168 Secondary - Complete 0.152 0.175 0.080 0.049 Further Education - Incomplete 0.027 0.033 0.005 0.002 Further Education - Complete 0.027 0.033 0.005 0.003 Boss 0.036 0.045 0.005 0.003 Boss 0.036 0.045 0.005 0.003 Self Employed 0.161 0.153 0.182 0.212 Informal Worker 0.256 0.231 0.348 0.328 Public Sector 0.107 0.116 0.084 0.044 Qualification: Operative 0.336 0.326 0.406 0.262 Qualification: Technician 0.111 0.135 0.032 0.012 Qualification: Professional 0.069 0.088 0.004 0.001 Spouse Not Present 0.354 0.376 0.242 0.369 Household Spouse Unemployed 0.021 0.054 <td>Unemployed</td> <td>0.060</td> <td>0.035</td> <td>0.102</td> <td>0.297</td>	Unemployed	0.060	0.035	0.102	0.297
Secondary - Incomplete 0.174 0.171 0.190 0.168 Secondary - Complete 0.152 0.175 0.080 0.049 Further Education - Incomplete 0.007 0.033 0.005 0.002 University Education 0.016 0.035 0.005 0.003 Boss 0.036 0.045 0.005 0.003 Self Employed 0.161 0.153 0.182 0.212 Informal Worker 0.256 0.231 0.348 0.328 Public Sector 0.107 0.116 0.084 0.044 Qualification: Operative 0.336 0.326 0.406 0.262 Qualification: Technician 0.111 0.135 0.322 0.012 Qualification: Professional 0.069 0.088 0.004 0.001 Spouse Not Present 0.354 0.376 0.242 0.369 Household Spouse Unemployed 0.021 0.054 0.074 Primary - Complete 0.113 0.107 0.150	Primary - Complete	0.313	0.283	0.432	0.386
Secondary - Complete 0.152 0.175 0.080 0.049 Further Education - Incomplete 0.009 0.010 0.005 0.002 Further Education - Complete 0.027 0.033 0.005 0.003 Boss 0.036 0.045 0.005 0.003 Self Employed 0.161 0.153 0.182 0.212 Informal Worker 0.256 0.231 0.348 0.328 Public Sector 0.107 0.116 0.084 0.044 Qualification: Operative 0.336 0.326 0.406 0.252 Qualification: Operative 0.354 0.322 0.012 Qualification: Technician 0.111 0.135 0.032 0.012 Qualification: Professional 0.069 0.088 0.004 0.001 Spouse Not Present 0.354 0.376 0.242 0.369 Household Spouse Unemployed 0.029 0.021 0.054 0.074 Primary - Complete 0.113 0.107 0.	Secondary - Incomplete	0.174	0.171	0.190	0.168
Further Education - Incomplete 0.009 0.010 0.005 0.002 Further Education - Complete 0.027 0.033 0.005 0.005 University Education 0.156 0.191 0.031 0.026 Boss 0.036 0.045 0.005 0.003 Boss 0.036 0.045 0.005 0.003 Boss 0.036 0.045 0.005 0.003 Self Employed 0.161 0.153 0.182 0.212 Informal Worker 0.256 0.231 0.348 0.328 Public Sector 0.107 0.116 0.084 0.044 Qualification: Professional 0.069 0.088 0.004 0.001 Spouse Not Present 0.354 0.376 0.242 0.369 Household Spouse U U 0.076 0.046 Secondary - Incomplete 0.113 0.107 0.150 0.096 Further - Incomplete 0.033 0.040 0.001 0.000	Secondary - Complete	0.152	0.175	0.080	0.049
Further Education - Complete 0.027 0.033 0.005 0.005 University Education 0.156 0.191 0.031 0.026 Boss 0.036 0.045 0.003 Self Employed 0.161 0.153 0.182 0.212 Informal Worker 0.256 0.231 0.348 0.328 0.044 Qualification: Operative 0.336 0.326 0.406 0.252 Qualification: Operative 0.336 0.326 0.406 0.252 Qualification: Technician 0.111 0.135 0.032 0.012 Qualification: Professional 0.069 0.088 0.004 0.001 Spouse Not Present 0.376 0.242 0.359 Household Spouse Unemployed 0.029 0.021 0.054 0.074 Primary - Complete 0.115 0.128 0.076 0.046 Secondary - Incomplete 0.113 0.107 0.150 0.096 Further - Complete 0.007 0.008 0.001 0.000 Set Employed 0.052 0.053 0.052	Further Education - Incomplete	0.009	0.010	0.005	0.002
University Education 0.156 0.191 0.031 0.026 Boss 0.036 0.045 0.005 0.003 Self Employed 0.161 0.153 0.182 0.212 Informal Worker 0.256 0.231 0.348 0.328 Public Sector 0.107 0.116 0.084 0.044 Qualification: Operative 0.336 0.326 0.406 0.262 Qualification: Technician 0.111 0.135 0.032 0.012 Qualification: Technician 0.111 0.135 0.032 0.012 Qualification: Professional 0.069 0.088 0.004 0.001 Spouse Not Present 0.354 0.376 0.242 0.369 Household Spouse Unemployed 0.029 0.021 0.054 0.074 Primary - Complete 0.115 0.128 0.076 0.046 Secondary - Incomplete 0.033 0.040 0.007 0.005 Further - Complete 0.033 0.040	Further Education - Complete	0.027	0.033	0.005	0.005
Boss 0.036 0.045 0.005 0.003 Self Employed 0.161 0.153 0.182 0.212 Informal Worker 0.256 0.231 0.348 0.328 Public Sector 0.107 0.116 0.084 0.044 Qualification: Operative 0.336 0.326 0.406 0.252 Qualification: Technician 0.111 0.135 0.032 0.012 Qualification: Technician 0.111 0.135 0.032 0.012 Qualification: Professional 0.069 0.088 0.004 0.001 Spouse Not Present 0.354 0.376 0.242 0.369 Household Spouse Unemployed 0.029 0.021 0.054 0.074 Primary - Complete 0.115 0.128 0.076 0.046 Secondary - Incomplete 0.113 0.107 0.150 0.096 Further - Complete 0.009 0.010 0.008 0.002 University 0.072 0.990 0.011<	University Education	0.156	0.191	0.031	0.026
Self Employed 0.161 0.153 0.182 0.212 Informal Worker 0.256 0.231 0.348 0.328 Public Sector 0.107 0.116 0.084 0.044 Qualification: Operative 0.336 0.326 0.406 0.262 Qualification: Technician 0.111 0.135 0.032 0.012 Qualification: Professional 0.069 0.088 0.004 0.001 Spouse Not Present 0.354 0.376 0.242 0.369 Household Spouse Unemployed 0.029 0.021 0.054 0.074 Primary - Complete 0.115 0.128 0.076 0.046 Secondary - Complete 0.115 0.128 0.076 0.046 Secondary - Incomplete 0.033 0.040 0.007 0.005 Further - Incomplete 0.009 0.010 0.008 0.002 University 0.072 0.900 0.011 0.002 Boss 0.007 0.064 0.023 <td>Boss</td> <td>0.036</td> <td>0.045</td> <td>0.005</td> <td>0.003</td>	Boss	0.036	0.045	0.005	0.003
Informal Worker 0.256 0.231 0.348 0.328 Public Sector 0.107 0.116 0.084 0.044 Qualification: Operative 0.336 0.326 0.406 0.262 Qualification: Technician 0.111 0.135 0.032 0.012 Qualification: Professional 0.069 0.088 0.004 0.001 Spouse Not Present 0.354 0.376 0.242 0.369 Household Spouse Unemployed 0.029 0.021 0.054 0.074 Primary - Complete 0.219 0.189 0.341 0.289 Secondary - Complete 0.113 0.107 0.150 0.096 Further - Complete 0.033 0.040 0.007 0.002 Iniversity 0.072 0.090 0.011 0.002 Boss 0.007 0.008 0.001 0.000 Sector 0.054 0.064 0.023 0.004 Qualification: Operative 0.054 0.064 0.023	Self Employed	0.161	0.153	0.182	0.212
Public Sector 0.107 0.116 0.084 0.044 Qualification: Operative 0.336 0.326 0.406 0.262 Qualification: Technician 0.111 0.135 0.032 0.012 Qualification: Technician 0.069 0.088 0.004 0.001 Spouse Not Present 0.354 0.376 0.242 0.369 Household Spouse U 0.029 0.021 0.054 0.074 Primary - Complete 0.219 0.189 0.341 0.289 Secondary - Complete 0.115 0.128 0.076 0.046 Secondary - Incomplete 0.113 0.107 0.150 0.096 Further - Complete 0.009 0.010 0.008 0.002 University 0.072 0.090 0.011 0.002 Boss 0.007 0.008 0.001 0.000 Sector 0.054 0.064 0.023 0.004 Qualification: Operative 0.064 0.071 0.044 <	Informal Worker	0.256	0.231	0.348	0.328
Qualification: Operative 0.336 0.326 0.406 0.262 Qualification: Technician 0.111 0.135 0.032 0.012 Qualification: Professional 0.069 0.088 0.004 0.001 Spouse Not Present 0.354 0.376 0.242 0.369 Household Spouse Unemployed 0.029 0.021 0.054 0.074 Primary - Complete 0.219 0.189 0.341 0.289 Secondary - Complete 0.115 0.128 0.076 0.046 Secondary - Incomplete 0.113 0.107 0.150 0.096 Further - Complete 0.009 0.010 0.008 0.002 University 0.072 0.090 0.011 0.002 Boss 0.007 0.008 0.001 0.000 Self Employed 0.054 0.064 0.023 0.004 Qualification: Operative 0.064 0.071 0.104 0.025 Qualification: Professional 0.055 0.068	Public Sector	0.107	0.116	0.084	0.044
Qualification: Technician 0.111 0.135 0.032 0.012 Qualification: Professional 0.069 0.088 0.004 0.001 Spouse Not Present 0.354 0.376 0.242 0.369 Household Spouse U 0.029 0.021 0.054 0.074 Primary - Complete 0.219 0.189 0.341 0.289 Secondary - Complete 0.115 0.128 0.076 0.046 Secondary - Incomplete 0.113 0.107 0.150 0.096 Further - Complete 0.033 0.040 0.007 0.005 Further - Incomplete 0.009 0.010 0.008 0.002 University 0.072 0.090 0.011 0.002 Boss 0.007 0.008 0.001 0.000 Self Employed 0.054 0.064 0.023 0.004 Qualification: Operative 0.064 0.071 0.044 0.025 Qualification: Professional 0.024 0.030 0	Qualification: Operative	0.336	0.326	0.406	0.262
Qualification: Professional 0.069 0.088 0.004 0.001 Spouse Not Present 0.354 0.376 0.242 0.369 Household Spouse 0.029 0.021 0.054 0.074 Primary - Complete 0.219 0.189 0.341 0.289 Secondary - Complete 0.011 0.007 0.006 Secondary - Complete 0.115 0.128 0.076 0.046 Secondary - Incomplete 0.113 0.107 0.150 0.096 Further - Complete 0.033 0.040 0.007 0.005 Further - Incomplete 0.009 0.010 0.008 0.002 University 0.072 0.900 0.011 0.002 Boss 0.007 0.008 0.001 0.000 Set f Employed 0.058 0.059 0.053 0.052 Informal Worker 0.107 0.107 0.109 0.102 Public Sector 0.054 0.064 0.023 0.004 Qualification: Technician 0.055 0.068 0.008 0.007 <td>Qualification: Technician</td> <td>0.111</td> <td>0.135</td> <td>0.032</td> <td>0.012</td>	Qualification: Technician	0.111	0.135	0.032	0.012
Spouse Not Present 0.354 0.376 0.242 0.369 Household Spouse	Qualification: Professional	0.069	0.088	0.004	0.001
Household SpouseUnemployed0.0290.0210.0540.074Primary - Complete0.2190.1890.3410.289Secondary - Complete0.1150.1280.0760.046Secondary - Incomplete0.1130.1070.1500.096Further - Complete0.0330.0400.0070.005Further - Incomplete0.0090.0100.0080.002University0.0720.0900.0110.002Boss0.0070.0080.0010.000Self Employed0.0580.0590.0530.052Informal Worker0.1070.1070.1090.102Public Sector0.0540.0640.0230.004Qualification: Technician0.0550.0680.0080.007Qualification: Technician0.0240.0300.0010.000Geographic locationUniversite0.0440.0340.0740.100Cuyo0.0610.0580.0750.0530.053Pampeana0.2300.2310.2270.234Patagonia0.0260.0280.0180.027	Spouse Not Present	0.354	0.376	0.242	0.369
Unemployed 0.029 0.021 0.054 0.074 Primary - Complete 0.219 0.189 0.341 0.289 Secondary - Complete 0.115 0.128 0.076 0.046 Secondary - Incomplete 0.113 0.107 0.150 0.096 Further - Complete 0.033 0.040 0.007 0.005 Further - Incomplete 0.009 0.010 0.008 0.002 University 0.072 0.090 0.011 0.002 Boss 0.007 0.008 0.001 0.000 Self Employed 0.058 0.059 0.053 0.052 Informal Worker 0.107 0.107 0.109 0.102 Public Sector 0.054 0.064 0.023 0.004 Qualification: Deprative 0.055 0.068 0.0007 0.000 Geographic location 0.024 0.030 0.001 0.000 Geographic location 0.044 0.034 0.074 0.100 Cuyo	Household Spouse				
Primary - Complete 0.219 0.189 0.341 0.289 Secondary - Complete 0.115 0.128 0.076 0.046 Secondary - Incomplete 0.113 0.107 0.150 0.096 Further - Complete 0.033 0.040 0.007 0.005 Further - Incomplete 0.009 0.010 0.008 0.002 University 0.072 0.990 0.011 0.002 Boss 0.007 0.008 0.001 0.000 Self Employed 0.058 0.059 0.053 0.052 Informal Worker 0.107 0.107 0.109 0.102 Public Sector 0.054 0.064 0.023 0.004 Qualification: Operative 0.064 0.071 0.044 0.025 Qualification: Professional 0.024 0.030 0.001 0.000 Geographic location 0.044 0.034 0.074 0.100 Qualification: Professional 0.024 0.030 0.001 0.000 <td>Unemployed</td> <td>0.029</td> <td>0.021</td> <td>0.054</td> <td>0.074</td>	Unemployed	0.029	0.021	0.054	0.074
Secondary - Complete 0.115 0.128 0.076 0.046 Secondary - Incomplete 0.113 0.107 0.150 0.096 Further - Complete 0.033 0.040 0.007 0.005 Further - Incomplete 0.009 0.010 0.008 0.002 University 0.072 0.090 0.011 0.002 Boss 0.007 0.008 0.001 0.000 Self Employed 0.058 0.059 0.053 0.052 Informal Worker 0.107 0.107 0.109 0.102 Public Sector 0.054 0.064 0.023 0.004 Qualification: Operative 0.055 0.068 0.008 0.007 Qualification: Technician 0.055 0.068 0.008 0.007 Qualification: Professional 0.024 0.030 0.001 0.000 Geographic location U U 0.024 0.034 0.074 0.100 Cuyo 0.061 0.058 0.075	Primary - Complete	0.219	0.189	0.341	0.289
Secondary - Incomplete 0.113 0.107 0.150 0.096 Further - Complete 0.033 0.040 0.007 0.005 Further - Incomplete 0.009 0.010 0.008 0.002 University 0.072 0.090 0.011 0.002 Boss 0.007 0.008 0.001 0.000 Self Employed 0.058 0.059 0.053 0.052 Informal Worker 0.107 0.107 0.109 0.102 Public Sector 0.054 0.064 0.023 0.004 Qualification: Operative 0.064 0.071 0.044 0.025 Qualification: Technician 0.055 0.068 0.000 0.000 Geographic location 0.024 0.030 0.001 0.000 Geographic location 0.080 0.068 0.121 0.122 Noreste 0.080 0.068 0.121 0.122 Noreste 0.044 0.034 0.074 0.100 C	Secondary - Complete	0.115	0.128	0.076	0.046
Further - Complete 0.033 0.040 0.007 0.005 Further - Incomplete 0.009 0.010 0.008 0.002 University 0.072 0.090 0.011 0.002 Boss 0.007 0.008 0.001 0.000 Self Employed 0.058 0.059 0.053 0.052 Informal Worker 0.107 0.107 0.109 0.102 Public Sector 0.054 0.064 0.023 0.004 Qualification: Operative 0.064 0.071 0.044 0.025 Qualification: Technician 0.055 0.068 0.000 0.000 Geographic location 0.024 0.030 0.001 0.000 Geographic location U U 0.024 0.030 0.001 0.000 Geographic location U U 0.024 0.034 0.074 0.100 Cuyo 0.061 0.058 0.075 0.053 0.053 Pampeana 0.230	Secondary - Incomplete	0.113	0.107	0.150	0.096
Further - Incomplete 0.009 0.010 0.008 0.002 University 0.072 0.090 0.011 0.002 Boss 0.007 0.008 0.001 0.002 Boss 0.007 0.008 0.001 0.002 Boss 0.007 0.008 0.001 0.000 Self Employed 0.058 0.059 0.053 0.052 Informal Worker 0.107 0.107 0.109 0.102 Public Sector 0.054 0.064 0.023 0.004 Qualification: Operative 0.064 0.071 0.044 0.025 Qualification: Technician 0.055 0.068 0.007 Qualification: Professional 0.024 0.030 0.001 0.000 Geographic location V V 0.034 0.074 0.100 Noroeste 0.080 0.068 0.121 0.122 Noreste 0.044 0.034 0.074 0.100 Cuyo 0.061 0.058<	Further - Complete	0.033	0.040	0.007	0.005
University 0.072 0.090 0.011 0.002 Boss 0.007 0.008 0.001 0.000 Self Employed 0.058 0.059 0.053 0.052 Informal Worker 0.107 0.107 0.109 0.102 Public Sector 0.054 0.064 0.023 0.004 Qualification: Operative 0.064 0.071 0.044 0.025 Qualification: Technician 0.055 0.068 0.000 0.000 Geographic location 0.024 0.030 0.001 0.000 Geographic location 0.080 0.068 0.121 0.122 Noreste 0.080 0.068 0.121 0.122 Noreste 0.044 0.034 0.074 0.100 Cuyo 0.061 0.058 0.075 0.053 Pampeana 0.230 0.231 0.227 0.234 Patagonia 0.026 0.028 0.018 0.027	Further - Incomplete	0.009	0.010	0.008	0.002
Boss 0.007 0.008 0.001 0.000 Self Employed 0.058 0.059 0.053 0.052 Informal Worker 0.107 0.107 0.109 0.102 Public Sector 0.054 0.064 0.023 0.004 Qualification: Operative 0.064 0.071 0.044 0.025 Qualification: Technician 0.055 0.068 0.008 0.007 Qualification: Professional 0.024 0.030 0.001 0.000 Geographic location Noroeste 0.080 0.068 0.121 0.122 Noreste 0.044 0.034 0.074 0.100 Cuyo 0.061 0.058 0.075 0.053 Pampeana 0.230 0.231 0.227 0.234 Patagonia 0.026 0.028 0.018 0.027	University	0.072	0.090	0.011	0.002
Self Employed 0.058 0.059 0.053 0.052 Informal Worker 0.107 0.107 0.109 0.102 Public Sector 0.054 0.064 0.023 0.004 Qualification: Operative 0.064 0.071 0.044 0.025 Qualification: Technician 0.055 0.068 0.008 0.007 Qualification: Professional 0.024 0.030 0.001 0.000 Geographic location Noroeste 0.080 0.068 0.121 0.122 Noreste 0.044 0.034 0.074 0.100 Cuyo 0.061 0.058 0.075 0.053 Pampeana 0.230 0.231 0.227 0.234 Patagonia 0.026 0.028 0.018 0.027	Boss	0.007	0.008	0.001	0.000
Informal Worker 0.107 0.107 0.109 0.102 Public Sector 0.054 0.064 0.023 0.004 Qualification: Operative 0.064 0.071 0.044 0.025 Qualification: Technician 0.055 0.068 0.008 0.007 Qualification: Professional 0.024 0.030 0.001 0.000 Geographic location 0.080 0.068 0.121 0.122 Noreste 0.044 0.034 0.074 0.100 Cuyo 0.061 0.058 0.075 0.053 Pampeana 0.230 0.231 0.227 0.234 Patagonia 0.026 0.028 0.018 0.027	Self Employed	0.058	0.059	0.053	0.052
Public Sector 0.054 0.064 0.023 0.004 Qualification: Operative 0.064 0.071 0.044 0.025 Qualification: Technician 0.055 0.068 0.008 0.007 Qualification: Professional 0.024 0.030 0.001 0.000 Geographic location 0.080 0.068 0.121 0.122 Noreste 0.044 0.034 0.074 0.100 Cuyo 0.061 0.058 0.075 0.053 Pampeana 0.230 0.231 0.227 0.234 Patagonia 0.026 0.028 0.018 0.027	Informal Worker	0.107	0.107	0.109	0.102
Qualification: Operative 0.064 0.071 0.044 0.025 Qualification: Technician 0.055 0.068 0.008 0.007 Qualification: Professional 0.024 0.030 0.001 0.000 Geographic location 0.080 0.068 0.121 0.122 Noreste 0.044 0.034 0.074 0.100 Cuyo 0.061 0.058 0.075 0.053 Pampeana 0.230 0.231 0.227 0.234	Public Sector	0.054	0.064	0.023	0.004
Qualification: Technician 0.055 0.068 0.008 0.007 Qualification: Professional 0.024 0.030 0.001 0.000 Geographic location 0.080 0.068 0.121 0.122 Noreste 0.044 0.034 0.074 0.100 Cuyo 0.061 0.058 0.075 0.053 Pampeana 0.230 0.231 0.227 0.234 Patagonia 0.026 0.028 0.018 0.027	Qualification: Operative	0.064	0.071	0.044	0.025
Qualification: Professional 0.024 0.030 0.001 0.000 Geographic location 0.080 0.068 0.121 0.122 Noroeste 0.044 0.034 0.074 0.100 Cuyo 0.061 0.058 0.075 0.053 Pampeana 0.230 0.231 0.227 0.234 Patagonia 0.026 0.028 0.018 0.027	Qualification: Technician	0.055	0.068	0.008	0.007
Geographic location 0.080 0.068 0.121 0.122 Noreste 0.044 0.034 0.074 0.100 Cuyo 0.061 0.058 0.075 0.053 Pampeana 0.230 0.231 0.227 0.234 Patagonia 0.026 0.028 0.018 0.027	Qualification: Professional	0.024	0.030	0.001	0.000
Noroeste 0.080 0.068 0.121 0.122 Noreste 0.044 0.034 0.074 0.100 Cuyo 0.061 0.058 0.075 0.053 Pampeana 0.230 0.231 0.227 0.234 Patagonia 0.026 0.028 0.018 0.027	Geographic location				
Noreste 0.044 0.034 0.074 0.100 Cuyo 0.061 0.058 0.075 0.053 Pampeana 0.230 0.231 0.227 0.234 Patagonia 0.026 0.028 0.018 0.027	Noroeste	0.080	0.068	0.121	0.122
Cuyo 0.061 0.058 0.075 0.053 Pampeana 0.230 0.231 0.227 0.234 Patagonia 0.026 0.028 0.018 0.027	Noreste	0.044	0.034	0.074	0.100
Pampeana 0.230 0.231 0.227 0.234 Patagonia 0.026 0.028 0.018 0.027	Cuyo	0.061	0.058	0.075	0.053
Patagonia 0.026 0.028 0.018 0.027	Pampeana	0.230	0.231	0.227	0.234
	Patagonia	0.026	0.028	0.018	0.027

Table C.4: Poverty Profile, Urban Argentina, October 1998

	Total	Non-poor	Poor	Extreme Poor
Log Normalised Income	0.452	1.024	-0.360	-2.254
Normalised Income	2.771	3.650	0.718	0.195
Demographic Composition				
Number of Infants (ages 0-5)	0.364	0.246	0.607	0.772
Number of Infants squared	0.618	0.365	1.111	1.546
Number of Children (ages 6-14)	0.579	0.352	1.018	1.420
Number of Children Squared	1.281	0.624	2.378	4.063
Number of Youths (ages 15-24)	0.642	0.516	0.942	0.998
Number of Youths Squared	1.324	0.972	2.079	2.479
Number of Adults (ages 25-64)	1.529	1.436	1.805	1.681
Number of Adults Squared	3.260	3.036	4.027	3.442
Number of Elderly (ages 65+)	0.345	0.413	0.208	0.110
Number of Elderly Squared	0.512	0.617	0.296	0.154
Household Head				
Age - 19 and younger	0.006	0.006	0.006	0.005
Age - 20-29	0.108	0.102	0.109	0.154
Age - 30-39	0.196	0.181	0.225	0.259
Age - 50-59	0.188	0.186	0.199	0.174
Age - 60 and older	0.289	0.337	0.185	0.133
Female	0.287	0.306	0.227	0.266
Recent migrant	0.055	0.062	0.036	0.038
Inactive	0.288	0.320	0.212	0.191
Unemployed	0.100	0.050	0.156	0.370
Primary - Complete	0.315	0.272	0.420	0.436
Secondary - Incomplete	0.162	0.155	0.192	0.152
Secondary - Complete	0.170	0.199	0.106	0.077
Further Education - Incomplete	0.011	0.013	0.007	0.004
Further Education - Complete	0.034	0.043	0.007	0.012
University Education	0.159	0.209	0.039	0.019
Boss	0.032	0.042	0.011	0.006
Self Employed	0.159	0.141	0.199	0.210
Informal Worker	0.256	0.222	0.351	0.324
Public Sector	0.107	0.121	0.080	0.053
Qualification: Operative	0.307	0.299	0.373	0.240
Qualification: Technician	0.109	0.139	0.037	0.016
Qualification: Professional	0.061	0.084	0.002	0.002
Spouse Not Present	0.370	0.402	0.276	0.322
Household Spouse				
Unemployed	0.039	0.029	0.058	0.076
Primary - Complete	0.207	0.162	0.327	0.305
Secondary - Complete	0.118	0.135	0.084	0.057
Secondary - Incomplete	0.104	0.089	0.143	0.137
Further - Complete	0.041	0.053	0.014	0.003
Further - Incomplete	0.010	0.011	0.009	0.003
University	0.07 7	0.101	0.017	0.010
Boss	0.007	0.010	0.001	0.000
Self Employed	0.051	0.046	0.065	0.060
Informal Worker	0.103	0.089	0.139	0.135
Public Sector	0.056	0.069	0.027	0.020
Qualification: Operative	0.066	0.075	0.051	0.031
Qualification: Technician	0.055	0.072	0.014	0.004
Qualification: Professional	0.025	0.034	0.002	0.000
Geographic location				
Noroeste	0.087	0.076	0.119	0.115
Noreste	0.047	0.036	0.066	0.094
Сиуо	0.064	0.062	0.073	0.059
Pampeana	0.236	0.239	0.225	0.233
Patagonia	0.030	0.034	0.020	0.018
Courses Authoris anti-untions hand an EDIT		A A A A A A A A A A A A A A A A A A A		

Table C.5: Poverty Profile, Urban Argentina, October 2001

	Total	Non-poor	Poor	Extreme Poor
Log Normalised Income	0.029	0.871	-0.379	-2.137
Normalised Income	1.975	3.039	0.704	0.202
Demographic Composition				
Number of Infants (ages 0-5)	0.368	0.201	0.527	0.699
Number of Infants squared	0.627	0.280	0.930	1.349
Number of Children (ages 6-14)	0.572	0.276	0.802	1.224
Number of Children Squared	1.269	0.473	1.693	3.269
Number of Youths (ages 15-24)	0.640	0.454	0.831	0.988
Number of Youths Squared	1.310	0.859	1.697	2.256
Number of Adults (ages 25-64)	1.534	1.400	1.729	1.713
Number of Adults Squared	3.304	2.977	3.878	3.613
Number of Elderly (ages 65+)	0.346	0.438	0.279	0.136
Number of Elderly Squared	0.517	0.651	0.444	0.183
Household Head				
Age - 19 and younger	0.006	0.005	0.005	0.008
Age - 20-29	0.118	0.104	0.128	0.151
Age - 30-39	0.192	0.166	0.218	0.240
Age - 50-59	0.189	0.186	0.194	0.189
Age - 60 and older	0.289	0.364	0.218	0.141
Female	0.287	0.323	0.225	0.253
Recent migrant	0.053	0.061	0.048	0.035
Inactive	0.289	0.341	0.233	0.196
Unemployed	0.124	0.052	0.137	0.339
Primary - Complete	0.311	0.250	0.393	0.403
Secondary - Incomplete	0.163	0.138	0.207	0.186
Secondary - Complete	0.173	0.211	0.142	0.093
Further Education - Incomplete	0.014	0.018	0.008	0.009
Further Education - Complete	0.035	0.050	0.019	0.009
University Education	0.162	0.239	0.068	0.036
Boss	0.025	0.038	0.009	0.003
Self Employed	0.150	0.127	0.162	0.207
Informal Worker	0.251	0.206	0.303	0.327
Public Sector	0.115	0.135	0.097	0.072
Qualification: Operative	0.294	0.277	0.372	0.249
Qualification: Technician	0.101	0.144	0.057	0.021
Qualification: Professional	0.058	0.097	0.007	0.002
Spouse Not Present	0.372	0.432	0.266	0.319
Household Spouse				
Unemployed	0.041	0.025	0.047	0.081
Primary - Complete	0.208	0.143	0.294	0.306
Secondary - Complete	0.122	0.137	0.118	0.080
Secondary - Incomplete	0.099	0.072	0.148	0.125
Further - Complete	0.041	0.058	0.022	0.011
Further - Incomplete	0.011	0.013	0.013	0.005
University	0.074	0.110	0.031	0.014
Boss	0.005	0.008	0.000	0.000
Self Employed	0.049	0.037	0.062	0.068
Informal Worker	0.103	0.078	0.136	0.141
Public Sector	0.057	0.073	0.035	0.033
Qualification: Operative	0.068	0.075	0.062	0.054
Qualification: Technician	0.050	0.076	0.018	0.008
Oualification: Professional	0.021	0.035	0.002	0.000
Geographic location				
Noroeste	0.086	0.069	0.111	0.109
Noreste	0.049	0.034	0.060	0.082
Cuyo	0.067	0.063	0.074	0.069
Pampeana	0.236	0.236	0.238	0.236
Patagonia	0.030	0.035	0.025	0.019
Source: Author's estimations based on EPH h	ousehold	survey data (INDEC)		

Table C.6: Poverty Profile, Urban Argentina, May 2002

APPENDIX D

FERTILITY AND LABOUR SUPPLY: ROBUSTNESS CHECKS

D.1 ROBUSTNESS CHECKS WITH DIFFERENT COVARIATES, SAMPLES AND VARIABLE DEFINITIONS

This Appendix provides additional findings and robustness checks of the benchmark results of Chapter 7 on the effects of fertility on female labour supply in Argentina. This Section corroborates that the results presented in Section 7.2.2 (page 176) are robust to differences in the underlying samples, in the specification of the model and in the definition of the variables.

The first set of robustness checks consists of adding covariates *X* to the estimation of Equation 5.7 (page 137). Table D.1 estimates the same model with two instruments as in Table 7.3 (page 181), with *Two boys* and *Two girls* as the instruments, and *Age*, *Age at first birth*, and the sex of the first two children as exogenous covariates, but it adds indicators for Argentina's 522 municipalities. These geographic controls capture the wide variation of socioeconomic conditions within the country, with a few relatively rich cities and extremely poor urban and rural areas. The results in Table D.1 are very close to those discussed in Section 7.2.2 (page 176): the coefficients are slightly smaller (in absolute value) once the geographic controls are introduced, but the differences are not significant. This finding reinforces the plausibility of the randomness of the instrument, since it indicates that it is orthogonal to the additional geographic controls, as indicated by the similarity of the coefficients with those in previous Tables and by the p-values of the Sargan overidentification tests.

Table D.2 presents the estimation of the same 2SLS model with demo-

	Instrumented: Number of children Instrumented: More than two ch					children		
Dependent variable:	Complete	Married	AE Sample	AE Sample.	Complete	Married	AE Sample	AE Sample,
Worked for pay	Sample		···· oumpre	married	Sample		·in sumpre	married
OLS estimates								
OLS	-0.0435	-0.0393	-0.0443	-0.0401	-0.0886	-0.0752	-0.0931	-0.0781
Control for sex of 1st child	[0.0005]***	[0.0006]***	[0.0005]***	[0.0006]***	[0.0013]***	[0.0014]***	[0.0013]***	[0.0015]***
OLS	-0.0435	-0.0393	-0.0443	-0.0401	-0.0886	-0.0752	-0.0931	-0.0782
Controls for sexes of 1st & 2nd children	[0.0005]***	[0.0006]***	[0.0005]***	[0.0006]***	[0.0013]***	[0.0014]***	[0.0013]***	[0.0015]***
IV Estimates			–					
IV: Same Sex	-0.0424	-0.0450	-0.0455	-0.0525	-0.0786	-0.0807	-0.0841	-0.0940
Controls for sexes of 1st & 2nd children	[0.0177]**	[0.0183]**	[0.0171]***	[0.0177]***	[0.0328]**	[0.0329]**	[0.0316]***	[0.0317]***
DWH p-value	0.9499	0.7552	0.9447	0.4821	0.7588	0.8667	0.7767	0.6159
IV: Two Boys and Two Girls	-0.0317	-0.0359	-0.0331	-0.0417	-0.0666	-0.0709	-0.0701	-0.0821
Control for sex of 1st child	[0.0165]*	[0.0173]**	[0.0159]**	[0.0167]**	[0.0318]**	[0.0320]**	[0.0306]**	[0.0308]***
Sargan p-value	0.0978	0.1253	0.0437	0.0621	0.1545	0.1888	0.0778	0.1075
DWH p-value	0.4746	0.8443	0.4798	0.9219	0.4875	0.8932	0.4520	0.8961
Spouse worked for pay								
OLS		-0.0059		-0.0051		-0.0085		-0.0065
Controls for sexes of 1st & 2nd children		[0.0002]***		[0.0003]***		[0.0006]***		[0.0006]***
IV: Same Sex		0.0029		-0.0014		0.0053		-0.0026
Controls for sexes of 1st & 2nd children		[0.0079]		[0.0074]		[0.0147]		[0.0132]
DWH p-value		0.2680		0.6189		0.3444		0.7655
Observations	653,213	497,194	599,941	456,437	653,213	497,194	599,941	456,437
Geographic Controls				522 Mun	icipalities	-		
Note: Standard errors in brackets. * s	ignificant at 1	0%; ** signi	ficant at 5%;	*** significat	nt at 1%. All 1	regressions ir	clude the ag	e and the ag

Table D.1: Robustness Checks: OLS and Two-Stage Least Squares Results with Geographic Controls

Note: Standard errors in brackets. * significant at 10%; ** significant at 5%; *** significant at 1%. All regressions include the age and the age at first birth of the women or her spouse, in addition to the sex of the first and second children (where indicated). The samples correspond to women aged 21-35 with two or more children aged 18 or younger from the extended questionnaire sample of the 1991 Census, as described in the text.

graphic and geographic controls, with the age variable replaced by cohort dummies and with additional covariates in X. The first panel in the Table presents the results with the three education level indicators of Section 7.3.2 (page 186) included as right hand side variables. Most of the coefficients are lower in absolute value than their counterparts in Table 7.3, but again these differences are not statistically significant. The main difference is that the Sargan exogeneity null is now rejected (at the 5 percent level) for four of the eight coefficients estimated, indicating that the education variables are probably endogenous in the labour supply model. The results in the other three panels are substantially the same: in the second panel, an indicator for rural areas is added to the demographic, educational and geographic controls, while the third panel adds a household indicator of structural poverty constructed by INDEC. Finally, the last panel computes the model with the woman and her husband's education status as right hand side variables. All these models are consistent with the findings presented in Table 7.3, although the overidentification tests indicate that the extra covariates are, in most cases, endogenous.

While the addition of covariates does not affect the essence of the benchmark results, there might still be an impact from the definitions of the underlying samples and the variables, and from measurement error. These issues are addressed in Table D.3.

In the first panel of Table D.3, the dependent variable *Worked for pay* was replaced by an indicator of labour force participation, *Active labour force status*, equal to 1 for women who worked for pay or were unemployed during the reference week, and 0 otherwise. It is thus possible to test the robustness of the main results with respect to the definition of the dependent variable. In the main results of Chapters 7 and 8, the *Worked for pay* indicator was employed because it was considered to be more reliable, since the distinction between unemployment and inactivity might not be reflected correctly in the data. This is because the 1991 Census data was collected mostly by teachers and other civil servants, unlike smaller household surveys like EPH that are carried out by specialised staff who are better trained and use a more specific questionnaire on labour market issues. Gregg and Wadsworth (1999a) provide a discussion of the sometimes subtle distinction between the unemployed and the inactive. In any case, the difference between the

	Instrumented: Number of children				Instrumented: More than two children			
Dependent variable: Worked for pay	Complete Sample	Married	AE Sample	AE Sample, married	Complete Sample	Married	AE Sample	AE Sample, married
IV: Two Boys and Two Girls Controls: education level and birth cohort Sargan p-value DWH p-value OLS with same controls	-0.0261 [0.0157]* 0.0436 0.4751 -0.0373 [0.0005]***	-0.0342 [0.0161]** 0.0777 0.9916 -0.0340 [0.0006]***	-0.0303 [0.0153]** 0.0144 0.6158 -0.0380 [0.0005]***	-0.0420 [0.0156]*** 0.0354 0.6367 -0.0347 [0.0006]***	-0.0561 [0.0301]* 0.0677 0.4740 -0.0776 [0.0012]***	-0.0679 [0.0298]** 0.1212 0.9853 -0.0684 [0.0013]***	-0.0652 [0.0292]** 0.0264 0.5765 -0.0814 [0.0012]***	-0.0828 [0.0288]*** 0.0651 0.6840 -0.0711 [0.0014]***
IV: Two Boys and Two Girls Controls: education level, birth cohort and urban dummy Sargan p-value DWH p-value OLS with same controls	-0.0265 [0.0157]* 0.0433 0.5034 -0.0370 [0.0005]***	-0.0344 [0.0161]** 0.0780 0.9721 -0.0339 [0.0006]***	-0.0307 [0.0153]** 0.0141 0.6421 -0.0377 [0.0005]***	-0.0422 [0.0156]*** 0.0352 0.6235 -0.0346 [0.0006]***	-0.0571 [0.0300]* 0.0677 0.4984 -0.0774 [0.0012]***	-0.0684 [0.0297]** 0.1221 0.9993 -0.0684 [0.0013]***	-0.0658 [0.0291]** 0.0260 0.5973 -0.0812 [0.0012]***	-0.0832 [0.0288]*** 0.0649 0.6745 -0.0711 [0.0014]***
IV: Two Boys and Two Girls Controls: education level, birth cohort, NBI and urban dummy Sargan p-value DWH p-value OLS with same controls	-0.0271 [0.0159]* 0.0445 0.4621 -0.0388 [0.0005]***	-0.0348 [0.0162]** 0.0800 0.9968 -0.0349 [0.0006]***	-0.0311 [0.0154]** 0.0144 0.5744 -0.0398 [0.0005]***	-0.0426 [0.0157]*** 0.0365 0.6614 -0.0358 [0.0006]***	-0.0574 [0.0302]* 0.0684 0.4965 -0.0779 [0.0012]***	-0.0685 [0.0298]** 0.1220 0.9952 -0.0683 [0.0014]***	-0.0661 [0.0292]** 0.0262 0.5899 -0.0819 [0.0012]***	-0.0833 [0.0288]*** 0.0651 0.6717 -0.0711 [0.0014]***
IV: Two Boys and Two Girls Controls: woman and spouse education, spouse age Sargan p-value DWH p-value OLS with same controls		-0.0297 [0.0155]* 0.0277 0.7961 -0.0337 [0.0005]***	а <u>н на на на на на на на</u>	-0.0416 [0.0156]*** 0.0365 0.6742 -0.0351 [0.0006]***		-0.0621 [0.0294]** 0.0434 0.8105 -0.0691 [0.0012]***		-0.0820 [0.0288]*** 0.0667 0.7085 -0.0713 [0.0014]***
Observations Geographic Controls	653,213	497,194	599,941	456,437 522 Muni	653,213 cipalities	497,194	599,941	456,437

Table D.2: Robustness Checks: OLS and Two-Stage Least Squares Results with Additional Covariates

Note: Standard errors in brackets. * significant at 10%; ** significant at 5%; *** significant at 1%. All regressions include the age (as cohort indicators in this Table) and the age at first birth of the women or her spouse, in addition to the sex of the first child. The education level indicators correspond to some primary, some secondary and some tertiary education. NBI is an indicator of structural poverty (basic needs) defined by INDEC. The samples correspond to women aged 21-35 with two or more children aged 18 or younger from the extended questionnaire sample of the 1991 Census, as described in the text.

two measures is relatively small, with 31.38 percent of the women in the complete sample working for pay and 33.82 percent in the labour force. The results with *Active labour force status* as the dependent variable are close (and qualitatively the same) to those using *Worked for pay* (as in Table 7.3), both in terms of levels and significance.

Another source of concern might be the narrow age group – 21 to 35 years old – for women in the final samples. The second panel in Table D.3 presents the results obtained by adding women in the age group 36-45 that complied with the selection criteria of Section 5.4.1 (page 142). This implies an additional 330,000 extra observations in the full samples, and around 280,000 in the married samples. These results (with *Worked for pay* as the dependent variable) are again similar to the those in Table 7.3. In fact, it can be noted that the increase in sample sizes boosts the precision of the estimates, since all the coefficients are significant at least at the 5 percent level. This suggests that the inclusion of the extra age group does not attenuate nor bias the results, but instead increases their precision.

Finally, some important adjustments were made to match women with their children in the Census data: Section 5.4.1 states that around 100,000 observations were discarded from the final sample because the total number of surviving children did not correspond to the number of matched children in the household. This was the most important adjustment in the sample, due to the lack of detail of the kinship variable. The final panel in Table D.3 estimates the same model as in Table 7.3 without discarding these 100,000 observations. The coefficients are smaller in absolute value than those in Table 7.3, but in the same range as those discussed in this Section. The measurement error in the fertility variables does affect the results in an important manner. Alternatively, this might be suggesting that children in the household seem to affect the labour supply of women even if they are not their own children.

D.2 ESTIMATION WITH AN ALTERNATIVE DATASET

Instrumental variables estimation with a relatively weak instruments such as the *Same sex* indicator usually requires large sample sizes to obtain coefficients estimated with precision, which is why Angrist and Evans (1998)

	Instrumented: Number of children				Instrumented: More than two children					
Dependent variable: Worked for pay	Complete Sample	Married	AE Sample	AE Sample, married	Complete Sample	Married	AE Sample	AE Sample, married		
Using labor force status instead of worked for pay as dependent variable										
IV: Same Sex	-0.0403 [0.0182]**	-0.0457 [0.0188]**	-0.0438 [0.0176]**	-0.0530 [0.0182]***	-0.0748 [0.0338]**	-0.0820 [0.0338]**	-0.0809 [0.0326]**	-0.0948 [0.0327]***		
DWH p-value	0.7211	0.8587	0.8189	0.5963	0.5396	0.9779	0.5430	0.7584		
OLS with same controls	-0.0468 [0.0005]***	-0.0423 [0.0006]***	-0.0478 [0.0005]***	-0.0433 [0.0007]***	-0.0956 [0.0013]***	-0.0811 [0.0015]***	-0.1007 [0.0013]***	-0.0847 [0.0015]***		
Observations	653,213	497,194	599,941	456,437	653,213	497,194	599,941	456,437		
	Women 21-45									
IV: Same Sex	-0.0386 [0.0126]***	-0.0462 [0.0129]***	-0.0403 [0.0123]***	-0.0499 [0.0127]***	-0.0719 [0.0235]***	-0.0840 [0.0236]***	-0.0749 [0.0229]***	-0.0907 [0.0231]***		
DWH p-value	0.7802	0.4824	0.8241	0.3464	0.7344	0.3551	0.6995	0.2641		
OLS with same controls	-0.0421 [0.0004]***	-0.0371 [0.0005]***	-0.0431 [0.0004]***	-0.0379 [0.0005]***	-0.0798 [0.0010]***	-0.0622 [0.0012]***	-0.0838 [0.0011]***	-0.0649 [0.0012]***		
Observations	989,113	774,059	932,166	730,542	989,113	774,059	932,166	730,542		
	Women with mismatching surviving children and children in household									
IV: Same Sex	-0.0350 [0.0171]**	-0.0442 [0.0179]**	-0.0378 [0.0167]**	-0.0501 [0.0174]***	-0.0698 [0.0341]**	-0.0827 [0.0335]**	-0.0732 [0.0328]**	-0.0920 [0.0323]***		
DWH p-value	0.8000	0.7852	0.9089	0.5567	0.4325	0.9721	0.3897	0.8243		
OLS with same controls	-0.0393 [0.0004]***	-0.0394 [0.0005]***	-0.0397 [0.0004]***	-0.0399 [0.0005]***	-0.0966 [0.0012]***	-0.0816 [0.0014]***	-0.1015 [0.0012]***	-0.0848 [0.0014]***		
Observations	748,157	541,529	690,103	499,000	748,157	541,529	690,103	499,000		
Note: Standard errors in brackets. * significant at 10%; ** significant at 5%; *** significant at 1%. All regressions include the age and the age at first birth of the women or her spouse, in addition to the sex of the first and second children (where indicated). The samples correspond to women aged 21-35 (except where noted) with two or more children aged 18 or younger from the extended questionnaire sample of the 1991 Census, as described in the text.										

Table D.3: Robustness Checks: Alternative De	ependent Variable and Different Sam	ple Definitions
--	-------------------------------------	-----------------

use datasets from the United States Census, and this Part employs the 1991 Census from Argentina. Applications of the "same sex" strategy to smaller datasets, as in Iacovou's (2001) study of the United Kingdom, tend to produce instrumental variables estimates with high standard errors.

This can be shown by following Dee and Evans' (2003) analysis of the power of the statistical significance test for IV coefficients. When a reduced form OLS regression of Y on Z generates unbiased estimates, the coefficient is $\phi_{RF} = E[Y|Z = 1] - E[Y|Z = 0]$. This estimate is statistically significant if the ratio of ϕ_{RF} and its standard error σ_{RF} is greater than a cut-off value η from the normal distribution – for instance, $\eta = 1.96$ for a 5 percent level of significance in a two-sided test. Defining $p_1 =$ E[Y|Z = 1] and $p_0 = E[Y|Z = 0]$, and with treatment and control groups defined by Z with $\frac{1}{2}N$ observations each, σ_{RF}^2 is approximately equal to $[p_1(1-p_1)+p_0(1-p_0)]/\frac{1}{2}N$. Since by definition $p_1 = p_0 + \phi_{RF}$, using these expressions in $|\phi_{RF}/\sigma_{RF}| > \eta$ allows to solve for the minimum number of observations N needed for a statistically significant value of ϕ_{RF} . For the Argentine Census dataset employed in Part II, there are just enough observations for testing the IV coefficients at the 1 percent level of significance for most samples, and there are largely more than enough observations for the usual 5 percent tests for all four samples.¹

However, using smaller alternative datasets covering the same region and period is valuable as a validation exercise, since at least the results from first-stages and OLS estimations can be compared with the main application. While no other Census dataset is available for Argentina, Table D.4 presents the estimation of a model equivalent to the one in Table 7.3 (page 181), but using a sample from INDEC's 1987 Expenditure Survey ("Encuesta Nacional de Gasto de los Hogares"–ENGH), which is the basis of the expenditure figures in Table 6.4 (page 166). While this survey did not cover smaller rural areas, it only excluded very small and isolated communities. It is thus broadly representative at the national level, and comparable to the 1991 Census data.² The four samples in the Table were constructed fol-

¹The ratios of available to needed observations for a two-sided 1 percent test are 0.98, 1.01, 1.09 and 1.37 for the *Complete, Married, AE Sample* and *AE Sample, married* respectively. For a two-sided 5 percent test, the ratios are 1.7, 1.73, 1.88 and 2.37.

²The data from the May 1991 Household Survey ("Encuesta Permanente de Hogares," presented in Section 1.2, page 22), corresponding to the month of the 1991 Census, is

lowing the criteria set in Section 5.4.1 (page 142), resulting in about 6,500 women in total and 4,500 in the married categories, which as discussed in the previous paragraph are not enough observations for obtaining precise IV estimates.

In Table D.4, the *Same sex* indicator has sizeable and significant first-stage effects on both the *Number of children* and *More than two children* variables. Moreover, these effects are close in size to those reported for the 1991 Census in Table 7.2 (page 179). The same is true for the OLS estimations of the effect of fertility on labour supply, where these two variables appear to have a significant negative effect on women's labour supply (*Worked for pay*). These effects are also within the range of the OLS coefficients in Table 7.3.

Finally, as expected from the small sample size, none of the instrumental variables estimates are significant, either with *Same sex* or *Two boys* and *Two girls* as instruments, but despite the lack of precision, the two-stage least squares coefficients are also in range with those in Table 7.3. These results suggest that the low significance of instrumental variables estimates is probably due to a problem of precision rather than the lack of a causal effect in the ENGH, and that the benchmark results of Section 7.2.2 (page 176) are not confined to one particular dataset for Argentina.

limited, with only seven urban areas with complete information (INDEC, 1996): Greater Buenos Aires, Greater Mendoza, Greater Córdoba, Neuquén, Jujuy, Río Gallegos and Santa Rosa. The resulting dataset of women with two children is too small, and it is not representative of the country. However, some of the results are still remarkably similar to the benchmark of this Chapter: in a sample of 1,398 women aged 21-35 with two or more children aged 18 or less, the average of the *Worked for pay* indicator is 0.38. The first-stage coefficient of *Same sex*, significant at the 5 percent level, is 0.0539, while the OLS coefficient of *More than two children* is -0.0438. The Conclusion to Part II (page 212) uses this data to discuss the implications of the main results in terms of poverty.

	Instrumented: Number of children				Instrumented: More than two children				
Dependent variable:	Complete Sample	Married	AE Sample	AE Sample	Complete	Married	AE Sample	AE Sample.	
Worked for pay			·	married	Sample		· in or in pro	married	
First stages									
Coefficient of Same sex	0.0738	0.0542	0.0772	0.0575	0.0535	0.0518	0.0552	0.0530	
	[0.0290]**	[0.0321]*	[0.0303]**	[0.0335]*	[0.0113]***	[0.0133]***	[0.0117]***	[0.0137]***	
OLS estimates									
OLS	-0.0553	-0.0491	-0.0567	-0.0502	-0.1023	-0.0731	-0.1080	-0.0778	
	[0.0052]***	[0.0065]***	[0.0053]***	[0.0066]***	[0.0134]***	[0.0158]***	[0.0137]***	[0.0161]***	
IV Estimates									
IV: Same Sex	-0.0686	-0.0079	-0.0717	-0.0345	-0.0945	-0.0083	-0.1004	-0.0374	
	[0.1632]	[0.2619]	[0.1605]	[0.2532]	[0.2257]	[0.2747]	[0.2254]	[0.2756]	
DWH p-value	0.9351	0.8745	0.9250	0.9507	0.9722	0.8129	0.9731	0.8834	
IV: Two Boys and Two Girls	-0.0716	-0.0863	-0.0648	-0.0735	-0.0987	-0.0171	-0.0906	-0.0444	
Controls for sexes of 1st & 2nd children	[0.1611]	[0.2443]	[0.1587]	[0.2344]	[0.2223]	[0.2745]	[0.2230]	[0.2752]	
Sargan p-value	0.9082	0.4042	0.7622	0.6860	0.9126	0.3670	0.7628	0.6285	
DWH p-value	0.9194	0.8787	0.9591	0.9208	0.9871	0.8378	0.9379	0.9031	
Observations	6,355	4,589	6,028	4,361	6,355	4,589	6,028	4,361	

Table D.4: OLS and Two-Stage Least Squares Results, 1987 Expenditure Survey

Note: Standard errors in brackets. * significant at 10%; ** significant at 5%; *** significant at 1%. All regressions include controls for the age and the age at first birth of the women and the sex of the first two children. The samples correspond to women aged 21-35 with two or more children aged 18 or younger from the Encuesta Nacional de Gasto de los Hogares, INDEC, 1987. The subsamples are analogous to those defined for the 1991 Census.

APPENDIX E

COMPUTATION OF RESULTS FOR COMPLIERS

E.1 COMPUTATION OF OUTCOMES FOR COMPLIERS BY REGRESSION METHODS

This Appendix provides the details of the computation of the parameters and tests presented in Chapter 8.

Proposition 8.2 (page 195) identifies the expectation of potential outcomes Y_1 and Y_0 for compliers, while Proposition 8.3 (page 196) shows how to compute the expectation of Y for the same group. Obtaining these averages and of the test in Equation 8.8 (page 198) from observed data is straightforward, but finding the correct standard errors of the estimates can prove more difficult. The results below are based on an analogy with LATE-Wald estimates of a binary instrument and one endogenous regressor by instrumental variables.

For any variable Ψ , with *Z* and *D* binary, the coefficient γ is given by:

$$\gamma = \frac{E[\Psi|Z=1] - E[\Psi|Z=0]}{E[D|Z=1] - E[D|Z=0]}$$
(E.1)

This coefficient can be obtained by means of an instrumental variable regression of Ψ on D and a constant with Z as an instrument for D (Proposition 5.5, page 135). The advantage of using a regression instead of Equation E.1 for finding γ is that the statistical software computes the standard errors for the parameter automatically.

 $E[Y_1^c]$ in Equation 8.3 can be obtained by setting Ψ in Equation E.1 to $\Psi = YD$: this provides the potential outcome and its standard error.

The parameter $E[Y_0^c]$ in Equation 8.2 can also be computed by defining

 $\tilde{D} = 1 - D$, using $\Psi = Y\tilde{D}$ and instrumenting \tilde{D} instead of D. However, a variable transformation ensures that it can also be obtained by instrumenting D, as shown by the following result:

$$E[Y_0^c] = \frac{E[Y(1-D)|Z=1] - E[Y(1-D)|Z=0]}{E[(1-D)|Z=1] - E[(1-D)|Z=0]}$$

= $-\frac{E[Y(1-D)|Z=1] - E[Y(1-D)|Z=0]}{E[D|Z=1] - E[D|Z=0]}$
= $\frac{E[Y(D-1)|Z=1] - E[Y(D-1)|Z=0]}{E[D|Z=1] - E[D|Z=0]}$

The same reasoning applies by setting $\Psi = Y(D - 1)$ in Equation E.1.

This result is useful for the computation of the average outcome for compliers from Proposition 8.3 and its standard error. Using the previous results:

$$\begin{split} E[Y^c] &= E[Y^c|Z=0]E[1-Z] + E[Y^c|Z=1]E[Z] \\ &= \frac{E[Y(D-1)|Z=1] - E[Y(D-1)|Z=0]}{E[D|Z=1] - E[D|Z=0]}E[1-Z] + \frac{E[YD|Z=1] - E[YD|Z=0]}{E[D|Z=1] - E[D|Z=0]}E[Z] \\ &= \frac{E[\tilde{Y}|Z=1] - E[\tilde{Y}|Z=0]}{E[D|Z=1] - E[D|Z=0]} \end{split}$$

where $\tilde{Y} = Y(D-1)E[1-Z] + YDE[Z]$. With large sample sizes, E[Z] and E[1-Z] can be considered non-stochastic. The last line in the previous Equation is analogous to Equation E.1, which implies that the expected outcome for compliers, $E[Y^c]$, can be obtained by setting $\Psi = \tilde{Y}$ and computing γ . Alternatively, it can be computed by a regression of \tilde{Y} on D and a constant with Z as an instrument for D.

E.2 COMPUTATION OF THE AUXILIARY TEST FOR COMPLIERS BY REGRESSION METHODS

Finally, Chapter 8 also presented a test for the generality of IV results for compliers. The test exploited the additional identification result presented in Proposition 8.3 (page 196), which identified the expected potential outcome for compliers. The intuition is that if the expected outcome for the general population is not different from that of compliers, then the two are similar in one important dimension, and it is thus more likely that other results for compliers concerning the outcome variable can be extrapolated to

the whole population. The test is in Equation 8.8, and given by:

$$\lambda = E[Y] - E[Y^c]$$

This test can be performed by means of the instrumental variable estimation described in the previous paragraph: the parameter γ in Equation E.1 estimates $E[Y^c]$ when $\Psi = \tilde{Y}$, which amounts to regressing \tilde{Y} on D with Zas an instrument for D. Using this regression, a t-test of whether $\gamma = E[Y]$ is equivalent to testing the hypothesis $\lambda = 0$ in Equation 8.8.

BIBLIOGRAPHY

- ABADIE, A. (2002): "Bootstrap Tests of Distributional Treatment Effects in Instrumental Variable Models," *Journal of the American Statistical Association*, 97(457), pp. 284–92.
- ——— (2003a): "Causal Inference and Public Program Evaluation," Unpublished lecture notes, CEMFI, Madrid.
- ——— (2003b): "Semiparametric Instrumental Variable Estimation of Treatment Response Models," *Journal of Econometrics*, 113(2), pp. 231–63.
- ABADIE, A., J. D. ANGRIST, AND G. W. IMBENS (2002): "Instrumental Variables Estimates of the Effect of Subsidized Training on the Quantiles of Trainee Earnings," *Econometrica*, 70(1), pp. 91–117.
- AHN, N. (1995): "Measuring the Value of Children by Sex and Age Using a Dynamic Programming Model," *Review of Economic Studies*, 62(3), pp. 361–79.
- ALBORNOZ, F., AND M. MENÉNDEZ (2004): "Income Mobility and Inequality: The Case of Argentina during the 1990's," Unpublished, DELTA, Paris.
- ANAND, S., AND J. MORDUCH (1998): "Poverty and the 'Population Problem'," in *Population and Poverty in Developing Countries*, edited by G. DE SANTIS AND M. LIVI BACCI, Clarendon Press, Oxford.
- ANGRIST, J. (2001): "Estimation of Limited-Dependent Variable Models with Dummy Endogenous Regressors: Simple Strategies for Empirical Practice," *Journal of Business and Economic Statistics*, 19(1), pp. 2–16.
- ANGRIST, J. D. (2004): "Treatment Effect Heterogeneity in Theory and Practice," *Economic Journal*, 114(494), pp. C52–C83.
- ANGRIST, J. D., AND W. N. EVANS (1998): "Children and Their Parents' Labor Supply: Evidence from Exogenous Variation in Family Size," *American Economic Review*, 88(3), pp. 450–77.
- ANGRIST, J. D., AND G. W. IMBENS (1995): "Two-Stage Least Squares Estimation of Average Causal Effects in Models with Variable Treatment Intensity," *Journal of the American Statistical Association*, 90(430), pp. 431– 432.

- ANGRIST, J. D., G. W. IMBENS, AND D. B. RUBIN (1996): "Identification of Causal Effects Using Instrumental Variables," *Journal of the American Statistical Association*, 91(434), pp. 444–55.
- ANGRIST, J. D., AND A. B. KRUEGER (1999): "Empirical Strategies in Labor Economics," in *Handbook of Labor Economics Volume 3A*, edited by O. ASHENFELTER AND D. CARD, pp. 1277–1366, Elsevier Science, North-Holland, Amsterdam.
 - ——— (2001): "Instrumental Variables and the Search for Identification: From Supply and Demand to Natural Experiments," *Journal of Economic Perspectives*, 15(4), pp. 69–85.
- ARROW, K. J. (1970): Essays in the Theory of Risk-Bearing, North-Holland, Amsterdam.
- ARZA, C. (2004): "Distributional Impacts of Social Policy: Pension Regimes in Argentina Since c.1944," Ph.D. thesis, Economic History Department, London School of Economics, University of London.
- ATKINSON, A. B. (1970): "On the Measurement of Inequality," *Journal of Economic Theory*, 2, pp. 244–263.
- (1987): "On the Measurement of Poverty," *Econometrica*, 55(4), pp. 749–64.
- ATKINSON, A. B., AND F. BOURGUIGNON, eds. (2000): Handbook of Income Distribution, North Holland, Amsterdam.
- ATKINSON, A. B., AND F. A. COWELL, eds. (1982): Panel Data on Incomes, ICERD, LSE, London.
- BASU, A., AND M. DAS GUPTA (2001): "Family Systems and the Preferred Sex of Children," in *International Encyclopedia of Social and Behavioural Sci*ences, edited by J. HOEM, Elsevier Science, North-Holland, Amsterdam.
- BAULCH, B., AND J. HODDINOTT (2000): "Economic Mobility and Poverty Dynamics in Developing Countries," *Journal of Development Studies*, 36(6), pp. 1–24.
- BECKER, G. (1991): A Treatise on the Family, Harvard University Press.
- BECKER, G. S. (1965): "The Economics of the Allocation of Time," *Economic Journal*, 75, pp. 493–517.
- BECKER, G. S., AND H. G. LEWIS (1973): "On the Interaction Between the Quantity and Quality of Children," *Journal of Political Economy*, 81(2), pp. 5279–88.
- BEN PORATH, Y., AND F. WELCH (1976): "Do Sex Preferences Really Matter?" Quarterly Journal of Economics, 90(2), pp. 285–307.

- BESLEY, T. (1995): "Savings, Credit and Insurance," in *Handbook of Develop*ment Economics Volume 3, edited by J. BEHRMAN AND T. N. SRINIVASAN, Chapter 36, pp. 2122–2207, Elsevier Science, North-Holland.
- BINSTOCK, G. (2004): "Continuity and Change: The Family in Argentina," Paper presented at the Conference on Ideational Factors in International Family Change, June 3 - 5, 2004, Ann Arbor, Michigan.
- BLACKORBY, C., AND D. DONALDSON (1987): "Welfare Ratios and Distributionally Sensitive Cost-Benefit Analysis," *Journal of Public Economics*, 34(1), pp. 265–290.
- BLUNDELL, R., AND I. PRESTON (1998): "Consumption Inequality and Income Uncertainty," *Quarterly Journal of Economics*, 113(2), pp. 603–40.
- BONVECCHI, A. (2003): "Business as Usual. Political Methods and Economic Normalcy in Argentine Fiscal Policymaking during Structural-Reforms Processes (1983-1999)," Ph.D. thesis, Department of Government, University of Essex.
- BROWNING, M. (1992): "Children and Household Economic Behavior," Journal of Economic Literature, 30(3), pp. 1434–75.
- BROWNING, M., AND A. LUSARDI (1996): "Household Saving: Micro Theories and Micro Facts," *Journal of Economic Literature*, 34, pp. 1797–1855.
- BUCHINSKY, M. (1994): "Changes in the US Wage Structure 1963-1987: Application of Quantile Regression," *Econometrica*, 62(2), pp. 405–58.
- CALVO, C., AND S. DERCON (2003): "Vulnerability: an Axiomatic Approach," Unpublished, Oxford University.
- CARRASCO, R. (2001): "Binary Choice with Binary Endogenous Regressors in Panel Data: Estimating the Effect of Fertility on Female Labor Force Participation," *Journal of Business and Economic Statistics*, 19(4), pp. 385– 94.
- CASE, A., AND A. DEATON (2003): "Consumption, Health, Gender, and Poverty," Working Paper No. 3020.
- CELADE (2004): "Population Estimations and Projections for Latin America and the Caribbean," *Demographic Bulletin*, 73. Some indicators were also obtained online from CELADE's website: http://www.eclac.cl/celade.
- CHAY, K. Y., AND J. L. POWELL (2001): "Semiparametric Censored Regression Models," *Journal of Economic Perspectives*, 15(4), pp. 29–42.
- CHETTY, R. (2003): "A New Method of Estimating Risk Aversion," NBER Working Paper Series Number 9988.

- CHUN, H., AND J. OH (2002): "An Instrumental Variable Estimate of the Effect of Fertility on the Labour Force Participation of Married Women," *Applied Economics Letters*, 9(10), pp. 631–4.
- CIGNO, A. (1991): Economics of the Family, Clarendon Press, Oxford.
- CLARK, S., R. HEMMING, AND D. ULPH (1981): "On Indices for the Measurement of Poverty," *Economic Journal*, 91, pp. 515–526.
- CORBACHO, A., M. GARCÍA-ESCRIBANO, AND G. INCHAUSTE (2003): "Argentina: Macroeconomic Crisis and Household Vulnerability," International Monetary Fund Working Paper 03/89.
- CORTÉS, R. (2003): "Mercado de Trabajo y Género. El Caso Argentino, 1994-2002," in *Mujeres, Pobreza y Mercado de Trabajo - Argentina y Paraguay*, Chapter 2, pp. 67–104, Organización Internacional del Trabajo, Oficina Regional para América Latina y el Caribe, Perú.
- COULTER, F. A. E., F. A. COWELL, AND S. P. JENKINS (1992): "Differences in Needs and Assessment of Income Distributions," *Bulletin of Economic Research*, 44, pp. 77–124.
- COWELL, F. A. (1979): "The Definition of Lifetime Income," Institute for Research on Poverty Discussion paper 566-79, University of Wisconsin, Madison.

—— (1982): "Introduction," in *Panel Data on Incomes*, edited by A. B. ATKINSON AND F. A. COWELL, ICERD, LSE, London.

—— (1989): "On the Distribution of Lifetime Income in a Lognormal Model," in *Essays on Income Distribution, Economic Welfare and Taxation,* edited by R. HAGFORS AND P. VARTIA, ETLA, Helsinki.

——— (1995): *Measuring Inequality*, London School of Economics Handbooks in Economics, Harvester Wheatsheaf/Prentice Hall, London.

—— (2000): "Measurement of Inequality," in *Handbook of Income Distribution*, edited by A. B. ATKINSON AND F. BOURGUIGNON, p. 16, Elsevier Science, North-Holland.

——— (2003): *The Economics of Poverty and Inequality*, Edward Elgar, Cheltenham.

------ (2005): *Microeconomics*. *Principles and Analysis*, Forthcoming, Oxford University Press.

- COWELL, F. A., AND G. CRUCES (2004): "Perceptions of Risk and Inequality," in Research on Economic Inequality, Volume 12. Studies on Economic Well-Being: Essays in the Honor of John P. Formby, edited by Y. AMIEL AND J. A. BISHOP, Chapter 4, pp. 99–132, Elsevier Science, North Holland.
- COWELL, F. A., AND K. A. GARDINER (2000): "Welfare Weights," Office of Fair Trading Economic Research Paper, 202, London.

- COWELL, F. A., AND M. MERCADER PRATS (1999): "Equivalence Scales and Inequality," in *Income Inequality Measurement: From Theory to Practice*, edited by J. SILBER, pp. 405–29, Kluwer, Dewenter.
- CRUCES, G., AND S. GALIANI (2003): "Generalizing the Causal Effect of Fertility on Female Labour Supply," Working Paper Number 625, William Davidson Institute, University of Michigan.
- CRUCES, G., P. MAKDISSI, AND Q. WODON (2004): "Poverty Measurement Under Risk Aversion Using Panel Data," Contributions to Economic Analysis & Policy, 3(1), p. Article 13.
- CRUCES, G., AND Q. WODON (2003a): "Argentina's Crises and the Poor, 1995-2002," Económica, 49(1-2).
 - —— (2003b): "Risk-Adjusted Poverty in Argentina: Measurement and Determinants," In World Bank (2003), Argentina Crisis and Poverty 2003. A Poverty Assessment, Volume 2. Also available as Distributional Analysis Research Programme Working Paper 72, STICERD, London.
- ——— (2003c): "Transient and Chronic Poverty in Turbulent Times: Argentina 1995-2002," *Economics Bulletin*, 9(3), pp. 1–12.
- ——— (2004): "Chronic Poverty, Transient Poverty and the Accounting Period," Unpublished, STICERD-LSE and World Bank.
- DAS, N. (1987): "Sex Preference and Fertility Behavior: A Study of Recent Indian Data," *Demography*, 24(4), pp. 517–30.
- DAS GUPTA, M. E. (2003): "Why Is Son Preference So Persistent in East and South Asia? A Cross-Country Study of China, India and the Republic of Korea," *Journal of Development Studies*, 40(2), pp. 153–87.
- DE FERRANTI, D., G. E. PERRY, I. S. GILL, AND L. SERVÉN (2000a): "Helping Poor Households Deal Better with Economic Crises," in Securing our Future in a Global Economy, Chapter 7, pp. 111–126, Office of the Chief Economist for Latin America and the Caribbean, World Bank, Washington DC.
 - ——— (2000b): "Helping Workers Deal with the Risk of Unemployment," in *Securing our Future in a Global Economy*, Chapter 6, pp. 89–109, Office of the Chief Economist for Latin America and the Caribbean, World Bank, Washington DC.
 - —— (2000c): "The Response of LAC Households to Economic Shocks," in *Securing our Future in a Global Economy*, Chapter 5, pp. 73–88, Office of the Chief Economist for Latin America and the Caribbean, World Bank, Washington DC.
- DEATON, A. S. (1997): *The Analysis of Household Surveys*, Johns Hopkins Press for the World Bank, Baltimore, Maryland.

- DEATON, A. S., AND J. MUELLBAUER (1980): Economics and Consumer Behavior, Cambridge University Press, Cambridge.
- DEE, T., AND W. EVANS (2003): "Teen Drinking and Educational Attainment: Evidence from Two-Sample Instrumental Variables Estimates," *Journal of Labor Economics*, 21(1).
- DERCON, S., AND P. KRISHNAN (2000): "Vulnerability, Seasonality and Poverty in Ethiopia," *Journal of Development Studies*, 36(6), pp. 25–53.
- DICKENS, R., P. GREGG, AND J. WADSWORTH, eds. (2004): *The Labour Market under New Labour. The State of Working Britain*, Palgrave Macmillan, London.
- DRÈZE, J., AND F. MODIGLIANI (1972): "Consumption Decisions Under Uncertainty," *Journal of Economic Theory*, V, pp. 308–35.
- ELBERS, C., AND J. W. GUNNING (2003): "Estimating Vulnerability," Presented at the conference "Staying Poor: Chronic Poverty and Development Policy," University of Manchester, April 7-9, 2003.
- ERMISCH, J. F. (2003): An Economic Analysis of the Family, Princeton University Press.
- FELLI, L. (2003): "Advanced Economic Theory, Lecture 1," Unpublished lecture notes, London School of Economics.
- FIEL, ed. (1999): *La Distribución del Ingreso en la Argentina*, Fundación de Investigaciones Económicas Latinoamericanas, Buenos Aires.
- FISZBEIN, A., P. GIOVAGNOLI, AND I. ADURIZ (2002): "Argentina's Crisis and its Impact on Household Welfare," *CEPAL Review*, 79, pp. 151–168.
- FOSTER, J., J. GREER, AND E. THORBECKE (1984): "A Class of Decomposable Poverty Measures," *Econometrica*, 52(3), pp. 761–66.
- FRIEDMAN, M., AND L. J. SAVAGE (1948): "The Utility Analysis of Choices Involving Risk," Journal of Political Economy, 56, pp. 1–23.
- FRIEND, I., AND M. E. BLUME (1975): "The Demand for Risky Assets," American Economic Review, 65(5), pp. 900–22.
- GALASSO, Protec-AND M. RAVALLION (2003): "Social E., tion in a Crisis: ArgentinaŠs Plan Jefes y Jefas," World Bank Economic Review, 18(3), pp. 367-399, available at http://ideas.repec.org/a/oup/wbrobs/v14y1999i1p31-48.html.
- GALIANI, S., AND P. GERCHUNOFF (2003): "The Labor Market," in A New Economic History of Argentina, edited by G. DELLA PAOLERA AND A. TAYLOR, Chapter 4, Cambridge University Press.
- GALIANI, S., D. HEYMANN, AND M. TOMMASI (2003): "Great Expectations and Hard Times: The Argentine Convertibility Plan," *Economia*, *The Journal of the Latin American and Caribbean Economic Association*, 3(2).

- GALIANI, S., AND H. A. HOPENHAYN (2003): "Duration and Risk of Unemployment in Argentina," *Journal of Development Economics*, 71(1), pp. 199–212.
- GASPARINI, L. (2003a): "Perfil Demográfico de Pobres y Nuevos Pobres," in *Características Demográficas y Pobreza en Argentina*, edited by CEN-TRO DE ESTUDIOS DISTRIBUTIVOS, LABORALES Y SOCIALES (CEDLAS), Chapter 4, Universidad Nacional de La Plata, and Inter-American Development Bank.

(2003b): "Vulnerabilidad y Crisis Económica. Un Análisis Exploratorio Sobre la Base de Datos de Panel," in *Características Demográficas y Pobreza en Argentina*, edited by CENTRO DE ESTUDIOS DISTRIB-UTIVOS, LABORALES Y SOCIALES (CEDLAS), Chapter 5, Universidad Nacional de La Plata, and Inter-American Development Bank.

- GASPARINI, L., AND M. MARCHIONNI (2003): "Tracing out the Effects of Demographic Changes on the Income Distribution. The Case of Greater Buenos Aires 1980-2000," in *Demographic Characteristics and Poverty in Argentina*, edited by CENTRO DE ESTUDIOS DISTRIBUTIVOS, LABORALES Y SOCIALES (CEDLAS), Universidad Nacional de La Plata, and Inter-American Development Bank.
- GASPARINI, L., M. MARCHIONNI, AND W. SOSA ESCUDERO (2001): La Distribución del Ingreso en la Argentina: Perspectivas y Efectos Sobre el Bienestar, Fundación Arcor, Premio Fulvio Salvador Pagani 2001, Córdoba, Argentina.
- GERCHUNOFF, P., AND L. LLACH (2004a): El Ciclo de la Ilusión y el Desencanto, Ariel, Buenos Aires.
 - (2004b): Entre la Equidad y el Crecimiento. Ascenso y Caída de la Economía Argentina, 1880-2002, Siglo XXI, Buenos Aires.
- GLEWWE, P., AND G. HALL (1998): "Are Some Groups More Vulnerable to Macroeconomic Shocks Than Others? Hypothesis Tests Based on Panel Data from Perú," *Journal of Development Economics*, 56(1), pp. 181–206.
- GREGG, P., AND J. WADSWORTH (1999a): "Economic Inactivity," in *The State of Working Britain*, edited by P. GREGG AND J. WADSWORTH, Chapter 3, pp. 47–57, Manchester University Press.
- GREGG, P., AND J. WADSWORTH, eds. (1999b): *The State of Working Britain*, Manchester University Press.
- GRONAU, R. (1973): "The Effect of Children on the Housewife's Value of Time," *Journal of Political Economy*, 81(2), pp. S168–99.

(1977): "Leisure, Home Production, and Work: The Theory of the Allocation of Time Revisited," *Journal of Political Economy*, 85(6), pp. 1099–124.
- HAMMOND, P. J. (1975): "A Note on Extreme Inequality Aversion," *Journal* of Economic Theory, 11, pp. 465–467.
- ——— (1981): "Ex-Ante and Ex-Post Welfare Optimality under Uncertainty," Economica, 48(191), pp. 235–250.
- HECKMAN, J., AND C. PAGES, eds. (2003): Law and Employment: Lessons from Latin America and the Caribbean, NBER, Cambridge.
- HECKMAN, J., J. L. TOBIAS, AND E. VYTLACIL (2003): "Simple Estimators for Treatment Parameters in a Latent-Variable Framework," *Review* of Economics and Statistics, 85(3), pp. 748–55.
- HILDRETH, C., AND G. KNOWLES (1982): "Some Estimates of Farmers' Utility Functions," Technical Bulletin 335, Agricultural Experimental Station, University of Minnesota.
- HULME, D., AND A. SHEPHERD (2003): "Conceptualizing Chronic Poverty," World Development, 31(3), pp. 403–423.
- IACOVOU, M. (2001): "Fertility and Female Labour Supply," Working Paper 2001-19, Institute for Social and Economic Research.
- IADB (2004): Good Jobs Wanted: Labor Markets in Latin America, Economic and Social Progress Report 2004, Inter-American Development Bank, Washington DC.
- ILO (2003): *Key Indicators of the Labour Markets*, International Labour Organisation, Geneve. Available online at: http://www.ilo.org/kilm.
- IMBENS, G. W., AND J. D. ANGRIST (1994): "Identification and Estimation of Local Average Treatment Effects," *Econometrica*, 62(2), pp. 467–75.
- IMBENS, G. W., AND D. B. RUBIN (1997): "Estimating Outcome Distributions for Compliers in Instrumental Variables Models," *Review of Economic Studies*, 64, pp. 555–74.
- INDEC (1996): *Marco de Muestreo Nacional Urbano para Encuestas a Hogares,* Serie Metodologías No. 12, Instituto Nacional de Estadísticas y Censos, Buenos Aires.

——— (2000): *El Ingreso y el Gasto de los Hogares,* Serie Perfil de Condiciones de Vida No. 1, Instituto Nacional de Estadísticas y Censos, Buenos Aires.

—— (2001): *Situación de la Mujeres en la Argentina,* Serie Análisis Social No. 1, Instituto Nacional de Estadísticas y Censos, Buenos Aires.

—— (2002): "Incidencia de la Pobreza y de la Indigencia en los Aglomerados Urbanos," Información de Prensa, Instituto Nacional de Estadísticas y Censos, Buenos Aires.

------ (2003a): Anuario Estadístico de la República Argentina, 2002-2003, Publicaciones del Instituto Nacional de Estadísticas y Censos, Buenos Aires. Some of the indicators were obtained online from INDEC's website: http://www.indec.gov.ar.

----- (2003b): "Incidencia de la Pobreza y de la Indigencia en los Aglomerados Urbanos - Anexo 2: Incidencia del Plan Jefes-Jefas," Información de Prensa, Instituto Nacional de Estadísticas y Censos, Buenos Aires.

—— (2004): "Incidencia de la Pobreza y de la Indigencia en los Aglomerados Urbanos," Información de Prensa, Instituto Nacional de Estadísticas y Censos, Buenos Aires.

JALAN, J., AND M. RAVALLION (1998): "Transient Poverty in Postreform Rural China," Journal of Comparative Economics, 26, pp. 338–357.

------ (1999): "Are the Poor Less Well-Insured? Evidence on Vulnerability to Income Risk from Rural China," *Journal of Development Economics*, 58, pp. 61–81.

——— (2000): "Is Transient Poverty Different? Evidence for Rural China," Journal of Development Studies, 36(6), pp. 82–99.

- KILLINGSWORTH, M. R., AND J. J. HECKMAN (1986): "Female Labor Supply: A Survey," in *Handbook of Labor Economics Volume 1*, edited by O. ASHENFELTER AND R. LAYARD, pp. 103–204, Elsevier Science, New York.
- KIMBALL, M. S. (1990): "Precautionary Savings in the Small and in the Large," *Econometrica*, 58(1), pp. 53–73.
- KUROSAKI, T. (2003): "Measurement of Chronic and Transient Poverty: Theory and Application to Pakistan," Unpublished, Institute of Economic Research, Hitotsubashi University, Japan.
- LANJOUW, P., AND M. RAVALLION (1995): "Poverty and Household Size," *Economic Journal*, 105(433), pp. 1415–34.
- LAVERGNE, N., S. LEVY, AND A. CATANZARO (1999): "Estudio Sobre la Variabilidad de los Ingresos Individuales," in *La Distribución del Ingreso en la Argentina*, edited by FIEL, Chapter III.1, Appendix 1, pp. 141–178, Fundación de Investigaciones Económicas Latinoamericanas, Buenos Aires.
- LEE, H. (2000): "Poverty and Income Distribution in Argentina: Patterns and Changes," World Bank.
- LELAND, H. E. (1968): "Saving and Uncertainty: The Precautionary Demand for Saving," *Quarterly Journal of Economics*, 82(3), pp. 465–473.
- LEUNG, S. F. (1991): "A Stochastic Dynamic Analysis of Parental Sex Preferences and Fertility," *Quarterly Journal of Economics*, 106(4), pp. 1063–88.
- LEWIS, C. M. (2002): Argentina: A Short History, Oxford: Oneworld.

LIGON, E., AND L. SCHECHTER (2003): "Measuring Vulnerability," Economic Journal, 113(486), pp. C95–110.

(2004): "Evaluating Different Approaches to Estimating Vulnerability," Unpublished, Department of Agricultural and Resource Economics, University of California, Berkeley.

- LIPTON, M. (1998): "Population and Poverty: How do they Interact?" in *Population and Poverty in Developing Countries*, edited by G. DE SANTIS AND M. LIVI BACCI, Clarendon Press, Oxford.
- MACKENZIE, D. (2004): "Aggregate Shocks and Labour Market Responses Evidence from Argentina's Financial Crisis," *forthcoming, Economic Development and Cultural Change.*
- MAKDISSI, P., AND Q. WODON (2003): "Risk-Adjusted Measures of Wage Inequality and Safety Nets," *Economics Bulletin*, 9(1), pp. 1–10.
- MAS-COLELL, A., M. D. WHINSTON, AND J. R. GREEN (1995): Microeconomic Theory, Oxford University Press, Oxford.
- MCCULLOCH, N., AND B. BAULCH (2000): "Simulating the Impact of Policy upon Chronic and Transitory Poverty in Rural Pakistan," *Journal of Development Studies*, 36(6), pp. 100–130.
- MILNE, F., AND H. M. SHEFRIN (1988): "Ex Post Efficiency and Ex Post Welfare: Some Fundamental Considerations," *Economica*, 55(217), pp. 63–79.
- MONTGOMERY, M., AND J. TRUSSELL (1986): "Models of Marital Status and Childbearing," in *Handbook of Labor Economics Volume 1*, edited by O. ASHENFELTER AND R. LAYARD, pp. 205–272, Elsevier Science, New York.
- MORALES, E. (1988): "Canasta Básica de Alimentos Gran Buenos Aires," Documento de trabajo no. 3., INDEC / IPA, Buenos Aires.
- MORDUCH, J. (1994): "Poverty and Vulnerability," American Economic Review, 84(2), pp. 221–225.
- MORENO, J. M., AND E. ROCA (2002): "Empleo No Registrado y Exclusión de la Seguridad Social," in *Informalidad y Exclusion Social*, edited by I. NOVACOVSKY, J. CARPIO, AND E. KLEIN, SIEMPRO-ILO, Fondo de Cultura Económica, Buenos Aires.
- NAG, M. (1991): "Sex Preference in Bangladesh, India, and Pakistan, and its Effect on Fertility," Research Division Working Paper No. 27, The Population Council.
- NEWBERY, D. M. G., AND J. E. STIGLITZ (1981a): "Supply with Risk-Averse Farmers," in *The Theory of Commodity Price Stabilization*, Chapter 6, pp. 97–110, Clarendon Press, Oxford.

(1981b): *The Theory of Commodity Price Stabilization*, Clarendon Press, Oxford.

- O'DONOGHUE, E. D., AND M. RABIN (1999): "Doing It Now or Doing It Later," American Economic Review, 89, pp. 103–124.
- PANTELIDES, E. A. (2002): "Completing the Fertility Transition: the Case of Argentina," Report of the Expert Group Meeting, Department of Economic and Social Affairs, Population Division, United Nations, New York.
- PAZ, J. A. (2002): "Una Introducción a la Dinámica de la Pobreza en la Argentina," Documento de trabajo 226, Universidad del CEMA, Argentina.
- PRATT, J. W. (1964): "Risk-Aversion in the Small and in the Large," *Econometrica*, 32(1-2), pp. 122–36.
- RAVALLION, M. (1988): "Expected Poverty under Risk-Induced Welfare Variability," *Economic Journal*, 98(393), pp. 1171–82.
 - ——— (1994): Poverty Comparisons: Fundamentals of Pure and Applied Economics, Harwood Academic Publishers, London.
- ——— (1998): "Poverty Lines in Theory and Practice," Living Standards Measurement Study Working Paper 133, World Bank, Washington DC.
- RAVALLION, M., AND J. JALAN (2003): "Estimating the Benefit Incidence of an Anti-Poverty Program," *Journal of Business and Economic Statistics*, 21(1), pp. 19–30.
- RAVALLION, M., AND Q. WODON (1999): "Poor Areas or Only Poor People," Journal of Regional Science, 39, pp. 689–711.
- ROSENZWEIG, M. R., AND K. I. WOLPIN (1980): "Life-Cycle Labor Supply and Fertility: Causal Inferences from Household Models," *Journal of Political Economy*, 88(2), pp. 328–48.

——— (2000): "Natural 'Natural Experiments' in Economics," *Journal of Economic Literature*, 38(4), pp. 827–74.

- ROTHSCHILD, M., AND J. E. STIGLITZ (1969): "Increasing Risk: A Definition and its Economics Consequences," Cowles Foundation Discussion Paper 275; published in two parts in the *Journal of Economic Theory*, Volumes 2 (1970) and 3 (1971).
- SANDMO, A. (1970): "The Effect of Uncertainty on Saving Decisions," *The Review of Economic Studies*, 37(3), pp. 353–360.
- SCOTT, C. D. (2000): "Mixed Fortunes: A Study of Poverty Mobility Among Small Farm Households in Chile, 1968-86," *Journal of Development Studies*, 36(6), pp. 155–80.
- SHORROCKS, A. F. (1978): "The Measurement of Mobility," *Econometrica*, 46, pp. 1013–1024.

- SIEMPRO (2003): *Informe de la Situación Social de las Provincias*, Sistema de Información, Evaluación y Monitoreo de Programas Sociales, Presidencia de la Nación, Buenos Aires, Argentina.
- SOBEK, M., S. RUGGLES, AND R. MCCAA (2003): "Integrated Public Use Microdata Series: Version 3.0," Minneapolis: Minnesota Population Center, University of Minnesota.
- THORBECKE, E. (2003): "Conceptual and Measurement Issues in Poverty Analisys," Presented at the UNU / WIDER Conference on Inequality, Poverty and Human Well-being, 30-31 May 2003, Helsinki.
- THORP, R. (1998): Progress, Poverty and Exclusion: An Economic History of Latin America, Inter-American Development Bank, Washington DC.
- UNITED NATIONS (2002): "Completing the Fertility Transition," Report of the Expert Group Meeting, Department of Economic and Social Affairs, Population Division, United Nations, New York.
- VON NEUMANN, J., AND O. MORGENSTERN (1944): *Theory of Games and Economic Behavior*, Princeton University Press, Princeton, New Jersey.
- WATTS, H. W. (1968): "An Economic Definition of Poverty," in *Understanding Poverty*, edited by D. P. MOYNIHAN, Chapter 11, pp. 316–329, Basic Books, New York.
- WILLIAMSON, N. (1983): "Parental Sex Preferences and Sex Selection," in *Sex Selection of Children*, edited by N. G. BENNETT, Academy Press, New York.
- WILLIS, R. J. (1987): "What Have We Learned from the Economics of the Family?" *American Economic Review*, 77, pp. 68–81.
- WORLD BANK (2000a): "Argentina: Poor People in a Rich Country," Report 19992-AR, Poverty Reduction and Economic Management, Washington DC.
 - ——— (2000b): "Managing Social Risks in Argentina," World Bank, Human Development Department, Social Protection Group, Latin America and the Caribbean Regional Office, Washington DC.

------ (2001): "Household Risk, Self-Insurance and Coping Strategies in Urban Argentina," Report 22426-AR, Social Protection, Washington DC.

—— (2003): "Argentina Crisis and Poverty 2003. A Poverty Assessment," Report 26127-AR in two volumes, Poverty Reduction and Economic Management, Washington DC.

—— (2004): *Inequality in Latin America: Breaking with History?*, Edited by De Ferranti, D., G. Perry, F. Ferreira and M. Walton. World Bank Latin American and Caribbean studies, Washington DC.

WRIGHT, P. G. (1928): The Tariff on Animal and Vegetable Oils, Macmillan, New York.

•

BIBLIOGRAPHY

STATEMENT CONCERNING CONJOINT WORK

CRUCES, GUILLERMO ANTONIO - LSE POVERTY, INCOME FLUCTUATIONS AND WORK: ARGENTINA, 1991-2002.

As required in the "UNIVERSITY OF LONDON REGULATIONS FOR THE DEGREES OF MPhil AND PhD," I hereby state that a part of Chapters 1, 3 and 4.of my thesis stems from joint research with Dr. Quentin Wodon from the World Bank, Washington DC, which have been published or submitted to publication to scholarly journals.

The Chapters as submitted represent adaptations and additions to this joint work with Dr. Wodon, and they form an integral part of the thesis. They consist of my own account of my investigations. The part that corresponds to me is two thirds of each of the three Chapters.

Signed by the candidate:

Date: 18/11/2004

G. CRUCES

Certified by the supervisor:

Date: 18/11/2004

F. COWELL